

# Assessment and Remediation of Contaminated Sediments

Edited by

## Danny Reible and Tomas Lanczos

NATO Science Series

IV. Earth and Environmental Sciences - Vol. 73

# Assessment and Remediation of Contaminated Sediments

#### **NATO Science Series**

A Series presenting the results of scientific meetings supported under the NATO Science Programme.

The Series is published by IOS Press, Amsterdam, and Springer in conjunction with the NATO Public Diplomacy Division

Sub-Series

I.	Life and Behavioural Sciences	IOS Press
II.	Mathematics, Physics and Chemistry	Springer
III.	Computer and Systems Science	IOS Press
IV.	Earth and Environmental Sciences	Springer

The NATO Science Series continues the series of books published formerly as the NATO ASI Series.

The NATO Science Programme offers support for collaboration in civil science between scientists of countries of the Euro-Atlantic Partnership Council. The types of scientific meeting generally supported are "Advanced Study Institutes" and "Advanced Research Workshops", and the NATO Science Series collects together the results of these meetings. The meetings are co-organized bij scientists from NATO countries and scientists from NATO's Partner countries – countries of the CIS and Central and Eastern Europe.

Advanced Study Institutes are high-level tutorial courses offering in-depth study of latest advances in a field.

Advanced Research Workshops are expert meetings aimed at critical assessment of a field, and identification of directions for future action.

As a consequence of the restructuring of the NATO Science Programme in 1999, the NATO Science Series was re-organised to the four sub-series noted above. Please consult the following web sites for information on previous volumes published in the Series.

http://www.nato.int/science http://www.springer.com http://www.iospress.nl

# Assessment and Remediation of Contaminated Sediments

edited by

### Danny Reible

The University of Texas, Austin, U.S.A.

and

#### **Tomas Lanczos**

Comenius University, Bratislava, Slovak Republic



Proceedings of the NATO Advanced Research Workshop on Assessment and Remediation of Contaminated Sediments Bratislava, Slovak Republic 18–21 May 2005

A C.I.P. Catalogue record for this book is available from the Library of Congress.

ISBN-10 1-4020-4958-7 (PB) ISBN-13 978-1-4020-4958-3 (PB) ISBN-10 1-4020-4957-9 (HB) ISBN-13 978-1-4020-4957-6 (HB) ISBN-10 1-4020-4959-5 (e-book) ISBN-13 978-1-4020-4959-0 (e-book)

Published by Springer, P.O. Box 17, 3300 AA Dordrecht, The Netherlands.

www.springer.com

Printed on acid-free paper

All Rights Reserved

© 2006 Springer

No part of this work may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, microfilming, recording or otherwise, without written permission from the Publisher, with the exception of any material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work.

#### **TABLE OF CONTENTS**

LIST OF CONTRIBUTORSvii
INTRODUCTION1
DANNY REIBLE AND TOMÁŠ LÁNCZOS
APPROACHES AND FRAMEWORKS FOR MANAGING CONTAMINATED SEDIMENTS - A EUROPEAN PERSPECTIVE
SABINE E. APITZ, JOS BRILS, ANTONIO MARCOMINI, ANDREA CRITTO, PAOLA AGOSTINI, CHRISTIAN MICHELETTI, ROBERTO PIPPA, PETRA SCANFERLA STEFANO ZUIN, TOMÁŠ LÁNCZOS, KATARÍNA DERCOVÁ, ANTON KOČAN, JÁN PETRÍK, PAVEL HUCKO AND PATRIK KUSNIR
PHYSICOCHEMICAL AND BIOLOGICAL ASSESSMENT AND CHARACTERIZATION OF CONTAMINATED SEDIMENTS
SAM BENTLEY, LOUIS THIBODEAUX, PETER ADRIAENS, MENG-YENG LI, MARÍA ROMERO-GONZÁLEZ, STEVEN A. BANWART, ZDENEK FILIP, KATERINA DEMNEROVA AND DANNY REIBLE
TREATMENT AND CONTAINMENT OF CONTAMINATED SEDIMENTS137
JEANNE E. TOMASZEWSKI, DENNIS W. SMITHENRY, YEO-MYOUNG CHO, RICHARD G. LUTHY, GREG V. LOWRY, DANNY REIBLE, TOMAS MACEK, MARTINA SURÁ, ZUZANA CHRASTILOVA, KATERINA DEMNEROVA, MARTINA MACKOVÁ, DANIELA PAVLIKOVÁ, MIKLOS SZEKERES AND MICHEL SYLVESTRE

#### 

SAÏD EL FANTROUSSI, SPIROS N. AGATHOS, DIETMAR H. PIEPER\*, ROBERT WITZIG, BEATRIZ CÁMARA, LOTTE GABRIEL-JÜRGENS, HOWARD JUNCA, GIULIO ZANAROLI, FABIO FAVA, JOSÉ R. PÉREZ-JIMÉNEZ, LILY Y. YOUNG, KELLY HAMONTS, RICHARD LOOKMAN, MIRANDA MAESEN, LUDO DIELS, WINNIE DEJONGHE, JOHN DIJK AND DIRK SPRINGAEL

#### 

FABIO FAVA AND SPYROS N. AGATHOS

#### LIST OF CONTRIBUTORS

#### **CO-DIRECTORS**

#### **Tomas Lanczos**

Natural Sciences, Comenius University PRIF UK Mlynská dolina 842 15 Bratislava 4 SLOVAK REPUBLIC Telephone: ++421 - 2 - 602 96 111 Facsimile: ++421 - 2 - 654 29 064 E-mail: lanczos@gsm.eurotel.sk URL: http://www.fns.uniba.sk/fns/info/0.php3

#### **Danny Reible**

Hazardous Substance Research Center S&SW The University of Texas at Austin Civil Engineering Department 1 University Station C1786 Austin, Texas 78712-0283 USA Telephone: 512-471-4642 Facsimile: 512-471-5870 E-mail: reible@mail.utexas.edu URL: http://www.ce.utexas.edu/profile.cfm?profilePK=172

#### PARTICIPANTS

#### **Peter Adriaens**

University of Michigan 1351 Beal Ave 180 EWRE Building Ann Arbor, MI USA Telephone: 734-763-8032 Facsimile: 734-763-2275 E-mail: adriaens@umich.edu

#### **Spiros N. Agathos**

Unit of Bioengineering (GEBI) University of Louvain Place Croix du Sud 2/19 B-1348 Louvain-la-Neuve BELGIUM Telephone: +32-10-47 36 44 (personal) +32-10-47 36 55 (secretary) Facsimile: +32-10-47 30 62 E-mail: agathos@gebi.ucl.ac.be URL: http://www.gebi.ucl.ac.be

#### Andrea Alessandrini

Environment Office Regione Del Veneto Calle Priuli – Cannaregio Venezia, Italia/Veneto ITALY Telephone: 0039 041 279 2037 Facsimile: 0039 041 279 2445 E-mail: andrea.alessandrini@regione.veneto.it

#### Sabine E. Apitz

Sea Environmental Decisions, Ltd. 1 South Cottages, The Ford Little, Hertfordshire UNITED KINGDOM Telephone: 01279 771890 Facsimile: 01279 771403 E-mail: drsea@cvrl.org URL: http://www.silsoe.cranfie

#### Steven A. Banwart

Groundwater Protection & Restoration Group Dept. of Civil & Structural Engineering University of Sheffield Sir Frederick Mappin Building, Room MEZ 4a Mappin Street Sheffield,

viii

S1 3JD UNITED KINGDOM Telephone: +44/0 114 222 5742 Facsimile: +44/0 114 222 5700 E-mail: s.a.banwart@sheffield.ac.uk URL: www.shef.ac.uk/gprg/

#### Walter Bellefontaine

GFI Environment Avenue Pasteur 9 Wavre, 1300 BELGIUM Telephone: +32 10 239 651 Facsimile: +32 10 239 659 E-mail: wb.gfi@forrestgroup.com

#### **Samuel Bentley**

Department of Oceanography & Costal Sciences Louisiana State University 302 Howe Russell Geosciences Complex Baton Rouge, LA 70803 USA Telephone: 225-578-2954 Facsimile: 225-578-2520 E-mail: sjb@lsu.edu URL: www.ocean.lsu.edu

#### Eva Berdisova

Natural Sciences, Comenius University PRIF UK Mlynská dolina 842 15 Bratislava 4 SLOVAK REPUBLIC Telephone: ++421 - 2 - 602 96 111 Facsimile: ++421 - 2 - 654 29 064 URL: http://www.fns.uniba.sk/fns/info/0.php3

#### **Vladimir Brenner**

Earth Tech Trojska 92 Prague 7 CZECH REPUIBLIC Telephone: +420283090653 Facsimile: +420283090658 E-mail: Vladimir.Brenner@Earthtech.cz

#### **Todd S. Bridges**

Center for Contaminated Sediments U. S. Army Engineer Research & Development Center 3909 Halls Ferry Rd. Vicksburg, MS USA Telephone: 601-634-3626 Facsimile: 601-634-3120 E-mail: Todd.S.Bridges@erdc.usace.army.mil

#### **Jos Brils**

TNO Built Environment & Geosciences Netherlands Organization for Applied Scientific Research Princetonlaan 6 P.O. Box 80015 Utrecht THE NETHERLANDS Telephone: +31 6 22799183 E-mail: j.m.brils@mep.tno.nl URL: www.tno.nl

#### Shamil Cathum

SAIC Canada 335 River Road Ottawa, Onterio K1A 0H3 CANADA Telephone: 613-990-6879 Facsimile: 613-991-1673 E-mail: shamil.cathum@saiccanada.com

#### Natalia Chubar

Institute for Sorption & Endoecology National Academy of Sciences of Ukraine General Naumov, 13 Kiev, 03164 UKRAINE Telephone: +38-044-4529329 Facsimile: +38-044-4529327 E-mail: chubar@ispe.ldc.net

#### Marinella Conte

Technical University of Bari – II Faculty of Engineering Viale del Turismo, 8 Taranto, 74100 ITALY Telephone: +393338766921 E-mail: m.conte@poliba.it

#### Laura D'Aprile

Italian Environmental Protection Agency APAT Via Vitaliano Brancati 48 Rome, Italy ITALY Telephone: 06 50072272 Facsimile: 06 50072048 E-mail: daprile@apat.it

#### Katerina Demnerova

ICTP, Inst. Of Chem. Technol. Prague Technicka 5 Prague 6, CZECH REPUBLIC Telephone: ++420220445172 Facsimile: ++420220443075 E-mail: demnerok@vscht.cz URL: http://www.vscht.cz/homepage/english/main

#### Katarina Dercova

Slovak University of Technology Faculty of Chemical & Food Technology Department of Biochemical Technology Radlinskeho 9 Bratislava SLOVAK REPUBLIC Telephone: 00 421 2 59 32 57 10 Facsimile: 00 421 2 52 96 70 85 E-mail: katarina.dercova@stuba.sk URL: http://www.chtf.stuba.sk/

#### **Ludo Diels**

Hoofd Milieu- en Procestechnologie Vlaamse instelling voor technologisch onderzoek (Vito) Boeretang 200 B - 2400 Mol, België BELGIUM TeleTelephone: + 32 (0)14 33 **69 24** Mobile: + 32 (0)477 622 304 Facsimile: + 32 (0)14 58 05 23 E-mail: ludo.diels@vito.be URL : http://www.vito.be

#### Fabio Fava

DICASM, Faculty of Engineering Alma Mater Studiorum-University of Bologna Viale Risorgimento 2 Bologna ITALY Telephone: 0039 051 2093212 Facsimile: 0039 051 2093220 E-mail: Fabio.fava@unibo.it

#### Zdenek Flip

Department of Biochemistry & Microbiology Institute of Chemical Technology Prague Technicka 3-5

xii

Prague CZECH REPUBLIC Telephone: +420 220 445 141 Facsimile: +420 220 445 167 E-mail: zdenek.filip@vscht.cz

#### **Roberta Guerra**

University of Bologna 163 Via S. Alberto Ravenna ITALY Telephone: +39 0544 937305 Facsimile: +39 0544 937303 E-mail: r.guerra@ambra.unibo.it

#### Joop Harmsen

Alterra, Wageningen University P.O. Box 47 Wageningen THE NETHERLANDS Telephone: \*31-317-474360 Facsimile: \*31-317-419000 E-mail: joop.harmsen@wur.nl URL: www.alterra.nl

#### **Don Hayes**

Civil & Environmental Engineering University of Utah, 122 S. Central Campus Dr. Suite 104, Salt Lake City, UT 84112; USA Telephone: 801-581-7110 E-mail: hayes@civil.utah.edu

#### **Mark Hodges**

Georgia Tech Research Institute Electro-Optical, Environment & Materials Laboratory Communication & Training Technologies Atlanta, GA 30332-0837 USA Telephone: 404-894-6987 E-mail: mark.hodges@gtri.gatech.edu URL: http://www.hsrc-ssw.org/hodges-bio.html

#### **Pavel Hucko**

Water Research Institute Nabr. Arm. Gen. L. Svobodu 5 Bratislava, SLOVAK REPUBLIC Telephone: 00421259343424, 00421259343477 Facsimile: 00421254411941 E-mail: Pavel\_Hucko@vuvh.sk URL: http://www.vuvh.sk

#### Marieta Ionescu

ICTP, Inst. Of Chem. Technol. Prague Technicka 5 Prague 6, Czech Republic CZECH REPUBLIC Telephone: ++420220445172 Facsimile: ++420220443075 E-mail: mariettuca22@yahoo.com

#### Elena Kamilova

Institute of Genetics & Plant Experimental Biology Academy Of Sciences O.Zakirova Ave., apt. 1-35 Tashkent UZBEKISTAN Telephone: + 9 98711564018 Facsimile: + 9 98711123271 E-mail: khelen@tkt.uz

#### Anton Kocan

Institute of Preventive & Clinical Medicine Slovak Medical University Limbova 14

xiv

833 01 Bratislava SLOVAK REPUBLIC E-mail: anton.kocan@szu.sk

#### **Alexander Kochetkov**

Scientific Production Association "Typhoon" Federal Government Institution Lenin Av. 82 Obninsk, Kaluga region RUSSIAN FEDERATION Telephone: (08439)7-11-96 Facsimile: (08439)7-12-42 E-mail: kochet@obninsk.com

#### Nedra Korevec

Louisiana State University, Hazardous Substance Research Center S/SW., 3221 CEBA Bldg. Nicholson Ext., Baton Rouge, LA 70803 USA Telephone: 225-578-6770 Facsimile: 225-578-5043 E-mail: nkorev1@lsu.edu URL : http://www.hsrc-ssw.org/korevec-bio.html

#### Katrína Krčmová

Natural Sciences, Comenius University PRIF UK Mlynská dolina 842 15 Bratislava 4 SLOVAK REPUBLIC Telephone: ++421 - 2 - 602 96 111 Facsimile: ++421 - 2 - 654 29 064 URL: http://www.fns.uniba.sk/fns/info/0.php3

#### Tamara Kukharchyk

Institute of Problems of Natural Resources Use & Ecology of National Academy of Sciences 10, Staroborisovski tract Minsk BELARUS Telephone: +375-17-290-34-27 Facsimile: +375-17-264-24-13 E-mail: kukharchyk@mail.ru

#### Patrik Kusnir

Water Research Institute Nabr. arm. gen. L. Svobodu 5, Bratislava SLOVAK REPUBLIC Telephone: +421 2 59343463 E-mail: Patrik\_Kusnir@vuvh.sk URL: http://www.vuvh.sk

#### **Gregory V. Lowry**

Carnegie Mellon University 5000 Forbes Ave. Department of Civil & Environmental Engineering Pittsburgh, PA USA Telephone: 412-268-2948 Facsimile: 412-268-7813 E-mail: glowry@cmu.edu URL: www.ce.cmu.edu/~glowry

#### **Richard G. Luthy**

Department of Civil & Environmental Engineering Terman Engineering Center, M-52 Stanford University Stanford, California USA Telephone: 650-723-3921 Facsimile: 650-725-8662 E-mail: luthy@stanford.edu URL: www-ce.stanford.edu/faculty/luthy/

#### **Thomas Macek**

Institute of Organic Chemistry & Biochemistry Flemingovo n.2 Prague, CZ- 166 10

xvi

#### CZECH REPUBLIC

Telephone: +420-220 183 340 Facsimile: +420-220 183 582 E-mail: tom.macek@uochb.cas.cz URL: www.uochb.cas.cz

#### Martina Mackova

Institute of Chemical Technology Prague Technicka 3 Prague, CZ-166 28 CZECH REPUBLIC Telephone: +420-220 445 139 Facsimile: +420-220 445 172 E-mail: martina.mackova@vscht.cz URL: www.vscht.cz

#### Antonio Marcomini

Ca Foscari Venice University & Consorzio Venezia Ricerche v. della Libertà 12 Marghera- Venice, Venice ITALY Telephone: +39 041 5093017 E-mail: sp.cvr@vegapark.ve.it

#### Vit Matějů

ENVISAN-GEM, a.s. Budova VUPP Radiova 7 Praha 10 CZECH REPUBLIC Telephone: +420-296792224 Facsimile: +420-296792223 E-mail: envisan@mbox.vol.cz URL : http://www.envisan.cz

#### Leigh McCook

Safety, Health & Environmental Technologies Georgia Tech Research Institute Georgia Institute of Technology Atlanta, GA 30332-0837 USA Telephone: 404- 894-7898 Facsimile: 404-894-9081 E-mail: leigh.mccook@gtri.gatech.edu URL: http://www.hsrc-ssw.org/mccook-bio.html

#### Maria Nikolopoulou

Technical University of Crete Department of Environmental Engineering Polytechniouplis Chania, Crete GREECE Telephone: +3 2821 0 37805 Facsimile: -3 2821 0 37846 E-mail: mnikolopoulou@isc.tuc.gr

#### Mikhail Novitsky

Typhoon Institute of Exp. Meterology Lenin pr.82 Obninsk, Kaluga reg. RUSSIA Telephone: +7 (08439) 71941 Facsimile: +7(08439)40910 E-mail: novitsky@typhoon.obninsk.org URL: typhoon-tower.obni

#### John Pardue

Civil & Environmental Engineering Louisiana Water Resources Research Institute Hazardous Substance Research Center S/SW 3221 CEBA Bldg. Baton Rouge, LA 70803 Louisiana State University USA Telephone: 225.578.6027 Facsimile: 225.578.5043 E-mail: jpardue@lsu.edu URL : http://www.lwrri.lsu.edu

xviii

#### Jan Petrik

Institute of Preventive & Clinical Medicine Slovak Medical University Limbova 14 833 01 Bratislava SLOVAK REPUBLIC E-mail: jan.petrik@szu.sk

#### Dietmar H. Pieper

German Research Centre for Biotechnology Mascheroder Weg 1 Braunschweig GERMANY Telephone: +49 531 6181 467 Facsimile: +49 531 6181 411 E-mail: dpi@gbf.de URL: www.gbf.de

#### Valentin Popov

Federal Government Institution Scientific Production Association Lenin St, 82 Obninsk, Kaluzhskaya oblast 249038 RUSSIAN FEDERATION Telephone: (08439)7-19-14 Facsimile: (08439)7-12-42 E-mail: vpopov2@obninsk.com

#### Maria Romero-Gonzalez

Groundwater Protection & Restoration Group Dept. of Civil & Structural Engineering University of Sheffield Sir Frederick Mappin Building, Room F111a Mappin Street Sheffield S1 3JD UNITED KINGDOM Telephone: +44/0 114 222 5746 Facsimile: +44/0 114 222 5700 E-mail: m.e.romero-gonzalez@shef.ac.uk URL: www.shef.ac.uk/gprg/

#### Marta Ruiz

Technical University of Crete Ploytecneioupolis Chaina, Crete 73100 GREECE Telephone: 00302821037819 Facsimile: 00302821037847 E-mail: marta@enveng.tuc.gr

#### Julia Russkova

Parsons Global Services, Inc. 23, 1<sup>st</sup> Tverskaya-Yamskaya Str. Moscow RUSSIA Telephone: 7(095)7558313,12128032027 Facsimile: 7(095)7558315,12124791935 E-mail: Julia.russkova@rcwdp.ru

#### Maroš Sirotiak

Natural Sciences, Comenius University PRIF UK Mlynská dolina 842 15 Bratislava 4 SLOVAK REPUBLIC Telephone: ++421 - 2 - 602 96 111 Facsimile: ++421 - 2 - 654 29 064 URL: http://www.fns.uniba.sk/fns/info/0.php3

#### Michal Šutriepka

Natural Sciences, Comenius University PRIF UK Mlynská dolina 842 15 Bratislava 4 SLOVAK REPUBLIC Telephone: ++421 - 2 - 602 96 111 Facsimile: ++421 - 2 - 654 29 064 URL: http://www.fns.uniba.sk/fns/info/0.php3

#### **Pavel Szerbin**

Paks Nuclear Power Plant POB 71 Paks 7031 HUNGARY Telephone: +36-75/514 2980 Facsimile: +36-75/414 653 E-mail: szerbin@npp.hu

#### Victor Tsvetkov

St.-Petersburg Institute of Technology Moscovsky pr., 26 St.-Petersburg RUSSIA Telephone: +7 (812)9325715 Facsimile: +7 (812)3151036 E-mail: tsvictory@mail.ru URL: www.gti.spb.ru

#### **Richard Wenning**

Environ 6001 Shellmound St. Suite 700 Emeryville, California USA Telephone: 510-420-2556 Facsimile: 510-655-9517 E-mail: rhwenning@environcorp.com URL: www.environcorp.com

#### **INTRODUCTION**

DANNY REIBLE The University of Texas at Austin, TX 78712 USA

TOMÁŠ LÁNCZOS Comenius University, Faculty of Natural Sciences, Dept. Geochemistry, Bratislava, Slavokia

On May 18-21, 2005, 56 scientists and engineers convened for an Advanced Research Workshop (ARW) sponsored by the North Atlantic Treaty Organization (NATO). The focus of the ARW was to define and evaluate progress in our ability to assess and remediate contaminated sediments. Contaminated sediments pose some of the most difficult site remediation issues today. They are typically the ultimate repository for contaminants in the environment as a result of runoff and deposition. As such they pose long-term sources of contaminants back to the environment after land contamination issues are identified and controlled. The environmental security of both NATO and partner countries is at risk due to the pervasive nature of sediment contamination of rivers, lakes and harbors.

Contaminated sediments typically reside in spatially variable and dynamic systems subject to seasonal flow variations and episodic storm events. The volume of sediments that must be managed at particular sites often exceeds one million cubic meters, dwarfing many contaminated soil sites. These sediments are also associated with equally daunting volumes of water and efforts to remove the contamination typically entrains even more water. The risks associated with these sediments depend on the processes controlling both contaminant release from the sediments and the transfer to benthic, aquatic, and land-based organisms. Observations of impairments in ecological or human health can indicate potential pollution problems; however, linking these adverse effects to contaminated sediments requires an understanding of the processes leading to exposure and uptake. In addition, the selection of cost-effective and environmentally protective remedial alternatives is dependent upon the ability to predict the risks during implementation and into the future. Due to the volume and complexity of contaminated sediment sites, few economic and effective solutions are available.

Many of the potential technologies for contaminated sediment management were initially developed to manage contaminated soils. Unfortunately, many of these technologies are difficult either to apply or impose potentially unacceptable risks when applied to contaminated sediments. Identifying, comparing, and selecting remedial options for contaminated sediment is complicated also by the multiple technologies often involved. For example, ex situ treatment or sediment disposal typically introduces a complete train of technologies, including removing material by dredging, temporarily storing or pretreating to reduce water content or volume, treating or disposing of final dredged material, and managing any residually contaminated materials. Large contaminated sediment sites generally require applying different options at different areas on-site, each containing multiple technologies. Therefore, identifying sediment management and remediation options must recognize the entire train of technologies that constitute each option and the interaction of these technologies with the natural sediment processes leading to exposure and risk. Risk reduction has been generally accepted as the metric by which various options are judged and selected. Use of this metric, however, places a premium on understanding the natural processes that link sediment contaminants to exposure and risk. Evaluating management or remedial options requires defining remedial action goals and objectives and developing a valid conceptual model of the sediment system to be remediated, an exceedingly difficult proposition in complex contaminated sediment sites.

The Advanced Research Workshop was directed toward identifying the current state of our understanding of these challenges and progress toward their solution. Many of the problems with managing contaminated sediments have been recognized only in past few years as large contaminated sediments sites such as the Hudson River in the United States have come under scrutiny. Our knowledge and understanding is relatively weak compared to other contaminated site problems. The goal of the Workshop was to ensure that the developing understanding of sediment processes on both sides of the Atlantic are shared and fully incorporated into the thinking of sediment scientists and the regulatory and management community that depend upon their guidance. In the United States, the US Army Corps of Engineers, the US Environmental Protection Agency and University consortia such as the Hazardous Substance Research Center/South and Southwest have focused attention and resources on improving the understanding of contaminated sediment processes and their management. In Europe, a loose organization of scientists and engineers, SEDNET, has led the way in organizing contaminated sediment information. Although there has been interaction among the various groups, the Advanced Research Workshop aided the exchange of information and, in addition, encouraged interactions with Eastern European and partner countries that have heretofore have not focused as much attention or resources on managing contaminated sediments.

#### INTRODUCTION

The Workshop was organized along the general lines of

- 1. Identifying the key contaminated sediment issues and approaches in US and Europe
- 2. Identifying the state of the art in physical, chemical and biological assessment of sediments
- 3. Identifying the state of the art in our understanding of key fate and transport processes in sediments for both organic and inorganic contaminants
- 4. Identifying the state of the art in both ex-situ and in-situ management of contaminated sediments
- 5. Integrating the information exchanged during the workshop to define research gaps

The introductory topic ensured that the participants had achieved some agreement as to the key sediment issues and concerns limiting their management. This was followed by a discussion of the tools available for sediment site characterization and analysis. The translation of these into a valid conceptual model of the system depends upon our understanding of the fate and transport processes influencing contaminants and the resulting exposure to human health and the environment. The Workshop then sought to identify the options available and the key technical issues remaining that limit the ability to effectively manage contaminated sediments. Finally, the Workshop sought to integrate the information to help focus attention on the technical deficiencies in both assessment and remediation that limit our ability to effectively respond to contaminated sediment problems.

It was recognized by those in attendance that in-situ management approaches to contaminated sediments hold significant advantages in complexity, cost and environmental consequences over ex-situ management approaches. As such, the workshop focused much of the discussion on in-situ assessment and remediation. Among the technologies that exhibit the greatest potential for management of contaminated sediments is biological transformation and degradation processes and a significant component of the workshop was devoted to their discussion. This is reflected in this book that summarizes the proceedings of the workshop. In addition, the uncertainties and research needs for realizing the potential of biotransformation and biodegradation processes were also highlighted.

It is hoped that the current volume will provide a window into the discussions held during the ARW for those that were unable to attend and will benefit professionals seeking to gain an understanding of the key issues limiting our ability to manage contaminated sediments. It is toward that goal that this volume is dedicated.

Acknowledgement

The Editors and Participants would like to express their appreciation to the North Atlantic Treaty Organization, the Assistant Secretary General for Scientific and Environmental Affairs, and the Advisory Panel on Environmental and Earth Science & Technology for support of the Advanced Study Institute under Grant EST.ARW. 981356.

The facilities and support of the host institution, the Comenius University in Bratislava, Slovakia is also gratefully acknowledged.

Finally, the Advanced Research Workshop would not have been possible without the support and assistance of Nedra Korevec of Louisiana State University, Vlasta Dideková of Commenius University, and Sharon Bernard and Marsha Betak of the University of Texas.

#### APPROACHES AND FRAMEWORKS FOR MANAGING CONTAMINATED SEDIMENTS - A EUROPEAN PERSPECTIVE

SABINE E. APITZ\* SEA Environmental Decisions, Ltd., Little Hadham, United Kingdom.

JOS BRILS Netherlands Organization for Applied Scientific Research - TNO, The Netherlands

ANTONIO MARCOMINI, ANDREA CRITTO, PAOLA AGOSTINI AND CHRISTIAN MICHELETTI *Ca' Foscari University, Venice, Italy* 

ROBERTO PIPPA, PETRA SCANFERLA AND STEFANO ZUIN Consorzio Venezia Ricerche, Venice, Italy

TOMÁŠ LÁNCZOS Comenius University, Faculty of Natural Sciences, Dept. Geochemistry, Bratislava

KATARÍNA DERCOVÁ Department of Biochemical Technology, Faculty of Chemical and Food Technology, Slovak University of Technology, Bratislava, Slovak Republic

ANTON KOČAN, JÁN PETRÍK Department of Toxic Organic Pollutants, Institute of Preventive and Clinical Medicine, Slovak Medical University, Bratislava, Slovak Republic

PAVEL HUCKO, PATRIK KUSNIR Water Research Institute Bratislava, Bratislava, Slovak Republic

\*To whom correspondence should be addressed: drsea@cvrl.org.

**Abstract-** Sediment is an essential, integral and dynamic part of the hydrologic system. In natural and agricultural basins, sediment is derived from the weathering and erosion of minerals, organic material and soils in upstream areas and from the erosion of river banks and other in-stream sources. As surface-water flow rates decline in lowland areas, transported sediment settles along the river bed and banks by sedimentation. However,

because sediments are the ultimate reservoir for the numerous potential chemical and biological contaminants that may be contained in effluents originating from urban, agricultural, and industrial lands and recreational activities, contaminated sediments in rivers and streams, lakes, coastal harbors, and estuaries have the potential to pose ecological and human health risks. The management of sediment quality and quantity in support of ecological and socioeconomic goals is a cause of extensive research, investment and regulatory and public scrutiny, both in Europe and internationally. This chapter examines the management of sediment from a European perspective. The first section discusses the role of sediment management in achieving European ecosystem objectives. Section two summarizes the findings of the EC-funded, demand driven European Sediment Research Network (SedNet). Section three describes a framework risk assessment and management in a major European region, the Venice Lagoon. The next three sections describe aspects of sediment management in an EC accession country, and the host country for this workshop -Slovakia. Section four describes river and lake sediment contamination and related legislation in Slovakia. Section five discusses the assessment and management of Persistent Organic Pollutants (POPs), a major risk driver in Slovakia. Section six discusses a specific class of POPs - PCBS in sediments of a specific region of Slovakia. Finally, Section seven discusses some barriers to successful sediment risk assessment and management, and lists science and infrastructure needs to address these barriers. Whilst all the authors of this chapter are living and/or working in Europe, and the focus is on European perspectives, much work has benefited from international collaborations and we feel that many of the observations and recommendations will have relevance to a broader audience.

**Keywords:** Sediment management; risk assessment; sediment reclamation; Water Framework Directive; Ecosystem; Basin-scale; SedNet; Venice Lagoon, biodegradation, bioremediation, persistent, pollutants, sediments, PCBs, POPs, Sediments quality, sediment contamination by PCBs, the Zemplinska Sirava Reservoir, Vicinity of Strazske

## 1. Sediment Management: Its role in achieving European ecosystem objectives

#### 1.1. THE EVOLUTION OF SEDIMENT MANAGEMENT

We have been dredging, and thus managing, sediments for centuries. Dredging technology is arguably one of the triumphs of human ingenuity in our battle to control nature. Notwithstanding some of its current press, it has given us our greatest cities, fuelled trade empires and continues to keep nations from disappearing beneath the sea. However, over time, this process grew more complex. The waste products of our other technological triumphs bound with the sediments that were being dredged Thus, dredged material (DM) slowly evolved from a precious resource to a waste material, and environmental assessment and control became a "burden" that was added onto long-established processes. Over time, dredgers developed containment, treatment and disposal technologies to handle these wastes, and thus clever engineering solutions have been developed and applied. However, costs became higher and options were restricted (Apitz 2004).

Now, the field of sediment management can be divided into two general categories, largely defined by the purpose for which they are being examined. The first, construction or navigational dredging, generally involves the assessment and removal of large volumes of sediment, in support of socioeconomic goals. The second type of sediment management, hotspot or environmental cleanup of contaminated sediments, generally addresses smaller volumes of sediment, though there are notable exceptions. The regulatory frameworks and technical communities that address these two sediment categories are often separate, with little or ineffective interaction. Assumptions, methods and frameworks designed to address one category may be inappropriate for the other (Apitz and Power 2002).

However, because the former category of sediment management was well-developed long before the latter was considered, we have historically addressed contaminated sediment management using a DM managementinfluenced approach. A decade or more ago, removal and treatment of contaminated sediments appeared to be the remedy of choice, regardless of any potential risks. This approach is a given for DM management, but one can call this a "Chemical engineering" approach to management. The consequence was that projected costs for managing contaminated sediments grew to the scale of national budgets. As a result, it is most likely that large volumes of contaminated sediments will be managed in place, and that source control is emerging as a major component of risk management. It can be argued that the goal of sustainable sediment management should be the selection of the least invasive, but sufficiently protective, management strategy. However, the way we assess and manage sediments in support of such an approach is different than if we are assessing the impact of DM disposal (Apitz 2004).

An emerging recognition of this issue has led, in the last decade, to what could be called the first revolution in scale for sediment assessment and management. Instead of treating sediments as volumes of material to be dumped, contained or treated, we now need to look at the micro- and mesoscale interactions. Scientists and decision makers began to consider how contaminants interact with sediments, how contaminants may move between sediments, water and biota and how contaminants might move over time. It is now clear that sediments can bind contaminants in different ways, depending upon sediment characteristics, geochemical conditions and even degree of aging. This can affect contaminant mobility, bioavailability, degradability, fate and risk. An examination of these issues resulted in the important consideration of site-specific Conceptual Site Models to evaluate these processes (Apitz et al. 2005c). Furthermore, it is clear that an understanding of pathways of contaminant transport (mode, media and mechanism) will inform CSMs, help put biological observations in context and help design management strategies (Apitz et al. 2005b). In recent years, scientists throughout the world have carried out extensive research examining pathways of contaminant - sediment - water - biota interaction, and determining how they might affect risks in sediments.

#### 1.2. THE ROLE OF SEDIMENTS IN BASIN-SCALE MANAGEMENT

However, contaminants transfer between all environmental media. We cannot manage one medium without taking this into account, nor can we manage connected sites in isolation. To reduce risk, we must assess and manage it holistically and at the basin scale. Sediment is part of the hydrodynamic continuum – actions on a sediment unit can affect other parcels. resulting in conflicting, counterproductive or inefficient management actions if not coordinated, regardless of goals. There is now a growing recognition that sustainable risk management implies assessment at the basin-scale, managed at the site-specific scale(Apitz and White 2003). A basin-scale risk management framework should be comprised of two principal levels of decision making: basin-scale evaluation (risk prioritization of sites for further evaluation and/or management) and an assessment of specific sites for risks and management options (site-specific risk ranking and management). The growing recognition of this issue can be viewed as a second revolution of scale (Apitz 2004).

Management of risk in a river basin demands that sediment risk management should be closely linked with the management of soil, water, and industrial and agricultural policy. Because of the complexity of interactions, conceptual models are required to identify, quantify and communicate the links between these processes and media (Apitz et al. 2005a). An understanding of the particle and contaminant flows and interactions within a river basin should inform basin-scale evaluation. This can be termed a Conceptual Basin Model (CBM). It describes how materials move and interact between sites and media, leads to increased knowledge about the river basin system and serves as an important communication tool between scientists, decision makers and stakeholders.

#### 1.2.1. The indivisibility of sediment and soil management

Development of CBMs makes clear that, as well as an integration of various aspects of sediment management, basin-scale management requires a better integration of sediment and soil science and management. Soil can be defined as "...the top layer of the earth's crust. It is formed by mineral particles, organic matter, water, air and living organisms. Soil is an complex. variable and living extremelv medium" (http://europa.eu.int/comm/environment/soil/). Sediment, on the other hand, has been defined by SedNet as "suspended or deposited solids, acting as a main component of a matrix which has been or is susceptible to being transported by water" (Brils 2003). The first thing one notices when comparing these definitions is that the above definition of soil does not specifically exclude sediments. A further examination of the SedNet sediment definition reveals that all soils, during the weathering and transport process, have been sediments, and also suggests that they will be sediments again during their lifetime. Clearly, both soils and sediments often have a common origin – the weathering products of rocks and organic material. Generally, if they are deposited (or formed) in a terrestrial setting, they are considered soils, while if deposited in an aquatic setting they are considered sediments. The greatest distinguishing factors are water and time. Although these differences will result in different biogeochemical, ecological and other behavior in soils vs sediments, in many cases, the distinction between these two materials will be subtle, transitory and, arguably, unnecessary.

#### 1.2.2. Differences between soil and sediment management

As urbanization and industrialization resulted in the release of contaminants into the environment, both soil and sediment science evolved to assess and manage the impact of these releases on the ecosystem. In terms of the management of contaminated soils and sediments, there are many commonalities, as they behave similarly: both are accumulators (and longterm donators) of persistent pollutants such as heavy metals and lipophilic organic compounds. However, contaminated sediment investigations have features that make them more complex than water evaluations and, to a lesser degree, soil or terrestrial investigations (NRC 2001, Apitz et al. 2005a). The simple fact that sediments lie under water implies that measurement, observation, and mapping of contaminant and ecosystem characteristics are technically challenging and expensive. Sediments integrate contaminant input from multiple sources within a watershed or coastal region, creating difficulties in tracking the potential sources of contamination. This can lead to ubiquitous, regional 'background' levels of anthropogenic contaminants that are difficult to separate from site specific sources (Crommentuijn et al. 2000). For the same reasons, sediments are, more often than soils, contaminated with multiple chemicals (Long et al. 1995), making risk assessment and management decision-making difficult and complex. The hydrodynamics and geochemistry of aquatic ecosystems are also quite different from those of terrestrial ecosystems. While soils and groundwater can often be isolated from receptors during remediation, similar isolation or removal approaches for contaminated sediments are more difficult to implement successfully; sensitive aquatic biota are more and at times unavoidably, directly affected likely. during the implementation of the remedy (USEPA 2002). Because the benthic community in direct contact with sediments is often near the base of the aquatic food chain, ecologically-based quality criteria can be orders of magnitude lower than those at most contaminated land sites. Together, these and other factors often push the limits of equivalent assessment methods and cleanup technologies for sediment and can at times increase costs significantly over what may be needed to address similar contaminant conditions in soils. On the other hand, some disposal and containment approaches, when available, compare favorably with soil cleanup technologies and some groundwater cleanups can be prohibitively expensive, so it is difficult to make sweeping statements about costs, which are always driven by site-specific conditions. While the benefits of ownership and cleanup of contaminated land, which can subsequently be sold or developed (or both) to offset the costs of remediation, are clear, such benefits are less obvious in aquatic ecosystems. Furthermore, although contaminated soils can often be left in place untreated (assuming there are no immediate ecological risks), the economic need for sediment dredging, whether for navigational or construction purposes, often requires that large volumes of sediments must be removed and managed even when there is no easily available or cost-effective disposal site or remedial technology. However, the management of sediments cannot be completely separated from that of soil and water, as these systems are interrelated and linked, hydrodynamically, even though not always in regulatory terms (Apitz and White 2003, Apitz et al. 2005a).

#### 1.2.3. Similarities between soil and sediment function

In recent years, there is a growing recognition that soils and sediments have important functions beyond the agricultural, transport and hydrological functions described above. In the development of the European Thematic Strategy for Soil Protection, it was recognized that soil has six main functions, three ecological and three technical. industrial and socioeconomic (Blum 2002). Although less explicitly laid out, the important functions of sediment have been pointed out by SedNet (SedNet 2004) and others as well. Upon examination, it becomes clear that the functions of soil do not differ greatly from the functions of sediment, with the proviso that sediments, unlike soils, are intrinsically linked with aquatic systems. One of the ecological functions of soil is as a substrate for the production of biomass, ensuring food, fodder, renewable energy and raw materials. Especially in shallow waters, sediment plays the same, albeit less obvious, role in aquatic systems, although a large part of many aquatic food chains also have primary producers in surface waters. Soils, sediments, and the organisms which live in them play major roles in the biosphere/atmosphere/land and aquatic biogeochemical balance, including filtering, nutrient regeneration, transformation, oxygen balance, buffering, etc. in freshwater and marine systems. Soil and sediments are biological habitats and form the biggest gene reserve on the globe. The major controlling factor of habitat type and health is soil or sediment type, whether the habitat is a healthy benthic community, eelgrass for fish breeding, meadows, desert, forest or farmland.

There are three main technical, industrial and socio-ecological functions of soil (Blum 2002), and these also are similar to the functions of sediments. Soils provide the physical basis for technical, industrial and socio-economic structures and their development, for instance industry, housing, transport, sports, recreation and the dumping of refuse. Sediments provide a similar substrate for most structures and developments in aquatic systems. As will be discussed later, a balanced soil and sediment supply is necessary to protect infrastructure, and, as sea levels rise and climate changes, our management of this cycle will be a primary line of defense. Soils and sediments are sources of raw materials, geogenic energy and water. Lastly, both soils and sediments are the memory of our geogenic and cultural heritage, forming an essential part of the landscape and concealing paleontological, stratigraphic and archaeological treasures. In spite of these critical functions, both soil and sediment scientists and managers have had difficulty engaging the interest of the regulatory community and the public. The public considers sediments invisible or, in the case of dredged material, a waste, whilst soils are taken for granted while they are eroded, covered, contaminated and destroyed. Extensive discussions were carried out in SedNet meetings about how to engage stakeholders and regulators so that the role of sediments was recognized (see www.sednet.org for workshop reports). Similarly, discussions of the European Soil Strategy grappled with how to make soil more 'sexy' to the same audiences (EEB 2002). Contributors to both the Soil Strategy and SedNet are generating documents and recommendations for the EC to convince them that the protection of soils and sediments are critical in holistic ecosystem management, but they are doing this largely in isolation from one another. Because soils and sediment management are, in reality, interdependent, these efforts should be combined.

#### 1.3. EMERGING ECOSYSTEM-BASED POLICY

It is not, however, just the different historical evolution of soil and sediment science that has hindered better coordination between these fields. Rather, those who work within the web of environmental law and regulation have also recognized that the funneling of problems within specific programmatic 'stove pipes', defined in terms of a specific medium, (e.g., water, soil, air), system (e.g., rivers, estuaries, marine systems, watersheds), contaminant (e.g., PCBs, pesticides, Hg), or activity (e.g., navigation dredging, waste water discharge, environmental cleanup), has in many cases presented impediments to achieving efficient, integrated solutions to environmental problems (Apitz and Power 2002, Bridges et al. 2005). As our understanding of ecological systems has evolved, it has become increasingly clear that effective and sustainable management strategies must focus on whole catchments and their interconnected media (chemicals. water, soil and sediment), rather than on one site or issue at a time. While conceptual approaches for addressing these disconnects are being developed (Apitz and White 2003, Heise 2005, in press), significant institutional barriers remain. However, the European Union has recently adopted several Directives, Strategies, Recommendations and agreements which will require a move from sectoral-based to more ecosystem-based, holistic environmental management (e.g. Apitz et al. 2005 in press, Borja 2005 in press, Ducroty and Elliott 1997, Elliott et al. 1999, Reader et al. 2001, Apitz et al. 2005b), which should make the integration of soil and sediment science and management (and many other fields) both simpler and more necessary.

The implementation of the Water Framework Directive (COM 2000) is changing the scope of water management from the local scale to basin (watershed or catchment) scale (often trans-boundary). It aims to establish a framework for the protection of ground waters and inland, transitional (i.e. fjords, estuaries, rias and lagoons) and coastal surface waters that prevents habitat deterioration and protects and enhances the status of aquatic ecosystems, as well as the terrestrial ecosystems and wetlands linked to them. There has also been a movement from addressing problems in isolation on land, in freshwaters, in estuaries or the coastal zone, to integrating these zones, and extending the ecosystem approach to whole shelf areas. The Integrated Coastal Zone Management Recommendation (ICZM) calls for the "...combination of instruments designed to facilitate coherence between sectoral policy objectives and coherence between planning and management" and "improved coordination of the actions taken by all the authorities concerned both at sea and on land, in managing the sea-land interaction", via a national stock-taking exercise, followed by the development of national strategies, international cooperation, reporting and review, ultimately leading to EC legislation on Coastal Zone management (COM 2002b). The adoption of the EU Marine Strategy (COM 2002a) and the recent suggestion of the need for an accompanying Marine Framework Directive will take integrated ecosystem management philosophies from the terrestrial and freshwater areas through the estuaries and coasts to the open sea, including to the shelf (200 nautical miles) areas. None of these initiatives deal explicitly with soil, sediment, or their relationships, but successful holistic management will require that these are managed as the interrelated issues that they are.

Enshrined in the European Treaty are the precautionary principle and the principles that pollution should be rectified at source, that the polluter should pay and that priority should be given to preventative action. As can be seen from the above, environmental management in Europe will now be based mainly upon biological and ecological (rather than physico-chemical) elements, with ecosystems at the centre of management decisions, applied to all European water bodies (Borja 2005, in press). An important part of achieving ecosystem management is the principle of integration embedded in the European Treaty which requires all other policy areas to take full and proper consideration of the European Community's environmental objectives when making policy decisions (COM 2001a). Thus, the Strategic Environmental Assessment (SEA) Directive (COM 2001b) was developed to ensure that environmental consequences of certain plans and programs are identified and assessed during their preparation and before their adoption. In contrast to Environmental Impact Assessments (COM 1985), required to determine the consequences to proposed projects, these will look at the impacts of decisions above or below the project level. Many activities have both intended and unintended impacts upon soil and sediment balance and function, as well as on the many systems with which they interact. If SEAs and EIAs are to properly address these issues, soil and sediment scientists must work together to provide better science and models in support of these goals.

#### 1.3.1. Implications of emerging policy

Notwithstanding the ambitions or expectations of the WFD and related policy, the impacts of centuries of human activities will not be eradicated (e.g., Malakoff 1997; Jackson et al. 2001), and the need for continuing sustainable development will remain (Sala et al. 2000). Thus, there is clearly a need for the development of ecological measures and technology that can evaluate the environmental status of aquatic ecosystems, and the potential impacts of both proposed developments and measures carried out to mitigate the impacts of past and projected activities (see, for example, Diaz et al. 2004). To be effective, however, new initiatives and technologies will need to apply realistic conceptual models that are applicable, irrespective of habitat status, and sufficiently sensitive to detect deleterious ecosystem change at a functional level (Diaz et al. 2004).

Although the value of aquatic ecosystems and the services that they supply is considerable (e.g., Costanza et al. 1997; Chee 2004), it is likely that, in many cases, considerations of overriding benefit and disproportionate cost will result in the licensing of activities and developments, in spite of their potential to damage the ecological status of the water body. In such cases, the impact of habitat loss may be minimized via either mitigation (the act of making impacts less severe) or compensation (the act of compensating for economic, resource, or ecological loss via a payment or, in the latter cases, improvement, elsewhere). However, whether due to mitigation or compensation, habitats in different times, locations, and/or conditions will differ in nature and function from those that existed in the past. By creating, restoring, or re-creating habitats, the overall aim should be sustainable and long-term preservation of ecosystem integrity. Elliott and Cutts (2004) suggest that this implies the maintenance or improvement of an ecosystem's function or carrying capacity. Sorensen et al. (2004) suggest the creation of new habitat, which they term compensatory restoration, can result in a net increase of ecological services; i.e., the processes and goods provided by ecosystems and their component species that ultimately affect human well being.

By adopting this definition, however, the concept of the ecosystem is broadened. In addition to the direct provision of goods such as food, drinking water, and pharmaceuticals, ecosystems also provide the
purification of air, sediments, and water; breeding grounds for local and migratory species; biological control of pest species; stabilization of climate; mitigation of flood and drought; and many cultural values, such as recreational and inspirational values. While most ecologists individually have a good idea of what ecological services and carrying capacity are, the science in support of their definition and measurement is often limited (e.g., Covich et al. 2004), dependent on current knowledge and trends in their field of research (Raffaelli et al. 2003), and thus must be refined within a wider context if this concept is to inform management decisions (Elliott and Lawrence 1998; Elliott and Cutts 2004).

Population growth and global change will increasingly impact marine and terrestrial ecosystems in undetermined ways. What is certain, however, is that, as a significant proportion of the human population within Europe live adjacent to waterways, estuaries, and coastal areas, aquatic systems as a whole are particularly vulnerable to disturbance.

Thus, European management is starting to concentrate less on pollution per se and more on the combined affects of multiple stressors at the ecosystem level with a view to ensuring their protection, restoration, and long-term viability. While chemical pollution is increasingly controlled in industrialized Europe, issues of habitat loss are increasing in importance. Elliott and Cutts (2004) propose that much of this habitat loss can be viewed in terms of physical pollution, analogous to the chemical and biological pollution that is more commonly addressed. However, whereas chemical pollution is (arguably) soluble, given a developed technology and finance, and there are more or less standard methods of assessment (e.g., Apitz et al. 2005a, 2005b), less is known about how to correctly assess and manage the impacts of habitat loss and other disturbances on the integrity and carrying capacity of the system (Elliott and Cutts 2004).

Addressing risk at the basin and ecosystem scale, as demanded by the WFD and related directives, requires an understanding of economic, ecological, hydrological, and other processes across many spatial and temporal scales. However, at present, there is a poor understanding, for example, of how aquatic ecosystems function to accurately assess ecosystem health on a site-specific basis (Germano 2001). Adding several layers of complexity and scale will further complicate this problem. Ecological assessment needs to move away from applying simple statistical methods (see, for example, critique in Diaz et al. 2004) that do not allow for full analysis of complex systems toward an approach more analogous to that of medical diagnostics (Germano 2001; Elliott and Cutts 2004; Galloway et al. 2004b). Implicit in the drive toward holistic assessment is the concept that, to have healthy ecosystems, the constituent biota must, in the main, be healthy. This ideal has been greatly aided by recent rapid

advances in diagnostic molecular technologies that make it possible to conduct health assessments of individual organisms in much the same way that we evaluate human health (Depledge and Galloway 2005).

It has been stated that "the identification of minor changes due to anthropogenic activities against the background of large scale natural changes will remain difficult, and at the opposite end, the quality of greatly modified areas also will be difficult to assess" (Read et al. 2001). Ecosystem management requires the assessment and definition of baseline conditions so that we can distinguish between the effects of natural and man-made factors (Lewis 1999; DEFRA 2004), and properly assess liability and sources of stresses. Furthermore, as the science to establish links between morphological change and ecology is weak, much more work is required to establish causality (Freeman 2005). The EU FP6-funded Coastal Ocean Benthic Observatories (http://www.cobo.org.uk) program integrates in situ observational and experimental systems to monitor marine benthic habitats in order to understand how anthropogenic impacts affect benthic ecosystem functioning in support, among other things, of the development of biogeochemical measures to assist in the characterization of ecological function, status, and potential in coastal benthic ecosystems. There is a need for similar experimental systems in various aquatic habitats to provide better scientific understanding of mechanisms linking anthropogenic activities and the biological communities that may be the focus of protection, at various spatial, temporal, trophic, and organizational scales (Levin 1992; Apitz et al. 2005b).

For successful, but pragmatic, environmental management to be achieved, 6 tenets must be fulfilled: actions should be environmentally sustainable, economically viable, technologically feasible, legislatively permissible, administratively achievable, and last, socially desirable and/or tolerable (Elliott and Cutts 2004). In the present climate, a 7th tenet may be added—that the actions will be politically expedient. Examination of these 7 tenets reveal that the focus of these requirements is on societal perceptions, wishes, and needs. Thus, while this article has identified the scientific and technical needs necessary to support basin-scale ecosystem management, the success of this more holistic approach to environmental management will depend greatly on how effectively scientists, regulators, stakeholders, and society in general communicate. Thus, a final research need is the development of decision and communication tools that link this complex science to the needs of society.

### 1.4. CONCLUSIONS

The future should lead towards a better integrated environmental policy, directly linked to the knowledge of 'environmental interfaces' (including soil-sediment, but not only) and how this knowledge could be used to environmental protection ensure an efficient policy. Successful implementation of ecosystem management and strategic assessment will unprecedented require integration to an degree: integration of environmental objectives from the catchment to the coast and, ultimately, to European seas; of the various water and land uses, functions and values: of different skills and disciplines; of previous and emerging legislation and policy into common and coherent frameworks: of technical, socioeconomic and legislative instruments; of stakeholders in decision-making; of the different decision making levels, affecting ecosystem and water status and management among the Member States (Borja 2005 in press, Apitz et al. 2005b). This integration will require extensive collaboration and research to adapt current systems of environmental assessment and management to the basin and ecosystem level.

Clearly, we have been, and continue to, reduce contaminant inputs into the system. However, deposited sediments reflect the history of past and present point and diffuse pollution. Point and diffuse sources continue to enter the system from ongoing agricultural and industrial practices, catastrophic spills and accidents and changes in erosional and depositional patterns due to climate change and anthropogenic activities. These are not transient problems that will soon cease to clutter "normal" sediment management. Rather, we need to re-think the links between our goals and activities, over time.

Risk management thus also requires a careful consideration in another scale – time. Contamination (and thus risk) can result from many types of sources. Unless all these risk sources are managed, risks will continue and spread. We must, then, project risk over time under various scenarios (using CBMs). We must consider the implications of these trends on our management strategies, but also on our criteria and standards (Apitz 2004).

Where are we that we weren't ten years ago? We have come very far. It is generally accepted that we can only manage risks in sediments if we understand the dynamics of contaminant behavior. Extensive research has supported these goals. It is beginning to be accepted that sediment quality and quantity issues cannot be addressed in isolation. There are many types of risk, and sites and media interact. It is now also generally accepted that we cannot sustainably manage sediments site-by-site but must manage risk in basins. We have begun to develop conceptual approaches to this problem. National and international networks have been established to address these issues. However, there are still significant barriers to sediment management in the manner described above. While we have conceptual approaches to basin-scale management we still lack joined-up policies, uniform datasets and modeling tools. There is a need to stop separating dredging and cleanup, sediments, soils and waters in environmental management, but the current practice of "regulatory stovepiping" (Bridges and Apitz 2005) creates considerable barriers to progress. Clearly, agricultural and industrial policy, global change, and changing priorities will all affect whether we can achieve our environmental and economic goals. The there is a need for a more joined-up approach to sediment management. We must move from incremental change in sediment management strategies ("evolution"), to a "revolution" in which we understand how particles and contaminants move from the micro- to the macro-scale, from source to sink, historically and into the future, in order to sustainably manage risk to both the environment and our socioeconomic goals (Apitz et al. 2005d).

# 2. The SedNet perspective on contaminated sediment management

### 2.1. SEDNET

The European Sediment Research Network - SedNet - was commissioned by the European Commission Directorate General Research (contract No. EVK1-CT-2001-20002) in order to (main objective) set up a thematic network, initially aimed at the assessment of fate and impact of contaminants in sediment and dredged material and aimed at sustainable solutions for their management and treatment. Hence, between 2002 and 2005 scientific, policy and regulatory aspects of contaminated sediments and dredged material were addressed in 17 workshops and 3 conferences organized by SedNet. Europe's leading scientists and major sediment managers contributed to these SedNet activities. The results are summarized in this section, which is extracted from the SedNet booklet "Contaminated Sediments in European River Basins" (Salomons and Brils, 2004). The comprehensive results can be found at the SedNet website (<u>www.SedNet.org</u>) and will be published in 2006 by Elsevier as a series of four books.

Since March 2005, SedNet, through its founding members (see website) succeeded to continue as a self-supporting European Sediment Network. SedNet continues a European network for environmentally, socially and economically viable practices of sediment management on river basin scales. SedNet wants to be a network of sediment professionals (multi-stakeholders & scientists) and to be the independent platform to expert-advice on any issues related to (sustainable) sediment management

challenges, positioned between science/knowledge providers and endusers/stakeholders. The SedNet focus is on sediment quality and quantity issues on a river basin scale, including: estuarine/marine sediments, the origin of sediment (soil erosion) and re-erosion of (contaminated) sediments.

## 2.2. SEDIMENTS AND ITS VALUE

## 2.2.1. Sediment

Sediment is an essential, integral and dynamic part of our river basins. In natural and agricultural basins, sediment is derived from the weathering and erosion of minerals, organic material and soils in upstream areas and from the erosion of river banks and other in-stream sources. As surface-water flow rates decline in lowland areas, transported sediment settles along the river bed and banks by sedimentation. This also occurs on floodplains during flooding, and in reservoirs and lakes. Often the natural sedimentation areas are severely restricted, e.g. because of embankments and the loss of flooding areas as a result of these embankments. At the end of most rivers, the majority of the remaining sediment is deposited within the estuary and in the coastal zone. Natural river hydrodynamics maintain a dynamic equilibrium, regulating small variations in water-flow and sedimentation by re-suspension and resettlement. In estuaries, sediment transport occurs both downstream and upstream, mixing fluvial and marine sediment as a result of tidal currents.

# 2.2.2. *Its value*

Sediment forms a variety of habitats. Many aquatic species live in the sediment. Microbial processes cause regeneration of nutrients and important functioning of nutrient cycles for the whole water body. Sediment dynamics and gradients (wet-dry and fresh-salt) form favorable conditions for a large biodiversity, from the origin of the river to the coastal zone. A healthy river needs sediment as a source of life. Sediment is also a resource for human needs. For millennia, mankind has utilized sediment in river systems as fertile farmland and as a source of construction material (Table 2.1).

Too much sediment	Too little sediment	Sediment as a resource
Obstruction of channels	Beaches erode	Construction material
Rivers fill and flood	Riverbanks erode	Sand for beaches
Reefs get smothered	Wetlands are lost	Wetland nourishment
Turbidity	River profile degradation	Agricultural soil enrichment

Table 2.1. Overview of sediment as a resource (Martin, 2002).

#### 2.3. CONTAMINATED SEDIMENTS IN EUROPEAN RIVER BASINS

### 2.3.1. Contamination

Sediment acts as a potential sink for many hazardous chemicals. Since the industrial revolution, human-made chemicals have been emitted to surface waters. Due to their properties, many of these chemicals stick to sediment. Hence in areas with a long record of sedimentation, sediment cores reflect the history of the pollution in a given river basin. Where water quality is improving, the legacy of the past may still be present in sediments hidden at the bottom of rivers, behind dams, in lakes, estuaries, seas and on the floodplains of many European river basins. These sediments may become a secondary source of pollution when they are eroded (e.g. due to flooding and channel bank erosion) and transported further downstream.

Along the course of the river to the sea, transportation, dilution and redistribution of sediment-associated contaminants occurs. Many relatively small inputs, all complying with emission regulations, accumulate to reach higher levels by the time sediment reaches the river delta. In the estuary, uncontaminated marine sediments are mixed with contaminated fluvial sediments. This natural 'dilution' decreases contamination level in a gradient towards the sea over short distances, but does not alter the actual transported quantity of contaminants.

Despite regular sediment quality assessment by member states, a reliable estimation of the overall amount of contaminated sediment in Europe is hard to give. The main reason for this is the absence of uniformity in sampling methods, analytical techniques and applied sediment quality standards or guideline values. This causes a lack of intercomparability. Typically, countries along the same river basin use different methods.

### 2.3.2. Adverse effects of contamination

Contaminants can be degraded or fixed to sediment components, thus decreasing their bioavailability. At a certain level, contaminants in sediment will start to impact the ecological or chemical water quality status and complicate sediment management. In the end, effects may occur such as the decreased abundance of sediment dwelling (benthic) species or a decreased reproduction or health of animals consuming contaminated benthic species. Contaminated sediments remain potential sources of adverse affects on water resources through the release of contaminants to surface waters and groundwater. Furthermore, contaminated material, e.g. in the case of dredging, is several times more expensive than handling clean material.

Clean sediment can also have environmental and socio-economic impacts. For instance, turbidity and excessive sedimentation have a physical effect on benthic life, too much sediment in navigation channels requires costly dredging, and sedimentation behind dams decreases the economic lifetime of that dam. Furthermore, dams decrease the supply of sediment needed to support downstream wetlands, estuaries and other ecosystems. SedNet focused on contamination issues, rather than on such sediment quantity issues.

### 2.3.3. Assessment of contaminated sediment

For the assessment of contaminated sediment, there is not one 'best' method available. Each specific management question requires a tailormade solution. Chemical analysis can be used to determine concentrations of selected hazardous chemicals and then it can be checked if the concentrations exceed pre-defined standards or guideline values. Using bioassays can test the toxic effects of sediment on organisms. Through a field inventory the long-term impact on sediment biota can be investigated. These assessment methods (chemical, bioassay, field) are complementary by giving a unique answer that cannot be given by any of the individual methods by themselves. But each method also has its own unique drawbacks and uncertainties. A simultaneous or tiered application of these three, complementary assessment methods (Figure 2.1) is commonly referred to as the Triad-approach (Chapman 1996).



*Figure 2.1.* An elaboration of the sediment Triad-approach, positioning hazard, risk and impact assessment and positioning the bridging tools Toxicity Identification Evaluation and Model Ecosystems.

### 2.4. SEDIMENT MANAGEMENT

## 2.4.1. Dredged-material management

Many water and port managers face the continuous effort of dredging in order to maintain the required water depth. Europe-wide, the volume of dredged material is very roughly estimated at 200 million cubic meters per year. There are three types of dredging: capital, maintenance and remediation dredging. Capital dredging is for example for land reclamation, deepening fairways, etc. Maintenance dredging is mainly to keep waterways at a defined depth to ensure safe navigation, and remediation dredging is to solve environmental problems of contaminated sediments. Contamination mainly leads to problems in maintenance dredging because given standards or regulations do not allow the free disposal in the aquatic system. In general with capital dredging old, uncontaminated sediments are being dredged. There is a lot of information on the different options of dredged-material management available.



*Figure 2.2.* The Catchment-Coast continuum and some of the upstream and downstream regulations and policies affecting sediment quantity and quality as well as those relevant for impact assessment downstream.

# 2.4.2. Legislation and guidance

Dredging is mainly done in the coastal or marine environment. Therefore, international guidance has existed for many years to minimize the ecological effects of dredging and open water disposal. European legislation for handling dredged material is complex, because dredged material is at the borderline of water, soil and waste policies. European

22

legislation affects the management of dredged sediments in upland areas, like the waste legislation (Directive 75/442/EEC; Decision 2000/532/EC), especially the EU Landfill Directive (Directive 1999/31/EC), and possibly the Habitats Directive (Directive 92/43/EEC), amongst others. These EU legislations do not (as yet) deal adequately with sediment (Figure 2.2).

The EU Water Framework Directive (WFD)(Directive 2000/60/EC), in force since the year 2000, does not specifically address sediment management. But it can be a tool to tackle the sources of sediment contamination. It offers an opportunity to further improve our knowledge about the relation between sediment quality and water quality and to harmonize quality assessment and sediment management on a river-basin scale.

### 2.4.3. Sediment management challenges

Sediment and dredged-material management challenges and problems relate to *quality* and *quantity* issues. Quality issues relate to contamination, legislation, perception, risk-assessment, source control and destinations of dredged material. Quantity issues mainly relate to erosion, sedimentation, flooding, the effects of damming and the resulting morphological changes downstream. Often quantity and quality aspects are interrelated: the overall umbrella is the river basin.

### Quality issues

The management of contaminated sediments in Europe has been mainly the direct concern of authorities dealing with navigable waterways. Contamination can inflict severely the management of dredged sediments. The costs for the removal of excess sediment increases when it is too contaminated for unrestricted relocation. Port managers are concerned that they have to bear the extra costs for managing contamination which is derived from contributions along the river basin. The 'polluter pays' principle is far from being applied. The problem is left for the problem owner and there is no link to those that have caused it.

Besides complicating dredging activities per se, contaminated sediment may pose ecological risks or risks to water quality. The relation between sediment quality and risks is complex and site specific, requiring assessment methods based on bioavailable contaminant fractions and bioassays rather than results based on the traditional total contaminant concentrations. However, if sediment quality impairs the chemical or ecological status, remediation measures may be needed. So far, only in a few Member States has contaminated sediment been managed due to its impact on the ecological quality of water bodies.

An integrated approach for sediment management is presently lacking. The WFD aims at source reduction, which in the long term may lead to 'clean' sediment quality. Next to the emissions of point and diffuse sources, a source of increasing importance in this respect is historic contamination, i.e. our legacy of the past. The diffuse sources in which such contamination is present in many European basins are becoming increasingly important. Even more now, since the risk of extreme river floods, that may wash the hidden pollution into the water system once again, seems to have been underestimated in the past.

# Quantity issues

Quantity aspects were not a predominant part of SedNet activities. However, they were addressed in several of the workshops since they influence the flux of (contaminated) sediments in river basins. A selection of the issues which were discussed and which have to be taken into account in basin-wide management are:

- the use of sediments in river basins for construction materials with the result of reduced sediment supply downstream, river bed incision and the associated impacts on infrastructure (e.g. bridges etc.);
- changing land use and effects of increased erosion on agricultural soils in particular;
- the effects of damming, reducing sediment supply downstream and resulting in morphological changes to floodplains and deltas;
- damming and the temporal storage of upstream pollutants in deposited sediment (legacy of the past), with the consequence of further transport downstream through erosion events;
- flood control measures, including controlled flooding of areas adjacent to the river, impacting sediment budgets.

Last but not least, climate change and its impacts on the hydrology at the river-basin scale will affect sediment fluxes and should be anticipated in a sediment management plan.

# 2.4.4. Management options

Costly end-of-pipe solutions may be unavoidable for sediment and dredgedmaterial management (Figure 2.3). Solutions like relocation into the aquatic system or placement on river embankments are the first options to consider, since they bring the sediment back to where it belongs. But these solutions are only acceptable if the contamination is below given standards. Depots for contaminated dredged material can be an option in this situation, but they are expensive, often lack public acceptance and are subject to complex legislation. Alternatives include treatment for beneficial use and controlled (confined) disposal. Treatment and re-use is politically encouraged, but is currently applied only at a small scale because of the higher costs compared



*Figure 2.3.* Treatment and disposal costs. The height of the columns represent average and the bar shows the range of costs (Netzband *et al.* 2002). Disposal options on the left and to the right treatment options vary in wide ranges due to different boundary conditions. (Note: Revenues or costs for application of the products are not included. The costs for sand-separation do not include dewatering and/or disposal; CDF = Confined Disposal Facility; LWA=Light Weight Aggregate; chem = chemical; immo = immobilization).

to disposal and the lack of product markets. However, in some cases treatment and beneficial use may be a competitive alternative for confined disposal. Confined disposal will remain the first choice solution for the time being. For the realization of new confined disposal sites (both upland and sub-aquatic), public involvement and support are needed.

In many cases the procedures are very time consuming (10-15 years) and/or the lack of public acceptance can complicate matters and their implementation.

#### 2.5. SEDNET RECOMMENDATIONS

### 2.5.1. Main recommendations

As a result of the activities of SedNet, and especially the workshops and conferences, SedNet has developed the following main recommendations:

Towards European policy development:

Further develop and eventually integrate sustainable sediment management into the European Water Framework Directive and related policy and legislation. SedNet viewpoint on sustainable sediment management is further described in the next section. Towards sediment management:

Find management solutions that carefully balance social, economic and environmental values and are set within the context of the whole river system,

## Towards research:

Improve our understanding of the relation between sediment contamination (hazard) and its actual impact to the functioning of ecosystems (ecological status) and develop strategies to assess and manage the risks involved.

# 2.5.2. Towards sustainable sediment management

Sediment issues occur in temporal (geological and seasonal cycles) and spatial scales (catchment area, river foreland, polder) which cross political and administrative boundaries. Yet boundaries tend to scatter sediment management responsibilities and in the end no single stakeholder or country feels responsible. Planning sediment management at the river-basin scale will urge co-operation between agencies and even countries.

At that scale, joint methods and strategies should be developed for sediment and dredged-material management that link to the EU WFD and to pilot projects on trans-boundary rivers. Such methods and strategies should preferably be shared between different basins so that we can learn from each other. It will also help to recognize the differences between basins. This will underline the need to develop tailor-made, realistic solutions towards the environmental and socio-economic management issues that are at stake at that specific basin, or more detailed solutions at specific sites in that basin. For instance the type of dredged material, and the type of contamination, varies considerably between basins and between specific sites within a basin.

Thus sediment and dredged-material management needs to be integrated into existing frameworks at this scale, such as river-basin management plans. An integrated approach is needed from inland (upstream) to coastal waters (downstream). This approach should respect the national and EU policy targets and comply with legislation.

Other recommendations towards sustainable sediment management are:

Find solutions in increase of the interaction with stakeholders. The perception of sediment depends on a variety of roles, values and definitions and is influenced by stakeholder interests. In order to maintain a dialogue, definitions and terms used to describe sediment must be neutral and allembracing, and sympathetic to stakeholder values and views.

Intervene in such a way that it does not result in unwanted impacts elsewhere in the river basin (up- or downstream), and should not have an adverse impact in the future. A basic understanding of the water system, its dynamics and of the functions of the bordering areas (populations, industries, agriculture) is essential for an effective decision-making process.

Look for integrated solutions that embrace the whole soil-water system. Sediment is part of the soil-water system. Management of sediments should be planned in the context of the soil-groundwater-water-sediment system.

Look for solutions that respect natural processes and functioning. Management strategies for sediment should respect nature: working with nature, not against it. Thus it is crucial to use and improve our understanding of river system functioning and the role of sediment in the processes involved (see further under research recommendations). For instance, taking sediment out of the system can cause sediment deficits resulting in habitat loss and destabilization of river system functioning. Therefore, sediment management must also consider the sediment balance and its dynamic role in the hydrological and geomorphologic processes operating within each river.

# 3. A framework of risk assessment and management for sediment reclamation in the Venice Lagoon

Persistent and toxic pollutants, as well as nutrients, of both industrial and municipal origin are distributed over the whole Venice lagoon sediment, and heavily contaminated hotspots were identified in the innermost, industrial harbour canals. Taking to account the Water Framework Directive and national/local regulations, the DPSIR framework, proposed by European Environmental Agency, was implemented to support the decision-making and management process. Driving forces, Pressures, and State were investigated through several research investigations and monitoring programs, including: analysis of well-known persistent pollutants as well as new classes of emerging pollutants (e.g. endocrine disrupting compounds); development of fate-transport and bioaccumulation models; application of specific bioassays and biomarkers. The research and monitoring results were integrated into an Environmental Risk Assessment to obtain a detailed investigation and evaluation of Impacts. Technological Responses were tested in specific pilot studies including seven individual/combined remedial techniques for heavily contaminated sediment (chemical stabilization by solidification, thermal desorption, solvent extraction, immobilization of heavy metals employing sulfatereducing bacteria, phytoremediation, metal high gradient magnetic separation and vitrification). Finally, in order to support the decisionmaking and management process, two GIS based Decision Support Systems (DSS) were defined, integrating the aforementioned tools, methodologies and experimental and technological activities. The GIS based DSS, called DESYRE, was designed for the rehabilitation of contaminated soils in the industrial area surrounding the Venice lagoon; while a specific GIS based DSS is currently being developed for the monitoring and management of Venice lagoon's environmental quality.

# 3.1. REGULATORY FRAMEWORK

# 3.1.1. European regulatory frameworks of environmental quality

There are a number of international and national conventions and regulations dealing with sediments, and in most cases with its quality (e.g. for dredging operations) such as London Convention revised in 2000, Oslo-Paris Convention revised in 2004 and Helsinki Convention (1992). Their purpose and primary aim is the environmentally sound disposal (relocation) of dredged material into the sea. Due to national implementation of international convention and EU Directives, the European member countries have developed special dredged material guidelines with different (limited) competences in practice (Sednet, 2004).

The recently adopted Water Framework Directive (WFD) (2000/60/EC) is an attempt to provide a more integrated approach to water management at the river basin level. Although the WFD provides a list with priority substances, sediment fluxes are not explicitly included, even though sediment fluxes have significant impacts on water quality. The WFD requires the adoption of specific measures to progressively reduce discharges, emissions and losses of priority hazardous substances (Decision 2455/2001/CE Annex X). This provision can help tackling existing pollution sources in European river Basins to reduce ongoing sediment contamination.

# 3.1.2. National and local regulatory priority substances frameworks for sediment reclamation in Venice Lagoon

Nowadays, in Italy, there is not a national legislation dedicated to sediment management. The only regulatory references are the decree (D.M. 24/1996) regulating the sea dumping of dredged material and the decree (D.M. 367/2003) implementing the WFD and the following Decision 2455/2001. In the D.M. 367/2003, quality standards are given for 26 priority substances likely to be found in sediment

Being the Venice Lagoon regulated by a special low, Sediment Quality Criteria (SQC) were enforced since 1993. They are based on the levels of contaminants (heavy metals, polycyclic aromatic hydrocarbons, total hydrocarbons and organohalogen compounds). These SQC are comparable with those in use in other European countries and with the criteria used in the USA. Basically, there are three classes (A, B, C) of sediments that can

be used inside the lagoon with increasing caution depending on the level of pollutant concentrations: A and B classes can be reused for the restoration of lagoon morphology, C class needs to be disposed of in confined facilities within the lagoon border. Pollutants exhibiting sediment concentrations higher than those of class C (also referred as  $C^+$  concentrations) cannot be reused within the Venice lagoon and must be landfilled after adequate treatment. The amount of sediment to be removed is quite relevant since it is asked by both purposes navigation maintenance and environmental restoration.

Moreover, in order to reduce entry of pollutants into the lagoon, water quality objectives were enacted (D.M. Ronchi-Costa 23/04/1998) together with maximum allowable loads from the catchment basin and treated effluents discharging directly into the lagoon.

# 3.2. IMPLEMENTATION OF THE DPSIR FRAMEWORK FOR THE VENICE LAGOON'S SEDIMENT MANAGEMENT

The sediment management in the lagoon of Venice was based on the DPSIR scheme (Figure 3.1), proposed by the European Environmental Agency (EEA, 1995) that includes the characterization of Driving forces (i.e. anthropogenic or natural factors triggering the environmental development such as population, industry, or tidal cycles), Pressures (i.e. how a driving force act on the environment: e.g. atmospheric emission, effluent discharge), State (i.e. result of pressures applied to biotic and abiotic resources), Impacts (i.e. alterations resulting from the effects of pressures on the state), and Responses (i.e. human actions on driving forces and/or pressures and/or state to reduce or eliminate the impacts).



Recently, a set of projects and monitoring activities were carried out in the lagoon of Venice, in order to identify the Pressures and to characterize the environmental State. Figure 3.2 shows a schematic summary of the investigations carried out in the lagoon of Venice over the last years.

As far as Pressures are concerned, two large projects were undertaken to quantify

Figure 3.1. DPSIR scheme (EEA, 1995).

the loads of organic and inorganic pollutants, as well as of nutrients, in the lagoon. In particular, the "DRAIN project" (MAV-CVN, 2001) investigated

the annual amounts of nutrients (i.e. nitrogen and phosphorus compounds), and organic (i.e. PCBs, PCDD/Fs, HCB, PAHs) and inorganic (i.e. arsenic, cadmium, copper, chromium, lead, mercury, nickel, zinc) chemicals from the catchment area (i.e. river loadings) to the lagoon. The contribution of atmospheric depositions of organic and inorganic pollutants, as well as nutrients, was estimated by the "2023 project" (MAV-CVN, 2000). Moreover, the temporal trends of dioxins (PCDD/Fs) sources were investigated both from experimental data and by applying fugacity models (Dalla Valle et al., 2005a and 2005b).



*Figure 3.2.* Schematic representation of monitoring and research activities carried out in the lagoon of Venice in the last 10 years (MAV-CVN, 2000).

As far the State of the lagoon is concerned, the spatial distribution of organic and inorganic pollutants in the surface sediment and in aquatic organisms was studied by the "MAP project" (MAV-CVN, 1999). Others projects investigated the distribution of PCDD/Fs between environmental compartments (e.g. atmosphere, sediment, water and organisms) (Dalla Valle et al., 2003). Emerging pollutants such as endocrine disrupting chemicals (EDC), especially estrogenic compounds, and pharmaceutical residues, were measured for the first time in sediments, water and organic and inorganic pollutants over the whole lagoon were monitored within the MELA project (MAV-CVN, 2005).

# 3.2.1. Role of environmental risk assessment within the DPSIR framework for the Venice Lagoon

Based on the Status analysis, the Impacts and Risks for the lagoon of Venice were investigated. In particular, the Ecological Risk Assessment (ERA) procedure (US-EPA, 1998) was applied to identify the chemical stressors of concern and the lagoon areas affected by the higher risk for the ecological resources, with the aim to address the monitoring activities and remediation actions. According to US-EPA procedure, the conceptual model of the Lagoon of Venice (Critto and Marcomini, 2001), defined in the Problem Formulation phase, allowed the selection of three risk hypotheses associated with the accumulation of persistent and toxic pollutants in the lagoon sediments. These three risk hypotheses focused on ecological entities such as the benthic community, the edible clam Tapes philipinarum, and the lagoon's food web (Critto and Marcomini, 2001). In order to characterize the potential risk to the benthic community, a screening ERA based on the quotient method (Jones et al., 1999) was developed (Critto et al., 2005). The exposure to different classes of pollutants (metals, chlorinated organic compounds and poly-nuclear aromatic hydrocarbons) was investigated in more than 100 sediment stations, distributed across the whole lagoon, and a map of the spatial distribution of the estimated risk for the benthic community was drawn (Figure 3.3).

Then, the ecological risk associated with bioaccumulation of both total and dioxin-like polychloro biphenyls (PCBs) and several inorganic pollutants (i.e. As, Cd, Cr, Cu, Hg, Ni, Pb, Zn) in clam Tapes philipinarum was estimated (Micheletti et al., 2004). A GIS-based map of risk was obtained comparing the pollutants bioaccumulation, estimated by regression models, with internal effects concentrations estimated according to the Tissue Screening Concentrations approach (Shephard, 1998). As example, Figure 3.4 shows the risk for clam associated with cadmium bioaccumulation, one of the main chemical of concern for clam. Finally, an Ecological Risk Assessment for the aquatic food web of the Venice lagoon was carried out, estimating risk posed by the bioaccumulation of organochlorine compounds (i.e. PCBs and PCDD/Fs) through the food web (Micheletti et al., 2005): bioaccumulation was estimated by means of a kinetic Food Chain model (Gobas, 1993) applied to a food web model representative for the Venice lagoon (Libralato et al., 2002). Moreover, the risk posed by the observed concentrations of EDCs in water for biota was analyzed and occurrence of a potential risk for the 24% of the water samples collected in the canals of the city of Venice was highlighted (Pojana et al., 2004).



Figure 3.3. Map of risk index spatial distribution for the benthic community (the risk was estimated through the quotient method and the ecotoxicological TEL benchmark). Each risk class represents the number of pollutants exceeding the corresponding TEL benchmark value. The lagoon zones characterized by higher and lower risk class are marked as I and II, respectively (Critto et al., 2005).

*Figure 3.4.* Map of the spatial distribution of risk (i.e. Hazard Quotients) posed by cadmium to clam (*Tapes philipinarum*) (Micheletti et al., 2005).

3.2.2. Responses by sediment remediation technologies: SeRTech project

With reference to DPSIR scheme, the Responses step (i.e. human actions undertaken on driving forces and/or pressures and/or state to reduce or eliminate the impacts) is a relevant aspect of the sediment management. Recently, the research project SeRTech (Sediment Remediation Technologies), funded by the Italian Ministry of Education and Scientific and Technology Research (MIUR), has been developed by the Venice Research Consortium (CVR) in order to define a technological competitive offer to the reuse/recycling/disposal of contaminated dredged sediment. The main objective of SeRTech project was to select a series of suitable

treatment techniques for the most contaminated dredged sediments of Venice industrial canals (over 1.5 million  $m^3$  classified as C<sup>+</sup>, see previous section) and to develop an integrated cost-effective treatment system to address the high and heterogeneous contamination compounds (heavy metals, polycyclic aromatic hydrocarbons and organo-cloride) and matrixes, such as those of the Venice lagoon. The re-use and economical potential of treatment by- and final-products has been assessed.

Due to heterogeneous contaminations and matrices of dredged sediments, seven treatment techniques (Figure 3.5) were selected over a large array to perform the laboratory tests (Pippa et al., 2003).



Figure 3.5. Selected remediation processes tested in the SeRTech project.

Two representative macro-samples (4 tons each one), were collected from one of the most polluted industrial canal: they were characterized and used to perform laboratory scale experiments.

Sediments characterization has included: full characterization of pollutants (heavy metals, organic compounds including PCDD/Fs and TBT), physical properties (density, organic matter content, water content, etc), mineralogical analyses, morphological characterization and metal speciation, microbiological characterization, various leaching tests, Microtox test, thermal analysis and other parameters such as nutrients content, salinity, sulfide content.

The assessment of individual treatment techniques (flux and material balance for the most representative contaminants, removal efficiency, energy balance and cost analysis) has led to the definition of an integrated system of technologies (Figure 3.6) to treat both organic and inorganic contaminants, which is based on the material ranking: Low Risk, Medium Risk and High Risk sediment.



Figure 3.6. The integrated system proposed in the SeRTech project.

The integrated system proposed provides three different technological lines according to different types of contamination level, with a innovative vitrification used as final step only for "High Risk" classified sediment. Different tests were performed to verify this integrated treatment system. experimentation led to an Italian Patent (patent number The VE2004A000050). The final products (glass and pellets) obtained from integrated system were characterized (mechanical properties, leaching tests) in order to verify their reuse in the building industry (i.e. coating materials), roadbeds, etc. A life cycle assessment (LCA) approach was also adopted to compare the different techniques, including environmental benefit and impacts, economical costs and social acceptance criteria.

### 3.3. DECISION SUPPORT SYSTEMS (DSS) DEVELOPMENT FOR DECISION MAKING AND MANAGEMENT PROCESSES

In the previous paragraphs, tools and methodologies for impacts analysis (i.e. application of environmental risk assessment) and for defining some technological responses (i.e. SeRTech project), have been presented. Nevertheless, more integrated tools are needed in order to assist the definition and planning of responses that deal with the identified pressures and impacts on the ecosystem.

A major current scientific and technical task is to provide the involved authorities with all the relevant information in a manageable format and easy-to-use structure, which integrates technological, economical, social, and political issues and includes sediment management in a more general framework. The development of GIS-based Decision Support Systems, identifying realistic choices and integrating different categories of information into a coherent framework suitable for analysis and selection of alternatives, has proved to be the most effective way (CLARINET, 2002). To this respect, two research projects were developed: one focusing on the definition of environmental quality indices for the Venice Lagoon and one for the definition of site-specific remediation strategies for contaminated sites. The first project was funded by Consortium for Lagoon of Venice Research (CO.RI.LA.), and involves the University of Venice, University of Padua and National Research Council. In order to support the management phase of the proposed DPSIR framework and guide the monitoring process, environmental quality indicators and indices were developed by integrating the information obtained through the application of the DPSIR framework. Exposure and effects data (i.e. lines of evidence) were integrated in the Weight of Evidence and TRIAD approaches, that allowed the formulation of specific indicators/indices of ecological status/integrity.

Within the second research project, funded by the Italian Ministry of Education and Scientific and Technology Research (MIUR), the DESYRE (DEcision Support sYstem for the REhabilitation of contaminated sites) DSS has been developed by the Venice Research Consortium (CVR) in collaboration with the University Ca' Foscari of Venice, Thetis Spa and CNR-ISE. DESYRE is a GIS-based Decision Support System, which assists the planning and the definition of site-specific remediation strategies for contaminated sites, by integrating environmental, socio-economic and technological information (Carlon et al., 2003). DESYRE, through a userfriendly interface, is addressed to two users groups: one is the multidisciplinary team of experts (such as risk assessors, socio-economists and technology engineers), who are facilitated in their evaluation work; the other one is the group of several stakeholders (such as public decision makers, planners, site owners, investors and associations), who participate to the evaluation and comparison of alternative remediation options. These options, or scenarios, are described by the system in terms of indices related to technological solutions, risk reduction, cost, time, environmental impacts and socio-economic benefits. As in Figure 3.7, the main interface of the system identifies the subsequent analytical steps. In fact, DESYRE DSS is structured into 6 modules, 5 analytical modules (socio-economic, characterization, pre- and post- risk assessment, technological) and a decision module. The Socio-economic Module addresses the socioeconomical constraints through a Fuzzy Logic analysis, until the selection of the best land use (Facchinetti et al., 2003). The site Characterization Module supports the analysis of contaminants spatial distributions by using geo-statistical methods, i.e. variography and Kriging. The Risk Assessment Module is divided into two phases: pre-remediation and post-remediation. In the Pre-remediation phase, an original procedure, based on RBCA methodologies (ASTM, 1998), allows to assess and visualize the spatial distribution of risks posed by contaminants in soil and groundwater. providing a risk-based zoning of the site. The Technological Module allows the definition of the technological solutions and the planning of interventions for the whole site. The system guides the user to the preliminary selection of technologies from a wider database and provides a comparison (through MCDA methodologies) of the selected technologies that support the experts in the definition of several technological sets (Carlon et al., 2004). In the Post-remediation Risk Assessment Module, a simulation of applied technologies provides residual risk maps with related uncertainty maps. Finally, in the Decision Module, alternative remediation scenarios are described by a set of indices. The indices represent the multiple factors derived by the previous analysis: risk, technological, socioeconomic, environmental impact, cost and time. The scenarios can be compared and ranked by using Multi-Criteria Decision Analysis methodologies, and presented to the relevant stakeholders in the form of histograms (Figure 3.7).



# SPECIFIC TOOLS: DESYRE

Figure 3.7. DESYRE DSS main deliverables.

DESYRE software was tested successfully in two areas (approximately 450 and 43 ha wide, respectively) of the mega-site of Porto Marghera (Venice, Italy), a large industrial contaminated site of national interest. The application to the different areas verified DESYRE potentialities in dealing with contamination problems at different scales.

### 3.4. CONCLUSION

In the near future, efforts will be oriented to obtain a Sustainable Management Support System for the lagoon of Venice capable to: integrate Environmental Risk Assessment and Management, improve stakeholders participation to the Risk Assessment process, reduce the uncertainty associated with the management decisions, and to increase the communication between the actors concerned with the lagoon management and the risk assessors. In order to develop such a type of Decision Support System (DSS), environmental quality, socio-economic and technological indices/indicators, need to be identified and quantified.

# 4. River and lake sediment contamination and related legislation in Slovakia

### 4.1. INTRODUCTION

Slovakia is a country without water bodies of considerable sizes. The country is without sea coast and big natural lakes, the biggest river in Europe the Danube is touching the southwestern border with Austria and Hungary on the length not bigger than 130 km. Following this information and considering the lack in environmental legislation in connection with sediment contamination, we could easily believe that issues concerning sediment contamination in Slovakia are purposeless. In despite to this Slovakia has a relatively dense smaller river and mountainous stream network because the dissected relief. The total water table area is increased by several dams.

Types of the contaminating chemicals are determined by contaminant sources. Even the anthropogenic pollution sources are very important; we have to take in account the geological background. The geology of the Western Carpathians occupying the bigger part of Slovakia is extremely various what have a very strong influence on the river and lake sediment composition. Regarding anthropogenic sources we may distinguish between industrial sites acting as pollution point sources and extensive agriculture using huge amount of chemicals during the time period between the 50's and 80's of the 20<sup>th</sup> century, acting as area pollution sources. The industry in Slovakia is more-less diverse, but in the second half of the 20<sup>th</sup> century

the dominating potential point sources became huge industrial sites producing chemicals and metal producing and processing enterprises. Maybe the oldest industry in Slovakia was metal ore mining and processing, especially in the area of the Spišské Rudohorie Mountains (see Figure 4.1). Consequently the main sediment contaminating species are different persistent organic pollutants, like chlorinated pesticides and PCBs as well different heavy metals like arsenic, mercury, lead, etc.



*Figure 4.1.* position of the Slovenské rudohorie Mts. and the area discussed more in details (see the square).

### 4.2. SLOVAK LEGISLATION RELATED TO SEDIMENT CONTAMINATION

At the present time neither act nor decree is dealing with sediment contamination issues in Slovakia. For this reason are often used limits adopted from abroad for this reason.

The only document dealing with sediments issued by a governmental institution is the Guideline No. 549/98-2 for the Contaminated River and Lake Sediment Assessment Methods. This document was issued by the Ministry of Environment of the Slovak Republic in August 27th 1998. The goal of this document is to unify risk assessment methods and analyses. Particular procedures in this guideline are forming an integrated system consisting of the following components:

sediment sampling,

chemical analyses of the sediment,

ecotoxicological tests,

biological analyses of the macrofauna living on the water bodie's bottom,

the highest priority contaminants selection.

The guideline is recommended to apply for:

watercourse and lake sediments pollution degree assessment,

further sediment pollution prevention leading to overreach acceptable environmental and health risks,

inventarisation of the sedimentation areas' contamination degree,

identification of the excessively contaminated areas, namely those which are causing endangering of the water ecosystems and people's health,

monitoring and surveying of sites with contaminated sediments.

This guideline was designed as a complex document dealing with data gathering, sediment sample's processing and analyses, ways and procedures of data processing and interpretation. Within the certain time period the document will be expanded by guidelines for particular scenarios, e.g.:

type of the contaminating species (e.g. heavy metals, pesticides, oil species)

type of the contamination source (point and area contamination sources, atmospheric deposition)

sediment use (recreation, protected areas, fishing, sediment exploitation ..) endangered target groups (water fauna and flora, predators, people ...)

The assessment methods are defined for the national level and for site specific risk assessment. On the *national* level the method are based on the concetration estimation in the sediment (*PEClocalsed*) and it's comparison with the *PNEC* = Predicted No Effect or with the *SQC* = Sediment Quality Criteria for the given species. The output is the relative risk score (*RQ*). The *Site Specific Risk Assessment* is based on chemical analyses, ecotoxicological test and the biological quality assessment.

# 4.3. OVERVIEW OF ARTICLES DEALING WITH POLLUTED SEDIMENTS IN SLOVAKIA

Several studies were published about sediment pollution in Slovakia. Most of them are related to particular case studies, especially related to river dam sediments, e.g. the Palcmanská Maša (Brehuv, 2002) or more case studies about the Šírava dam (see the next article in this book).

Distribution of the contaminants in sediments all over Slovakia is described in regional works, e.g. the distribution of the pesticides was elaborated by Korenková and Slobodník (2001), the relationships between the suficial water and sediments in the Slovak watercourses was evaluated by Hucko (1994).

The most comprehensive description of metal distribution in stream sediments all over Slovakia is provided by the Geochemical Atlas of Slovakia, Part Stream Sediments (Bodiš and Rapant, 1996), published by the Dionýz Štúr Geological Institute in Bratislava. The Atlas was published together with the Geochemical Atlases for groundwater, soil, forest biomass, rocks and natural radioactivity. The Atlas contains maps of distribution for the following elements: Li, Na, K, Rb, Be, Mg, Ca, Sr, Ba, Y, Ti, Zr, V, Cr, Mo, W, Mn, Fe, Co, Ni, Cu, Zn, Cd, Hg, B, Al, Ga, Tl, Sn, Pb, P, As, Sb, Bi, Se. These elements were analyzed in the stream sediment samples of fraction minus 0.125 mm, mineralogically containing clay, primary silicate minerals, rocks, oxides and hydroxides. The sampling density was 1 sample/2km<sup>2</sup>, what means 24 432 samples, sampled during the time period 1991 – 1995.

### 4.4. CONTAMINATION SOURCES

As I mentioned in the introduction, as important contamination sources we are considering chemical production sites and metal processing and production sites, including metal ore mines.

The rapid changes in the political system and economy in the former Czechoslovakia brought also changes in the industrial production. Typical features of those changes were a shift from the prevailing heavy industrial production to smaller enterprises, the overall production rate dropped also. The pesticide consumption within agricultural production was dropping dramatically due the insolvency of the farmers. Due those circumstances the total influx of contaminants to sediments dramatically decreased, however the accumulated amount of persistent pollutants decreases very slowly.

Polluting species originating from the chemical industrial sites are diverse. The most serious sediment pollution case is the huge PCB pollution of sediments in the Šírava dam in Eastern Slovakia. The pollution source is a huge chemical industrial site owned by the Chemko Company in the city of Strážske. This case is described in details in another contribution of this book.

The metal ore deposits in Slovakia are located in more regions, but the most important region is the Slovenské rudohorie Mountains (Slovak Ore Mts.), especially its easternmost part, also called the Spiškogemerské rudohorie Mts. Following the earliest written evidences the mining activities started in the 12<sup>th</sup> century (Chalupecký, 1981). Most of the metal ore mines were closed and flooded in the beginning of the 90's, consequently the production of the metal production and processing plants bound to the metal ore mines decreased dramatically and reduced mainly to scrap processing and in some cases to ore processing imported from abroad. Those events also caused decreasing of the heavy metal flux in to sediments. However, if we take in consideration that the mining activities

started at least 800 years ago, the sediments contains a huge amount of heavy metals accumulated during this time period.

The character of the mine pollution is strongly bound to the geological background. The deposits are characterized mainly as hydrothermal ore veins, mineralogically built up by siderite, calcite, quartz, baryte and different sulphide minerals, especially by pyrite, chalcopyrite, tetrahedrite, locally occurring also cinnabarite.

# 4.5. SELECTED HEAVY METAL POLLUTION SOURCES DESCRIPTION

As representatives of typical heavy metal pollution point sources in the Slovenské Rudohorie Mts. three source groups were selected:

- **Rudňany** iron and mercury ore mine, ore processing plant and sludge deposit,
- **Slovinky** copper ore mine, ore processing plant and sludge deposit and the
- **Kovohuty a.s.** copper smelter and related industrial waste deposit in the city of Krompachy.

These sites are located in the northern part of the Slovenské rudohorie Mts. and belong to the upper part of the Hornád river basin.

The Rudňany ore deposit was exploited for several centuries until the beginning of the '90s of the 20<sup>th</sup> century. The mineral composition of the ore veins is represented by barite, siderite, sulphides (pyrite, chalcopyrite, tetrahedrite, cinnabarite). The deposit was used for iron ore (siderite) exploitation, since the 17<sup>th</sup> century also copper, silver and mercury. At the end of the 20<sup>th</sup> century also barite was exploited from the upper part of the deposit (Rapant et al. 2003).

The first written evidence about ore exploitation in the Slovinky ore deposit originates in the 13<sup>th</sup> century (Chaluopecký, 1981). The deposit is formed again by the typical hydrothermal ore veins containing quartz, siderite with low content of iron, ankerite, calcite and different sulphides: mainly chalcopyrite, pyrite and tetrahedrite. The most important exploited metal was copper gained by chalcopyrite processing, in the past also gold and silver (Rapant et al. 2003). The mining was stopped in 1991.

The processed ore material – copper concentrate was further processed in the Kovohuty copper smelter in the adjacent city of Krompachy, placed on the alluvium of the Hornád river. Although the copper was produced in smaller smelters within the wider area earlier, this smelter started as iron and steelworks in 1835 and finished in 1918 as the biggest iron and steel producing factory in the former Austrian Hungarian Monarchy. In the 20's of the 20<sup>th</sup> century started with copper producing only and until the beginning of the 90's processed the copper concentrate in Slovinky. In the 60's the production was diversified by manganese, zinc and sulphuric acid production. In present time the smelter is processing copper scrap only (Hyde, 2004a).

The huge production is connected with production of enormous amount of industrial waste. The produced slag in the 19<sup>th</sup> century was deposited in the adjacent area of the factory. As the factory was growing, the new buildings and plants were built on the slag deposit, placed on the alluvium. In the present time all the industrial area of the city is built on the slag. The production diversification in the 60's was connected with production of different kinds of sludge deposited on the Halňa waste deposit, placed at the easternmost part of the industrial area, also on the alluvium of the Hornád river. The waste deposit was used in the beginning for iron smelting slag deposition, since the 60's until 2000 when the deposition was stopped also for different kinds of industrial sludge and municipal waste (Hyde, 2004a).

### 4.6. THE ENVIRONMENTAL LOAD AND SEDIMENT POLLUTION CONNECTED WITH THE DESCRIBED SITES

All the described sites are serious heavy metal pollution sources especially for water environment. For the documentation of their impact I used data from the stream sediment sample analyses database created for the purposes of Geochemical Atlas of Slovakia part Stream Sediments (Bodiš and Rapant, 1966). The samples were collected during the time period 1991 – 1992.

The way of the transportation of the metals is various and dependent on local conditions and the character of a particular source. In the past, during the operation of the ore processing plant in Rudňany huge amounts of mercury vapors were emitted to the atmosphere causing visible damage of the surrounding vegetation and contaminating the soil on the adjacent hill slopes. We should take in consideration this way of transportation, but the leachates formed in mine tailings and sludge deposit contains also considerable amounts of mercury. In comparison with the next two sites, the stream sediments around Rudňany contain highest mercury concentration (see Figure 4.2), up to 127.4 mg.kg<sup>-1</sup>.

One of the common features for both of the Rudňany and Slovinky hydrothermal ore deposits is the presence of barite (barium sulphate) in the upper parts of the deposit. However, if we compare Ba concentrations in stream sediments close to Rudňany with stream sediments close to Slovinky, we will observe incomparably higher Ba concentrations in Rudňany, up to 18 010 mg.kg<sup>-1</sup> (see Figure 4.3).



*Figure 4.2.* Mercury concentrations in stream sediment samples, following data in database created for the purposes of Geochemical Atlas of Slovakia, Part Stream Sediments (Bodiš and Rapant, 1996), gathered from the Geofond archive, Bratislava.



*Figure 4.3.* Barium concentrations in stream sediment samples, following data in database created for the purposes of Geochemical Atlas of Slovakia, Part Stream Sediments (Bodiš and Rapant, 1996), gathered from the Geofond archive, Bratislava.

The opposite situation is apparent following comparison of copper concentrations. The copper is extracted from chalcopyrite (CuFeS<sub>2</sub>) occurring in both of the deposits, however the highest concentrations are occurring in stream sediments close to the mine tailings related to the Slovinky ore deposit (up to 2898 mg.kg<sup>-1</sup>, see Figure 4.4). This difference is again caused by copper utilization in Slovinky. Higher concentrations of copper were observed also in sediments below the Kovohuty copper smelter and belonging Halňa waste deposit.



*Figure 4.4.* Copper concentrations in stream sediment samples, following data in database created for the purposes of Geochemical Atlas of Slovakia, Part Stream Sediments (Bodiš and Rapant, 1996), gathered from the Geofond archive, Bratislava.

Arsenic is often found in sulphidic minerals substituting sulphur in theirs crystal lattice, but the main arsenic containing minerals in the Slovinky ore deposit are arsenopyrite (FeAsS) and tetrahedrite ((Cu,Fe,Ag,Zn)<sub>12</sub>Sb<sub>4</sub>S<sub>13</sub>.) with high As content (verbal information from Mr. Otto Stettner, former chief geologist of the Slovinky ore deposit). Arsenic in water solution is occurring as arsenate (III or IV) anions, what means that is more mobile under more alkaline conditions (Appelo, Postma, 1993). Also in groundwater polluted by sludge deposit leachate of the Halňa waste deposit was reported a correlation between As concentrations and pH values, where the As is the main pollutant (Hyde, 2004b). This is probably the reason of occurring of As polluted sediments in the Slovinsky creek below the sludge deposits and also increased As concentrations in the sediments below the Halňa waste deposit (see Figure 4.5).



*Figure 4.5.* Arsenic concentrations in stream sediment samples, following data in database created for the purposes of Geochemical Atlas of Slovakia, Part Stream Sediments (Bodiš and Rapant, 1996), gathered from the Geofond archive, Bratislava.

The tetrahedrite is probably source of antimony in sediments nearby mine tailings of both of the deposits (see Figure 4.6). Antimony is transported by similar processes as arsenic; the different distribution between the two deposits is caused probably by lower As concentration in minerals of the Rudňany waste deposit.



*Figure 4.6.* Antimony concentrations in stream sediment samples, following data in database created for the purposes of Geochemical Atlas of Slovakia, Part Stream Sediments (Bodiš and Rapant, 1996).

### 4.7. CONCLUSIONS

This section presented information related to sediment pollution in the region where mining and metal processing activities were running for centuries. My goal was to point on a need for individual approach to different sites; even they are geographically close to each other and on the sites were running similar activities. Those differences are determined by complicated geological background and dissected terrain. Unfortunately, regarding mobilization of heavy metals, their transport in aquatic environment and bond to the sediments, eventually remobilization, in most cases we are only guessing.

The discussed area is mountainous area with fast flowing creeks and small rivers. During storm events and snowmelt periods these watercourses has significantly increased discharge what is causing also moving the sediments downward. Definition of sections were the sediments are transported during these events and sediment transport barriers where the sediments are accumulated should be the another task of investigation.

It is symptomatic, that the serious sediment pollutions in Slovakia presented in this and the next paper dealing with the PCB pollution of the Sírava dam are related to stopped industrial activities. This situation is widespread and raises questions about responsibilities and remediation funding. In the case of heavy metal pollutions caused by mining activities it is more complicated, as these activities were running for several centuries. Moreover funding of remedial projects by different financial mechanisms is mostly bound to clarified landowner relationships what in some situations is unsolvable obstacle. Helping this situation will need some legislative interventions, to define responsibilities and ways of funding investigations and remedial actions.

# 5. Problems of persistent organic pollutants (POPS) in Slovakia: environmental and technical aspects of PCBs disposal, monitoring and remediation

The National Implementation Plan of the Stockholm Convention commitments and Non-combustion Decontamination Project are discussed. Slovak Republic, a part of former Czechoslovakia, belonged to the eight world's largest producers of PCBs. Safe disposal of PCB wastes from the production, PCB-containing equipment, and environmental parts has to be carried out. The pollution of an open wastewater canal sediments in Strážske locality and a resulting pollution of the Laborec River and Zemplínska Šírava water reservoir sediments constitute major problems. Strategies and action plans at the national level are discussed as well.

# 5.1. PERSISTENT ORGANIC POLLUTANTS: INTERNATIONAL COMMITMENTS

Persistent organic pollutants (POPs) or persistent bioaccumulative toxic substances (PBT) are carbon-based chemicals that resist degradation in the environment and accumulate in the tissues of living organisms, where they can produce undesirable effects on human health and the environment at certain exposure levels. Concerning such characteristic properties of POPs as *low degradability, low water solubility, high lipid (fat) solubility, bioaccumulation in living organisms, toxicity, semi-volatility, high sorption ability as well as due to migration of POPs at long distances from their original source, it is necessary to deal with the release of POPs to the environment at multinational level.* 

Two international conventions are presently adopted, which have the goal to protect human health and environment against the adverse effects of POPs: the *Protocol on POPs to the UN ECE Convention on Long-range Trans-boundary Air Pollution* (1998) and the *UN Stockholm Convention on POPs* (2001). Slovakia is party to both international commitments. The European Union is party to these conventions as well, and in addition adopted also the Community Policy for POPs Management, which upon joining the EU has become mandatory also to Slovak Republic (SR).

Stockholm Convention concerning persistent organic pollutants requires from the signatories to take measures for source elimination of the 12 substances (so called "dirty dozen") belonging to persistent organic pollutants. The measures include: a) direct production control of POPs pesticides (Aldrin, Dieldrin, DDT, Endrin, HCH, Chlordecon, Mirex, Toxaphene); b) elimination of use of equipment and wastes containing industrially produced PCBs (Delor, Hydelor, Delotherm); c) minimization of unintentionally produced POPs emissions (PCDD/F, HCB, PCBs, PAHs).

### 5.2. NATIONAL IMPLEMENTATION PLAN: STRATEGIES AND ACTION PLANS

Through Regional Centre of UNDP, Slovakia has obtained a GEF grant to support implementation of the Stockholm Convention commitments and to elaborate the *National Implementation Plan (NIP)*. In compliance with the GEF Guidelines, NIP considers all POPs-related international commitments relevant to the SR.

Slovak Republic, formerly Czechoslovakia, is one of the countries where production of the mentioned dangerous substances, specifically PCBs, took place in the past. SR belonged to the eight world largest producers of PCB commercial mixtures (formerly producer Chemko Strážske). Thus, safe disposal of PCBs wastes from the production and PCB-containing equipment and environmental parts, namely so called "old environmental burdens", has to be done. The Ministry of the Environment of the SR (ME SR) in cooperation with the Slovak Hydrometeorological Institute (SHMU) and other stakeholders were responsible for the project *Initial assistance to the Slovak Republic to meet its obligations under the Stockholm Convention on Persistent Organic Pollutants (POPs).* 

NIP contains the following action plans and strategies:

- 1. POPs Pesticides (Aldrin, Dieldrin, DDT, Endrin, HCH, Chlordecon, Mirex, Toxaphene);
- 2. Equipment and wastes containing industrially produced PCBs (Delor, Hydelor, and Delotherm);
- 3. Unintentionally produced POPs (PCDD/F, HCB, PCBs, PAHs);
- 4. Contaminated sites and releases from stockpiles and wastes;
- 5. Monitoring;
- 6. Reporting and information exchange;
- 7. Raising public awareness of POPs;
- 8. Institutional and legal measures;
- 9. Research and development.

# 5.2.1. POPs pesticides

*Hexachlorobenzene:* was manufactured at a Bratislava chemical plant in the 60's and 70's. It was used mainly as a pesticide until its use was banned in 1976.

*DDT:* was manufactured at a Bratislava chemical plant in the 1960's and 70's. Its use in agriculture was banned in 1976, however, in spite of that fact, it was used until the depletion of DDT stockpiles.

*Other chlorinated pesticides (Aldrin, Chlordane, Dieldrin, Endrin, Heptachlor, Mirex,\_Toxaphene):* have neither been produced nor used (perhaps only in small amounts, e.g. Toxaphene) in Slovakia. Their use is currently prohibited. Storage and destruction of obsolete stockpiles remain a problem. An inventory of obsolete POPs pesticide stockpiles in the years 2000 and 2003 documented more than 28 tons of stored POPs pesticide preparations in Slovakia.

# 5.2.2. Equipment and wastes containing PCBs

Altogether, more than 21,000 tons of PCBs were produced in Slovakia during the period 1959-1984, and broadly utilized in former Czechoslovakia mainly for production of capacitors, paints, and varnishes. Basing on extensive inventories, which were carried out in the years 2000 and 2002, current existence of about 3,500 tons of PCBs may be assumed in the territory of Slovakia. Another 900 tons of PCB containing wastes are

stored at the landfill near by the factory. It is assumed that the total amount of identified PCB wastes will still increase, after the new waste management legislation will come in force.

# 5.2.3. Unintentionally produced POPs

The release of unintentionally produced POPs in Slovakia had decreasing trends in the period 1990-2001. The PCDD/PCDF emissions decreased by 76 %, the PAHs by 65 %, HCB by 74 %, and PCBs by 85 %. Gradual implementation of environmental measures in the main polluting sectors (metallurgy, waste incineration, energy sector, and transportation) triggered by the introduction and application of the new BAT/BEP based environment protection legislation, were the main reasons for this decrease.

# 5.2.4. Contaminated sites and releases from stockpiles and wastes

The pollution of an open wastewater canal in Strážske locality and resulting pollution of the Laborec River and the Zemplínska Šírava water reservoir, as a consequence of previous PCB production, constitutes a major problem. This contamination resulted in increased values of PCB content in the monitored components of the environment as well as in the population of Michalovce District, comparing with the rest of Slovakia.

Another important group of contaminated areas polluted with PCBs includes those located in the vicinity of asphalt-gravel mixing plants. Their contamination is caused by improper handling of PCB containing heat-exchange fluids.

# 5.2.5. Monitoring

POPs, in particular PCBs, DDT, and HCB, are monitored in an uncoordinated manner in all the components of the environment, living organisms (including humans), and food chain. None of the monitoring programs covers the whole area of Slovakia. The highest POPs levels were detected in the human population and house animals in Michalovce District (Eastern Slovakia). Regarding the components of the environment, the highest POPs concentration levels were detected in soils and in certain localities (Strážske region) also in sediments. In general, decreasing tendency of POPs presence prevails in all monitored matrices.

# 5.2.6. Institutional and legal measures

Legal regulations concerning POPs are intersecting through several government departments (healthcare, environment, agriculture, and financecustoms). The main driving force influencing the formulation of the new legislation is, besides demands of the international treaties, also a necessity to harmonize the Slovak legislation with that of the EU, in particular with the Proposal for Regulation of the European Parliament and of the Council on POPs and amending Directives 79/117/EEC and 96/59/EC (Regulation 491, 2002).

# 5.2.7. Research and development

State R&D programs and R&D according to state order are providing the most important platform for research and development concerning POPs issues. State programs are development programs, which should contribute to socio-economic development of Slovakia. They should become the most preferred research tasks in the future, onto which also the financial support should focus. However, general structural problems, caused by a long-term lack of state as well as private financial sources (according to the EU their ratio should be 1/3:2/3) supporting R&D in Slovakia, constitute a potential risk also for implementation of POPs specific project tasks. Besides insufficient financing, also pertaining problems with current legislation, which is not consistent enough with the program financing method pursuant to Law on Science and Engineering, as well as with the Law on Support Agency for Science and Engineering (APVT), are enhancing the risks. Possible tools for implementation of POPs related projects are: utilization of the existing system of R&D financing through APVT (inclusion of POPs into the priority list of the state program of research and development); participation of Slovakia in the 6th EU Framework Program for Science and Engineering or other international cooperation.

The following priorities at the national level were pointed out:

Finalizing the process of adoption of the necessary legal documents for POPs management, setting up the necessary institutional framework and their consistent enforcement; Inventory and environmentally sound destruction of wastes and equipments containing PCBs; Inventory and environmentally sound destruction of obsolete POPs pesticides: Decontamination of the waste water canal from Chemko Strážske; Raising of broad public awareness about hazards connected with open uncontrolled burning; POPs monitoring for the purposes of international reporting. POPs management at national level, and information dissemination to public; Gradual decontamination of all other identified contaminated sites; Research and development in the relevant fields.

# 5.3. NON-COMBUSTION DEMONSTRATION PROJECT IN SLOVAKIA: DECONTAMINATION OF PCBs

As has been mentioned above, the most urgent and relevant issue to be solved in Slovakia is the environmental contamination with PCBs. For this
reason, the Slovak Republic has decided to participate in another global program Demonstration of Viability and Removal of Barriers that Impede Adoption and Effective Implementation of Available, Non-combustion Technologies for Destroying Persistent Organic Pollutants (Short title: Noncombustion Project).

*Non-combustion Project* is a part of a global program and is primarily oriented towards assistance with PCB stockpile and other POPs wastes destruction using non-combustion technology of high technical level. Technology unit delivery and installation will be supported by the Project Establishment of the conditions for environmentally grant. safe decontamination of polluted sediments and soil is an integral component of the project efficacy. In this demonstration project, UNDP acts as an implementation agency and UNIDO as an executive agency. During the demonstration phase of the project, destruction of an initial volume (1,500 tons of PCB wastes) will be carried out. Project duration together with a preparatory phase, including 2 years of technology performance and results dissemination activities is suggested to be 4 years. The project will be performed in collaboration between public and private sector. Slovak Republic co-financing is aimed on the effective project support, site preparation, operation of the unit, destruction of PCB stockpile etc. (Murín, 2004).

At present, the project document and detailed materials were revised, with the aim to satisfactorily address all GEF Council members' comments. Within the scope of the project document, project technical and financial efficacy, securing of co-financing and roles of the stakeholders should be defined.

Realization of the *Non-combustion Project* will positively affect the PCB decontamination program and PCB-containing equipment phase-out in the Slovak Republic. Besides other measures to be undertaken, safe destruction of PCBs by the year 2010 is required of the EU Member States.

Under the terms of the project, a unit for PCB extraction from the solid matrix (sediment and soil) and a unit for the destruction of PCBs will be provided by the GEF grant:

PCB destruction in a destruction unit will have a projected capacity of 750 tons a year. Sediment treatment includes environmentally safe sediment excavation, storage close to the extraction unit, drying of sediment, extraction and transportation of the extract to the place of final destruction. Cleaned sediments will be placed at the waste dump taking into consideration the residual PCB concentration up to 1 - 2 ppm (mg/kg).

#### 5.3.1. Production and properties of polychlorinated biphenyls (PCBS)

In Slovakia, PCBs were produced (21,482 t, in total) at the factory Chemko Strážske in 1959–1984 as commercial PCBs mixtures under the brand names Delor, Hydelor, and Delotherm (Figure 5.1). About 46 % of the PCB production was exported mainly to former East Germany. The rest (11,613 t) was used in the territory of former Czechoslovakia (ca. 5,500 t in SR, 6,000 t in CR) as dielectric fluids for transformers and power capacitors, as heat exchanger and hydraulic fluids, as paint additives and lubricants.



*Figure 5.1.* Amounts of PCB formulations manufactured by Chemko Strážske in 1959-1984.

PCBs are oily liquids, which contain two aromatic rings, and according to a location of the chlorine atoms they can occur in up to 209 different chemical compounds combinations (congeners), which have different toxicity. PCB production started in the USA 1929 and PCBs have found a broad range of applications in industry (transformer contents, capacitor contents, hydraulic liquids, heat transfer medium, plasticizers, lubricants, impregnates, dyes, glues, additives to construction materials, sealing liquids, burning inhibitors, pesticides etc.). PCBs are emitted into atmosphere via evaporation from free materials and refuse heaps contaminated with PCBs. In water, they are generally contained in sediment which has strong adsorption properties. In soils, they have no tendency to a noticeable spread due to adsorption and low solubility. PCB degradation is very slow - the more chlorinated substance, the slower is the degradation. Biodegradation by microorganisms is slow as well. PCBs can also get into the plants, and some plants have ability to accumulate PCBs. From plant food and also directly from the sea-water, these substances can get into animals.

In humans, PCBs are cumulated mostly in adipose tissue and can be gradually long-term released from adipose cells into blood. The important fact is also, that they are contained in mother's milk, so these toxic chemicals can be received already by newborns. Besides, the health effects are also ascribed to a simultaneous content of dioxin-like compounds, such as PCDFs, in mixtures. Toxic effects are assumed responsible for respiratory, digestive, and liver functional disorders, neurological and process changes (growth inhibition of children, low birth weight, psychomotor process arrest, as well as a possible intelligent quotient decrease). They may also lead to changes in skin pigmentation and to rashes of acne type. There was no carcinogenicity for humans found, however, PCBs are considered to be potential carcinogens. Finally, PCBs are referred to as the substances that impair human immune system, socalled endocrine disruptors.

#### 5.3.2. Situation with the Slovakia's inventory of POPS: PCB stockpiles

According to the available data (PCB-containing equipment inventory in years 2001 - 2003), total amount of PCB wastes and material in Slovakia is estimated to be 3,500 tons. Out of those:

- 1,000 tons PCB wastes from Chemko Strážske company (residues and wastes from production): various wastes from PCB production, contaminated clothes, waste material etc. is kept relatively safely in premises inside the factory. The major part is comprised of hardly manipulable material of a semifluid consistence.

- 1,000 tons PCB-containing equipment – transformers, capacitors and other equipment: at present, there are registered: 400 pieces of transformers, 30,000 pieces of capacitors, and 400 pieces of other equipment.

- 1,500 tons various wastes: these originate largely from the agricultural sector and include stockpiles of contaminated hydraulic and transformer oils with PCB contents, scraps of PCB - dyes, contaminated concrete pieces, PCB-containing equipment etc.

Additionally to the documented quantity, it is estimated that at Pláne waste dump there are additional ca. 900 tons of PCB contaminated wastes from production.

In the near future, new legislation will be applied in Slovakia focused on PCB wastes and equipment management. Thus, that total waste volumes will be finally higher comparing with the current estimates. The results of the pilot inventory of PCB formulations used in Slovakia are presented in Figure 5.2 (Kočan et al., 1998).





#### 5.3.3. Possible sources of POPs

In Slovakia, PCBs are still used in power capacitors. Such capacitors can contain from 1.4 to 20 kg of PCBs. As their casing is fragile, PCBs from damaged devices can easily enter the environment.

It seems that the fillings of transformers and heat exchangers operating in Slovakia have already been refilled with non-PCB containing fluids. However, in the past, especially in the case of the heat exchangers, a large amount of PCBs leaked during their operation. Several tens (hundreds) of tons of PCBs were disposed of by incineration either in cement kilns in Slovakia or abroad in hazardous waste incinerators.

Because PCBs were also used as a paint additive (their content was up to 21%), old paint coatings can also release these pollutants.

A layer of mud in the 5.3 km long Strážsky Creek which an effluent canal of the former Slovak PCB producer flows into, containing 15 years after the termination of the production about 3 kg PCBs in 1 ton of dry mud, represents serious environmental source of PCBs. This sediment is an abundant PCB source causing the long-term contamination of the waters of eastern Slovakia (Kočan et al., 1999). Figure 5.3 shows PCB concentrations determined in sediment samples taken in the polluted area (Michalovce District) and in a comparative one (Stropkov District). As expected, the highest value was found in a muddy part of the Strážsky Creek (one of the samples contained up to 5 g/kg, i.e. 0.5%). It is doubtless that the polluted creek emptying into the Laborec River has caused the contamination. The Zemplínska Šírava water reservoir (33.5 km<sup>2</sup> surface area) that is partly filled from the Laborec River contained several hundred times higher PCB levels (100-2000 times) in comparison to a similar water reservoir

(Domaša) in the comparative area. Basing on these findings, one can estimate that at least tons of PCBs are still adsorbed in the sediments of the effluent canal, Laborec River and Zemplínska Šírava water reservoir.

are potential sources Waste disposal sites of environmental contamination. PCBs determined in soil samples collected in the neighborhood of the waste disposal sites of Chemko and plants manufacturing asphalted gravel contained PCB levels much higher than those found in agricultural and forest soil taken in areas remote from possible sources. The evidence of this is provided in Figure 5.4. Those factories often situated in the vicinity of the guarries, i.e. in the mountains on rocky ground, have been manufacturing gravel coated with asphalt to be used for road construction. It is known that due to untightness in heat exchanging systems filled with PCBs, hundreds of tons of PCBs leaked out. PCB concentrations peaked at 53,000 mg/kg in a soil sample taken under one of the heat exchangers. However, high PCB levels were also observed in agricultural (35 and 38 mg/kg) and forest soil (3.9 and 7.5 mg/kg) taken near the plants. As the soil in the vicinity of a major Chemko site, which could be contaminated only by air transported particles, contained increased PCB levels (0.4 to 5.8 mg/kg), it is likely that this waste dump may contain large quantities of PCBs. Similarly, PCB waste might be present at the municipal dump of Michalovce (district town about 20 km from Chemko), since increased PCB concentrations were found in a sample collected close to the dump (0.17 mg/kg).



*Figure 5.3.* PCB levels (the sum of all congeners) in bottom sediment samples taken from some watercourses in the districts of Michalovce and Stropkov.



*Figure 5.4.* PCB levels (the sum of all congeners) in spill samples collected in the vicinity of asphalt/gravel mixing plants and the waste disposal sites of the Chemko chemical factory.

#### 5.3.4. PCB contamination in Slovakia

Contamination at the factory Chemko Strážske surroundings belongs to the so-called "old environmental burdens". It directly relates to the former PCB production. Contaminated areas are found inside the factory area and in wider surroundings as well. The contamination is primarily spread through surface water by gradual release from contaminated sediments of an open sewer that leads from the factory to Laborec River, and subsequently through the inlet canal contaminates the water reservoir – Zemplínska Šírava Lake. The contamination is also slowly reaching other areas located lower in the direction of river flow.

Area of contamination by PCBs is partly documented by the previous studies under the project Environmental and Human Population Stress in Area contaminated by PCBs (Kočan et al., 1998), which was financed by ME SR. Recently, a study co-financed by WHO has been completed, and data concerning PCB exposure of population is prepared for publication. PCB concentrations in industrial canal fluctuate from grams to tens of grams of PCBs per kg of sediment dry weight, in some places in Laborec at hundreds of milligrams and in sediment of Zemplínska Šírava in milligrams per kg of sediment dry weight. Another problematic area is a waste dump Pláne, where in accordance with the information from Strážske municipality, up to 900 tons PCB wastes could be stored. PCB waste was dumped there during a period of production at Chemko factory. Complex monitoring of PCB contamination in the area surrounding a former production site was not carried out until now. Only partial data exists. For this reason, it is not possible to estimate exactly the amount of contaminated sediment and soils in precise concentration layers. Initial approximation assumes that the open industrial canal may contain up to 40,000 tons highly contaminated material. For final problem resolution of soils and sediment contamination, it is needed to develop a complex strategy considering various concentrations of PCB contamination, technical and economical decontamination aspects, and the timetable.

Water pollution has resulted in a big difference of PCB levels between fish caught in Michalovce waters and fish from a comparative area (Stropkov District). The PCB pollution of Michalovce District was manifested also in the exposure of home-raised animals resulting in increased PCB levels in their meat or eggs. It should be stressed that, in general, PCB levels in Slovak human population (not only of that of polluted area) are higher than those in other countries (with the exception of CR).

PCB contents of the sediments from water bodies from other areas of Slovakia range between 0.01-2.8 mg/kg. Samples with high content of PCBs come mainly from the areas of high industrial activities. The measurements conducted in 1983-2001 documented that 24% of surface water samples (SHMU, 2003) contained more than 10 ng PCBs/L - a recommended maximum concentration limit.

Sediments analyzed within a study performed in 2002 were sampled from different water bodies from all regions of Slovakia. As expected, the highest values of PCBs were determined in sediments from the industrial effluent canal from the factory of Chemko Co. (Table 5.1). The increased

Table 5.1. PCB	concentrations	in	sediments.
----------------	----------------	----	------------

	plan-PCBs	mono-ortho PCBs	PCBs
	[WHO-TEQ,	pg/g <sub>D.W.</sub> ]	[mg/kg <sub>D.W.</sub> ]
Ondava (upstream from Bukocel Co.)	0.17	0.032	0.019
Ondava (downstream from Bukocel Co.)	0.53	0.11	0.027
Ondava (downstream from Kijov. Creek)	0.81	0.16	0.040
Effluent canal (downstream from Chemko Co.)	230	110	109
Effluent canal (downstream from WWT plant)	1960	1240	733
Effluent canal (road Strážske-Michalovce)	1990	1040	567
Laborec (Voľa)	11	6.0	1.4
Laborec (Nacina Ves)	6.8	3.6	0.97
Laborec (Petrovce)	710	430	98
Zemplínska Šírava (Biela hora)	53	20	7.3
Zemplínska Šírava (Medvedia hora)	74	22	5.5
Zemplínska Šírava (Kusín)	27	7.6	2.3
V. Domaša (dam)	0.2	0.039	0.036
V. Domaša (Holčíkovce)	0.16	0.029	0.016
V. Domaša (Nová Kelča)	0.16	0.038	0.011
Nitra (upstream from NChZ Co.)	0.26	0.05	0.022
Nitra (downstream from NChZ Co.)	0.41	0.11	0.13
Nitra (Chalmová)	0.57	0.11	0.074
Váh (Lisková)	1.1	0.64	0.012
Váh (Černová)	1.0	1.3	0.20
Váh (Krpeľany reservoir)	0.46	0.23	0.045

values were also observed in sediments from the Laborec River and Zemplínska Šírava. Other watercourses (Nitra and Váh) showed PCB values at the background levels in accordance with those determined in the previous study (Kočan et al., 1999).

From the comparison of water samples from Zemplínska Šírava and other Slovak water bodies collected in 1997/98 and 2002 one can evidently conclude that only the samples taken in the surroundings of the former PCB producer reach values above the recommended limits for surface waters. Although the PCB production was ceased almost 20 years ago, the industrial effluent canal from the factory (Table 5.1) is the most probable source of contamination with PCBs in this area. PCBs accumulated in sediments are only slowly mobilized into water. This situation will last probably until the start of remediation (Petrík et al., 2001).

#### 5.3.5. Hot spots and urgent problems

As regards PCBs, the established highly polluted sites, such as the former producer's effluent canal and some watercourses contaminated by this canal (Laborec River and Zemplínska Šírava water reservoir) can be considered as hot spots and present an urgent problem in Slovakia. Since estimated tens of tons of PCBs are adsorbed in the sediment of these water reservoirs, a flood, for example, might cause an ecological disaster. A possible solution involves the detailed examination of an extent of the pollution of the water reservoir, the removal of the contaminated sediment and the decomposition or retrieval of PCBs by a suitable technology, e.g. incineration, thermal desorption, chemical dehalogenation, solvent extraction, or bioremediation.

#### 5.3.6. The state of legislation

In Slovakia, it is prohibited to use PCBs in open systems, however, in the closed ones such as capacitors or transformers, they still can be used. At present, there are legally established limit values for: ambient air, work atmosphere, soils, irrigation water, surface water, foods, and feeds. According to the Directive 473/2000 of Code of Laws of the Slovak Government waste oil containing more than 10 ppm PCBs/PCT shall not be combusted (except cement kilns where exists 50 ppm limit). Waste containing PCBs at range of 50 to 150 mg/kg can be burnt in cement kilns at specified conditions.

In the individual countries there are different limits for specific components of environment and for foodstuffs. In Slovakia, the highest admissible concentrations in working atmosphere are stated in amendment No. 7/1978 of Ministry of Health of SR official bulletin (average 8-hour = 0.5 mg/m<sup>3</sup>, boundary = 1 mg/m<sup>3</sup>). For free atmosphere, the maximum average daily concentration of PCBs is 0.08  $\mu$ g/m<sup>3</sup> (IHE Prague, 1991). Since the primary way of PCB penetration into humans is through food, the content of selected individual PCB congeners was limited in Food Code of Slovak Republic (act of MH SR and MFA SR No. 981/1996 quoted in the official bulletin of MFA SR No. 7/1996) for appointed food commodities (milk and milk products, baby and children food, various meat types, eggs, fish and fish products). In 1990, WHO established a tolerable daily intake (TDI) of 10 pg/kg body weight/day for 2,3,7,8-TCDD, which was revised in 1998 to 1-4 pg/kg<sub>bw</sub>/day for dioxins and dioxin-like PCBs expressed as WHO-TEQ. The upper range pf the TDI of 4 pg TEQ/kg<sub>bw</sub>/day should be considered as a maximal tolerable intake on a provisional basis and the ultimate goal is to reduce human intake levels below 1 pg TEQ/kg<sub>bw</sub>/day.

# 5.3.7. Available technologies and facilities for PCBS and pesticides destruction

In Slovakia, there is no hazardous waste incinerator able to meet criteria for the thorough decomposition of POPs. As already mentioned above, PCBcontaining waste (50-150 mg/kg) is allowed to be burnt in cement kilns. The former Slovak producer of organochlorine pesticides can perform decomposition of limited amounts of organochlorine pesticides by alkaline dechlorination. Bioremediation technologies are currently in the stage of laboratory or pilot experiments.

# 5.4. RESEARCH AND DEVELOPMENT: BIODEGRADATION AND BIOREMEDIATION OF PCBS

The research at Laboratory of Environmental Biotechnology, Department of Biochemical Technology, Faculty of Chemical and Food Technology, Slovak University of Technology has been focused on theoretical aspects of biodegradation of chloroaromatics, namely on the factors that should be preferentially considered when looking for strategies how to improve their biodegradation.

The goals of our work can be summarized as follows: 1. isolation and development of PCB degrading microorganisms with a high degradation and survival potential; 2. to design a PCB distribution model describing the concurrent processes of evaporation, biosorption, and biodegradation in a suspension of PCB-degrading bacteria; 3. to use the distribution model for the determination of the primary biodegradation rate constants of individual PCB congeners, in active bacterial suspension; 4. to conduct an empirical structure-degradability analysis of the obtained data; 5. to improve aerobic

biodegradation by chemical oxidation; 6. to enhance the efficacy of PCB degradation using naturally occurring inducers; 7. to study biodegradation of PCBs in soil; 8. to study adsorption-desorption properties of chlorinated POPs during their immobilization on organomineral complexes (lignite humic acids bound to zeolite) as an alternative decontamination method; 9. to study effect of aromatics and chloroaromatics on the membrane lipids and fluidity of bacterial cells (Dercová et al., 1995, 1996, 1999, 1999, 1999a, 2001; Tandlich et al., 2001; Vrana et al., 1995, 1996, 1996a, 1998; Dercová et al., 2003, 2004).

Biodegradation of PCB using degradation activity of microorganisms (biostimulation and/or bioaugmentation strategy) represents a more ecological and economical way of PCBs elimination in comparison with a combustion process in contaminated localities with low concentration of contaminants. It is possible to suppose that the improvement of biodegradation in localities with higher contaminant concentration can be reached using a combination of physico-chemical and biological processes.

#### Acknowledgements

Financial support from the Scientific Grant Agency of Ministry of Education SR (Grant No. 1/1309/04) is gratefully acknowledged.

# 6. PCBs in sediments of the Zemplinska Sirava Reservoir and the Laborec River in vicinity of Strazske

In this section the results of sediment quality monitoring in the Zemplinska Sirava Reservoir in 2003 are presented, which are compared with those ones obtained in years 1997 – 1999. Furthermore the results of sediment quality in vicinity of Strazske from 2004 are evaluated. The presented outcomes show that bottom sediments in the Zemplinska Sirava Reservoir are substantially contaminated by PCBs, namely by Delors 103 and 106 and by PCB congeners. The values were measured in the order of thousands  $\mu$ g/kg. In comparison to sediments from the reservoir especial attention is paid to PCBs content in sediments from Strazsky Canal. The values from Strazsky Canal were one order of magnitude higher (in the order of ten thousands  $\mu$ g/kg) than the values in sediments of the Zemplinska Sirava Reservoir. The results are fully confirmed by other authors, where PCBs content has been measured on comparable or even higher concentration level.

#### 6.1. INTRODUCTION

The Water Research Institute Bratislava (Slovak Republic) had been dealing with sediment quality of the Zemplinska Sirava Reservoir within the national scientific and technical project (STP) for three years (1997 – 1999) (Hucko et al., 1999a). Obtained results were published, or used for proposal of sediments disposal (Hucko et al. 1999b; Hucko et al., 2000; Hucko, 2001; Hucko and Sumna, 2000). This issue was handled again in 2003 in connection with a verifying of methodical procedures for sediment disposal from water reservoirs (Hucko and Sumna, 2003) and with additional surface water and sediment quality monitoring of selected specific organic pollutants as well (Makovinska, 2004). The results were published in 2004 (Hucko, 2004). In the same year the attention was also focused on bottom sediment quality in vicinity of Strazske (Laborec – Krivostany, Laborec – Petrovce and the Strazsky Canal) (Hucko et al., 2004; Hucko and Kusnir, 2005, Kusnir and Hucko, 2005).

The Zemplinska Sirava Reservoir is situated at the Laborec River at southern piedmont of the Vihorlat Mountain. Its function is to secure a sufficiency of water for the Vojany Thermal Power Station and for irrigations. Another function is also a flood run-off regulation at the Laborec River and partly in area of the whole Eastern Slovakia Lowland. The further aim is a relatively strong application for holiday and recreation (Hucko, 2001). The rest of the reservoir attributes is given in Table 6.1.

PARAMETER	UNIT	VALUE
Reservoir Surface Area*	km <sup>2</sup>	32.8
Length*	km	11.0
Width*	km	3.5
Depth - maximal	m	14.0
Depth - average	m	9.5
Total Capacity	mil. m <sup>3</sup>	334.0
Altitude*	m above see level	117.5
River Basin Surface Area	km <sup>2</sup>	1 400.0

Table 6.1. Basic attributes of the Zemplinska Sirava Reservoir.

\* By maximal working level

According to last measurement, the quantity of sediments in the Zemplinska Sirava Reservoir has been approximately 6.9 million m<sup>3</sup> (Hucko and Kusnir, 2005).

The reservoir was substantially polluted by waste waters from the whole river basin in the past and even though significant measures by individual producers of pollution were taken, this load has been persisting till now.

The pollution played the important part on a decrease of quality of surface water and ground water in the river basin. The great progress in improvement of the Laborec River water quality protection was achieved by waste water cleaning technology in the Chemko Strazske enterprise. In those time, own waste water treatment plant was built and diversion of cleaned waste waters off the river basin was applied (into to the Ondava River through the Posa sludge storage lagoon). Only a small part of waste waters (non-contaminated cooling waters, precipitation waters, specific portion of domestic sewage) are discharged via two emergency holding tanks by the outlet canal to the Laborec River. This polluter is emphasized on this spot because of his important participation in the contamination of the Laborec River and the Zemplinska Sirava Reservoir in the past.

#### 6.2. METHODOLOGY

Samples were taken from the Zemplinska Sirava Reservoir in period 1997 – 1999 and also in 2003, during which time they were taken from 4 cross profiles. Sediments from the Strazsky Canal were sampled in 1999 and in 2004 (close to the Vola village).

In case of sediments from the Laborec River, sampling was performed in July and October 2004 at 2 sampling sites:

Krivostany (above the outlet of the Strazsky Canal to the Laborec River)

Petrovce (below the outlet of the Strazsky Canal to the Laborec River) Overview of all sampling sites is presented in Figure 6.1.

The sediment corer sampler from UWITEC Corp./AUSTRIA was used as a sampling device (see Figure 6.2), which was a transparent plastic tube allowing a visual examination and also a sample partition by layers.

After sample processing these parameters among many others were determined:

Delor-103 and Delor-106 in the area 1997 – 1999,

PCB Congeners # 8, 28, 52, 101,118, 138, 153, 180, 203 in 1999, 2003 a 2004.

Selection of the monitored parameters was done according to the Guideline of MoE SR No. 549/98-2 Risk Assessment of Contaminated Sediments of Rivers and Water Reservoirs and Act No. 188/2003, on Application of Sludge and Bottom Sediments on Soil (Guideline of MoE, 1998; Act No. 188/2003).

The aim of the guideline is to unify the procedures in risk assessment arising from influence of sediment on qualitative composition of upper bed water column what represents water in surface watercourses and water reservoirs.

The act 188/2003<sup>13</sup> specifies conditions of application of wastewater sludge and bottom sediments on agriculture soil and forest soil so as to exclude their detrimental effect on soil nature, plants, water and on human and animal health.



*Figure 6.1.* Location of the sampling sites in the Zemplinska Sirava Reservoir (ZSR) and vicinity of Strazske

- 1 The Strazsky Canal
- 2/1, 2/2, 2/3 West part of the ZSR (cross profiles in 2003)
- 3/1, 3/2 The ZSR Stone-pit locality (cross profiles in 2003)
- 4/1, 4/2 Middle part of the ZSR, Kaluza (cross profiles in 2003)
- 5 East part of the ZSR, Kusin
- 6 Laborec Krivostany
- 7 Laborec Petrovce



Figure 6.2. Sampling device Corer.

#### 6.3. RESULTS AND DISCUSSION

#### 6.3.1. Sediment quality in the Zemplinska Sirava Reservoir

In the past there was monitored Delors occurrence (D-103, D-106) in a larger measure in this reservoir (in water and sediments), because of their importance with regard to the Slovakian industry production and use. That is the reason why they were monitored first of all on the beginning of our trials. In 1999 and continuing in 2003, the monitoring was extended by other PCB congeners, which had started to have a higher environmental importance (PCB 8, 28, 52, 101, 118, 153, 180 and 203) (Guideline of MoE, 1998).

When we compare all sampling periods in the Zemplinska Sirava Reservoir in years 1997-1999 and 2003, the highest values of D-103 and D-106 were detected in 1999, whereas this kind of analysis was not already performed in 2003.

With regard to a sum of PCB congeners it was approximately three-fold higher in 2003 in comparison with these values from 1999. The highest value was measured for congener # 180 in 1999 and for congener # 118 in 2003.

It is necessary to be stated that here presented (compared) values are averaged from all measurements obtained in respective sampling period on the Zemplinska Sirava Reservoir (e.g. all values obtained from the individual cross sections from 2003 were put into average one).

#### 6.3.2. Sediment quality in vicinity of Strazske (1999)

The results of PCBs sediment analysis from the Strazsky Canal (close to Vola village) showed, that here measured values were a few orders of magnitude higher than values from the Zemplinska Sirava Reservoir (sum of congeners in the canal 65 619  $\mu$ g/kg towards 1 124.18  $\mu$ g/kg in the reservoir). Delors (D-103 and D-106) in sediments from the Strazsky Canal highly exceeded the values from the reservoir (D-103 76 852  $\mu$ g/kg and D-106 57 097  $\mu$ g/kg towards 734.2  $\mu$ g/kg for D-103 a 985.9  $\mu$ g/kg pre D-106). The results are fully confirmed by other authors, where PCBs content has been measured on comparable or even higher concentration level (Petrik et al., 1999).

#### 6.3.3. Sediment quality in vicinity of Strazske (2004)

In addition to the Strazsky Canal, the results of sediment samples analysis from the Laborec River in Krivostany and Petrovce were also evaluated.

PCB congeners were detected in sediments of the Strazsky Canal in July (34 765  $\mu$ g/kg) and in October (18 644.6  $\mu$ g/kg) as well. This detection was also confirmed in sediments from Krivostany in July at 1 451  $\mu$ g/kg (congeners # 52 and # 153 – 1 180  $\mu$ g/kg) and similarly in October at 429.9  $\mu$ g/kg (congeners # 52 and # 153). Values being measured at these localities were 24 and 43-fold lower in comparison with those ones in the Strazsky Canal.

There were also detected higher PCB congener's values in the Petrovce sampling site (1701.4  $\mu$ g/kg) in comparison with Krivostany and lower ones in relation to the Strazsky Canal. The results show the highest values at congeners # 8 and # 28.

In case of suspended matters (silt content), which were sampled in July, lower PCB congeners values were measured (102.3  $\mu$ g/kg), in relation to the remainder.

Obtained results of PCBs content in analysed sediments and suspended matter are showed in Figures 6.3 - 6.5.

In order to focus our research not only on monitoring of sediment quality, in the process of leachibility (desorption) and mobility we tried in laboratory condition to characterize the retroactive impact and behaviour of selected organic compounds accumulated in sediments in the water/sediment interface. Among others we used a source of contaminated sediment by PCB from the Strazsky Canal (July 2004) and also from the Laborec Petrovce (October 2004).

As we found out, PCBs together with selected polycyclic aromatic hydrocarbons (PAHs), non-polar extractable substances (NES) and hexachlorobenzene (HCB) showed very poor leachibility confirming their strong binding on sediment particles<sup>9, 10, 11</sup>.



*Figure 6.3.* PCBs in sediments of the Zemplinska Sirava Reservoir (ZSR) and the Strazsky Canal (SC).



*Figure 6.4.* PCBs in sediments in vicinity of Strazske in July (7) and October (10) 2004: Laborec - Petrovce (LP), Laborec - Krivostany (LK), the Strazsky Canal (SC).



*Figure 6.5.* PCBs in sediments of Strazsky Canal in July (7) and October (10) 2004.

#### 6.3.4. Conclusions

The PCBs in sediments from the Strazsky Canal reached the values being a few orders of magnitude higher than those values from the Zemplinska Sirava Reservoir.

There was shown a significant degree of pollution by PCBs, namely by Delors D-103 and D-106 and by PCB congeners in sediments of the reservoir relative to sediments in the canal (such substantial degree of pollution did not appear for the rest of parameters, e.g. heavy metals, PAHs).

The higher content of PCB congeners was detected by an increased water stage during flood period of the Strazsky Canal, which could have been probably caused due to PCBs washing out of the bed.

In 2004 the set of laboratory experiments with sediments from the Laborec – Petrovce and the Strazsky Canal was performed. In the process of leachability and mobility, the behaviour of PCBs accumulated in these sediments in the water/sediment interface was evaluated. There was typical very poor desorption of PCBs (together with other organic compounds) from sediments into water.

### 7. Recommendations

#### 7.1. WORKGROUP OBJECTIVES

Discussion of the approaches for managing contaminated sediments as described in the preceding section led to recognition for the need to answer the questions:

- What are the *major barriers* to successful sediment risk assessment and management? and
- What are the science and infrastructure needs to address these barriers?

This section summarizes the results of that discussion and preliminary efforts to answer those questions.

## 7.2. BACKGROUND: CONTAMINATED SEDIMENT ASSESSMENT AND MANAGEMENT

The magnitude of the challenge posed by contaminated sediment is large. The U.S. Environmental Protection Agency (USEPA) has estimated that in the U.S. more than 1 billion cubic yards of surficial sediment are sufficiently contaminated so as to pose a potential risk to human health and the environment (USEPA 1997). While relatively little quantitative information exists on the extent of the global scope of the problem, there is little reason to think that the situation in the U.S. is markedly different in other industrialized countries. Europe-wide, the volume of dredged material (for capital, maintenance and remediation purposes) can be very roughly estimated at 200 million cubic metres per year. If the US estimate that at least 10% of dredged material is too contaminated for open disposal, this results in 20 million cubic meters of contaminated dredged sediment. However, given that a large proportion of European rivers and ports are in highly populated and industrialized areas, the proportion can most likely be expected to be higher. Little information exists to allow estimates of the extent of sediment contamination throughout Europe, but emerging data suggest a vast and complex problem, particularly in some of the former Eastern Block nations that are currently joining Europe (Bridges et al. 2005).

Given the role sediments play as both a sink and a source of contaminants in aquatic systems, numerous government and private organizations are working to develop the policies and the technical means to address the challenge of how best to manage contaminated sediments, but the policy, approaches, science and frameworks to address these problems differ vastly both regionally and nationally. Over the last 30 years, several countries have proposed or promulgated environmental laws

and regulations that address a broad range of stresses imposed by human activity on the environment. For example, sediment quality is addressed in the U.S. by several state and federal programs and regulations, including most prominently the National Environmental Policy Act, the Clean Water Act, the Marine Protection, Research, and Sanctuaries Act and the Comprehensive Environmental Resource, Compensation, and Liability Act (NRC 1997). Whilst many countries and regions in Europe have or are developing sediment quality standards and assessment programs, the state and details of these differ from country to country. Within the European Union, the Water Framework Directive contains far-reaching provisions intended to secure and manage water resources, and largely by implication. sediments at the river basin scale (Brils 2003; Brooke 2004a; Brooke 2004b), but sediments are not explicitly addressed. It is of note that the WFD does not recognize the important role of sediments in many natural systems, even treating suspended solids as a pollutant, whether contaminated or not. Sediment management is directly or indirectly included in different European directives such as the Waste Directive and Habitats Directive (Köthe 2003). Guidelines for management of coastal dredged material have been developed by several international maritime conventions, including the London Convention, OSPAR and the Helsinki Convention (Bergmann and Maass 2005, in press; den Besten et al. 2003).

Sustainable sediment management will require extensive collaboration and research to adapt current systems of environmental assessment and management to the basin and ecosystem level. Therefore, an integrated, multi-stressor, multi-use/user, multi-media management approach should be developed which encompasses a decision-making hierarchy. This should aim at setting priorities at a basin scale followed by site-specific risk assessment at a local level and control of point and diffuse contaminant sources (Apitz et al. 2005, in press), and other pressures (Landis 2005). A description and model, whether conceptual or quantitative, of the mass flow of water, contaminants and particles within a river basin, or Conceptual Basin Model, (CBM: (Apitz and White 2003), if thoughtfully developed, is a critical part of effective risk assessment for a particular site, and for basinscale management as a whole. Such a CBM should also be put in terms of the ecological receptors in aquatic ecosystems, and in terms of other drivers of ecosystem impact (e.g., habitat loss, saline incursion, thermal change, sea level rise, abstraction, and so on). If, by any process occurring within the river basin, contaminants, activities or other pressures pose a risk to ecosystems or river functions at other sites, a control of that risk source is a vital part of an effective management plan of the overall basin.

7.3. BARRIERS TO SUCCESSFUL SEDIMENT RISK ASSESSMENT AND MANAGEMENT

There seemed to be two general approaches to these questions, one can be categorized as the top-down approach (barriers perceived by information users such as regulators and decision makers) and the other as the bottom-up approach (barriers perceived by information generators such as scientists and site assessors). In spite of this, there were very common themes.

- > There is a difficulty in establishing links...
  - ...between contaminant presence and bioavailability (i.e., between hazard and risk)
    - There is a lack of common strategies/procedures to reliably estimate the actual bioavailability of priority pollutants in sediments and in soils.
      - Some felt that the fact that many sites are prioritized not based upon actual causeand-effect relationships within an ecosystem but by conservative methods employing conventional sum-parameters for pollutant content determination posed a significant barrier to progress.
      - Others felt that the use of risk-based evaluations (which are often not straightforward) rather than strict sediment quality guidelines (SQGs) impeded progress
      - Some panellists felt that SQGs should be the same internationally (or throughout regions or continents), others felt that there needed to be national (or regional) development and validation.
    - There are no common strategies for addressing non-priority pollutants or the effects of noncontaminant causes of stress such as organic loading, thermal stress, habitat loss, disturbance, etc.
  - o ... between upstream sources and downstream sinks
  - ... between a determination of risk and a decision to manage
    - There are technical, financial and policy gaps between assessment/monitoring and remediation
      - There are no standard methods for prioritizing sites for management. This

70

problem is exacerbated by a lack of funds, and a perception that politics, rather than risk, drive prioritization in many cases

- It is not always clear, once a risk is determined, how to manage that risk (particularly at large scales)
- In many cases, there is no policy or public commitment to the cleanup of contaminated sites, even if risks are established
- ...between universities (scientists) and agencies (decision makers)
  - Both scientists and decision makers voiced frustration on this front. Researchers felt that decision makers were not using their science, either at all or appropriately, and decision makers felt that scientists were either not carrying out research of relevance to their real-world problems, or they were not communicating the results of that research in terms that made them useful
- o ... between various agencies
  - There is a need for various government agencies to develop better communication in support of common policies and goals. This lack of interaction was observed within a given country (either from various regions or with different responsibilities (e.g., coastal, water, soil and river management), and between countries (both within continents or internationally)
- ...between various disciplines (e.g., hydrology, toxicology, chemistry, etc.)
  - Soil, water and sediment management (as well as freshwater and marine sediment management) are treated as unrelated issues, making effective, basinscale management impossible
  - A particular barrier is the balance and communication between the technical and socioeconomic issues driving sediment management

Scientists and decision makers all pointed to money as the biggest barrier to effective management - there is not enough to fund all relevant research, nor to clean up all sediments. At times, there is not even enough to carry out sufficient monitoring.

- 7.4. SCIENCE AND INFRASTRUCTURE NEEDS TO ADDRESS THESE BARRIERS
  - There is a need for methods, models and policies in support of basin-scale risk management
    - There is a need for more extensive cooperation between the EU Environmental Protection Agencies and a more extensive cooperation between these and the USEPA structures, in order to develop uniform common and unique limits of acceptance of pollutants in sediments
      - (there was no consensus on this statement others felt the need for national standards independent of those of other nations)
    - There is a need for a comparative evaluation, selection and standardization of a limited group of procedures for estimating the bioavailability of pollutants in sediments
    - There is a need for the development of innovative and effective (bio)analytical strategies for the estimation and monitoring of emerging pollutants, such as hormones, endocrine disrupters, pharmaceutical products, in sediments. This might enhance the potential of actual and future risk analyses procedures
    - There is a need for strategies to identify contaminant sources, predict sediment transport, deposition and erosion within the stream bed considering both continuous and sporadic processes (such as storm events, floods and snowmelt), and how this affects risk distribution at the basin scale
      - There is a need for methods to determine contaminant transport pathways and processes from the sediment to the endangered subject (processes of the contaminant release from the sediment and its transport to risk recipient)
      - There is a need for methods to define sediment vulnerability areas for particular streams (areas from where the contaminant is transported), including possible transport processes – as a way of "source control"
      - There is a need to identify the sediment accumulation sections within the streambeds as well as the final sediment transport barriers (e.g. dams), in connection with the possible contaminant sources

- There is a need for quantitative Conceptual Basin Models to achieve this, and thus for research on physical system function in rivers, estuaries and coasts
- There is a need for research into how non-priority pollutants and the effects of non-contaminant causes of stress such as organic loading, thermal stress, habitat loss, disturbance, etc., affect ecosystem status and function, to ensure that management strategies are properly addressing the causes of impact
- There is a need for interdisciplinary collaboration and communication as well as regional, national and international networks to develop uniform approaches, models, policies and monitoring tools in support of these goals
- There is a need for research in support of standard methods and frameworks, and thus for the development of decision and communication tools that link this complex science to the needs of society
  - There is a need to determine what aspects of sediment risk assessment and management (tools, standards and approaches) are universal, and which require regional or national validation.
  - There is a need for large-scale validations of remedial technologies. These must not just demonstrate site-specific technical feasibility but should provide information that might be generalizable to other sites or nations. For example, in Italy, policy states that they can "inject only substances specifically necessary for remediation scope", but few demonstrations have helped develop tools for such questions.
  - There is a need for common strategies/procedures to reliably estimate the bioavailability of priority pollutants in sediments and in soils, allowing for the prioritization of sites
  - There is a need for transparent, science-based decision frameworks for assessing risk, prioritizing sites, determining the necessity of remediation and selection of the most suitable management methodologies.
    - This should ensure the allocation of funds to sites that require urgent management activities.

- There is a need for these decision tools frameworks to allow for a transparent balance of scientific, technical, regulatory and socioeconomic issues
  - In support of this, there is a need to "translate" science into the language of decision makers

#### References

Act No. 188/2003, on Application of Sludge and Bottom Sediments on Soil.

- Apitz S.E., Carlon C., Oen A., White S., 2005, in press. Strategic Frameworks for Managing Sediment Risk at the Basin and Site-Specific Scale. In: Heise S., editor. Sediment Risk Management and Communication: Elsevier.
- Apitz S.E., Davis J.W., Finkelstein K., Hohreiter D.L., Hoke R., Jensen R.H., Kirtay V.J., Jersac J., Mack E.E., Magar V. et al. (2005a): Assessing and Managing Contaminated Sediments: Part I, Developing an Effective Investigation and Risk Evaluation Strategy. Integrated Environmental Assessment and Management 1, 2–8.
- Apitz S.E., Davis J.W., Finkelstein K., Hohreiter D.L., Hoke R., Jensen R.H., Kirtay V.J., Jersac J., Mack E.E., Magar V., Moore D., Reible D., Stahl R., 2005b. Assessing and managing contaminated sediments: Part II, evaluating risk and monitoring sediment remedy effectiveness. Integ Environ Assess Manag 1(1):e1–e14.
- Apitz S.E., Elliot M., Fountain M., Galloway T., 2005 (submitted). European Environmental Management: Moving to an Ecosystem Approach. Integrated Environmental Assessment and Management:submitted.
- Apitz S.E., Power B. (2002): From Risk Assessment to Sediment Management: An International Perspective. JSS J Soils & Sediments 2 (2) 61–68.
- Apitz S.E., Power B., 2002. From Risk Assessment to Sediment Management: An International Perspective. Journal of Soils and Sediments 2(2):61–68.
- Apitz S.E., White S. (2003): A Conceptual Framework for River-Basin-Scale Sediment Management. JSS J Soils & Sediments 3 (3) 132–138.
- Appelo C.A.J., Postma, D. (1993): Geochemistry, Groundwater and Pollution. Balkema, Rotterdam, 536 pp.
- ASTM American Society for Testing and Materials (1998) "Standard Provisional Guide for Risk-Based Corrective Action PS 104". Philadelphia.
- Bergmann H., Maass V., 2005, in press. Diversity of sediment regulations and monitoring programmes in Europe. In: Heise S, editor. Sediment Risk Management and Communication: Elsevier.
- Blum W.E.H. (2002): How can we achieve a holistic and integrated approach towards soil protection? EEB workshop on the Thematic Strategy for Soil Protection: Workshop Report, 27 October 2002.
- Bodiš, D., Rapant, S. (1996): Geochemical Atlas of Slovakia, Part Stream Sediments. MŽP SR, GÚDŠ, Bratislava, 120 pp.
- Borja A. (2005, in press): The European Water Framework Directive: a challenge for nearshore, coastal and continental shelf research. Continental Shelf Research.
- Borja A., 2005, in press. The European Water Framework Directive: a challenge for nearshore, coastal and continental shelf research. Continental Shelf Research.
- Brehuv, J. (2002): Kontaminácia nánosov v nádrži vodného diela Palcmanská Maša niektorými kovmi. Vodohospodársky spravodajca 4/2002, p.13–14.

- Bridges T., Apitz S.E., Evison L., Keckler K., Logan M., Nadeau S., Wenning R.J. (2005, in press): Making Decisions to Manage Risks from Contaminated Sediments. Integrated Environmental Assessment and Management.
- Brils J.M. (2003): The SedNet Strategy Paper: The opinion of SedNet on environmentally, socially and economically viable sediment management. SedNet <www.SedNet.org>.
- Brooke J. The EU Water Framework Directive A Wake Up Call; 2004a July 2004; The University of York, UK. Defra, Flood Management Division, Area 3D, Ergon House, Horseferry Road, London, SW1P 2AL, UK. Tel: +44 (0) 20 7238 6178.
- Brooke J. The EU Water Framework Directive One Year On; 2004b 29th September 2004; Hamburg, Germany. CEDA Secretariat, Radex Building, Rotterdamseweg 183c, 2629 HD Delft, The Netherlands. Tel: +31(0)15 268 2575.
- Brussels: The European Environmental Bureau (EEB), pp 31–33.
- Carlon, C., Critto A., Nadal N., Samiolo M., Marcomini A. and Petruzzelli G.A. (2003) DESYRE – DEcision Support sYstem for REhabilitation of contaminated sites. Annokkée, G.J., Arendt, F., Uhlmann, O. (Ed.) Consoil 2003 Conference Proceedings, 2987–2993. Forschungszentrum Karlsruhe publishing, Karlsruhe.
- Carlon, C., Giove, S., Agostini, P., Critto, A. and Marcomini, A. (2004) The role of Multi-Criteria Decision Analysis in a DEcision Support sYstem for REhabilitation of contaminated sites (the DESYRE software). *In* Pahl-Wostl, C., Schmidt, S., Rizzoli, A.E. and Jakeman, A.J. (eds), Complexity and Integrated Resources Management, Transactions of the 2nd Biennial Meeting of the International Environmental Modelling and Software Society, iEMSs: Manno, Switzerland, 2004. ISBN 88–900787–1–5.
- Čertík, M., Dercová, K., Finďová, M., Sejáková, Z. and Jakubík, M., 2003, Effect of polyaromatic hydrocarbons (PAHs) on the membrane lipids of bacterial cell. *Biológia* 58: 1115–1121.
- Chalupecký, I. (1981): Dejiny Krompách. Mestský úrad Krompachy, pp. 346.
- Chapman, P.M., 1996. Presentation and interpretation of Sediment Quality Triad data. *Ecotoxicology* 5:327–339.
- Chee Y.E., 2004. An ecological perspective on the valuation of ecosystem services. Biol Conserv 120:549–565.
- CLARINET (2002) Review of Decision Support tools for contaminated land management, and their use in Europe. A report from the Contaminated Land Rehabilitation Network for Environmental Technologies, pp. 180.
- COM (1985): Council Directive 85/337/EEC of 27 June 1985 on the assessment of the effects of certain public and private projects on the environment. Official Journal of the European Communities L 175, 05/07/1985:0040–0048.
- COM (2001a): Communication from the Commission to the Council, the European Parliament, the Economic and Social Committee and the Committee of the Regions. On the sixth environment action programme of the European Community 'Environment 2010: Our future, Our choice' – The Sixth Environment Action Programme – Proposal for a Decision of the European Parliament and of the Council Laying down the Community Environment Action Programme 2001–2010. Brussels. Report nr COM (2001) 31 final 2001/0029 (COD). 24.1.2001.
- COM., 1992. Council Directive 92/43/EEC of 21 May 1992 on the conservation of natural habitats and of wild fauna and flora. Official Journal of the European Communities L 206, 22.7.1992:7.
- COM., 1996. Council Directive 96/61/EC of 24 September 1996 concerning integrated pollution prevention and control. Official Journal of the European Communities L 257, 10/10/1996:0026–0040.
- COM., 2000a. Communication from the Commission on the precautionary principle. Brussels: Commission of the European Communities. Report COM(2000) 1 final.

- COM., 2000b. Directive 2000/60/EC of the European Parliament and of the Council of 23 October 2000 establishing a framework for community action in the field of water policy. Official Journal of the European Communities L 327, 22/12/2000:0001–0073.
- COM., 2001b. Directive 2001/42/EC of the European Parliament and of the Council of 27 June 2001 on the assessment of the effects of certain plans and programmes on the environment. Official Journal of the European Communities L 197/30, 21.7.2001.
- COM., 2002a. Communication from the Commission to the Council and the European Parliament: Towards a strategy to protect and conserve the marine environment. Brussels: Commission of the European Communities. Report COM(2002) 539 final.
- COM., 2002b. Recommendation of the European Parliament and of the Council of 30 May 2002 concerning the implementation of Integrated Coastal Zone Management in Europe (2002/413/EC). Official Journal of the European Communities L 148/24, 6.6.2002.
- COM., 2004. Directive 2004/35/CE European Parliament and of the Council of 21 April 2004 on environmental liability with regard to the prevention and remedying of environmental damage. Official Journal of the European Communities L 143/56, 30.4.2004.
- Costanza R., d'Arge R., deGroot R., Farber S., Grasso M., Hannon B., Limburg K., Naeem S., O'Neill R.V., Paruelo J., Raskin R.G., Sutton P., van den Belt M., 1997. The value of the world's ecosystem services and natural capital. Nature 387:253–260.
- Covich A.P., Austen M.C., Barlocher F., Chauvet E., Cardinale B.J., Biles C.L., Inchausti P., Dangles O., Solan M., Gessner M.O., Statzner B., Moss B., 2004. The role of biodiversity in the functioning of freshwater and marine benthic ecosystems. Bioscience 54(8):767–775.
- Critto A. and Marcomini A., 2001. Ecological risk and chemical pollution in Venice Lagoon. Cà Foscarina, Venezia.
- Critto A., Carlon C. and Marcomini A., 2005. Screening Ecological Risk Assessment for the Benthic Community in the Venice Lagoon (Italy). *Environment International*, in press.
- Crommentuijn T., Sijm D., de Bruijn J., van den Hoop M., van Leeuwen K., van de Plassche E. (2000): Maximum permissible and negligible concentrations for metals and metalloids in the Netherlands, taking into account background concentrations. Journal of Environmental Management 60 (2) 121–143.
- Dalla Valle M., Marcomini A., Jones K.C. and. Sweetman A.J., 2005b. Reconstruction of historical trends of PCDD/Fs and PCBs in the Venice lagoon, Italy. *Environment International*, in press.
- Dalla Valle M., Marcomini A., Sfriso A., Sweetman A.J., Jones K.C., 2003. Estimation of PCDD/F distribution and fluxes in the Venice Lagoon, Italy: combining measurement and modelling approaches. *Chemosphere* 51: 603–606.
- Dalla Valle M., Marcomini A., Sweetman, A.J. and Jones K.C., 2005a. Temporal trends in the sources of PCDD/Fs to and around the Venice lagoon. *Environment International*, in press.
- Decision 2000/532/EC. Commission Decision of 3 May 2000 replacing Decision 94/3/EC establishing a list of wastes pursuant to Article 1(a) of Council Directive 75/442/EEC on waste and Council Decision 94/904/EC establishing a list of hazardous waste pursuant to Article 1(4) of Council Directive 91/689/EEC on hazardous waste (European Waste Catalogue). European Union Commission Decision, 2000D0532.
- DEFRA., 2004. Understanding of Undesirable Disturbance in the Context of Eutrophication, and Development of UK Assessment Methodology for Coastal and Marine Waters. Stage 1. Project report: What is undesirable disturbance? Edinburgh: Brussels. Prepared by: Napier University, Edinburgh; Centre for Environmental, Fisheries and Aquacultural Science, Lowestoft; Department of Agriculture and Rural Development, Belfast; Heriot–

Watt University, Edinburgh; Liverpool University, Port Erin Marine Laboratory, Isle of Man.

- den Besten P.J., de Deckere E., Babut M.P., Power B., DelValls T.A., Zago C., Oen A.M.P., Heise S., 2003. Biological effects-based sediment quality in ecological risk assessment for European waters. Journal of Soils and Sediments 3(3):144–162.
- Depledge M., Galloway T., 2005. Healthy animals, healthy ecosystems. Front Ecol Environ 3:251–258.
- Dercová, K., Baláž, Š., Haluška, Ľ., Horňák V. and Holecová, V., 1995, Degradation of PCBs by bacteria isolated from long-time contaminated soil. *Int. J. Environ. Anal. Chem.* 58: 337–348.
- Dercová, K., Baláž, Š., Vrana, B. and Tandlich, R., 2003, Aerobic biodegradation of polychlorinated biphenyls (PCBs): The fate, distribution, kinetics and enhancement of PCB biodegradation efficacy in the bacterial cell suspension of *Pseudomonas stutzeri*. In: *"The Utilization of Bioremediation to Reduce Soil Contamination: Problems and Solutions"* (V. Šašek, P. Baveye, J.A. Glaser, eds.), p. 95–113. Kluwer Academic Publishers, NATO Science Series, Series IV: Earth and Environmental Sciences Vol. 19. Dordrecht, NL.
- Dercová, K., Čertík, M., Maľová, A. and Sejáková, Z., 2004, Effect of chlorophenols on the membrane lipids of bacterial cells. *Inter. Biodegrad. Biodeter.* 54: 251–254.
- Dercová, K., Vrana, B. and Baláž, Š., 1999, A kinetic distribution model of evaporation, biosorption and biodegradation of polychlorinated biphenyls (PCBs) in the suspension of *Pseudomonas stutzeri*. Chemosphere 38: 1391–1400.
- Dercová, K., Vrana, B., Baláž, Š. and Šándorová, A., 1996, Biodegradation and evaporation of polychlorinated biphenyls (PCBs) in liquid media. *J. Ind. Microbiol.* **16**: 325–329.
- Dercová, K., Vrana, B., Tandlich, R. and Šubová, Ľ., 1999a, Fenton's type reaction and chemical pretreatment of PCBs. *Chemosphere* **39**: 2621–2628.
- Diaz R.J., Solan M., Valente R.M., 2004. A review of approaches for classifying benthic habitats and evaluating habitat quality. J Environ Manage 73(3):165–181.
- Directive 1999/31/EC. Council Directive 1999/31/EC of 26 April 1999 on the landfill of waste (European Landfill Directive). Official Journal of the EC, L182.
- Directive 2000/60/EC. Of the European Parliament and of the Council of 23 October 2000 establishing a framework for Community action in the field of water policy (Water Framework Directive). Official Journal of the EC, L327.
- Directive 75/442/EEC. Council Directive 75/442/EEC of 15 July 1975 on waste (Waste Directive). Official Journal EC, L194.
- Directive 92/43/EEC. Council Directive 92/43/EEC of 21 May 1992 on the conservation of natural habitats and of wild fauna and flora (Habitats Directive). Official journal of the EC, 1992L0043.
- Ducrotoy J.-P., Elliott M. (1997): Interrelations between science and policy-making: the North Sea example. Marine Pollution Bulletin 34, 686–701.
- Ducrotoy J.-P., Elliott M., 1997. Interrelations between science and policy-making: The North Sea example. Mar Pollut Bull 34:686–701.
- EEA, 1995. The DPSIR Framework For Reporting on Environmental Issues. European Environment Agency, Copenhagen.
- EEB (2002): EEB workshop on the Thematic Strategy for Soil Protection: Workshop Report, 27 October 2002. Brussels: The European Environmental Bureau (EEB). 38 pp.
- Elliott M., Cutts N.D., 2004. Marine habitats: loss and gain, mitigation and compensation. Marine Pollution Bulletin 49:671–674.
- Elliott M., Fernandes T., Jonge V.D. (1999): The impact of recent European Directives on estuarine and coastal science and management. Aquatic Ecology 33, 311–321.

- Elliott M., Lawrence A.J., 1998. The protection of species versus habitats—Dilemmas for marine scientists. Mar Pollut Bull 36:174–176.
- Facchinetti, G., Mastroleo, G., Mannino, I., Soriani, S. (2003) A fuzzy expert system for the socio-economic analysis of contaminated sites. Extended Proceedings ACS 2003, 10th International Multi-Conference on Advanced Computer Systems, 22–24 October, Miedzyzdroje, Poland, 2003.
- Freeman S., 2005.WFD—Implications of physical modifications and possible new chemical standards for the navigation sector in estuaries and coastal waters; 2005 31 January, 2005; ICE, London (UK).
- Germano J., 2001. Reflections on statistics, ecology and risk assessment. In: Aller J.Y., Woodin S.A., Aller R.C., editors. Organism-sediment interactions. Columbia (SC): University of South Carolina Press. p 33–42.
- Gobas A.P.C.F., 1993. A model for predicting the bioaccumulation of hydrophobic organic chemicals in aquatic food-webs: application to Lake Ontario. *Ecological Modelling*, 69: 1–17.
- Haluška, Ľ., Barančíková, G., Baláž, Š., Dercová, K., Vrana, B., Furčiová, E., Paz-Weisshaar, M. and Bielek, P., 1995, Degradation of PCBs in different soils by inoculated *Alcaligenes xylosoxidans. Sci. Total Environ.* 175: 275–285.
- Harkness, M.R., McDermott, J.B., Abramowicz, D.A., Salvo, J.J., Flanagan, W.P., Stephens, M.L., Mondello, F.J, May, R.J., Lobos, J.H., Caroll, K.M., Brennan, M.J., Bracco, A.A., Fish, K.M., Warner, G.L., Wilson, P.R., Dietrich, D.K., Lin, D.T., Morgan, C.B. and Gately, W.L., 1993, In situ stimulation of aerobic PCB biodegradation in Hudson River sediments. *Science* 259: 503–507.
- Heise S. (2005, in press): Sediment Risk Management and Communication: Elsevier.
- Hucko, P. (1994):, Prehodnotenie kvality povrchových vôd a sedimentov v tokoch. Záverečná správa čiastkovej úlohy č. 01.01 VT-projektu ZP 531–050 Vodohospodárske aspekty ekologizácie krajiny. VÚVH, Bratislava, 168 pp.
- Hucko, P. and Kusnir, P.: Bottom Sediments Source of Secondary Contamination of Surface Waters. In: Proceedings of the International conference Sediments of rivers and reservoirs. Bratislava, 11 – 12 May 2005. Slovak Water Management Association of WRI Bratislava, 2005, pp. 47–62.
- Hucko, P. and Sumna, J.: Water Reservoir Sediments Disposal Verification. Final Report of WRI Bratislava, Bratislava, 2003.
- Hucko, P. and Sumna, J.: Water reservoirs Sediment Quality analysis and Proposal of their Disposal. Final Report of WRI Bratislava, Bratislava, 2000.
- Hucko, P., Weigeltova, S. and Kusnir, P.: Influence of Surface Water Quality and Sediment Quality by Organic Compounds from point source of pollution in selected areas. Final Report of WRI Bratislava, Bratislava, 2004.
- Hucko, P., Holubova, K. and Szolgay, J.: STP No. 514–78 Research of drinking Water Treatability and Environmental Aspects of Water Streams. Final report Phase 03.05 Influence of bed-loads and suspended solids in Areas Effected by waterworks on Water Quality. WRI Bratislava, Bratislava 1999a.
- Hucko, P., Hudec, I. and Luther, S.: Influence of the Zemplinska Sirava Reservoir on Water Quality Forming. In: Proceedings of the 12th Limnological Conference. Limnology on Turning Point of Millennium. Kouty nad Desnou, 18 -22 September 2000. Olomouc, University of Palackeho, 2000, pp. 287–290.
- Hucko, P., Luther, S., Elexova, E., Tothova, L. and Hudec, I.: The Present State of Water and Sediment Quality in the Zemplinska Sirava Reservoir. In: Proceedings of the 15th Conference the Present Tasks of Water Supply Biology, Prague, 3 - 4 February 1999, Czech Scientific-technical Water Management Association, 1999b, pp. 94–98.

- Hucko, P.: Influence of Anthropogenic Activity in Catchment's area of the Zemplinska Sirava Reservoir on Water and sediment Quality. In: 5th International Conference Influence of Anthropogenic Activities of Water Regime of Lowland Territory. Michalovce, Zemplinska Sirava 25 - 27 May 2004. IH SAS Bratislava, Research Base of Michalovce, 2004.
- Hucko, P.: Sediment Chemistry of the Zemplinska Sirava Reservoir. In: Proceedings of the 5th District Water Days Conference. Michalovce, Zemplinska Sirava, 5 6 April 2001. IH SAS Bratislava, Research Base of Michalovce, 2001, pp. 47–52.
- Hull S.C., Freeman S.M., Rogers S.I., Ash J., J.B., Elliott M., 2004. Methodology for the Provisional Identification and Formal Designation of Heavily Modified Water Bodies in UK Transitional and Coastal Waters under the EC Water Framework Directive. Bristol (UK): Environment Agency.
- Hyde, G. (2004a): Halna Landfill, Krompachy, Environmental Base Line Study, Part A: Collection and Evaluation of Existing data. DANCEE, Copenhagen, 17 pp.
- Jackson J.B.C., Kirby M.X., Berger W.H., Bjorndal K.A., Botsford L.W., Bourque B.J., Bradbury R.H., Cooke R., Erlandson J., Estes J.A., Hughes T.P., Kidwell S., Lange C.B., Lenihan H.S., Pandolfi J.M., Peterson C.H., Steneck R.S., Tegner M.J., Warner R.R., 2001. Historical overfishing and the recent collapse of coastal ecosystems. Science 293(5530):629–638.
- Jones D.S., Barnthouse L.W., Suter II G.W., Efroymson R.A., Field J.M. & Beauchamp J.J., 1999. Ecological risk assessment in a large reservoir: 3.Benthic invertebrates. *Environmental Toxicology & Chemistry*, 18: 599–609.
- Kočan, A., Drobná, B., Chovancová, J., Kočan, J., Petrík, J. and Szabová, E., 1998, Burden of the environment and human population in the area contaminated with PCBs. Interim report, Project No. 105/98–2.4. IPCM Bratislava, November, 113 pp. (*in Slovak*).
- Kočan, A., Petrík, J., Chovancová, J. and Drobná, B., 1994, Method for the group separation of non-ortho-, mono-ortho- and multi-ortho-substituted polychlorinated biphenyls and polychlorinated dibenzo-p-dioxins/polychlorinated dibenzofurans using activated carbon chromatography. J. Chromatography 665: 139–153.
- Kočan, A., Petrík, J., Drobná, B., Chovancová, J., Jursa, S., Pavúk, M., Kovrižnych, J., Langer, P., Bohov, P., Tajtáková, M. and Suchánek, P. 1999, Burden of the environment and human population in the area contaminated with PCBs. Final report, Project No. 105/98–2.4., IPCM Bratislava, February, 217 pp. (*in Slovak*).
- Korenková, E., Slobodník, J. (2001): Prieskum a zhodnotenie výskytu pesticídnych prípravkov a nebezpečných látok v podzemných a povrchových vodách a v sedimentoch Slovenskej republiky so zameraním na látky kontrolované Smernicou Európskej únie, Environmental Institute, Okružná 784/42, 972 41 Koš. 80 pp.
- Köthe H., 2003. Existing Sediment Management Guidelines: An Overview. What will happen with the sediment/dredged material? Journal of Soils and Sediments 3(3): 139–143.
- Kusnir, P. and Hucko, P.: Mobility of Substances accumulated in Sediments into Water Environment. In: Proceedings of the International conference Sediments of rivers and reservoirs. Bratislava, 11 – 12 May 2005. Slovak Water Management Association of WRI Bratislava, 2005, pp. 63–76.
- Landis W., 2005. Regional scale ecological risk assessment using the relative risk model. Boca Raton (FL): CRC Press.
- Landis W., 2005. Regional Scale Ecological Risk Assessment Using the Relative Risk Model. Boca Raton: CRC Press.
- Levin S.A., 1992. The problem of pattern and scale in ecology. Ecology 73:1943–1967.

- Lewis J.R., 1999. Coastal zone conservation and management: a biological indicator of climatic influences. Aquat Conserv 9(4):401–405.
- Libralato S., Pastres R., Pranovi F., Raicevich S., Granzotto A., Giovanardi O., & Torricelli P., 2002. Comparison Between the energy flow networks of two habitats in the Venice Lagoon. *Marine Ecology*, PSZN, 23, Suppl. 1:228–236.
- Long E.R., Macdonald D.D., Smith S.L., Calder FD (1995): Incidence of Adverse Biological Effects within Ranges of Chemical Concentrations in Marine and Estuarine Sediments. Environmental Management 19 (1), 81–97.
- Makovinska, J.: Monitoring of Selected Specific Pollutants for Supplementary Monitoring of Surface Waters and Sediments Quality. Final Report of WRI Bratislava, Bratislava, 2003.
- Malakoff D., 1997. Extinction on the high seas. Science 277:486-488.
- Martin L.R., 2002. Regional Sediment Management. Institute for Water Resources. Report 02-PS-2. Prepared for US Army corps of engineers.
- MAV-CVN, 1999. Mapping the lagoon bottom contamination (in Italian). Final Report. Consorzio Venezia Nuova, Venice, Italy.
- MAV-CVN, 2000. Project 2023 results (in Italian). Final report. Consorzio Venezia Nuova, Venice, Italy.
- MAV-CVN, 2001. Determination of the pollutant load discharged into the Venice Lagoon by the drainage basin. DRAIN project workshop, 14–15 June 2001, Venice, Italy.
- MAV-CVN, 2005. Final results of 2001–2002–2003 MELa 1 project. MELa 1 workshop, 31 May 2005, Venice, Italy.
- McLusky D., Elliott M., 2004. The estuarine ecosystem; Ecology, threats and management. 3rd ed. Oxford (UK): OUP. 216 p.
- Micheletti C., Critto A., Carlon C., Marcomini A., 2004. Ecological Risk Assessment of Persistent Toxic Substances for the Clam *Tapes philipinarum* in the lagoon of Venice. *Environmental Toxicology & Chemistry*, 23(6): 1575–1582.
- Micheletti C., Critto C., Semenzin E. and Marcomini A., 2005. A methodology for estimating ecological risk from bioaccumulation of POPs in a marine food web (lagoon of Venice, Italy). In: Scientific Research and Safeguarding of Venice. Research Programme 2001-2003, Volume III. 2003 results. Edited by Campostrini P.
- Ministerstva životného prostredia Slovenskej republiky z 27. augusta 1998 č. 549/98 2 na hodnotenie rizík zo znečistených sedimentov tokov a vodných nádrží.
- Ministry of Environment of Slovak Republic (1998): Metodický pokyn.
- Murín, M., 2004, Global program to demonstrate the viability and removal of barriers that impede adoption and successful implementation of available, non-combustion technologies for destroying persistent organic pollutants (POPs). Ecotoxicological Centrum, Bratislava.
- Netzband A., Hakstege A.L., Hamer K., 2002. Treatment and confined disposal of dredged material. Report, part two, of Dutch-German exchange on dredged material (DGE). 36 pages, Bonn, Den Haag.
- NRC (2001): A risk-management strategy for PCB-contaminated sediments, National Research Council. Washington, D.C.: National Academy Press.
- NRC., 1997. Contaminated Sediments in Ports and Waterways: Cleanup Strategies and Technologies. Washington, D.C.: National Academy Press.
- Petrik, J., Jursa, S. and Kocan, A.: Polychlorinated Biphenyls (PCB) and Organochlorinated Pesticides in Water Ecosystems of East Slovakia Region. In: Proceedings of the International conference Sediments of rivers and reservoirs. Bratislava, 20 – 21 October 1999. Slovak Water management Association of WRI Bratislava, 1999, pp.116–125.

- Petrík, J., Kočan, A., Jursa, S, Drobná, B., Chovancová, J. and Pavúk, M., 2001, Polychlorinated biphenyls in sediments in Eastern Slovakia. *Fres.Environ. Bull.* 10: 375–380.
- Pippa R., Scanferla P., Zuin S., Cammarata F., Pavoni B., Hreglich S. and Avezzù, F., 2003, An integrated treatment system for contaminated sediments, in: *Proceedings of the Second International Conference on Remediation of Contaminated Sediments*; Venice, Italy, 30 Sep - 3 Oct 2003; Eds. M. Pellei and A. Porta, Battelle Press, Part D-06.
- Pojana G., Bonfà A., Busetti F., Collarin A., Marcomini A., 2004. Estrogenic potential of the Venice lagoon waters. *Environmental & Toxicological Chemistry* 23(8): 1874–1880.
- Raffaelli D., Emmerson M., Solan M., Biles C., Paterson D., 2003. Biodiversity and ecosystem processes in shallow coastal waters: an experimental approach. J Sea Res 49(2): 133–141.
- Rapant, S., Cicmanová, S., Mackových, D., Lučivjanská, V., Bodiš, D. (2003): Zhodnotenie potenciálneho vplyvu geochemického prostredia na zdravotný stav obyvateľstva v oblasti Spišsko-gemerského rudohoria, regionálny geologický výskum. MŽP SR, ŠGÚDŠ, Bratislava. 39 pp.
- Read S., Elliott M., Fernandes T. (2001): The possible implications of the Water Framework Directive and the Species & Habitats Directive on the Management of Marine Aquaculture. In: Read Pea, editor. The Implications of Directives, Conventions and Codes of Practice on the Monitoring and Regulation of Marine Aquaculture in Europe (MARAQUA). Aberdeen: Publ. Fisheries Research Services. Pp 58–74.
- Regulation 491, 2002, Regulation of the Government of the Slovak Republic from July 17, 2002 (Statute Book No. 491/2002).
- Sala O.E., Chapin III F.S., Armesto J.J., Berlow E., Bloomfield J., Dirzo R., Huber-Sanwald E., Huenneke L.F., Jackson R.B., Kinzig A., Leemans R., Lodge D.M., Mooney H.A., Oesterheld M., Poff N.L., Sykes M.T., Walker B.H., Walker M., Wall D.H., 2000. Global biodiversity scenarios for the year 2100. Science 287:1770–1774.
- Salomons, W., Brils J.M. (eds.) 2004. Contaminated Sediments in European River Basins. Freely available/downloadable through: www.SedNet.org.
- SedNet (2004): Contaminated Sediments in European River Basins. Salomons W, Brils J, editors. The Netherlands: TNO.
- Sednet "Contaminated Sediments in European River Basins" 2004, 3<sup>rd</sup> International Conference 25-26 November 2004 Venice.
- Shephard B.K., 1998. Quantification of Ecological Risks to Aquatic Biota from Bioaccumulated Chemicals. In Proceedings, National Sediment Bioaccumulation Conference, Bethesda, Maryland, USA, September 11-16 1996, pp 2.31–2.52.
- SHMU, 2004, Initial Assistence to the Slovak Republic to meet its Obligations under the Stockholm Convention on Persistent Organic Pollutants (POPs). Proposal of the National Implementation Plan under the Stockholm Convention on POPs in Slovakia. Project SLO/01/G31, Technical Report No. 5 (Final version), SHMU, Bratislava, April, 131 pp.
- Sorensen M.T., Gala W.R., Margolin J.A., 2004. Approaches to ecological risk characterization and management: Selecting the right tools for the job. Hum Ecol Risk Assess 10:245–269.
- Tandlich, R., Brežná, B. and Dercová, K., 2001, The effect of terpenes on the biodegradation of PCBs by *Pseudomonas stutzeri*. *Chemosphere* **44**: 1547–1555.
- The Guideline of MoE SR No. 549/98–2 Risk Assessment of Contaminated Sediments of Rivers and Water Reservoirs. Bulletin of MoE SR, Annual volume VI, Part 5, 1998.

- USEPA (2002): Contaminated Sediment Remediation Guidance for Hazardous Waste Sites. United States Environmental Protection Agency, Office of Solid Waste and Emergency Response. Report nr OSWER 9355.0–85.
- US-EPA, 1998. "Guidelines for Ecological Risk Assessment. Washington DC: Risk Assessment Forum". EPA/630/R-95/002F. Environmental Protection Agency, Washington, DC.
- USEPA., 1997. The incidence and severity of sediment contamination in surface waters of the United States: Volume 1 national sediment quality survey EPA-823-R-97–006. Washington DC: U.S. Environmental Protection Agency.
- Vrana, B., Dercová, K. and Baláž, Š., 1995, Monitoring evaporation polychlorinated biphenyls (PCBs) in long-term degradation experiments. *Biotechnol. Tech.* 9: 333–338.
- Vrana, B., Dercová, K. and Baláž, Š., 1996, Evaporation kinetics of polychlorinated biphenyls (PCBs) during biodegradation experiments. *Biotechnol. Tech.* 10: 37–40.
- Vrana, B., Dercová, K., Baláž, Š. and Ševčíková, A., 1996a, Effect of chlorobenzoates on the degradation of polychlorinated biphenyls (PCBs) by *Pseudomonas stutzeri*. World J. Microb. Biot. 12: 323–326.
- Vrana, B., Tandlich, R., Baláž, Š. and Dercová, K., 1998, The aerobic biodegradation of polychlorinated biphenyls by bacteria. *Biológia* 53: 251–256.

### PHYSICOCHEMICAL AND BIOLOGICAL ASSESSMENT AND CHARACTERIZATION OF CONTAMINATED SEDIMENTS

SAM BENTLEY\* AND LOUIS THIBODEAUX Louisiana State University, Baton Rouge, LA 70803

PETER ADRIAENS AND MENG-YENG LI University of Michigan, Ann Arbor, MI

MARÍA ROMERO-GONZÁLEZ\* AND STEVEN A. BANWART University of Sheffield, United Kingdom

ZDENEK FILIP AND KATERINA DEMNEROVA Institute for Chemical Technology, Prague, Czech Republic

DANNY REIBLE The University of Texas at Austin, TX 78712

\*To whom correspondence should be addressed.

Abstract- Physicochemical and biological processes that influence contaminant fate and transport in sediments are diverse, and include physical chemical behaviour, biological processes at microbial and macro scales, as well as pore water and solid phase transport. Spatial and temporal variability in these environmental characteristics and processes are large, and can be difficult to portray and predict. We herein present a series of case studies that evaluate environmental processes and properties that must be assessed with respect to studies of contaminated sediments. These topics are: chemical behaviour of metals in the environment; biological processes and sediment properties influencing contaminant release from sediments; the biogeochemistry and adsorptive capacity of ambient organic compounds in aquatic environments: and new approaches to modeling and scaling the spatial distribution of relevant processes and sediment properties. In a concluding assessment of our state of understanding for these processes and properties, we find that significant knowledge gaps remain in terms of both causal and statistical uncertainty, and these uncertainties should be the focus of new research and development in assessment and characterization of contaminated sediments.

**Keywords:** advection, diffusion, reduction-oxidation, bioturbation, dissolved contaminants, particulate contaminant, biogeochemistry, redox, metal complexation, speciation, geostatistics, scaling, porosity, stability.

### 1. Chemical behavior of metals in the environment

The three main processes that govern the chemical characteristics of contaminant behavior in the environment are: oxidation-reduction reactions, adsorption processes and dissolution-precipitation reactions. These processes also play an important role in the fate of metals. Redox reactions control chemical speciation of many metals affecting their transport in the environment. Although many redox transformations are biologicallymediated as irreversible reactions, the focus in this chapter is on redox and speciation equilibria, particularly how to handle chemical complexity in real systems. Microbiologically-mediated reactions are addressed more comprehensively in Chapter X (Agathos). In the following sections we first present the problems associated with predicting the rates of chemical weathering at mine sites and how to overcome these problems by predicting rates for mineral weathering from laboratory and field data. The second section presents metal speciation and the role of sorption on complex natural sorbents as a control on the mobility of metals within aqueous systems. The final section outlines the value of understanding these processes in the natural environment and the implications for risk assessment.

### 1.1. MINERAL WEATHERING AS A SOURCE OF METAL POLLUTION

Mining activities worldwide contribute significantly to the solute loads of receiving streams and aquifers. Mining activities contribute with an estimate of 12% to the fluvial sulfate to the sea (Nordstrom and Southam, 1997) and release dissolved metal ions and acidity as potential environmental hazards. Therefore, understanding how fast sulphide minerals generate acidity and metal loads to receiving waters and sediments is crucial. The dissolution of iron sulphide minerals disrupts the acid-base balance in the vicinity of mine sites, as expressed by the iron sulphide (pyrite) dissolution reaction:

$$FeS_{(s)} + 7.5 O_2 + H_2O \rightarrow Fe^{2+} + 2SO_4^{2-} + 2H^+$$
 (1)

Pyrite weathering produces sulfate, releases soluble ferrous iron (Fe<sup>2+</sup>) and produces acidity, represented in equation (1) by the production of protons. Transformation of soluble ferrous iron by oxidation into soluble

ferric iron (Fe<sup>3+</sup>) needs also be accounted for (Younger *et al.*, 2002). The net impact on the acid-base balance depends on the rates of alkalinity production from dissolution of carbonate and silicate minerals.

At the same time, the dissolution of other metal sulphides such as ZnS(s) (sphalerite) and PbS(s) (galena) releases potentially hazardous metals to the environment, but does not necessarily produce acidity.

$$ZnS_{(s)} + 2O_{2(aq)} \rightarrow Zn^{2+} + SO_4^{2-}$$
 (2)

$$PbS_{(s)} + 2O_{2(aq)} \rightarrow Pb^{2+} + SO_4^{2-}$$
(3)

These chemical processes can be quantified from laboratory data and the associated weathering rates can be calculated by normalizing the data to the mass of reacting mineral or rock sample. A crucial factor to quantify in order to extrapolate weathering rates from laboratory experiments to field scale is the reacting surface area of the minerals at the respective scales of observation (Banwart et al., 2002). The value of specific surface area of a mine site is generally not known, leading to very uncertain estimates of contaminant loadings and contaminant weathering rates at field scale. Surface area values obtained by empirical site-specific estimates at field sites are generally two-three orders of magnitude lower than the specific area determined by laboratory studies (Banwart et al., 2002). The challenge for risk assessment is therefore to develop a generic method to predict chemical processes rates which: (i) determine contaminant source strength and longevity of acidity and metal ions; (ii) attenuate acidity loads through neutralization along transport pathways and (iii) attenuate metal ion loads through sorption and precipitation reactions. Such method will allow the estimation of environmental impact from desktop where limited funding for site investigation or limited data is available. Although research to date has focused on metals release rates from mine sites, the concepts are equally applicable to metals sources, such as dredged sediments or other metalsbearing waste, stored above the water table or deposited in locations with hydrological transport pathways to surface waters.

#### 1.1.1. Determination of weathering rates by aqueous chemical methods – Prediction of field weathering rates from laboratory data

A potentially general scaling method to predict field-weathering rates from laboratory data was developed for a mining site at Aitik, Sweden (Malström et al., 2000). The scaling algorithm quantifies for different observations scales (lab vs field) the effects on weathering rates from differences in temperature, physical surface area, mass of flowing water and rock reacting and relative abundance of reactive minerals. For laboratory assessment of weathering, the reactions are generally studied using either of two main experimental approaches, namely, closed batch reactors or flowthrough column reactors (see Banwart et al., 2002). Batch reactor experiments are a simple approach to determine the potential of a soil or rock sample to generate contamination. The reactors are set up with a known mass of solid material and a known volume of aqueous solution. The reactor is run under controlled laboratory conditions and parameters such as ionic strength solution, pH and  $O_2(aq)$  concentration are fixed. Batch reactors can be set up with large numbers of replicate experiments that can be run over periods of months. The progress of the reaction is followed by periodic sampling and monitoring of chemical composition. In this type of experiment the increasing mass of solutes present in the aqueous phase can be related to the source minerals present. The weathering reaction rate is then determined as moles of mineral dissolving over time. Laboratory rates determined in this fashion can be then used. with an appropriate scaling method such as described above, to predict source strength at field sites.



*Figure 1.1.* Comparison between batch mineral rates and field dissolution rates. (a) Temperature; (b) pore water pH; (c) particle size distribution; (d) mineral content; (e) water flow patterns.
Column experiments may be set up under hydraulically -saturated or unsaturated conditions, offering the opportunity to simulate realistic solid to water ratios, gas transport such as  $O_2(g)$  ingress, and element release as a function of water flow rate. This includes assessing whether solubility equilibria between effluent solutes and source or secondary minerals occurs. It also evaluates whether ion release is controlled by chemical kinetics or by mass-transfer diffusion.

Comparison between mineral weathering rates determined in batch laboratory experiments and field rates are shown in Figure 1.1. The solid lane represents the ideal case where batch test rates equal field rates. Batch rates calculated without taking into account the effect of factors such as temperature, pH, particle size distribution, mineral content or water flow patterns are 2-3 order of magnitude higher than rates determined from field data. Progressively accounting for differences in the factors (a)-(e) at lab and field scale, to a large degree, resolves the discrepancy. These results shows that scale-dependence of these rates is predictable to a large degree by quantification of the effects of a few bulk physico-chemical characteristics of a field site. The fact that this scale dependence is consistent with other laboratory studies (discussed in Malmström et al., 2000) suggests the general applicability of extrapolating weathering rates from the laboratory to many types of field sites.

## 1.2. REDOX/BIO-GEOCHEMISTRY OF METALS

The main chemical transformation reactions that metals undergo in the environment are oxidation-reduction (redox) reactions. Redox reactions change the speciation and solubility of many elements, create new compounds and affect the biogeochemistry of soils. Redox reactions are not uniform in soil and sediment matrices and depend on the dynamics of soil oxygen and other dominant redox-reactive components. Figure 1.2 shows a scale for some of the major redox couples in the environment. The oxidized form of each element is shown on the left side of the scale and the reduced form on the right. Any oxidized form can react with the reduced form of another couple located below it on the scale. These reactants correspondingly form their reduced and oxidized forms as reaction products.

Of particular importance are reactions that are related to dissolution or precipitation of mineral phases since this transformation radically alters the mobility and bioavailability of these substances. For example, manganese is sparingly soluble under oxic conditions as  $MnO_2(s)$  but is reductively dissolved under reducing conditions thus liberating  $Mn^{2+}$  to solution. The ion is difficult to remove from mine waters to the concentration levels that

are often required by regulatory control (see Younger *et al.*, 2002). Another example is given by metal sulphide phases such as FeS(s) that are sparingly soluble under reducing conditions, but release soluble metals and sulphate ion when oxidized.

A particularly important case is given by the oxidative precipitation/reductive dissolution of iron oxyhydroxide minerals. These minerals are abundant in mining environments due to release of Fe by pyrite weathering (Eq. 1).



Figure 1.2. Redox ladder showing common redox couples in the environment.

Figure 1.3 shows the Pourbaix diagram for iron and water in soil and sediment with the stability limits for water (see Garrels and Christ, 1965) and the stability fields for various Fe mineral phases and dissolved forms. Since the oxidation of  $Fe^{2+}$  to  $Fe^{3+}$  is independent of pH a straight line with intercept at 0.771 volts implies that the activity ratio of the two iron species is always 1. There are no redox reactions involved in the transformation of  $Fe^{3+}$  into  $Fe(OH)_3$  giving a straight line parallel to the Eh axis. At pH of natural environments the iron speciation is dominated by the presence of  $Fe^{2+}$  at low Eh values (anoxic environment) and the oxidation and subsequent precipitation of  $Fe^{2+}$  into  $Fe(OH)_3$  (oxic environment). Other phases as iron -carbonates, -sulfides and silicates can be added to the

diagram as necessary for the system considered. With respect to other redox couples, similar diagrams can be constructed for other elements in soil and sediments systems, including potentially toxic metals such as chromium, arsenic or uranium which are affected by redox transformations and changes in pH.

In particular to the case of iron is the role of  $Fe(OH)_3(s)$  to act as a strong sorbent for many types of metal ions. The presence of  $Fe(OH)_3(s)$ thus acts to scavenge metals from the water column or pore waters, and can release these sorbed contaminants if redox transformation to soluble  $Fe^{2+}$ occurs under reducing conditions. In the following session, more attention is given to the behavior of  $Fe(OH)_3(s)$  to scavenge metals from solution.



*Figure 1.3.* Pourbaix diagram for iron in relation to Eh and pH boundaries in soil and sediments.

#### 1.3. SORPTION OF METALS

The ultimate fate of metals released into the environment depends on the transport pathways, the redox dynamics and the chemical transformations occured such as sorption and solubility. As discussed above, in the case of iron, oxidative precipitation produces solid surfaces that may act as sorbent for other metals. In order to assess the contribution of sorption processess on iron oxyhydroxide mineral surfaces to the fate and transport of metals, it is possible to estimate sorption using existing chemical equilibrium models that have been tested in a wide range of laboratory studies (see Dzombak and Morel, 1991).

# 1.3.1. Sorption and metal speciation in the presence of iron hydroxide mineral surfaces

The term species describes the form in which a metal ion is present in aqueous solution. A metal ion may exist in solution as one or more positively or negatively charged species, as an organic complex, ion pair complex, hydrated complex or hydroxide (Stumm and Morgan, 1996). Understanding the distribution and relative concentrations of metal species, for example as a function of pH, is crucial in order to predict the reactions or type of bonding that the metal may undergo in the natural environment. The extent of adsorption on  $Fe(OH)_3(s)$  as a function of pH is compared for a range of metal contaminants in Figure 1.4. A key point is that adsorption of all of these metals increase strongly with pH, where the pH threshold for adsorption is unique for each contaminant. Lowering pH can therefore strip adsorbed metals from the mineral surface and lead to release of previously scavenged metals contamination to solution. Another important point is that cationic metals species are able to adsorb strongly on positively charged surfaces. The surface of Fe(OH)<sub>3</sub>(s) exhibits variable charge with pH and has zero net charge at circumneutral pH. Below this pH, the mineral surface is positively charged. It is clear from inspection of Figure 4 that  $Cu^{2+}$ ,  $Pb^{2+}$ and Hg<sup>2+</sup> form dominant surface species at pH values where the mineral surface would otherwise be positively charged.

The results from Figure 1.4 demonstrate influence of pH and speciation in sorption behavior for relatively clean systems that are amenable to study by setting up controlled laboratory experiments with synthetic minerals in clean aqueous systems.

A more general overview of the theory and the methods to study these reactions are presented in Banwart (1997). The experimental design for these types of studies provides information about the kinetics of the reaction, species present and sorption reactions occurring at variable or fixed pH. The advantage of these systems is that they allow quantification



Figure 1.4. Adsorption and acid-base behavior of some common metals in the environment.

of the effect of changing one or more known variables systematically, thus allowing investigators to unravel details of the sorption process occurring in the system. In the following section, a case is presented on how these types of aqueous chemistry studies can be combined with advanced surface analysis techniques based on spectroscopy of micro-imaging, in order to obtain thermodynamic data and detailed speciation and chemical bonding information.

## 1.3.2. Sorption and metal speciation in the presence of complex sorbents

Aqueous chemistry methods described previously are usually sufficient to explain speciation and complex formation in controlled clean systems and as first-order estimates for metals scavenging (or release by desorption) in natural waters if the reaction of interest has been demonstrated to dominate. However, in complex adsorption systems such as in soil, or in biofilms on sediment surfaces, the information provided in the existing literature, most generally obtained from studies on single, synthetic minerals, is often not sufficient to describe the reactions occurring. In this case is therefore necessary to combine aqueous chemistry studies with measurements on the sorbent surface. These are observed as a function of changes in the aqueous chemical composition.

In a recent study, surface spectroscopy has been used along with bulk aqueous speciation methods to elucidate the sorption mechanism of metals to a natural biosorbent (Romero-González et al., 2001 and 2003). Information obtained by potentiometric titrations and isotherm studies provided the average stoichiometry of the ion exchange reaction and the nature of the functional groups. This data allowed investigators to determine the thermodynamic constants for formation of the metal complexes on the surface of the biosorbent. Surface and microspectroscopy provided confirmation of the composition and distribution of the functional groups and metal-ligand stoichiometry. It also provided information on the metal coordination environment and bond distances. In turn, a chemical structure of the metal-biosorbent complex may be proposed and metal-binding preferences can be predicted. Figure 1.5 shows the proposed structure for Cd<sup>2+</sup> sorbed to dealginated seaweed waste, a biosorbent that is being studied for its potential as a sorption barrier and waste treatment material for metals.



*Figure 1.5.* Proposed chemical structure for Cadmium sorbed to the surface of dealginated seaweed waste.

The stoichiometry and metal chemical speciation were obtained by ion selective electrode (ISE) and the dissolved concentration by ICP-OES. Dealginated seaweed surface composition and major functional groups were identified by potentiometric titrations and FTIR-ATR. Spatial distribution of metal-biosorbent complexes was obtained by ESEM. Bond distances and coordination chemistry of the structure were elucidated by means of EXAFS. The thermodynamic data produced by this experimental approach can be applied in geochemical speciation models and included in reactive transport models in order to help assess the fate of metals in soil and sediments. An example of existing software that utilizes this approach is the PHREEQC freeware available from the United States Geological Survey (http://water.usgs.gov/software/phreeqc.html). This simulation code contains databases for adsorption reactions, and has the capability for users to additionally define new reactions for simulation of speciation including incorporation into models for reactive transport.

## 1.3.3. Implications for risk assessment

The previous section highlighted the value of characterizing in detail the chemical composition and speciation of the metal and soil or sediment in order to identify and quantify the chemical reactions that are occurring in order to help understand the fate of the contaminant. Such detailed information is necessary to understand the underlying controls of the metal transport and fate in the environment. However, due to the uncertainty that arises from the complexity that exists at field scale, this level of detail is not always directly applicable to risk assessment.

Of crucial importance is understanding sources of uncertainty in an assessment of environmental risk. When making quantitative predictions of metals fate and transport, the types of chemical information described above can help provide important first-order information such as.

- 1. The degree of variability in the dominant soil properties that control adsorption, and
- 2. The sensitivity of metals fate (sorption) to these controls.

Such knowledge helps the practitioner understand how error in site data that is used to estimate sorption and metals fate can propagate through the risk assessment procedure. This allows better-informed decisions and more optimal targeting of the site investigation budget for activities that best help to reduce uncertainty.

Figure 1.6 is a diagram of a tiered risk assessment framework that shows how increased data costs are incurred at higher-tiered risk assessment stages, in order to reduce uncertainty. This helps make the point that the most crucial value of detailed chemical models for adsorption is to provide more general conceptual and mathematical models for this important class of reactions. If improved estimates of metals adsorption can be made using general models and databases, with minimal site-specific data, then site investigation costs drop accordingly without a loss of data quality.



Figure 1.6. A diagram of the tiered risk assessment framework.

For the fate of metals in the environment, this has already been achieved with chemical speciation models. Using general thermodynamic models and databases, and using only site-specific data on water composition and temperature, a full chemical model for the distribution of dissolved species can be carried out. Adsorption data for metals on many pure mineral phases is also now largely available in the scientific literature. As more studies are carried out on adsorption on complex organic and mixed-sorbent systems, it is likely that general speciation models will be able to utilize this information. If this is possible, it should allow better estimates of adsorption processes at the level of desk-studies, thus meeting higher data quality objectives for less cost at lower-tiered risk assessment.

As shown in the figure, the best way to reduce uncertainty in risk assessment is by increasing the quality of the data, at a higher cost an effort. However, the amount of effort and costs in developing conceptual models and remediation design is improving the quantitative analysis of the data. The benefit then of developing more general understanding of the processes that produces contamination will contribute to develop general mathematical descriptions, more fundamental data and parameters that are generally applicable to site-specific conditions.

# 2. Particle and soluble release of organic contaminants from the sediment bed

By definition, sediment contaminants are strongly associated with the solid phase. Thus a first approximation to the transport characteristics of the contamination is the transport behavior of the sediments themselves. This has led to a focus on sediment erosion and deposition as the primary means of contaminant exchange across the sediment-water interface. Typically, however, contaminated sediments have accumulated over time in relatively low energy environments where erosion is not significant or occurs only rarely. The subject of this section is identification of the particle and soluble release processes under such conditions. That such conditions occur is illustrated in Figure 2.1, in which measurements of the vertical profiles of Cs-137, Pb-210 and Be-7 in the sediments of the Anacostia River in Washington DC are shown.



Figure 2.1. Radionuclide profiles in Anacostia River.

The radionuclide profiles shown in Figure 2.1 indicate typical mixing behavior in the surficial sediments and the long term stability of the sediments. Be-7 is a short lived radionuclide and its decrease in the upper 4 cm of sediment implies a rate of mixing of that layer (6 cm/yr). The long

term stability is shown by the linear decrease in excess Pb-210 with depth, which suggests a steady rate of deposition without evidence of any erosional events. This is further indicated by Cs-137, in that the presence of this radionuclide indicates sediment at the surface since 1954 when this radionuclide was first introduced to atmosphere by nuclear testing. The release of this radionuclide reached a maximum in 1963, identifying the date associated with the peak in the Cs-137 prfofile. Both the Cs-137 profile and the rate of burial of Pb-210 suggest a steady sediment deposition rate of 0.44-0.66 cm/yr.

In steady depositional situations such as this there is little or no erosion. Because only situations in which there is no net loss of particles are considered, the abiotic release of contaminants across the sediment-water interface is controlled by soluble release. Benthic organisms that live at the sediment-water interface may cause significant reworking of particles within the surficial sediments but are assumed to cause no net release of particles from the bed. Organisms can, however, provide another mechanism for movement across the sediment-water interface by accumulating contaminants within their body mass and introducing these contaminants into the food chain via predation.

Thus, the movement of contaminants into the water column in a stable sediment environment is controlled by the soluble release and by that fraction of contaminants that enter the food chain via accumulation in benthic organisms. The controlling factors and the rate of these processes will be discussed in this section. It will be concluded that both mechanisms are ultimately controlled by the soluble fraction of contaminants and not the bulk sediment concentrations.

## 2.1. TRANSPORT PROCESSES WITHIN SEDIMENTS

In a stable sediment, that is, one not subject to erosion, there still exist mechanisms for movement of chemicals via pore water processes such as advection and diffusion, as well particle reworking. Of these, any mechanism that causes particle reworking will dominate over pore water processes due to the sorbing nature of most sediment contaminants. Movement of gas or pore water at sufficient rates can move particles but the flows necessary to cause such behavior are not generally observed. Benthic organisms and bioturbation, the mixing process associated with the normal burrowing and feeding activities of these organisms, generally represent the dominant sediment reworking mechanism in sediments. Many of these organisms process sediment directly for food and thus in one feeding cycle may cause movement of chemicals that might require months or years by pore water processes. In freshwater systems, for example, tubificid oligochaetes that are a few cm long and weighing but a few mg, can be found in concentrations approaching 100,000 organisms/m<sup>2</sup> and each may ingest up to 10-20 times their weight in sediment each day. These organisms are generally limited to the upper 10-15 cm of the sediments and their greatest activity is limited to the upper 5 cm. The penetration of Be-7 into the Anacostia River sediments as shown in Figure 2.1 is likely due to the action of these organisms. Thus the Be-7 profile shows both the depth of activity (~4 cm) as well as the rate of activity (~6 cm/yr) resulting from these organisms.

#### 2.2. SOLUBLE RELEASE FROM STABLE SEDIMENTS

In a stable sediment bed, that is, one not subject to net particle loss, contaminant movement into the water column is limited to soluble release. Other processes may rework the surficial sediments and control the rate at which the sediment surface is renewed with contaminants but the rate at which contaminants at this surface move into the water is controlled by soluble processes. Included in soluble release is contamination that is truly dissolved as well as contamination that is associated with colloidal matter that is operationally defined as dissolved. There is an important difference between these two components, however, in that only the truly dissolved fraction is available to directly partition into air, water or organic matter. That fraction associated with colloidal material must normally desorb back to truly dissolved form before partitioning.

Note that the focus on no net loss of sediment still allows short term resuspension of particles, for example by organisms that eject particles forcibly into the water column which settle back quickly to the adjacent sediment surface. In addition, tidal or wave action can cause similar shortterm resuspension of sediments that can quickly settle back to the bed surface. The significance of these processes for contaminant release is still dependent upon desorption into the water column and the rate of soluble release during their duration of the particle's exposure to the water column.

The net rate of contaminant release through the sediment water interface can be conceptualized as a process of mass transfer through the upper sediment layers followed by a process of mass transfer through the benthic boundary layer. These two processes operate sequentially and the two-film model can be adapted to describe transport through the sediment-water interface. Conceptually, this approach recognizes that under steady conditions, the flux to the sediment-water interface is equal to the flux away from this interface. Thus the flux,  $N_A$ , through the sediment-water interface can be written

$$\frac{N_A}{C_{pw} - C^*} = k_{sw} = \frac{1}{1/k_w + 1/(K_{sw}k_s\rho_s)}$$

Where  $C_{pw}$  is the pore water concentration of the contaminant below the sediment surface layer,  $C^*$  is the concentration that would be in the pore water if it were in equilibrium with the water (0 if the water were devoid of significant contamination). The mass transfer coefficients,  $k_w$ ,  $k_s$ , and  $k_{sw}$ , are defined in the water film, surface sediment layers and overall across the sediment-water interface, respectively.  $K_{sw}$  is the effective partition coefficient between the solid phase and the pore water in the sediment.  $\rho_s$  is the average density of the particles in the bed.

The above equation can be used to determine the relative importance of the mass transfer resistance posed by the water or sediment side processes. The values of the water-side mass transfer coefficient, k<sub>w</sub>, is a relatively well understood function of flow in the overlying water. It is typically of the order of 1 cm/hr and Thibodeaux et al. (2002) determined a range of 12.3-33.3 cm/day for several rivers, including rivers subject to erosion and rivers in which the sediment bed was essentially stable. The value of k<sub>s</sub> must be estimated on the basis of measurements of sediment reworking rates and Thibodeaux et al. (2002) estimated a range of 2-3.6 cm/yr for the same rivers. As indicated previously the Be-7 profile in the Anacostia River core referenced earlier was consistent with an effective mass transfer coefficient of 6 cm/yr. Although the numerical value of the coefficient is small, it is based on solid phase concentrations which are considerably higher than pore water or soluble concentrations. The presence of K<sub>sw</sub> in the above equation is the result of placing the two mass transfer coefficients on the same basis. Assuming a solid phase density of 1 g/cm<sup>3</sup> and a  $K_{sw}$  of  $10^3$ - $10^5$  L/kg, the k<sub>s</sub> range of 2-3.6 cm/yr causes the term K<sub>sw</sub>k<sub>s</sub> $\rho_s$  to range between 5.5 cm/day and 1000 cm/day. At the lower end of this range, associated with low sediment reworking rates and low preference for sorption to the solid phase, sediment side mass transfer resistances can control contaminant release. At the higher end of this range, associated with high sediment reworking rates and very hydrophobic and sorbing compounds, the water side mass transfer resistances control contaminant release.

### 2.3. CONTAMINANT MIGRATION AS A RESULT OF BIOTURBATION

We thus have a picture of the stable sediment-water interface in which the upper layers of the sediment are relatively rapidly mixed and the benthic boundary layer resistance may control mass transfer into the overlying water. The question arises as to the cause of the relatively rapid mixing in the surficial sediment layers. Advective and diffusive processes are operative only on contaminants present in pore water and thus strongly sorbed contaminants are unlikely to be mixed at the observed rates by these processes. Bioturbation, however, the sediment mixing processes associated with the normal burrowing and foraging activities of benthic organisms, can move sediment and associated contaminants at these rates. Deposit feeding tubificid worms, for example, typically are present in densities approaching 100000 m<sup>2</sup> and process 1-5 mg/worm per day in the upper 5 cm of sediment. This is the equivalent of 100-500 g of sediment being reworked per day per  $m^2$  of sediment. Assuming that the sediment has a dry density of 1 g/cm<sup>3</sup>, this implies that the upper 5 cm of sediment (50000 cm<sup>3</sup> per m<sup>2</sup>) is completely processed in 100-500 days, or an effective reworking rate of 3-15 cm/yr. Thus bioturbation is consistent with the observed rates of reworking of a stable sediment. By comparison, a relatively rapid groundwater seepage rate of 1 cm/day with a compound exhibiting an effective partition coefficient of 1000 L/kg, would only account for a sediment reworking rate of approximately 0.3 cm/yr, smaller than only account for a sediment reworking rate of approximately 0.3 cm/yr, smaller than typical sediment mixing rates. Lower seepage rates would give rise to even lower effective sediment mixing rates.

We thus have an image of a stable sediment as one that is dominated by organism associated sediment reworking within the sediment layer and contaminant migration into the overlying water limited by the rate of sediment reworking and transport through the benthic boundary layer. Bioturbation also has other influences on contaminant fate in the sediments as illustrated by a laboratory study of contaminant depletion with time in a sediment exhibiting an initial uniform concentration of pyrene. The concentration profile initially (broken line) is contrasted with the concentration profile after approximately 3 months with (filled bars) and without (open bars) the presence of benthic organisms (a tubificid oligochaete) is shown in Figure 2.2. Significant depletion of pyrene is observed to extend deeper in the organism populated cells, presumably as a result of deeper oxygen penetration and aerobic degradation of pyrene in those cells.



Figure 2.2. Depletion of pyrene in a laboratory microcosm with and without faunal influence (tubificid oligochaetes).

## 2.4. CONTAMINANT ACCUMULATION IN BENTHIC ORGANISMS

In addition to providing a mechanism for rapid reworking of the surficial sediment and transport of any associated contaminants, benthic organisms also introduce contaminants into the food chain. Organisms ingest or absorb contaminants and these accumulated contaminants can be transferred into the food chain when fish or other predators feed on these organisms. The amount of contaminant that accumulates in these organisms also relates directly to any toxic effects on these organisms. Assessment of contaminant burden that can accumulate in benthic organisms is thus a good indication of the risk associated with the sediment contamination.

Potentially complicating contaminant accumulation in benthic organisms is the strongly sorbed nature of some sediment contaminants. Contaminant sorption or desorption often tends to display biphasic character with a fraction of contaminants readily desorbable and a fraction of contaminants which exhibit desorption that is limited by rate and extent. The question arises as to what extent this desorption resistant fraction of contaminants is available for accumulation in benthic organisms.

This has been studied extensively by Lu et al. (2003) and this work has shown that despite the route of exposure, benthic organisms can ultimately accumulate a contaminant burden that is related to the pore water concentration of contaminant. That is, to the extent that pore water concentrations are reduced by the presence of desorption–resistance phenomena, this would directly reduce steady state accumulation and uptake of contaminants into a benthic organism.

This is illustrated in Figure 2.3, in which the steady state uptake of a variety of polynuclear aromatic hydrocarbons into a tubificid oligochaete is summarized. In every case, the steady state body burden in the organism correlated well with the dissolved phase porewater concentration of the PAH contaminant.



*Figure 2.3.* Steady state uptake of selected PAHs in tubificid oligochaetes compared to uptake predicted by dissolved phase pore water concentration of the PAHs.

## 2.5. SUMMARY

This section has demonstrated that the migration and fate of sediment contaminants is often controlled by soluble release. Although sediment contaminants are typically strongly associated with the solid phase, the accumulation of these contaminants is often in locations that are not subject to erosion. In a stable sediment environment, contaminant migration is typically controlled by the reworking of the surface layers of the sediment by the benthic organisms. This process of bioturbation results in rapid movement of contaminants to the sediment surface followed by dissolution and soluble release to the overlying water. In addition, contaminants in this

101

surface layer can be taken up directly by the benthic organisms and introduced into the food chain. Experiments show, however, that even ingestion of strongly hydrophobic contaminants in sediments results in body burdens in the benthic organisms that can be related to the soluble contaminant levels in the pore water.

Thus contaminant movement from sediment into the overlying water can be controlled and defined by soluble release and uptake, at least under conditions such that significant erosion of the sediments does not occur. This soluble component controls or at least indicates both movement into the overlying water by direct release from the sediment bed as well as by introduction into the food chain through accumulation in benthic organisms. Movement within the sediment may be controlled by particle reworking associated with bioturbation but movement into the overlying water in a stable sediment must then be via the soluble fraction.

# **3.** Biogenic influences on sediment stability and rheology: examples from long-term mesocosm study

## 3.1. INTRODUCTION

The upper few centimeters of aquatic sediments are characterized by extreme gradients in sediment porosity and strength. These gradients exert strong influences on the response of sediments to hydrodynamic and mechanical stresses, and on the exchange of pore fluids and particles (including contaminants) with overlying water, as discussed in Section 2.0. A large body of research suggests that bioturbation by infauna exerts significant influence on shallow-seabed physical properties (e.g., Rhoads and Young, 1970; Rhoads and Boyer, 1982; Richardson et al., 1983; Meadows and Tait, 1989; Jones and Jago, 1993; Rowden et al., 1998) and dissolved and solid-phase chemical transport (Berner, 1980; Boudreau, 1997, and references therein). However, little is known about the feeding and burrowing activities of most benthic macrofauna, and even less is known of how macrofaunal behavior can influence these sediment characteristics. Sediment porosity and shear strength are fundamental properties of the seabed, and understanding biogenically induced variability for these properties is critical to many investigations of seabed stability. sediment dynamics, and contaminant fate and transport. Bioturbation of the seabed has been shown to influence sediment shear strength and erodibility (Rhoads et al., 1978; Dade et al., 1992; Wiberg, 1999), however, both our basic understanding of these influences and our ability to model them remain minimal.

In this study, we examine the temporal and spatial influence of bioturbation on sediment shear strength and porosity using long-term mesocosm observations of a subsurface-deposit feeding enteropneust (Schizocardium sp.), and a surface-deposit feeding ophiuroid (Hemipholis elongata), placed in compacted, muddy sediment. Specifically, we wanted to know if these organisms are capable creating the vertical porosity and strength distribution typical of modern surficial marine muds (e.g., Dade et al., 1992; Boudreau and Bennett, 1999). The burrowing and feeding styles of these species are typical for two major groups of benthic invertebrates. Schizocardium is a funnel-feeding deposit feeder (Fig. 3.1), similar to numerous species of polychaetes, enteropneusts, and holothurians that are distributed from estuarine to bathyal sediments. Hemipholis elongata is a surface-deposit feeder that dwells in a simple burrow (Fig. 3.1); many common bivalves and polychaetes have similar lifestyles (see Bromley, 1996, for an excellent review). Both species are common at high population densities (Flint and Kalke, 1986; Valentine, 1991; Rowden et al., 1998), and are probably the most important bioturbators at our collection sites.

## 3.2. METHODS

Sediment and organisms were collected by box-core from stations in Mississippi Sound (*Schizocardium*: 89° 20.0'N, 30° 14.0'W, 3 m depth; *Hemipholis*: 30° 16.0'N, 88° 46.0'W, 9 m depth). Sediment was sieved at 0.5 mm, placed in glass mesocosm tanks (60 cm wide x 60 cm long x 45 cm deep), and allowed to settle for 3 weeks. Sediment was compacted with ~90 kg of lead to reach a porosity of 0.65-0.7. Sediment for a macrofauna-free mesocosm was frozen to eliminate meiofauna, dewatering the sediment to a porosity of ~0.77. This sediment was not compacted additionally, because it was also intended as a control for other coincident experiments. Mesocosm populations for *Schizocardium* (420 m<sup>-2</sup>) and *Hemipholis* (300 m<sup>-2</sup>) were established at mean field densities, and were fed small quantities of fish food biweekly. Salinity (*Schizocardium*: 17 psu; *Hemipholis*: 27 psu) and temperature (19-20° C) were monitored and adjusted twice weekly. Timed fluorescent lamps were illuminated over the tanks for 10 hours per day. Water was filtered through external sumps filled with aragonite gravel.

Burrow morphology and changes in surface-sediment relief were documented by photography throughout the experiment. Surface relief was also assessed during resistivity profiling.

Resistivity was measured using a small, Wenner-type probe (Keller and Frishknecht, 1966). The probe was made of stainless-steel electrodes potted in epoxy resin, with dimensions of 3 x 10 mm, mounted on 3-mm o.d. x 60-cm tube. The signal (100Hz, 1.0 V square wave) was supplied

with a Hewlett Packard HP3312A function generator. Voltage from measurement electrodes was measured with a true-rms multimeter. Measurements were made 2 cm above the sediment-water interface, and at 1-cm vertical intervals within the sediment. The probe was calibrated in seawater (17 and 27 psu) prior to each series of measurements. Porosity was estimated using the Archie relationship (e.g., Jackson et al., 1978; Briggs et al., 1998),  $FF = R_s/R_f = \phi^{-m}$ , where FF is the formation factor,  $R_s$  and  $R_f$  are resistivity of saturated sediment and overlying water (or measured voltages), respectively,  $\phi$  is porosity, and *m* is a constant derived from probe calibration, (for this study, m = 1.69 ± 0.16). Porosity estimated from FF was validated against porosity measured by water loss at the start and end of the experiment.

Shear-strength profiles were collected with a hand-held shear vane (4 blades, 1.9 x 1.9 cm diameter x height) attached to a torque-measuring screwdriver (Rhoads and Boyer, 1982; Briggs and Richardson, 1996). Measured torque was corrected for sediment-rod friction (Briggs and Richardson, 1996). Shear strength ( $\tau_f$ ) was calculated from (Holtz and Kovacs, 1981):  $\tau_f = (6T_{max})/(7\pi D^3)$ , where  $T_{max}$  is measured torque, and D is vane height and width. Shear-strength and resistivity profiles were collected at 10-cm intervals along transects located to avoid previous measurements.

## 3.3. RESULTS

## 3.3.1. Biogenic sedimentary features

Generalized burrow morphology for *Schizocardium* and *Hemipholis* are shown in Figure 3.1. A typical *Schizocardium* burrow consists U-shaped burrow extending >10 cm deep, with a feeding pit (1-3 cm deep) at one opening of the burrow and a fecal mound (1-4 cm tall) at the other (Fig. 3.1), creating total relief of 2-7 cm. As observed through tank walls, the basal portions of burrows are relatively long-lived features (stable for months), whereas the locations of burrow openings are shifted on the order of 1 cm every few days.

A *Hemipholis* burrow is a simple feature, shaped like an inverted bell extending 5-9 cm into the substrate (Fig. 3.1). The central disc of the ophiuroid remains hidden at the base of the burrow, while arms are extended into the water column and across the sediment surface to gather food particles. Occasionally, arms are extended through the main burrow wall, exiting the sediment away from the primary burrow opening. Mounds created by burrow excavation (and possibly feeding activity) produce gentle relief of 1-2 cm on the sediment surface. Burrow locations are not

permanent, but the mechanism of burrow relocation (i.e., sideways excavation versus emergence and excavation of an entirely new burrow) is not clear. Experiments documenting more detailed burrow geometry and





bioturbation rates for both *Schizocardium* and *Hemipholis* are presently underway.

Figure 3.1. Illustrations of Schizocardium and Heminholis burrows. (A) Schematic illustration of entire burrow for each organism (scale is approximate). (B) Photo of *Hemipholis* organisms and burrows through aquarium wall. Hemipholis arms are frequently extended into the water column, evidently searching for food particles, but are also seen sifting through surficial sediment. (C) Photo of Schizocardium feeding pit through aquarium wall, showing clear geochemical gradients in the sediment. Schizocardium appears ingest both consolidated to sediment from walls of the feeding pit, as well as material that falls into the pit. Sediment is egested to the fecal mound as puffs of mucous-laden flocs. with no pelletization or coiled structure.

## 3.3.2. Porosity

The macrofauna-free mesocosm had a vertical initial porosity profile that remained constant for the lower 12 cm of the sediment column throughout the study. However, the upper 2 cm increased in porosity and thickness after day 44, due to inadvertent introduction of meiofauna (very fine burrows were visible in the upper 2-3 cm of sediment). Porosity profiles from Day 98 onward development of a high-porosity layer in the upper 2 cm, with respect to initial conditions (Table 3.1, Fig. 3.2).



Figure 3.2. Time-series profiles of formation factor and estimated porosity (m=1.69) for experimental mesocosms (mean and standard deviation of 8-11 profiles). Vertical datum is the tank floor, and ISS refers to the sedimentsurface elevation at initiation of the experi-ment. In each case, bio-turbation resulted in dilation of some portion of the sediment column, increasing total sediment thickness. The basal gradient the in Hemipholis resistivity profiles results from differential settling of sand during substrate preparation. Sediment from the Hemipholis collection location is ~30:70% sand: whereas sediment mud. Schizocardium used for and macrofauna-free mesocosms is  $\sim 10:90\%$  sand: mud.

In the *Schizocardium* mesocosm, the initial nearly vertical profile ( $\phi \sim 0.7$ ) evolved to an exponential-type profile with high surface porosity ( $\phi > 0.9$ ), and slightly increased basal porosity ( $\phi \sim 0.72$ )(Fig. 3.2). Sediment thickness increased by ~4 cm, due to the dilation of sediment excavated from feeding pits and ejected onto fecal mounds (Fig. 3.2, Table 3.1). Formation of pits and fecal mounds also increased relief. Dilation (or excavation) was most pronounced in the upper ~7 cm (depth of Z<sub>50</sub>, Table 3.1), but occurred throughout the sediment column, increasing the volume of the upper 12cm (depth to Z<sub>90</sub>, Table 3.1) by a factor of ~1.4 during the experiment.

Porosity profiles from the *Hemipholis* mesocosm were strongly influenced by the presence of a basal sandy layer that settled out of suspension during substrate preparation (Fig. 3.2). The formation of a high-porosity surface layer is the most obvious development during the experiment, during which the volume of the upper ~4 cm (depth to  $Z_{90}$ , Table 3.1) increased by a factor of ~1.25.

For all mesocosms, sediment inventories indicate close mass balance (and so no measurement artifacts) between initial and final measurements (Table 3.1). However, measurements in the *Schizocardium* (Day 240) and *Hemipholis* (Day 198) tanks at intermediate times produced anomalously high mean porosities and low mass inventories for both tanks (Fig. 3.2). The cause of this is not known, and these data are not used for calculation here.

### 3.3.3. Shear Strength

Shear strength increased by  $\sim 0.5$  kPa at all depths in the macrofauna-free mesocosm during the experiment, reaching an apparently stable condition after Day 101 (Fig. 3.3). In the both *Schizocardium* and *Hemipholis* mesocosms, shear strength decreased above 9 cm elevation, and increased below 9 cm (Fig. 3.3), but the magnitude of change was greatest in the *Schizocardium* mesocosm. Both populated mesocosms reached apparently stable conditions after 3-4 months of bioturbation (Fig. 3.3).



*Figure 3.3.* Time-series profiles of shear strength in mesocosm profiles (mean and standard deviation of 5-8 profiles). Vertical datum is the tank floor, and ISS refers to the sediment-surface elevation at initiation of the experiment. In each case, macrofaunal bioturbation resulted in sediment weakening with respect to the macrofauna-free sediment, which strengthened during the measurement period.

### 3.4. DISCUSSION

These results document the biogenic conversion of a relatively firm, homogenous substrate into а very watery. weak, and spatially heterogeneous substrate typical of the muddy seafloor (e.g., Dade et al., 1992; Boudreau and Bennett, 1999). Schizocardium is the most effective of the two taxa at modifying both porosity and shear strength, probably because it ingests large volumes of sediment below the sediment surface, egesting the particles as a dilute sediment suspension. (In contrast, Hemipholis appears to be a particle-selective surface feeder [Gielazyn et al., 1999; this study]). In the Schizocardium mesocosm, the largest changes of shear strength and porosity occur in the upper 6-8 cm of the substrate, although burrows extend much deeper (Fig. 3.1). This probably indicates that Schizocardium effects on shear strength and porosity are most closely tied to depth and rate of particle ingestion and egestion (feeding pit and fecal mound formation) rather than burrow depth (Table 3.1, Figs. 3.1-3.3).

Schizocardium Hemipholis Macrofauna-free Duration of Experiment (d) 240 336 281 Sediment Inventory Day 0  $11.8\pm0.09$  $8.25 \pm 0.06$  $13.05 \pm 0.26$  $(g \text{ cm}^{-2})$ Sediment Inventory at 11.94±0.35 12.72±0.34 8.20±0.13 Experiment End (g cm<sup>-2</sup>) Sediment Mass Lost  $0.33 \pm 0.13$  $0.93 \pm 0.36$  $0.70\pm0.43$ Below Initial Sediment Surface (ISS)  $(g \text{ cm}^{-2})$ Mass transferred above ISS 0.29 0.38  $0.86 \pm 0.04$  $(g \text{ cm}^{-2})$ Maximum Elevation 1 1 4 Gained Above ISS (cm) Z<sub>50</sub>: Depth Below ISS to < 1 7 2 50% of Sediment Loss (cm) Z<sub>90</sub>: Depth Below ISS to 2 12 4 90% of Sediment Loss (cm) % Dilation for sediment 129% 272% 189% moved from Z<sub>50</sub> to sediment surface, based on observed porosities.

*Table 3.1.* Sediment mass balance and transfer due to bioturbation, estimated from porosity profiles (Fig. 3.2). Vertical datum for comparison is elevation of initial sediment surface (ISS) from tank floor. Particle density of 2.6 g cm-3 was used for calculations.

The effects of *Hemipholis* bioturbation on shear strength (Fig. 3.3) do not appear to be tied to porosity changes or feeding depth. Because the burrow depth (Fig. 3.1) and depth of shear-strength influence (Fig. 3.3) are similar, this biogenic remolding is probably related to the rate and depth of burrow formation (not feeding) by the *Hemipholis* population, in contrast to *Schizocardium*.

These observations have several important consequences. First, specific biological activities can dilate and weaken sediments in opposition to gravitational consolidation. These opposing processes appear to reach a steady state after 3-4 months. Second, because a functional relationship appears to exist between depth/rate of biogenic activity and depth/intensity of effect, these influences can probably be incorporated into porosity and consolidation models (e.g., Toorman, 1996; Boudreau and Bennett, 1999) in terms similar to those used in diagenetic models (biodiffusion coefficient, mixing/ingestion depth, ingestion rate: Boudreau, 1986; Rice,

1986; Robbins, 1986; Wheatcroft et al., 1990) to portray work done by organisms to create and maintain sediment in a dilated or remolded state.

Future experiments will look at bioturbation rates and depths, and changes in electrical resistivity (porosity/density) and shear strength in initially *unconsolidated* sediment populated by *Schizocardium* and *Hemipholis*. We hypothesize that porosity and strength profiles similar to Figures 3.2 and 3.3 will be established in spite of very different initial conditions. This leads to the prediction of species/sediment specific steady-state profiles of sediment properties in nature.

## 4. Biogeochemistry of humic substances in a salt marsh estuary

## 4.1. INTRODUCTION

Salt marshes represent natural ecosystem-level interfaces between terrestrial, fresh and salt water environments. Three major components of this interface are sediment, tidal sea water, and many autochthonous organisms. A major plant component is a vascular plant *Spartina alterniflora* (Loisel.). This smooth cordgrasss densely occupies salt marshes at the East and Gulf Coast of the USA, and is responsible for very high primary production, which amounts approximately 1400 g C m<sup>-1</sup> y<sup>-1</sup> (Pomeroy et al., 1981). Although large amounts of different natural and anthropogenic organic and inorganic materials, including pollutants are permanently introduced into salt marshes especially by rivers draining agricultural and forested watersheds, these natural environments remain keeping high productivity and other environmentaly important functions. Apparently, some site-specific components such as humic substances may support these valuable natural properties of a salt marsh estuary (Alberts et al., 1994).

Humic substances have been shown repeatedly to represent a significant fraction of the natural organic matter in both terrestrial and aquatic environments. Also, HS are known to interact with a wide spectrum of inorganic and organic compounds and elements through an array of chemical reaction and physical binding that can influence bioavailability and toxicity of different pollutants (Frimmel et al., 2002; Senesi, 1992). Such a capacity have been repeatedly described for HS in seawater and sediments (Ertel and Hedges, 1983; Harvey and Boran, 1985; Vanderbroucke et al., 1985). In addressing geochemical interactions in the salt marsh, Pellenbarg (1985) examined how *S. alterniflora* litter can interact with waters and sediments, including interaction with some potentially toxic elements.

Since both the tidal water and salt marsh sediment usually consist HS different in concentration and quality, we performed extended studies over years to elucidate the origin and fate of HS in a salt marsh estuary. In our experiments a possible contribution of *S. alterniflora*-plants to the HS-pool was emphasized. It is the purpose of this report to summarize the findings achieved. In its individual sections, the respective references will be indicated.

### 4.2. S. ALTERNIFLORA AS A SOURCE OF HUMIC SUBSTANCES

In a primary study devoted to the topic we investigated whether *S. alterniflora*-biomass consists of constituents which might resemble to HS from a salt marsh sediment (Filip et al., 1988). For this reason, samples of fresh and standing-dead leaves and culms of *S. alterniflora* were extracted under N<sub>2</sub> with 0.1 *M* NaOH + 0.1 *M* Na<sub>4</sub>P<sub>2</sub>O<sub>7</sub> mixture (1:1 v/v) for 24 h. The supernatant was acidified with HCl to pH 1 and HS were allowed to precipitate overnight. They were than separated by centrifugation, dialyzed against deionised water and freeze dried. Salt marsh sediment was sampled from a depth of 0-15 cm at the same site as the plants, dried at 60°C, finely ground, and extracted in the same way as the plant material. In Table 4.1 the amounts and chemical composition of HS yielded are presented.

Sample	Humic substances	Ele	)		
	(g dry wt./100g)	С	Н	Ν	0
fresh	0.56	46.0	5.9	5.8	42.3
dead	1.09	52.0	5.6	3.6	33.8
Sediment	2.17	40.4	4.5	2.5	52.2

*Table 4.1.* The yield and elemental composition of humic substances extracted from the biomass of S. alterniflora, and from salt marsh sediment (Filip et al., 1988).

It was calculated that 0.64% of the C of the fresh plant biomass was extracted as HS, and that in the dead plant biomass, and salt marsh sediment this percentages increased to 1.42 and 9.38 %, respectively. Similarly, Ertel and Hedges (1985) found that humic acids accounted for 1-2% of the carbon from an oak wood. In our experiments, HS were extracted under  $N_2$  in order to minimize a possible autooxidative polymerization of some phenolic plant constituents.



*Figure 4.1.* FTRI spectra of the S. alterniflora-dead plant biomass (A), and a humic substance extracted from the same material (B). (Filip et al., 1988).

The FTIR spectra (Fig. 4.1) added support to the idea that HS were present in plant tissues before the extraction: Although there are some quantitative differences, almost all the absorption peaks observed in the dead plant-related HS have their counterparts in the spectrum of the parent plant material. The similarity of the individual HS-preparations were evidenced also by UV/VIS spectroscopy. In addition, and irrespective of their origin, the individual HS were shown as concentrating similar amounts of metals such as Cd, Cr, Cu, Mo, Ni, Pb, Sn, Zn. FTIR spectral analyses of different artificially prepared complexes of HS with Ag, Au, Cu, Hg, Mn and Pb suggested that oxygen- and nitrogen-containing functional groups are involved in the metal-binding reactions (Alberts and Filip, 1998).

## 4.3. MICROBIAL CONTRIBUTION TO THE FORMATION OF S. ALTERNIFLORA-RELATED HUMIC SUBSTANCES

In situ both the green and senescent leaves and stems of *S. alterniflora* are inhabited by numerous microscopic fungi that participate in the decomposition of plant biomass (Gessner and Goos, 1973; Newell, 1984; Leuchtmann and Newell, 1991). Among them, the Deuteromycete *Epicoccum nigrum* was found already earlier to produce dark polymers resembling soil humic substances (Filip et al., 1972). A question arose whether some of the Ascomycetes such as *Phaeosphaeria spartinicola* and *Phaeosphaeria halima* which usually colonize *S. alterniflora* plants are also capable of forming HS-like polymers. Filip and Alberts (1993) performed laboratory experiments of up to one year in duration to elucidate this. They found that fungi did not form dark in color pigments when grown in an

artificial, full-nutrient medium. However, if the nutrient solution was enriched with a water extract of *S. alterniflora*-biomass, appreciable amounts of dark colored substances were produced. The FTIR spectra of the respective compounds (Fig. 4.2) demonstrated a high degree of similarity especially between the polymer isolated from the *P. halima*-culture and the HS extracted from *S. alterniflora*-dead plant biomass.



*Figure 4.2.* FTIR spectra of (A) polymer isolated from culture of P. spartinicola grown in a nutrient solution amended with S. alterniflora extract; (B) polymer isolated from culture of P. Halima grown as under (A); (C) humic substance extracted from S. alterniflora-dead plant biomass (Filip and Alberts, 1993).

## 4.4. THE RELEASE OF HUMIC SUBSTANCES FROM *S. ALTERNIFLORA* INTO SEAWATER

Different simple in structure organic compounds, e.g., monosaccharides, amino acids, and phenols, have been detected in samples of both fresh and dead *S. alterniflora,* and also in HS related to the respective plant biomass (Alberts et al., 1992). Several authors have found that in part these compounds are leached from the plants into seawater by the rising tide, and

may influence carbon metabolism of the autochthonous microbial communities (Fallon and Pfaender, 1876; Gallagher et al., 1976; Pakulski, 1986). In our experiments (Filip and Alberts, 1988), we wished to find out whether, perhaps, also HS from *S. alterniflora* plant biomass can be released into seawater and whether microorganisms indigenous to a salt marsh environment are capable of influencing such a leaching. For this reason equivalent amounts of fresh and dead plant material were incubated in seawater under either sterile conditions or in the presence of different autochthonous microorganisms. Within 10 months the plant material inoculated by *S. alterniflora* epiphytic microbial populations or by a population indigenous to salt-marsh sediment developed a milky skin on the surface, and a dense turbidity in seawater. As shown in Table 4.2, different amounts of HS were obtained from the individual samples of seawater (that was made free of particulate material prior to the isolation of HS).

Sample and inoculum	Humic substances (dry weight)			
	mg $l^{-1}$	%		
Seawater with S. alterniflora fresh				
Sterile control	198.7	100.0		
Epiphytic microbial population	994.3	500.4		
Microbial population from sediment	324.7	163.4		
Seawater with S. alterniflora dead				
Sterile control	24.2	100.0		
Epiphytic microbial population	417.8	1726.4		
Microbial population from sediment	301.6	1246.3		

*Table 4.2.* Yield of humic substances from the seawater long-term incubated with S. alterniflora plant material added. (Filip and Alberts, 1988).

The data indicate that the fresh plant material can release up to 820% more HS into seawater than the dead one. However, if related to the sterile control, the effectiveness of the microbial populations was much greater with the dead plant material. Nevertheless, both fresh and dead biomass of *S. alterniflora* is apparently capable of contributing not only some simple organic compounds but also considerable amounts of HS to the pool of dissolved organic carbon in seawater. In addition, the remaining fresh and dead plant material appeared another possible source of HS to the salt marsh environment. If extracted by alkali (see in Paragraph 1) appreciable

amounts of HS could be additionally obtained (Filip and Alberts, 1989). According data published by Moran and Hodson (Moran and Hodson, 1990) up to 20% of the total dissolved organic matter appears in the form of HS on the southeastern USA continental shelf.

## 4.5. MICROBIAL UTILIZATION OF SALT MARSH-RELATED HUMIC SUBSTANCES

A principal question in biogeochemical studies concerns the stability of organic substances in a diagenetic environment. According to Rashid (1985), humification, i.e. formation of HS, represents the major process in early diagenesis. Thus, it should be recognized that *S. alterniflora* plants represent an important factor in diagenetic processes occurring in a salt marsh estuary. Usually, HS are considered as refractory organic matter (Frimmel et al., 2002), and accordingly, salt marsh-related HS were considered as little degradable by microbial activities (Pomeroy and Imberger, 1981; Fogel et al., 1989; Esham et al., 2000).

In a specific experimental approach, we wished to established, whether S. alterniflora-related HS, and those extracted from salt marsh sediment can resist to attack by microorganisms indigenous to that environment (Filip and Alberts, 1994). The HS were used as additional sources of nutrients or as single sources of either C or N in a nutrient broth. The individual preparations were incubated for 15 days (aerobic cultures) or 12 months (semi-anaerobic cultures) with a complex population of salt marsh sediment-related microorganisms. Depending on cultivation conditions and type of HS, between 27% and 100% of the added HS were utilized. The highest utilization rates were obtained with HS related to S. alterniflora fresh and used as the sole source of N, but also sediment-related HS served as efficient nitrogen sources in aerobic cultures (78.1% utilized). The remaining HS-preparations recovered from aerobic cultures exhibited elemental and structural changes that are characteristic for diagenetic transformations of sedimentary HS: Increase in C content, C:H and C:N ratios, and IR absorption typical for aromatics. The slopes of the spectra in the visible range of light indicate an increasing optical density, and thus somewhat higher degree of "structural condensation" for the HS preparations re-isolated from microbial cultures (Fig. 4.3). Under semianaerobic cultural conditions the diagenetic transformations of HS did not occur in the same way as in the aerobic cultures. Rather, within a year the HS from S. alterniflora fresh was completely utilized by microorganisms. Esham et al. (2000) found a bacterial community from seawater capable of mineralizing only up to 3% of HS obtained from a decomposing S. alterniflora biomass. However, these authors used only indirect methods such as  $CO_2$  measurements and a loss of light absorbance as an index of HS-degradation which may have influenced final results as recognized earlier (Filip et al., 1998).



*Figure 4.3.* Visible spectra of humic substances (A) extracted from salt-marsh sediment; (B) the same substances incubated for 15 days under aerobic conditions in a full-strength nutrient broth; (C) the same substances incubated as a sole source of carbon; (D) the same substances incubated as a sole source of nitrogen (Filip and Alberts, 1994).

## 4.6. ADSORPTION AND TRANSFORMATION OF HUMIC SUBSTANCES BY QUARTZ AND CLAY MINERALS

The adsorption of organic material dissolved in seawater by suspended mineral particles may prove to be a very significant process in removal of such organic matter from water and of incorporating it into the bottom deposits. Concerning salt marsh-related HS, one can assume that their adsorption on sedimentary particles may play a role in supporting their protection from biological decomposition, and thus making them geochemically more stable.

Rivers deliver large quantities of mineral materials into the estuaries, from which about 30% are clay minerals, particularly montmorillonite and

kaolinite (Bader et al., 1960). In addition, the tidally-influenced water bodies receive an important influx of clays from shorline erosion and bottom scouring. Botom sediments of Sapelo Sound (which was our main experimental site) contain a clay fraction composed of 30-45% kaolinite, and 50-60% of expandable clays (Pinet and Morgan, 1979). Some other salt marsh sediments were found to contain up to 82% quartz sand (Bradley and Morris, 1999). Therefore, we considered it important to investigate whether HS related to salt marshes are capable of forming stable complexes with some clay minerals and quartz, and whether their elemental and structural characteristics change as a result of adsorption on these minerals (Filip and Alberts, 1994). The amounts of salt marsh-related HS which were adsorbed on quartz, kaolinite and montmorillonite are shown in Table 4.3.

*Table 4.3.* Adsorption of salt marsh-related humic substances (HS) on quartz and clay minerals (HS in mg TOC  $l^{-1}$ ) (Filip and Alberts, 1994).

Source of HS Orig.HS HS ads. on quartz			HS ads. on kaolinit		HS ads. on montm.	
C	TOC	%	TOC	%	TOC	%
S. alternifl. fresh 95.8	11.3	11.8	77.3	80.6	80.7	84.2
S. alternifl. dead 89.9	9.9	11.0	72.1	80.2	71.8	79.9
Salt marsh sed. 98.4	9.3	9.4	83.2	84.6	88.5	89.9

The rate of adsorption did not differ substantially for the individual HS preparations. The lowest amounts of HS were adsorbed on quartz but still, they comprised up to 11% of HS added. Surprisingly, the amounts of HS adsorbed on montmorillonite were similar to those adsorbed on kaolinite. This indicates that very little interlayer adsorption apparently occurred on montmorillotite. The C contents generally increased in HS which remained in solution after mineral-bound HS-fractions were removed by centrifugation. The differences were higher after adsorption on kaolinite and ranged from 13.5% for HS from *S. alterniflora* fresh, to 7.9% for HS from *S. alterniflora* dead, and 4.3% for those from sediment. These features indicate a preferential adsorption of low in C structural components of HS, i.e., some aliphatic structural units.

The FTIR spectra shown in Fig. 4.4 indicate, that no significant change in the structure of *S. alterniflora*-fresh related HS occurred due to the adsorption on quartz. The HS remaining in the solution after a portion was bound on kaolinite and montmorillonite show more differences in comparison to the original preparations. Among the most striking features observed was the decrease in intensity of the aliphatic C–H stretching at 2925 cm<sup>-1</sup>, a shift of carbonyl C–O stretch in amide I band at 1654 cm<sup>-1</sup> to an aromatic C=C stretch at 1615 cm<sup>-1</sup>, and loses in intensity of the amide II and amide II bands at 1517 and 1230 cm<sup>-1</sup>. Since the absorption at 3340 cm<sup>-1</sup> which can be attributed to H-bonded OH groups and partially to NH groups also lost intensity, one can assume that aliphatic chains such as in alkanes and some primary and secondary amides with their carbonyls, and further carboxyl and OH groups are involved in the HS-clay interactions which occurred in our experimental system. An increase in aromaticity of HS remaining after interaction with montmorillonite was indicated by a strong absorption band at 803 cm<sup>-1</sup> which accounts for C–H bonding absorption of substituted benzoic acid (Bellamy, 1975). Structural changes of HS are likely to have ecological consequences by modifying the solubility of contaminants, biotransformation processes, and the carbon cycle in soils and sediments (Myeni et al., 1999).



*Figure 4.4.* FTIR spectra of humic substances (HS) isolated from S. alterniflorafresh biomass; (B) HS-fraction remaining in solution after adsorption on quartz; (C) HS-fraction remaining in solution after adsorption on kaolinite; (D) HS-fraction remaining in solution after adsorption on montmorillonite (Filip and Alberts, 1994).

An attempt was made to remove the adsorbed HS from the complexes with minerals by acidic or alcalic extraction. When using  $0.1 N H_2SO_4$ , up to 20.2% of HS bound on quartz but only up to 6.7% of HS bound on montmorillonite were recovered. The desorption was more complete when using a mixture of 0.1 M NaOH + 0.1 M Na<sub>4</sub>P<sub>2</sub>O<sub>7</sub>. A maximum yield of 41.9% of salt marsh sediment-related HS was extracted from quartz-HS complexes. Other maximum yields were 48% for HS from S. alternifloradead adsorbed on kaolinite and 68.4% for the same HS adsorbed on montmorillonite. All HS re-extracted from kaolinite or montmorillonite contained a large proportion of ash (between 50.2 - 88.7%) which was prevalently composed of the respective clay minerals as evidenced by FTIR spectroscopy (not shown). This typical feature and the fact that only some amounts of HS could be re-extracted from complexes with minerals indicate appreciable amounts of HS as to be transformed in a fraction of humins, i.e., organic-mineral complexes that are known for their long-term stability in both aquatic and terrestrial environments (Hatcher et al., 1985).

#### 4.7. CONCLUSIONS

Using the annual primary production data (approx. 1400 g C m<sup>-2</sup> y<sup>-1</sup>) for the temperate salt marshes and the fact that 85% of that production is in aboveground S. alterniflora biomass (Pomeroy et al., 1981), it may be estimated that about 66 g C m<sup>-2</sup> y <sup>-1</sup> are associated with HS related to S. alternilora fresh material (Alberts et al., 1988). These HS comprise mainly of photosynthetically bound bioelements (plant nutrients) and energy, and apparently exist in the original plant material of S. alterniflora. On the other hand, they also consist of polymeric substances produced from water soluble components of the plant biomass by different epiphytic fungi. Due to the activity of complex populations of microorganisms indigenous to salt marsh environments, these HS can be released into seawater and salt marsh sediment. Many evidences exist on the similarity between sedimentary HS and those ones related to S. alterniflora fresh or dead plant biomass. Complex microbial populations indigenous to salt marsh sediment are capable of utilizing HS as additional sources of nutrients, and even more intensely as sole sources of carbon or nitrogen. In the presence of quartz particles or clay minerals typically occurring in salt marsh sediments, HS form organic-mineral complexes which in part can resist to the solubilization by acids or alkali.

According to Malcolm (1990), a uniqueness of humic substances is a typical feature in soil, stream and marine environments. Due to their specific origin, chemical composition and structure, and their diverse fate, salt marsh related HS may represent an example of the uniqueness of such substances in a salt marsh estuary. The formation, degradation, transformation, and complexation processes of these HS account for an important part of the biogeochemical cycling of organic matter in a salt marsh ecosystem.

## 5. Scaling contaminant distributions and contaminant processes

## 5.1. BACKGROUND

Risk assessment is an integral component of decision making in contaminated sediment management, and an important driving force is the presence of uncertainty. Uncertainties accumulate at all stages of site assessment and modeling and are magnified during scaling of cause-andeffect relationships from laboratory to field scale, from site scale to regional scale, and during extrapolation of relationships beyond their range of input measurements or beyond the site where the relationships/models were established. Even when preventive measures are taken by following sound planning, sampling (Batley, 2002) and analysis methods to eliminate and minimize some sources of error and bias, uncertainties always remain. Nevertheless, the element of uncertainty in the management of contaminated sediments is often treated inadequately (Suter, 1999) or not addressed (e.g. "Critical Issues for Contaminated Sediment Management", Apitz et al., 2002). For example, unquantified uncertainties can force extremely conservative estimates (Linkov et al., 2002), which can lead to high inefficiencies in the allocation of sparse resources for remediation. Most frequently, uncertainty is expressed in the form of confidence intervals, or by incorporation of safety factors based on best "professional judgment", such as a factor of 1.5 suggested by Palermo et al. (2002) in the design of cap thickness for contaminated sediments. But these measures and strategies incorporate assumptions about the nature (such as normal distributions) and magnitude of uncertainty, which are rarely tested and which vary along with sampling and measurement methods, models, and site variability and other site characteristics.

Several approaches are available for system modeling and decisionmaking under uncertainty, depending on the data, the magnitude and the acceptable level of uncertainty. Weight-of-evidence (WOE) approaches deal with large uncertainties by integrating multiple lines of evidence into a decision measure using methods ranging from best professional judgment to weighted scoring and statistical summaries/comparisons to reference situations (Chapman et al., 2002). WOE is most often applied in risk assessment, but its methods are valuable in any situation where uncertainties are large and/or multiple lines of evidence for the same decision requirement are available. When systems are better understood and mechanistic models available, the inputs can be treated as random variables. Uncertainties of these inputs (parameters/coefficients and variables) can be expressed in the form of probability distributions (using regression, analytical error parameters, or best professional judgment in the context of Bayesian prior and posterior probabilities). These uncertainties are then propagated to the model output via mathematical analysis (e.g. ISO/GUM) for simple models and via stochastic simulation: Monte Carlo analysis, Latin hypercube sampling, and bootstrap simulation (Kleijnen, 1997).

The distribution of input *parameters* must often rely on professional judgment, in particular when values of parameters are scattered in the literature and were derived in different experiments (or sites), by different investigators, under different experimental (or site) conditions (Linkov et al., 2002; Steinberg et al., 1997). When experiments are repeated, (non-) linear regression of a functional relationship will yield regression estimates and standard errors used as parameter estimators in appropriate distributions, e.g. the Gaussian distribution (Goovaerts et al., 2001). In the case of variables (i.e. field data), probability distributions can be generated using, for example, the known precision of the instrument. If the sampling is spatially distributed, local distributions at all sampled and unsampled locations can be derived using kriging methods. Once input distributions are quantified, values for each input variable and model parameter are randomly drawn from their probability distributions using, for example, a Monte Carlo approach, preserving correlations among the inputs if applicable (e.g. Figure 5.1).



*Figure 5.1.* Principles of uncertainty imparted by information loss during point sampling, and the need for probability distribution functions to make uncertainty-based predictions.

The resulting combination of input values are used in the model (e.g. sediment fate and transport, contaminant flux) to generate a realization (Papadopoulos and Yeung, 2001; Clarke and McFarland, 1999; Rossi et al., 1993). The process is repeated, while the nature of the sampling ensures that the realizations are equally probable, allowing a statistical summary of the model outputs. Uncertainties associated with scaling (e.g. laboratory derived contaminant degradation used in site models) or aggregation can be assessed in the context of validation, weight-of-evidence approaches (Steinberg et al., 1997), and by gathering additional data that represent the larger-scale patterns of variability (Loxmoore et al., 1991; Barabas et al., 2003ab).

The applications of geostatistical techniques to spatial analysis of sediments are recent and few. Barabás et al. (2001) modeled spatial uncertainty in dioxin contamination in three dimensions using indicator kriging in Passaic River sediments, Murray et al. (2002) mapped the extent of DDE contaminated sediment thickness on the Palos Verdes shelf using sequential indicator simulation, while Fioole et al. (1998) developed the SURFIS computer program, to integrate geostatistical methods in order to account for random error in the optimization of dredging of contaminated sediments with digital terrain models. The basis of geostatistics is the semivariogram, which is a mathematical expression of dissimilarity (e.g. of concentrations) as a function of separation distance between any two points in space (Goovaerts, 1997). The geostatistical approach models an attribute z at each unsampled location **u** as a random variable  $Z(\mathbf{u})$  with a ccdf  $F(\mathbf{u};$ z|(n), conditional to *n* neighboring data  $z(u_a)$ . The ccdf fully captures the uncertainty at **u** since it gives the probability that the unknown is no greater than any given threshold z (Goovaerts, 1997):  $F(\mathbf{u}; z|(n)) = \text{Prob } \{Z(\mathbf{u}) \leq z \}$ z|(n). Parametric and non-parametric methods exist to model attribute ccdfs. Indicator kriging (IK) is a non-parametric approach, with the advantages that the shape of the ccdfs is derived using spatially varying information rather than assuming a global analytical expression (such as a parametric Gaussian model). IK is also able to integrate these with uncertainties due to analytical methods (Saito and Goovaerts, 2002), as well as to incorporate additional information on correlated secondary attributes and soft information (such as soil maps, or qualitative field observations) into the modeling procedure. Least squares interpolation algorithms such as kriging tend to smooth out local details of the spatial variation of the attribute with small values typically overestimated and large values underestimated. To avoid these errors, geostatistical stochastic simulation is increasingly preferred to kriging for environmental assessment applications such as delineation of contaminated areas (Debarats, 1996; Goovaerts, 1997b; Kyriakidis, 1997; Naber et al., 1997), or the modeling of
groundwater flow in heterogeneous aquifers (Dagan, 1982; Desbarats and Srivastava, 1991; Kitanidis, 1997; Morgan et al., 1993; Desbarats, 1996; McKinney and Lin, 1996).

The set of alternative realizations generated provides a quantitative measure of spatial uncertainty. Features that appear consistently on most of the simulated maps are deemed certain, as expressed by a corresponding local conditional cumulative distribution function (ccdf) summarizing the probability distribution of the attribute at each simulated grid point. The uncertainty can then be summarized into probability maps, risk maps, and maps of false positives and false negatives (Goovaerts, 1999). Such results can aid in a more scientifically grounded assessment of safety factors and a more realistic and comprehensive formulation of confidence intervals. These maps are also well suited to the communication of uncertainties to non-scientific communities.

#### 5.2. METHODS

The statistical premise of the M-Scale model is that the variation of values at different spatial scales is heuristically related. In other words, if samples are collected at the "point scale" (e.g. single cores selected from large sites), the variation of parameters collected within the core are propagated to higher scales by taking into account the correlations of the local averages between scales, as shown in Figure 5.2. These correlations are informed by the variation of local averages at each scale. Taken together, the contributions of variation at each scale are weighted and aggregated into a best linear unbiased estimator, as shown in Figure 5.2c.

The model not only serves as a tool to evaluate parameter relationships over different scales by their covariances and data uncertainty, but also makes further use of these covariances and data uncertainty as the basis for a precision-optimized estimator. Information from each scale is weighted by the projected similarity to the scales of interest, with adjustments incorporating the different precision they provide. Unlike conventional geostatistic tools that are based on point-to-point spatial structures, the multi-scale model introduces a new framework for spatial analysis in which regional values at different scales are anchored by the correlations of each other.

Both normally distributed and bi-modal datasets, representing environmental contamination and microbial impact on contamination through dechlorination were evaluated. These datasets include: 1. A 396 sample characterization in three dimensions of dioxin contamination in the Passaic River (NJ), with emphasis on 2,3,7,8-tetra-chloro-dibenzo-p-dioxin (TCDD), previously mapped using geostatistical techniques (Barabas et al., 2001, Figure 5.3), and 2. A derived map of microbial dechlorination

#### S. BENTLEY ET AL.

signatures of dioxins in the sample locations (Barabas et al., 2004ab). These signatures were obtained using polytopic vector analysis (PVA), a multi-variate statistical technique which allows for unmixing of a complex mixture of contaminants in its weighted source contributions.



*Figure 5.2.* Schematic components of the M-Scale model: A. Integration of average values at different scales; B. Determination of variances between scales; C. Correlation between scales at sampled locations to estimate variation at each scale.

Both datasets were analyzed using ordinary kriging and M-Scale, and the quantile-quantile plots mapped to compare both methods for their predictive capability. To ensure robustness of the comparison, both crossvalidation (removal and re-estimation of one sample point at a time) and bootstrapping (removal and re-estimation of multiple datapoints) approaches were employed. Only cross-validation data will be shown.

#### 5.3. RESULTS AND DISCUSSION

The Passaic River has been contaminated with dioxins from a wide range of point and diffuse sources of industrial and urban (combined sewage overflows) origin. Multiple datasets in xyz-direction have been collected, and the 1996 dataset is presented. The histogram indicates that the data are log-normally distributed. The process data, based on microbial fingerprints observed in the samples, explaining up to 7% of the pattern variance, were bi-modally distributed. For brevity of this document, these data will not be presented here; the dioxin distribution information should be viewed as illustrative of the results gained from M-Scale and OK analysis.



*Figure 5.3.* Original dioxin distribution and data histogram for 396 TCDD measurements.

A comparison of Ordinary Kriging estimate (left) with M-Scale estimation (right) for two selected cross-sections of the river (Fig. 5.4) indicates that M-Scale appears to preserve local features in contamination, and that OK smoothens out the concentration distributions.



Figure 5.4. Estimation Maps by Ordinary Kriging (OK) and M-Scale Models.

When developing a new estimator for spatially-interpolated data, it is important to report the estimation uncertainty as well. Figure 5.5 (computed for transect 1), shows a correlation between sample density and reduction of uncertainty, as the lower left quandrant in Figure 5.5 exhibits higher uncertainty. The estimation uncertainty map was similar for the OK-based data interpolation (not shown).



Figure 5.5. Spatial distribution of estimation uncertainty for M-Scale.

The Q-Q plot of the estimated cross-validation results (Figure 5.6) represents graphically how well the model estimates predict the actual values (deviation from the 90° line). The results further confirm that the M-Scale estimation generally reflects the data across the entire magnitude range, while Ordinary Kriging tends to underestimate higher values and overestimate lower values.



Figure 5.6. Q-Q Plots from Ordinary Kriging (OK) and M-Scale Models.

## 5.4. CONCLUSIONS

The currently developed methodology for spatial interpolation represents an innovative approach to include multiple scales to estimate attributes in unsampled locations. This capability is of considerable use to inform riskbased decision-making, as the approach allows for the preservation of the overall value distribution (i.e. good coverage of information across scales). The applications, demonstrated here for contaminant distribution as part of include characterization. also estimation of processes site (e.g. biodegradation reactions, microbial respiratory activity, ebullition fluxes, etc...) in environmental systems. Its particular appeal is apparent for the opportunity to quantify spatial uncertainty associated with in situ implementation of remediation technologies, as the accuracy of value distributions allows for technology scaling from pilot demonstration. The method development requires further validation against variogram-based spatial uncertainty estimation for different spatial data structures and histograms (value distribution) to demonstrate robustness and application limitations

# 6. Appraisal of the problem: what are the gaps in our understanding?

Through consideration of the preceding studies and their implications, we have identified two major types of uncertainty in our understanding of contaminated sediment assessment and characterization. The first we refer to as causal uncertainty, which refers to the difficulty in conclusively demonstrating that a specific observation is the result of a specific process operating at a particular rate (such as the role in macrofaunal biodiffusion in controlling fair-weather pollutant concentrations in a riverine water column). The second type we refer to as statistical uncertainty associated with quantitative predictions of pollutant distribution, fate, and transport, based on either statistical or computational process models. We feel that need for new understanding in these two areas should drive new research and development in contaminated sediment characterization and assessment.

New research, development, and demonstration efforts should focus on issues where reduction in causal and/or statistical uncertainty would make a significant difference in the decision-making process. With respect to contaminant assessment, fate, and transport, both causal and statistical uncertainty exist in the following general areas: (a) process understanding (e.g., measurement and prediction of critical parameters and rates for individual processes such as sediment erosion, consolidation, or biogenic mixing); (b) relationships and interactions among processes (for example, the influence of bioturbation on sediment erodibility or sediment-water

128

contaminant transfer); and (c), forecasting, modeling, and integrating tools (for example, development of transient computational models to predict sediment dynamics, biogenic mixing, and contaminant transfer between the sediment bed and water column). Following is a brief summary of research needs in this area.

# 6.1. SPECIFIC AREAS OF UNCERTAINTY

- There are no standard or universally accepted approaches for measuring or predicting critical shear stresses and erosion rates for cohesive sediments, including the evaluation of biogenic effects on these sediment properties.
- The dynamics of pollutant binding with humic substances (free and mineral association) and the dynamics of pollutant adsorption/-desorbtion to/from humic organomineral complexes are not well understood.
- No standard approach exists for quantification of spatial (xyz) microbial attributes using surrogate parameters.
- Present *in situ* validated correlations for microbial activity are not adequate.
- Temporal trends of chemical distributions and microbial impact on their fate are not well understood.

# 6.2. SPECIFIC RESEARCH NEEDS

These specific needs are not limited to topics discussed in this chapter. All measurement and predictive tools should provide estimates of statistical uncertainty.

- Develop, evaluate, and validate standard tools and techniques to reduce uncertainty in measurement and prediction of critical shear stress, erosion rate, mode of transport, depth of physical reworking, and depositional processes for cohesive sediment, including biogenic effects on critical shear stress and erosion rates.
- Develop, evaluate, and validate a standardized approach to propagating uncertainty and representing sensitivity of the processes and interactions controlling sediment stability and contaminant transfer.
- Develop, validate, and distribute standard computational transient fate and transport models that incorporate the above approaches for characterizing sediment dynamics and contaminant mobility, including macrofaunal biogenic effects.

- Evaluate estuarine production and cycling of humics, with a focus on the role of free humics and mineral-associated humics on pollutant fate and availability.
- Develop understanding of how sediment biogeochemical sediment composition influences contaminant partitioning and bioavailability.
- Quantify exchange processes with overlying water and groundwater.
- Investigate impact of sparse sampling and variable histograms/spatial distribution on model robustness
- Evaluate approaches for data fusion (incorporation of additional information, such as inexpensive proxy estimates for more difficult/expensive analyses) on predictive reliability
- Investigate optimization schemes based on trading of scale with minimization of risk classification (i.e. what are misclassification rates as a function of probability thresholds)
- Develop robust approaches to deal with outlying data points.

# Acknowledgements

SB acknowledges the Office of Naval Research, the Naval Research Laboratory, Stennis Space Center and the Consortium for Oceanographic Research and Education for funding his work, with valuable guidance from Michael Richardson and Yoko Furukawa, and taxonomic assistance from Ed Ruppert (Clemson University), Gary King (University of Maine) and Richard Heard (University of Southern Mississippi).

PA and M-YL acknowledge the DODDOE-EPA Strategic Environmental Research and Development Program (SERDP) for funding this work.

ZF and KD gratefully acknowledge support obtained from the European Commission in Brussels through granting a Marie Curie Chair in Environmental Microbiology and Biotechnology to the Institute of Chemical Technology, Prague, Czech Rep.

# References

Alberts, J.J. and Filip, Z., 1994 Humic substances in rivers and estuaries of Georgia, USA. Trends in Chem. Geol. 1, 143–162.

Alberts, J.J. and Filip, Z., 1998 Metal binding in estuarine humic and fulvic acids: FTIR analysis of humic acid-metal complexes. Environ. Technol. 19, 923–931.

- Alberts, J.J., Filip Z., Price, M.T., Hedges, J.I. and Jacobsen, T.R., 1992 CuO-oxidation products, acid hydrolysable monosaccharided and amino acids of humic substances occurring in a salt mash estuary. Org. Geochem. 18, 171–180.
- Alberts, J.J., Filip, Z., Price, M.T., Williams, D.J. and Williams, M.C., 1988 Elemental composition, stable carbon isotope ratios and spectrophotometric properties of humic substances in a salt marsh estuary. Org. Geochem. 12, 455–467.
- Apitz, S.E., Davis, J.W., Finkelstein, K., Hohreiter, D.L., Hoke, R., Jensen, R.H., Jersak, J., Kirtay, V.J., Mack, E.E., Magar, V., Moore, D., Reible, D., Stahl, R. (2002) Critical Issues for Contaminated Sediment Management, MESO-02-TM-01. <u>http://meso.spawar.navy.mil/docs/MESO-02-TM-01.pdf</u>.
- Bader, R.G., Hood, D.W. and Smih, J.B., 1960 Recovery of dissolved organic matter in seawater and organic sorption by particulate material. Geochim. Cosmochim. Acta 19, 236– 243.
- Banwart, S., 1997, Aqueous speciation at the interface between geological solids and groundwater. Invited and peer-reviewed contribution in *Modelling in Aquatic Chemistry*, I.\_Grenthe and I. Puigdomenech eds., The Nuclear Energy Agency of the OECD, Paris, pp. 245–287.
- Banwart, S.A. and Malmström, M.E., 2001, Hydrochemical modeling for preliminary assessment of mine water pollution, *J. Chem. Exp.* **74**: 73–97.
- Banwart, S.A., Evans, K.A. and Croxford, S., 2002, Predicting mineral weathering rates at field scale for mine water risk assessment, article in: *Mine water hydrogeology and geochemistry*, P.L. Younger and N.S. Robin, ed., Geological Society special publications, London, pp. 137–157.
- Barabás, N., Goovaerts, P., Adriaens, P. (2001) Geostatistical assessment and validation of uncertainty for three-dimensional dioxin data from sediments in an estuarine river. Environmental Science and Technology, 35, 3294–3301.
- Barabas, N., Goovaerts, P. and Adriaens, P., 2004. Modified polytopic vector analysis to identify and quantify dioxin dechlorination signatures in sediments. 1. Theory. Environ. Sci. Technol., 38: 1813–1820.
- Barabas, N., Goovaerts, P. and Adriaens, P., 2004. Modified polytopic vector analysis to identify and quantify dioxin dechlorination signatures in sediments. 2. Application to the Passaic River Superfund site. Environ. Sci. Technol., 38: 1821–1827.
- Batley, G.E., Burton, G.A., Chapman, P.M. and Forbes, V.E. (2002) Uncertainties in sediment quality weight-of-evidence (WOE) assessments. Human and Ecological Risk Assessment 8, 1517–1547.
- Bellamy, L.J., 1975 The Infrared Spectra of Complex Molecules. Chapman and Hall, London.
- Berner, R.A. Early Diagenesis: A Theoretical Approach. Princeton, New Jersey: Princeton University Press. 241 p.
- Best, .I. and Gunn, D.E., 1999, Calibration of marine sediment core loggers for quantitative acoustic impedance studies. Marine Geology 160: 137–146.
- Boudreau and Bennett, 1999, New rheological and porosity equations for steady-state compaction. American Journal of Science 299: 517–528.
- Boudreau, B.P., 1997, Diagenetic models and their implementation: Berlin, Springer, 410 p.
- Bradley, P.M. and Morris, J.T., 1999 Physical characteristics of salt marsh sediments: ecological implications. Mar. Ecol. Prog. Ser. 61, 245–252.
- Briggs, K.B. and Richardson, M.D., 1996, Variability in in situ shear strength of gassy muds. Geo-Marine Letters 16: 189–195.

- Briggs, K.B., Jackson, P.D., Holyer, R.J., Flint, R.C., Sandidge, J.C. and Young, D.K., 1998, Two-dimensional variability in porosity, density, and electrical resistivity of Eckernförde Bay sediment. Continental Shelf Research 18: 1939–1964.
- Bromley, R.G., 1996. Trace Fossils: Biology and Taphonomy. Unwin Hyman. 280 pp.
- Chapman, P.M., McDonald, B.G. and Lawrence, G.S. (2002) Weight-of-evidence issues and frameworks for sediment quality (and other) assessments. Human and Ecological Risk Assessment, 8, 1489–1515.
- Clarke, J.U., McFarland, V.A. (2000) Uncertainty analysis for an equilibrium partitioningbased estimator of polynuclear aromatic hydrocarbon bioaccumulation potential in sediments. Environmental Toxicology and Chemistry 19, 360–367.
- Dade, W.B., Nowell, A.R.M. and Jumars, P.A., 1992, Predicting erosion resistance of muds. Marine Geology 105: 285–297.
- Dagan, G. (1982) Stochastic modeling of groundwater flow by unconditional and conditional probabilities, 1. Conditional simulation and the direct problem. Water Res. Research 18, 813–833.
- Deng, Y., Stjernström, M., Banwart, S., 1996, Accumulation and remobilization of aqueous chromium (VI) at iron oxide surfaces: application of a thin-film continuous flow-through reactor, J. Contam. Hydrol., 21: 141–151.
- Desbarats, A.J. (1996) Modeling spatial variability using geostatistical simulation. In: Rouhani, S., Srivatsava, R.M., Desbarats, A.J., Cromer, M.V. and Johnson, A.I. (eds), Geostatistics for Environmental and Geotechnical Applications. American Society for Testing and Materials STP 1283, Philadelphia, pp. 32–48.
- Desbarats, A.J., Srivastava, R.M. (1991) Geostatistical characterization of groundwater-flow parameters in a simulated aquifer. Water Resources Research 27, 687–698.
- Dzombak, D.A. and Morel, F.M.M., 1990, *Surface Complexation Modelling. Hydrous Ferric Oxide*. John Wiley and Sons, New York.
- Ertel, J.R. and Hedges, J.I., 1983 Bulk chemical and spectroscopic properties of marine and errestrial humic acids, melanoidins and catechol-based synthetic polymers. *In* Aquatic and Terrestrial Humic Materials. Eds. R.F. Christman and E.T. Gjessing. pp 143–163. Ann Arbor Science, Ann Arbor, Mich.
- Ertel, J.R. and Hedges, J.I., 1985 Sources of sedimentary humic substances: Vascular plant debris. Geochim. Cosmochim. Acta 49, 2097–2107.
- Esham, E.C., Ye, W. and Moran, M.A., 2000 Identification and characterization of humic substances degrading bacterial isolates from an estuarine environment. FEMS Microbiol. Ecol. 34, 103–111.
- Fallon, R.D. and Pfaender, F.K., 1876 Carbon metabolism in model microbial systems from a temperate salt marsh. Appl. Environ. Microbiol 31, 959–968.
- Filip, Z. and Alberts, J.J., 1988 The release of humic substances from *Spartina altrniflora* (Loisel.) into sea water as influenced by salt marsh indigenous microorganisms. Sci. Total Environ, 73, 143–157.
- Filip, Z. and Alberts, J.J., 1989 Humic substances isolated from *Spartina alterniflora* (Loisel.) following long-term decomposition in sea water. Sci. Total Environ. 83, 273– 285.
- Filip, Z. and Alberts, J.J., 1993 Formation of humic-like substances by fungi epiphytic on *Spartina alterniflora*. Estuaries 16, 385–390.
- Filip, Z. and Alberts, J.J., 1994 Adsorption and transformation of salt marsh related humic acids by quartz and clay minerals. Sci. Total Environ. 153, 141–150.
- Filip, Z. and Alberts, J.J., 1994 Microbial utilization resulting in early diagenesis of saltmarsh humic acids. Sci. Total Environ, 144, 121–135.

- Filip, Z., Alberts, J.J., Cheshire ,M., Goodman, B.A. and Bacon, J.R., 1988 Comparison of salt marsh humic acid with humic-like substances from the indigenous plant species *Spartina alterniflora* (Loisel.). Sci. Total Environ. 71, 157–172.
- Filip, Z., Dippell, G. and Claus, H., 1998 Abbau von Huminstoffen durch Mikroorganismen eine Übersicht. Z. Pflanzenernähr. Bodenkd. 161, 605–612.
- Filip, Z., Haider K. and Martin J.P., 1972 Influence of clay minerals on the formation of humic substances by *Epicoccum nigrum* and *Stachybotrys chartarum*. Soil Biol. Biochem. 4, 147–154.
- Fioole, A., Houwing, E.J. and van der Heijdt, L.M. (1998) SURFIS: A tool for designing and optimizing dredging schemes. Water Science and Technology, 37, 103–107.
- Flint, R.W. and Kalke, R.D., 1986, Biological enhancement of estuarine benthic community structure. Marine Ecology Progress Series 31: 23–33.
- Fogel, M.L., Sprague, E.K., Gize, A.P. and Frey, R.W., 1989 Diagenesis of organic matter in Georgia salt marshes. Estuar. Coast. Shelf. Sci. 28, 211–230.
- Frimmel, F.H., Abbt-Braun, G., Heumann, K.G., Hock, B., Lüdemann, H.-D. and Spiteller, M. (Eds.) 2002 Refractory Organic Substances in the Environment. Wiley-VCH Weinberg, 546 pp.
- Gallagher, J.L., Pfeiffer, W.J. and Pomeroy, L.R., 1976 Leaching and microbial utilization of dissolved organic carbon from leaves of *Spartina alterniflora*. Estuar. Coast. Marine Sci. 4, 467–471.
- Garrels, R.M., Christ, C.L., 1965, *Solutions, Minerals and Equilibria*. 2<sup>nd</sup> ed., Freeman and Cooper, San Francisco, pp. 160.
- Gessner, R.V. and Goos, R.D., 1973 Fungi from decomposing *Spartina alterniflora*. Can. J. Bot. 51, 51–55.
- Gielazyn, M.L., Stancyk, S.E. and Piegorsch, W.W., 1999, Experimental evidence of subsurface feeding by the burrowing ophiuroid Amphipholis gracillima. Marine Ecology-Progress Series 184: 129–138.
- Goovaerts, P., Semrau, J., Lontoh, S. (2001) Monte Carlo analysis of uncertainty attached to microbial pollutant degradation rates. Environmental Science and Technology 35, 3924– 3930.
- Goovaerts, P. (1997b) Kriging, vs. stochastic simulation for risk analysis in soil contamination. In: geoENV I – Geostatistics for Environmental Applications, A. Soares, Gómez-Hernández, J. and Froidevaux, R. (eds). pp. 247–258. Kluwer Academic Publishers, Dordrecht.
- Goovaerts, P. (1999) Geostatistics in soil science: state-of-the-art and perspectives. Geoderma, 89, 1-45.
- Goovaerts, P., 1997. Geostatistics for Natural Resources Evaluation. Oxford University Press, New York.
- Harvey, G.R. and Boran, D.A., 1985 Geochemistry of humic substances in seawater. *In* Humic Substances in Soil, Sediment, and Water. Eds. G.R. Aiken, D.M. McKnight, R.L. Wershaw and P MacCarthy. pp. 233–247. Wiley, New York.
- Hatcher, P.G., Breger, I.A., Maciel, G.E. and Szeverenyi, N.M., 1985 Geochemistry of humin. *In* Humic Substances in Soil, Sediment, and Water. Eds. G.R. Aiken, D.M. McKnight, R.L. Wershaw and P. MacCarthy. pp.275–325. Wiley, New York.
- Holtz, R.D. and Kovacs, W.K., 1981, Introduction to geotechnical engineering. Englewood Cliffs, N.J.: Prentice-Hall, 732 p.
- Jackson, P.D., Taylor Smith, D. and Stanford, P.N., 1978, Resistivity-porosity-particle shape relationships for marine sands. Geohphysics 43: 1250–1268.
- Jones, C.D. and Jackson, D.R., 1997, Temporal fluctuation of backscattered field due to bioturbation in marine sediments. In Pace, N.G., Pouliquen, E. and Lyons, A.P., eds.,

High Frequency Acoustics in Shallow Water, NATO SACLANT Undersea Research Center, La Spezia, Italy, p. 275–282.

- Jones, S.E. and Jago, C.F., 1993, In situ assessment of modification of sediment properties by burrowing invertebrates. Marine Biology 115: 133–142.
- Keller, G.V. and Frischknecht, F.C., 1966, Electrical methods in geophysical prospecting. Oxford: Pergamon, 512 p.
- Kitanidis, P.K. (1997) Introduction to Geostatistics: Applications in hydrogeology. Cambridge University Press, New York.
- Kleijnen, J.P.C. (1997) Sensitivity analysis and related analyses: A review of some statistical techniques. Journal Of Statistical Computation and Simulation 57, 111–142.
- Kyriakidis, P.C. (1997) Selecting panels for remediation in contaminated soils via stochastic imaging. In: E.Y. Baafi and N.A. Schofield (eds) Geostatistics Wollongong '96, 2, 973– 983. Kluwer Academic Publishers, Dordrecht.
- Thibodeaux, L.J., Reible, D.D. and Valsaraj, K.T. "Non-particle Resuspension Chemical Transport from Stream Beds," Chapter 7 in Chemicals in the Environment: Fate, Impacts and Remediation, Ed. R. Lipnick, R. Mason, M. Phillips and C. Pittman, ACS Symposium Series 806 (2002)
- Leuchtmann, A. and Newell, S.Y., 1991 Phaeosphaeria spartinicola, a new species on Spartina. Mycotaxon XLI, 1–7.
- Linkov, I., Burmistrov, D., Cura, J., Bridges, T.S. (2002) Risk-based management of contaminated sediments: Consideration of spatial and temporal patterns in exposure modeling. Environmental Science & Technology 36, 238–246.
- Lu, X.X., D. Reible, "Linking sediment exposure with effects: modeling techniques, organic availability and uptake", Int'l Journal of Sediment Research, 18, 2 208–213 (2003)
- Luxmoore, R.J., King, A.W., Tharp, M.L. (1991) Approaches to scaling up physiologically based soil plant-models in space and time. Tree Physiology 9, 281–292.
- Lyons, A.P. and Orsi, T.H., 1998, The effect of a layer of varying density on high-frequency reflection, forward loss, and backscatter. I.E.E.E. Journal of Oceanic Engineering 23: 411–422.
- Malcolm, R.L., 1990 The uniqueness of humic substances in each soil, stream and marine environments. Anal. Chim. Acta 232, 19–30.
- Malmström, M. and Banwart, S., 1997, Biotite dissolution at 25°C: The pH dependence of dissolution rate and stoichiometry, *Geochim. Cosmochim. Acta.*, 61(4): 2779–2799.
- Malmström, M., Destouni, G., Banwart, S.A., Strömberg, B.H.E., 2000, Resolving the scaledependence of mineral weathering rates, *Env. Sci. Technol.*, 34(7): 1375–1377.
- McKinney, D.C., Lin, M.D. (1996) Pump and treat ground-water remediation system optimization. Journal of Water Resources Planning and Management-ASCE 122, 128–136.
- Meadows, P.S. and Tait, J., 1989, Modification of sediment permeability and shear strength by two burrowing invertebrates. Marine Biology 101: 75–82.
- Moran, M.A. and Hodson R.E., 1990 Bacterial production on humic and nonhumic components of dissolved organic carbon. Limnol Oceanogr 34, 1744–1756.
- Morgan, D.R., Eheart, J.W., Valocchi, A.J. (1993) Aquifer remediation design under uncertainty using a new chance constrained programming technique. Water Resources Research 29, 551–561.
- Murray, C.J., Lee, H.J., Hampton, M.A. (2002) Geostatistical mapping of effluent-affected sediment distribution on the Palos Verdes shelf. Continental Shelf Research, 22, 881– 897.
- Myneni, S.C.B., Brown, J.T., Martiney, G.A. and Meyer-Ilse, W., 1999 Imaging humic substance macromolecular structures in water and soils. Science 286, 1335–1337.

- Naber, S.J., Buxton, B.E., Bertoni, M.J., Scheibe, T.D. and Warren, J. (1997) U.S.EPA computer system for simulating site characterization activities at Superfund cleanup sites.. In: E.Y. Baafi and N.A. Schofield (eds) Geostatistics Wollongong '96, 2, 1066– 1074. Kluwer Academic Publishers, Dordrecht.
- Newell, S.Y., 1984 Bacterial and fungal productivity in the marine environment: A contrastive overview. Bacteriologie Marine (CNRS, Paris) 1984, 133–139.
- Nordstrom, D.K. and Southam, G., 1997, Geomicrobiology of sulfide mineral oxidation. Geomicrobiology: interactions between microbes and minerals. *Rev. Mineral.* **35**: 361–390.
- Pakulski, J.D., 1986 The release of reducing sugars and dissolved organic carbon from *Spartina alterniflora* Loisel in a Georgia salt marsh. Estuar. Coast. Shelf Sci. 22, 385– 394.
- Palermo, M.R., Thompson, T.A., Swed, F. (2002) White Paper No. 6B Is-Situ capping as a remedy component for the lower Fox River. Response to a document by the Johnson Company: Ecosystem-based rehabilitation plan – An integrated plan for habitat enhancement and expedited exposure reduction in the lower Fox River and Green Bay. December 2002.
- Papadopoulos, C.E., Yeung, H. (2001) Uncertainty estimation and Monte Carlo simulation method. Flow Measurement and Instrumentation 12, 291–298.
- Pellenbarg, R.E., 1985 <u>Spartina alterniflora</u> litter in salt marsh geochemistry. *In* Marine and Estuarine Geochemistry. Eds. A C Sigleo and A Hattori. pp. 305–314. Lewis, Chelsea, Mich.
- Pinet, P.R. and Morgan, Jr. W.P., 1979 Implication of clay-provenance studies in two Georgian estuaries. J. Sediment. Pet. 49, 575–580.
- Pomeroy, L.R. and Imberger, J., 1981 The physical and chemical environment. *In* The Ecology of a Salt Marsh. Eds. L.R. Pomeroy and R.G. Wiegert. pp 21–27. Springer, New York.
- Pomeroy, R.L., Darley, W.M., Dunn, E.L., Gallagher, J.L., Haines, E.B. and Whitney, D., 1981 Primary production. *In* The Ecology of a Salt Marsh. Eds. L R Pomeroy and RG Wiegert. pp. 39–67. Springer, New York.
- Rashid, M.A., 1985 Geochemistry of Marine Humic Compounds. Springer, New York, 300 pp.
- Rhoads D.C. and Young D.K., 1970. The influence of deposit-feeding organisms on sediment stability and community trophic structure. Journal of Marine Research 28: 151–178.
- Rhoads, D.C., McCall, P.L. and Yingst, J.Y., 1978, Disturbance and production on the estuarine seafloor: American Scientist, v. 66, p. 577–586
- Rhoads, D.C. and Boyer, L.F., 1982, The effects of marine benthos on physical properties of sediments; a successional perspective. In: McCall, P.L. and Tevesz, M.S., eds., Animal-Sediment Relations: the Biogenic Alteration of Sediments. New York: Plenum, p. 3–52.
- Rice, D.L., 1986, Early diagenesis in bioadvective sediments: Relationships between the diagenesis of <sup>7</sup>Be, sediment reworking, and the abundance of conveyor-belt deposit feeders. Journal of Marine Research 44: 149–184.
- Richardson, M.D., Young, D.K. and Briggs, K.B., 1983, Effects of hydrodynamic and biological processes on sediment geoacoustic properties in Long Island Sound, U.S.A. Marine Geology 52: 201–226.
- Robbins, J.A., 1986, A model for particle-selective transport of tracers in sediments with conveyor-belt deposit feeders. Journal of Geophysical Research, 91: 8542–8558

- Romero-González, M.E., Williams, C.J., Gardiner, P.H.E, Gurman, S.J., Habesh, S., 2003, Spectroscopic studies of the biosorption of Au(III) by dealginated seaweed waste, *Environ. Sci. Technol.* 37:4163–4169.
- Romero-González, M.E., Williams, C.J., Gardiner, P.H.E., 2001, Study of the mechanisms of cadmium biosorption by dealginated seaweed waste, *Environ. Sci. Technol.*, 2001, 35: 3025–3030.
- Rossi, R.E., Borth, P.W., Tollefson, J.J. (1993) Stochastic simulation for characterizing ecological spatial patterns and appraising risk. Ecological Applications. 3, 719–735.
- Rowden, A.A., Jago, C.F. and Jones, S.E., 1998, Influence of benthic macrofauna on the geotechnical and geophysical properties of surficial sediment, North Sea. Continental Shelf Research 18: 1347–1363.
- Saito, H., Goovaerts, P. (2002) Accounting for measurement error in uncertainty modeling and decision-making using indicator kriging and p-field simulation: application to a dioxin contaminated site. Envirometrics, 13, 555–567.
- Senesi, N., 1992 Metal-humic substance complexes in the environment. Molecular and mechanistic aspects by multiple spectroscopic approach. *In* Biogeochemistry of Trace Metals. Ed. D C Adriano. pp. 429–496. Lewis, Boca Raton.
- Steinberg, L.J., Reckhow, K.H., Wolpert, R.L. (1997) Characterization of parameters in mechanistic models: A case study of a PCB fate and transport model. Ecological Modeling 97, 35–46.
- Stumm, W., Morgan, J.J., 1996 Aquatic Chemistry, Chemical Equilibria and Rates in Natural Waters, 3<sup>rd</sup> ed. John Wiley and Sons, Inc, USA, pp. 257–258.
- Suter, G.W. (1999) Statistics and risk assessment. 1999 Lukacs Symposium. Frontiers of Environmental and Ecological Statistics for the 21st Century - Synergistic Challenges, Opportunities and Directions for Statistics, Ecology, Environment, and Society Department of Mathematics and Statistics, Bowling Green State University, Bowling Green, Ohio, USA. April 23–25, 1999. <u>http://wwwmath.bgsu.edu/symposium/abstract/node4.html</u>.
- Toorman, E.A., 1996, Sedimentation and self-weight consolidation: general unifying theory. Geotechnique 46: 103–113.
- Valentine, J.F., 1991, Temporal variations in populations of the brittlestars Hemipholis elongata (Say, 1825), and *Microphiopholis atra* (Stimpson, 1852) in eastern Mississippi Sound. Bulletin of Marine Science 48: 597–605.
- Vandenbroucke, M., Pelet, R. and Debyser, Y., 1985 Geochemistry of humic substances in marine sediments. *In* Humic Substances in Soil, Sediment, and Water. Eds. G R Aiken, D.M. McKnight, R.L. Wershaw and P. Mac Carthy. pp. 249–273. Wiley, New York.
- Walker, D.J., Lambert, D.N., Young, D.C. and Stephens, K.P., 1997, Mapping sediment acoustic impedance using remote sensing acoustic techniques in a shallow-water carbonate environment. Geo-Marine Letters 17: 260–267.
- Wheatcroft, R.A., Jumars, P.A., Smith, C.R. and Nowell, A.R.M., 1990, A mechanistic view of the particulate biodiffusion coefficient: step lengths, rest periods, and transport directions: Journal of Marine Research, v. 48, p. 177–207.
- Wiberg, P., 1999, Coupling of sediment resuspension and post-depositional bed modifications in fine-grained sediments. Eos 80: OS251.
- Younger, P.L., Banwart, S.A., Hedin, R.S., 2002, *Mine Water, Hydrology, Pollution and Remediation*, Kluwer academic publishers, The Netherlands, pp.67–76.

# TREATMENT AND CONTAINMENT OF CONTAMINATED SEDIMENTS

JEANNE E. TOMASZEWSKI, DENNIS W. SMITHENRY, YEO-MYOUNG CHO, RICHARD G. LUTHY<sup>\*</sup> Stanford University, Stanford, CA 94305

GREG V. LOWRY Carnegie Mellon University, Pittsburgh, PA

DANNY REIBLE The University of Texas at Austin, Austin, TX

TOMAS MACEK Department of Natural Products, Institute of Organic Chemistry and Biochemistry, Czech Academy of Sciences, Flemingovo n. 2, 166 10 Prague, Czech Republic

MARTINA SURÁ, ZUZANA CHRASTILOVA, KATERINA DEMNEROVA and MARTINA MACKOVÁ Dept. of Biochemistry and Microbiology, Faculty of Food and Biochemical Technology, ICT Prague, Technická 3, 166 28 Prague, Czech Republic

DANIELA PAVLIKOVÁ, Department of Agrochemistry, Faculty of Agronomy, Czech Agricultural University, 160 00 Prague - Suchdol, Czech Republic

MIKLOS SZEKERES, Biological Research Center, Hungarian Academy of Sciences, 6701 Szeged, Hungary

MICHEL SYLVESTRE INRS - IAF, 245, Boul. Hymus, Pointe-Claire, Québec, H9R 1G6 Canada

**Abstract-** Several approaches to the containment and treatment of contaminated sediment were evaluated, including the efficacy of adding activated carbon to sediment as an in-situ stabilization method and the use

of conventional and innovative treatment caps. The applicability of phytoremediation for dredged sediments and sediments in shallow water or wetlands was also explored. The effectiveness of any treatment relies on successful application in the field and the emphasis herein is on evaluation of these approaches in the field. A demonstration at Hunters Point, San Francisco Bay, California strives to prove the efficacy of activated carbon treatment, and stands as an example for applications elsewhere. Simultaneous containment and treatment of sediment contaminants is being demonstrated in the Anacostia River, Washington, DC using both organic and metal sequestering agents incorporated into a cap. Both conventional placement and placement in a laminated mat were demonstrated to provide an array of placement approaches. The laminated mat allowed placement of thin layers of high value material at specific location, enabling the use of high value "active" capping materials such as activated carbon and microscale iron. Phyto- and rhizoremediation using plants and related bacteria is a promising approach for treatment of contaminated sediments, but has its inborn limitations. To overcome the slow performance of the process, transgenic plants were evaluated that express the bacterial gene responsible for cleaving PCBs, or metal binding proteins to provide additional metal binding capacity. Continued development of cap and sediment treatments, and approaches to assess effectiveness and long-term reliability are encouraged.

**Keywords:** capping, activated carbon, reactive core mat, bioavailability, phytoremediation, transgenic plants

# 1. In-situ treatment of contaminated sediments by activated carbon amendment

# 1.1. THE LIMITATIONS OF CURRENT APPROACHES FOR REMEDIATION OF ORGANIC CONTAMINANTS IN SEDIMENTS

Hydrophobic organic pollutants, such as polychlorinated biphenyls (PCBs), polycyclic aromatic hydrocarbons (PAHs), and dichlorodiphenyltrichloroethane (DDT) are well-known for their toxicity to humans and wildlife, wide geographic distribution, and slow degradation. Once deposited in water, these chemicals bind with organic matter and, via sediment association, provide a continuous source of such contamination to water and biota. Numerous remedial strategies are available for management of contaminated sediments, each with their own limitations. Dredging is the prevalent ex-situ approach of sediment remediation, which destroys benthic habitat, may incompletely remove contaminated sediments, and may resuspend contaminated fines that migrate to surrounding waters. Further, the disposal of contaminated dredged sediment presents significant problems, both technical and political. The current alternative to dredging is capping of sediments. This method relies exclusively on adequate application and long-term structural integrity, and therefore may prove problematic over time. Mandates against filling productive wetlands also exist in areas such as San Francisco Bay that make capping inappropriate.

Such limitations of remediation alternatives materialized at Lauritzen Channel in Richmond, CA, an EPA Superfund site. The sediment is highly contaminated with DDT, a chemical that causes death and sterility of birds and fish, and possible carcinogenic and endocrinal effects in humans. The channel is an active industrial waterway near a marina and the Rosie the Riveter National Historic Park. Before remediation, resident mussels contained the highest levels of DDT and dieldrin measured in California (Kohn and Evans, 2004). The selected sediment remediation remedy prescribed in the Record of Decision in October 1995 specified dredging and dewatering of all soft bay mud material with off-site disposal and placement of clean material (sand) cap to enhance habitat value (USEPA, 1996). Disposal of the sediments was planned to occur in Colorado Springs, Colorado, but public outcry moved the location to Mobile, Arizona, After dredging was completed in 1997, the sediments were hauled by train to Arizona, but growing public protests in Arizona eventually diverted some of the sediment to a site in Utah (Bruggers, 1997).

Other problems with dredging were evident at Lauritzen Channel even after finding an adequate disposal site. Post-dredging marine monitoring of water and fauna by Pacific Northwest National Laboratory (PNNL) found residual DDT contamination levels exceeding by nearly 200 fold the water remediation goal of 0.59 ng/L (Antrim and Kohn, 2000). Sediment cores sampled by us in 2003 exhibited high DDT concentrations (250 mg/kg sediment maximum) and a stratification of contamination. Some cores showed the remnants of the applied sand layer overlain with highly contaminated sediment. These data suggest the sand layer was not evenly distributed or washed away, and that contaminated sediment remains above and below the applied sand layer. Several biological studies with Lauritzen sediment following dredging found remaining toxicity to amphipods, with several sampling stations rated among the most toxic of coastal sites measured nationwide (Anderson et al., 2000; Weston et al., 2002). The remedial action therefore proved ineffective, and the remaining contamination poses a risk to biota and human health. Currently, the California Department of Toxic Substances Control advises against eating fish from the Richmond Harbor area, and the EPA and Army Corps of Engineers (ACE) are studying alternatives for the next remedial action.

As illustrated by the efforts at Lauritzen Channel, an easy solution for remediating contaminated sediments does not exist. Possible solutions lie in the non-removal treatments of contaminated sediments, such as reactive caps, monitored natural recovery, and in-situ stabilization. Each of these alternative sediment management approaches has its own advantages and is currently being evaluated at sites in the United States. The remainder of this section will focus on the use of in-situ stabilization for PCB-contaminated sediment at Hunters Point Naval Shipyard, San Francisco Bay, CA.

### 1.2. IN-SITU STABILIZATION: PROOF OF CONCEPT AT HUNTERS POINT

As dredging methods prove ineffective to deal with contaminated sediments, recent research at Stanford University has tested an in-situ alternative to achieve the ideals of high containment effectiveness, limited taxing of the ecosystem, and low costs. The alternative is based on in-situ stabilization with activated carbon amendment. This concept was formulated from the field observation that PCBs naturally, over time, accumulate on black carbonaceous particles in the sediment at Hunters Point. We propose that by mixing activated carbon into the biologically active upper layer of sediment, PCBs will repartition and be sequestered in the carbon, thus reducing PCB bioavailability and release to water. The path forward for advancing this new treatment relies on relating sediment chemistry and effects, on both the micro-scale level of contaminant binding and the macro-scale level of bio-uptake, modeling carbon dose and time effects, transitioning the technique to the field, and engaging regulators and advisory groups.

## 1.3. DISTRIBUTION OF PCBS AND ADSORPTION EFFICIENCY

Hunters Point is an inter-tidal zone in the South Basin of San Francisco Bay historically contaminated with PCBs (0.2-65 mg/kg total PCBs). In a micro-scale study of the sediment, carbonaceous particles such as coal, coke, char and charcoal (see Figure 1.1) contributed 5-7% of total mass and held about two-thirds of the PCBs (Ghosh et al., 2003). This and related work (Ghosh et al., 2001) showed that compared to other types of particles, PCBs and PAHs accumulate to a much greater extent in black carbonaceous particles and become more strongly bound and less bioavailable. The black carbonaceous particles comprise a portion of the total organic carbon that



oxidized coal

charcoal

coke

*Figure 1.1.* Petrographic identification of organic carbon particles in sediment. Images shown are under reflected light in oil at 500x magnification.(Ghosh et al., 2003).

strongly affects the partitioning of PCBs and PAHs, because of its high surface area and adsorption affinity.

Bioavailability, as measured by the absorption efficiency of 2,2',5,5'tetrachlorobiphenyl (PCB-52) by clams (*Macoma balthica*), varies among the different carbonaceous particles (McLeod et al., 2004). The absorption efficiency is the highest from wood and diatoms and the lowest from activated carbon, about 1.4% (Figure 1.2). This shows that PCBs bound to activated carbon are not readily absorbed even if ingested.



*Figure 1.2.* Absorption efficiency of PCB-52 varies significantly among different particle types. Error bars represent 95% confidence intervals.(McLeod et al., 2004).

#### 1.4. LABORATORY FEASIBILITY TESTING

After initial findings pointed to the effectiveness of in-situ stabilization with activated carbon, several laboratory feasibility tests were performed with Hunters Point sediment. These tests investigated how activated carbon affected PCB bioaccumulation by several benthic organisms, physico-chemical system properties, and sediment erosion.

Bioaccumulation studies were designed to measure PCB accumulation in three benthic marine organisms plus survival, growth, reproduction, and activity. In the studies, activated carbon was mixed with sediment from Hunters Point for one or six months. After mixing, clams (*Macoma balthica*), polychaetes (*Neanthes arenaceodentata*), and amphipods (*Leptocheirus plumulosus*), were placed in the sediment, and bioaccumulation and biological effects were evaluated in 28 or 56 day tests (Millward et al., 2005; Zimmerman et al., 2005). Results were compared to those obtained using untreated Hunters Point sediment.

A significant decrease in the PCB bioaccumulation as compared to the untreated sediment was noted in all organisms. In 56 day exposure tests, reductions of 90% and 93% were measured for the amphipod and worm, respectively. A clear dose-response was also noted for these organisms in the 56 day test (Figure 1.3). PCB body burden significantly decreased as the dose of activated carbon increased. Similar trends were noted for the clam.



*Figure 1.3.* PCB accumulation by *Neanthes arenaceodentata* and *Leptocheirus plumulosus* during 56-day exposure tests. Sediment was contacted for one month with varying doses of TOG carbon before tests (Zimmerman et al., 2005).

Exposure to activated carbon amended sediment resulted in no significant effect on the survival, growth, lipid content, and fecundity in the amphipod or survival and lipid content in the polychaete. The polychaete's wet weight did decrease by approximately 50%, but direct toxicity was discounted as an explanation for the effect. All these tests point to the effectiveness of activated to carbon to sequester PCBs from sediment, thus reducing PCB bioavailability.

positive bioaccumulation tests The results were mirrored in physicochemical tests performed in a laboratory. After Hunters Point sediment was contacted with activated carbon in well-mixed systems, the aqueous equilibrium concentration was tested. Reductions up to 92%, as compared to untreated sediment, were measured for a six-month contact time (Zimmerman et al., 2004). As with bioaccumulation, a clear dose response was observed for aqueous equilibrium concentrations (Figure 1.4). In addition, more efficient reduction was noted for lesser chlorinated PCBs. Thus, the repartitioning of PCBs from sediment to activated carbon may be faster for the lower chlorinated and less hydrophobic congeners. The differences observed for aqueous equilibrium concentrations among PCB homologues may be a matter of slower mass transfer kinetics.



*Figure 1.4.* Aqueous PCB concentrations for Hunters Point sediment contacted with varying doses of activated carbon for one month (Zimmerman et al., 2005).

Uptake of PCBs by semi-permeable membrane devices (SPMDs), biomimetic devices for lipid uptake in fatty tissue, and measurement of quiescent flux also followed the same dose response to activated carbon treatment. These tests utilized sediment mixed with activated carbon for 1 month. The uptake and flux were reduced with increasing activated carbon doses. Maximum reductions of 73% and 81% were obtained for SPMD uptake and quiescent flux, respectively (Zimmerman et al., 2005).

As shown in both physicochemical and bioaccumulation tests, increasing the carbon dose increases treatment effectiveness. Increased reductions in aqueous equilibrium concentration, SPMD uptake, quiescent flux, and bioaccumulation by a worm, amphipod and clam were all noted with increasing dose. Similar effectiveness in reducing aqueous equilibrium concentration and SPMD uptake was measured for a freshwater PCB-contaminated sediment obtained from a rural site at Lake Hartwell, SC (Werner et al., 2005) and in ongoing tests at the DDT-contaminated industrial site of Lauritzen Channel, CA.

In addition to assessing the effectiveness of activated carbon for in-situ stabilization, issues related to how this technique might be applied in the field are important considerations. To investigate if activated carbon mixed into sediment would stay in place at Hunters Point, a Sedflume device was used to measure the critical shear and stress needed to erode sediment. Results indicated that the sediment at Hunters Point is cohesive, the activated carbon is retained in the sediment under normal tidal conditions, and the critical shear stress is not decreased by mixing carbon into the sediment. Collectively, these laboratory tests suggest that the use of activated carbon stabilization of PCB-contaminated sediment in the field is feasible in terms of effectiveness of treatment and application.

### 1.5. FIELD DEMONSTRATION

Laboratory-scale bioaccumulation and physicochemical testing provide an important weight of evidence for the efficacy of the proposed activated carbon treatment. However, field demonstration is necessary to evaluate the efficacy and limitations of such innovative treatment technique. Field testing at Hunters Point started in the summer of 2004 and will continue through 2008 (Battelle et al., 2004; Cho et al., 2005). Challenges in the field include effective carbon deployment, minimization of sediment resuspension and mobilization, and adequate field monitoring tools. To meet these challenges, the field approach contains four goals, delineated in the following sections.

# 1.5.1. Demonstrate and compare effectiveness of two large-scale activated carbon mixing technologies

The cohesive sediment hinders the use of heavy equipment on the mudflats at Hunters Point. To combat this challenge, large-scale mixers were identified that either float to position during high tide or could effectively mix from a barge on the shoreline. The Aquamog (Aquatic Environments, Concord, CA), is a large barge-like craft with a rotovator attachment. The craft moves into position during high-tide, and at low tide settles onto the sediment. The rotovator would mix activated carbon applied on the surface into the sediment. As an alternative, the slurry injection system (Compass Environmental Services Inc., Stone Mountain, GA), comprises a mountable base and a long metal arm ending in prongs containing injectors/mixers. Through these mixers, a slurry of activated carbon can be injected and mixed into the sediment. Testing of both mixing technologies will not only show the effectiveness of use at Hunters Point, but also provide important information for use at other contaminated sites.

The effectiveness of the two mixing technologies will be compared by measuring that uniformity of activated carbon mixing and the effects to the indigenous community structure. Five plots will be used to assess success: 1. control, 2. Aquamog mixed, no activated carbon addition, 3. slurry injection system mixed, no activated carbon addition, 4. Aquamog activated carbon application, 5. slurry injection system activated carbon application. Application success will be measured by taking cores in all five plots before and after treatment. The amount of homogenous mixing and any detrimental effects on the biological community will then be quantified. Initial assessment trials showed well-mixed activated carbon resulting from surface carbon application and mixing by the Aquamog (Battelle et al., 2004; Cho et al., 2005).

### 1.5.2. Demonstrate reduced PCB availability

Bioaccumulation and physicochemical testing will be used to measure the success of stabilization of PCBs in Hunters Point sediment during field treatments. Sediment collected before and after mixing will be used to measure reductions in SPMD uptake, aqueous equilibrium concentrations, and sediment PCB desorption rates. PCB bioaccumulation in field-deployed clams (*Macoma nasuta*) and resident amphipods (*Corophium* spp.) will be examined once before and twice after treatment. These data in combination with laboratory feasibility testing mentioned previously should provide an engineering evaluation of the efficacy for field application of PCB availability reduction using activated carbon in-situ stabilization.

# 1.5.3. Evaluate possible sediment resuspension and PCB release

During field activated carbon application, sediment resuspension and PCB release should be minimized to ensure that contamination does not spread to adjacent areas. Before and after activated carbon application, suspended solids and dissolved PCBs will be measured over each test plot. The effect of mixing on suspension can then be quantified, and changes made in application protocols if necessary. Preliminary assessment trials measured no resuspension of PCBs in the water column after mixing of carbon with the Aquamog.

# 1.5.4. Coordinate and gain approvals from regulatory agencies

Hunters Point, as many contaminated areas, is a Superfund site, and activities are regulated by several agencies including the EPA, the Navy, and the California Water Quality Control Board. Any work at Hunters Point, therefore, requires a large amount of coordination among different groups to obtain an approved field test plan (Battelle et al., 2004). In order to achieve success as field trials continue, the coordination with agencies, public interest groups, and other consultants working at the site must also continue.

# 1.6. CONCLUSIONS

Current remediation techniques for management of contaminated sediment have significant limitations, including those of dredging for which disposal is a significant problem and residual contamination leads to continued contaminate availability to water and biota, and therefore humans. In new research, however, a look at the indigenous components of sediment that sequester hydrophobic organic contaminants has led to intense investigation of the efficacy of adding activated carbon to sediment as an in-situ stabilization method. Numerous laboratory feasibility studies found increasing the dose of activated carbon mixed with contaminated sediment strongly affects the degree of reduction in PCB availability. Similar work shows the possibility of use at other sites contaminated with hydrophobic chemicals. The effectiveness of any treatment, however, relies on successful application in the field. Success is quantified by measurements of mixing, reduced PCB availability, and minimal resuspension. Engaging regulators early in the treatment evaluation process helps ensure the development of a successful testing program and allows for ease of transition to the field.

The continuing work at Hunters Point strives to prove the efficacy of treatment, and stands as an example for applications elsewhere. As work

146

continues, the long-term stability and fate of PCBs, application to others sites and contaminants, and possible activated carbon retrieval after deployment will be studied. The work thus far is a case showing how being outcome-oriented during feasibility testing and engaging regulators early can allow for a timely transition of a new technology to the field.

# 2. Cap and treat technologies

Sediment contaminants such as polychlorinated biphenyls (PCBs) and heavy metals are a widespread and costly environmental problem. Source controls can reduce the input of contaminants into ecosystem, but the slow and continual release of sediment contaminants to the benthic community can significantly degrade a riverine or estuarine ecosystem. Human exposure to PCBs is believed to be predominantly through the food chain (trophic transfer) via fish consumption (Thomann and Connolly, 1984). In situ capping (ISC) can be a useful tool for managing contaminated sediments and improving the overall quality of the aquatic habitat, and ISC overcome some of the problems associated with dredging (e.g. residuals and ex-situ storage and treatment costs) and monitored natural attenuation (e.g. insufficient rate of burial). ISC eliminates contact between the benthic community and the underlying contaminated sediment with a physical barrier of clean material, usually sand. The objectives of ISC are to isolate and stabilize the underlying contaminated sediment, reduce contaminant flux into the biologically active portion of the sediment, and create new habitats for aquatic organisms (Reible et al., 2003). Layered sediment cap designs have been proposed (Palermo et al., 1998), which incorporate active layers to treat porewater transporting through the cap (Figure 2.1). This active layer may consist of sorbents to retard PCB porewater transport or media that provides biotic or abiotic dechlorination. Active layers in the sediment cap provide the ability to simultaneously isolate PCBs from the benthic community for long times and provide PCB mass removal.

# 2.1. CAP COMPONENTS

Sediment capping with sand or clean sediment has been used to improve sediment stability and isolate contaminants for many decades (Talbert et al., 2001). Each layer of a layered sediment cap serves a specific function. The uppermost layer is the habitat and armoring layer. This layer is specifically designed to stabilize the cap and protect it against erosion from waves, currents, ice, prop wash, and other erosion processes. It is also designed to encourage colonization of a new benthic community. From a chemical isolation point of view, this portion of the cap is compromised as this region of the cap may be continually reworked by benthic macrofauna. Mixing from bioturbation in this layer decreases the effectiveness of this portion of the cap. The habitat zone is very important, however, as the choice of substrate in this region of the cap strongly influences the ability to restore ecological function in the sediment bed. The isolation layer provides physical and chemical isolation of contaminants in the underlying sediment to the newly colonized benthic community in the habitat layer. The increased path length for diffusion, and retardation caused by contaminant sorption to solids in the isolation layer provides decades to centuries of chemical isolation time and reduced contaminant flux. A portion of the isolation layer at the sediment/cap interface will be compromised during cap placement via mixing of contaminated sediment into the cap material during placement, or by the rapid infusion of contaminated porewater squeezed out during consolidation of the underlying sediment layer. This cap layer is termed the mixing and variation layer.



Figure 2.1. Cap layers and conceptual cap design.

# 2.2. ACTIVE CAP LAYERS

Materials that sequester sediment contaminants or degrade them to nontoxic products can be included in the isolation layer to improve the effectiveness of sediments caps over that provided by sand caps. Amendments designed to control groundwater seepage through a cap, sequester heavy metals or organics, or degrade organic contaminants are available. Sediment caps containing these amendments, or a suite of these amendments can significantly lower the contaminant flux from the sediments to the overlying benthic community and water column. Caps

#### TREATMENT AND CONTAINMENT

employing multiple active materials may provide the ability to address multiple contaminants. Finally, caps containing amendments capable of degrading sediment contaminants to non-toxic products can provide simultaneous isolation and in situ treatment of contaminated sediments and eliminate the negative consequences of cap failure. Some potential amendments that can be added to sand caps are presented in Table 2.1, along with the fate process encouraged by each amendment.

*Table 2.1.* Potential Sediment Cap amendments to Encourage Contaminant Sequestration or Degradation.

Seepage	Sequestration of	Sequestration	Degradation of
Control	organics	of metals	organics
Aquablok™	Activated Carbon Coke/Ambersorb/XAD-2 Organo modified clay	Apatite Zero valent iron	Bion Soil Zero valent iron

#### 2.3. KEY ISSUES IN DESIGNING "ACTIVE" SEDIMENT CAPS

The ability of a sediment cap to reduce the contaminant flux from the underlying sediment into the habitat zone depends on the length of time the cap can isolate contaminants (isolation time), the overall reduction in the maximum rate of contaminant flux into the bioactive zone, and the rate of contaminant attenuation occurring in the underlying sediment. Contaminant transport through the cap occurs due to a chemical potential gradient across the cap (diffusion), and in the presence of groundwater seepage, also due to advective transport. Encouraging contaminant sorption or degradation within the cap can significantly decrease advective and diffusive transport of contaminants through the cap. Assessing the ability of a thin sorbent layer imbedded in a sand cap to mitigate contaminant transport through the cap therefore requires a quantitative description of the chemical fate and transport processes including sorption, diffusion, advection, hydrodynamic dispersion, and reaction in the underlying sediment and the cap.

There are several key issues in designing and implementing "active" sediment caps. Each of these issues is described briefly here, and further explored in later sections. Designing a sediment cap requires that the following be considered.

Defining appropriate performance criteria.

Identifying appropriate cap material to meet these criteria.

Evaluating the physicochemical and biological mechanisms affecting cap effectiveness.

Evaluating appropriate cap placement approaches. Evaluating required armoring layer characteristics. Developing habitat restoration criteria.

There are several ways to define the performance criteria of a sediment cap. For example, the cap can be designed to mitigate contaminant flux to some acceptable level defined by the local, state, and federal regulators. The flux may be chosen to ensure that the criterion critical concentration (CCC) for a specific sediment contaminant is never reached. Alternatively, a cap can be designed to provide a minimum isolation time (e.g. 30 years) before contaminants break through the cap layer.

Appropriate "active" cap materials must be identified for each site. The material, or combination of materials must be compatible with the contaminants present, and with the physical, chemical, and biological conditions at the site. Often, bench scale treatability studies are required to verify the sorption, degradation, and lifetime of each of the amendments being considered. The biological and physicochemical mechanisms affecting cap effectiveness must also be evaluated. For example, the rate of pore water advection due to groundwater seepage, or the effects of the sediment matrix components on the sorption strength or reactivity with sediment contaminants must be evaluated.

Once appropriate cap materials are identified, cap placement approaches must be determined. Placement techniques such as particle broadcasting, or the use of geotextiles to place "active" layers of high value material are potential placement techniques. The ability to use these methods will often be site specific. In addition to placing the cap, the required armoring layer characteristics must also be determined. The armoring characteristics will depend on site specific properties such as flow, ice, etc.

The overall goal of capping is to improve the quality of the aquatic habitat. As such, integrating sediment caps with habitat restoration is an important benefit of in situ capping. The criteria for habitat restoration must be identified and included in the design stages.

#### 2.4. PERFORMANCE EVALUATION

For hydrophobic organic contaminants such as PCBs and PAHs, the dominant factor influencing the rate of migration through the cap is the sorption strength of the sorbent layer in the cap (retardation factor). Sorbent layers such as high organic-content soil, and thermally altered carbonaceous materials (TACMs) such as activated carbon and coke have been considered. The TACMs are known to very strongly adsorb contaminants, and PAHs and PCBs bound to these materials are less bioavailable (Talley

#### TREATMENT AND CONTAINMENT

et al. 2002, McLeod et al., 2004). Recently, the ability of a thin layer of an organic-rich soil, activated carbon, and coke at mitigating PCB transport through a sediment cap amended with each material was conducted (Murphy et al., 2005). These treatability studies determined the Freundlich parameters for sorption of 2,4,5-PCB to coke, soil, and activated carbon,

$$C_{i,solid} = K_f C_{i,water}^n \tag{1}$$

151

where C<sub>i, solid</sub> and C<sub>i, water</sub> are the sorbed and aqueous phase concentrations of sorbate i, K<sub>f</sub> and n are the Freundlich parameters, and the physical properties relevant to transport through a layer of the material (effective porosity and dispersivity). The Freundlich parameter, log K<sub>f</sub>, ranged from 4.7 for coke to 6.2 for activated carbon. The organic-rich soil (log  $K_f=4.9$ ) was very similar to coke. These properties were then used in numerical models to predict the flux of 2,4,5-PCB through a 1.25-cm thick layer of each sorbent (Figure 2.2). This provides a conservative estimate for PCB flux through the cap as most PCB congeners present in sediments are higher chlorinated (more hydrophobic) PCB congeners and will have higher values of log K<sub>f</sub>. In the absence of groundwater seepage, a 1.25-cm layer of soil, coke, and AC all provided isolation times greater than ~500 years and lowered PCB flux into the bioactive zone compared to sand. With moderate groundwater seepage (U=1cm/day), strongly sorbing high capacity sorbents such as activated carbon can still provide long isolation times (>1000 years).



Figure 2.2a. Flux of 2,4,5-PCB for each sorbent type in the absence of advection.



*Figure 2.2b.* Flux of 2,4,5-PCB for each sorbent type with groundwater advection (Darcy velocity=1cm/d).

It must be noted that these simulations employed many simplifying assumptions to assess contaminant transport (e.g. equilibrium sorption, no sorption competition), but is useful for comparing the relative effectiveness of different sorbents and providing a sense of the isolation times that may be achieved with sorbent-amended "active" sediment caps. Potentially important processes that were not addressed in this analysis include competitive sorption, sorption kinetics, and particle-facilitated transport of strongly sorbing contaminants. Competitive sorption between PCBs and dissolved organic carbon in sediment porewater may decrease  $K_f$  and lower the sorbents effectiveness. Slow sorption kinetics of PCBs onto cap sorbents may also limit effectiveness. For example, the assumption of equilibrium sorption may not be valid at high groundwater seepage velocities because the sorbent layer is thin and the residence time within the "active" layer may be too short to achieve a sorption equilibrium.

#### 2.5. CAP PLACEMENT

In situ capping requires that a layer of clean or "active" material be placed over the contaminated sediments. Cap placement should be able to be done quickly with available construction tools (e.g. clam shell), and result in a cap of relatively uniform thickness that does not contain significant amounts of entrained contaminated sediments. The region of the cap where contaminated sediment has mixed with clean cap material will lose its effectiveness and thus must be minimized. The cap thickness and integrity must therefore be monitored during or after cap placement. The leading method to place subaqueous caps is by particle broadcasting. In particle broadcasting, a crane and clam shell is used to distribute a layer of high density granular material (e.g. sand) over contaminated sediment. Factors limiting cap placement by this methods are that thin layers (e.g. <6 inches) are difficult to place uniformly and with certainty, and that highly porous low bulk density materials (e.g. TACMs) can be difficult to place as their settling velocity is low and variable and some fraction of the material may not readily sink.

Recently, four 1/4-acre test plots were placed in the Anacostia River (Wash. DC) in April 2004 to evaluate different placement and monitoring methods under realistic in-situ conditions methods, and to compare the effectiveness of traditional (sand) and innovative "active" caps relative to uncapped areas. A clam shell was used to place sand and apatite (for heavy metal sequestration) caps, and a clay Aquablock<sup>TM</sup> cap (hydraulic barrier to control tidal seepage). A 1.25 cm thick coke-filled geotextile (24 kg/m<sup>2</sup>) accurately deployed a thin coke layer into a 15-cm subaqueous sediment cap without releasing fines or larger low density (floatable) coke particles. The sorbent layer (Figure 2.3) was placed over the contaminated sediment, followed by a 15-cm layer of clean sand to secure it and to allow a place for the benthic community to colonize. Approximately 100  $m^2/hr$  was placed using this method, and mixing between the contaminated sediment and the cap were minimal. This is a reliable method that can be used to place thin layers of high value (e.g. nanoiron) reactive material into caps. All caps placed using the clam shell, sand, sand overlying the coke layer, apatite, and Aquablock<sup>TM</sup> were accurately placed as determined by sonar scans conducted after placement (Figure 2.4). Each material was placed uniformly and with an accuracy of  $\sim \pm 10$  cm, but with a standard deviation of less than 5 cm in all active cap layers.



*Figure 2.3.* Coke-filled geotextile placed in the Anacostia River.



#### 2.6. CAP AND TREAT SCENARIOS

Capping with traditional sand caps and with sorbent amended caps can provide very long chemical isolation times before contaminants break through the cap. A potentially highly effective approach to in situ management of sediments is to combine containment with contaminant degradation to non-toxic products. There are several ways in which cap and treat scenarios can be employed. First, the cap itself can contain "reactive"



Figure 2.4. Cap heights after placement determined by sonar.

materials that degrade contaminants. This scenario is essentially a permeable reactive barrier technology similar to that used at a variety of contaminated groundwater sites. The performance issues important in groundwater sites (e.g. lifetime, production of toxic reaction intermediates, etc.) are also important in a cap application and less well understood. Second, the cap can be used to contain contaminants in the underlying sediment for long times (decades to centuries) while the contaminants in the underlying sediment are degraded through natural biological degradation, or through enhanced biological or abiotic degradation. Also, the ability to combine in situ treatment with habitat restoration is an attractive cap and treat scenario. Each of these scenarios is discussed.

#### 2.7. IN-CAP TREATMENT

In an in-cap scenario, contaminants are degraded as they transport into the sediment cap. In theory, this type of "active" sediment cap can be used to treat the sediment porewater as it migrates through the cap and mitigates contaminant transport from the underlying sediment to the habit zone indefinitely. The concept is similar to the in situ permeable reactive barriers used in groundwater treatment (Starr and Cherry, 1994). The difficulty of this approach lies in the ability to accurately place thin layers of high value reactive material, identifying materials that afford contaminant degradation rates raid enough for use in a thin layer that may have limited reaction time, and identifying materials that remain active for extended periods of time (years to decades), or an easy method to replace the reactive material. Each of these challenges must be addressed before a robust in cap treatment method will be available.

One candidate for reactive media is zerovalent iron (ZVI). ZVI, nanoscale ZVI, or some other reducing agent can be used to encourage the reductive dechlorination of chlorinated aromatics, chlorinated solvents (Liu et al., 2005) and PCBs (Lowry and Johnson, 2004), the reduction of perchlorate (Moore, A. et al., 2003), and the reduction and immobilization of hexavalent chromium (Alowitz and Sherer, 2002). Controlled placement using a geotextile (Murphy et al, 2005) makes iron or nanoiron a viable option. Exhausted iron in a geotextile could be easily exchanged for fresh iron, especially if constructed in a funnel and gate arrangement where the active treatment area is small. A second option is to use an engineered system to provide a redox gradient from reducing to oxidizing conditions at the sediment-water interface. Composted soils or creation of a biologically active layer through the introduction of nutrients or other reactants (e.g. hydrogen) could be used to create a highly bioactive region in a cap that can biodegrade sediment contaminants as they migrate through the cap.

#### 2.8. SUB-CAP TREATMENT

A second alternative is to use sediment caps amended with activated carbon or some other highly sorptive material, coupled with in situ treatment of sediment contaminants in the underlying sediment. The very long chemical isolation times afforded by sorbent-amended caps (Figure 2.5) allows adequate time for slow natural attenuation processes such as biodegradation that can further mitigate the risks to benthic and aquatic organisms. Engineered methods to increase the rate of biotic or abiotic attenuation (complete destruction) in underlying sediments could provide a costeffective means to contain and treat PCB-contaminated sediments in situ. For example, contaminant attenuation rates as slow as  $10^{-3}$  yr<sup>-1</sup> (t<sub>1/2</sub>=700 yr) in the underlying sediment can lead to significant (>90%) mass removal in the underlying sediment and decreases the maximum contaminant mass flux into the bioactive zone by a factor of 3 for an AC-amended sediment cap in the scenario evaluated in Figure 2.5). To be effective, methods to deliver and mix amendments into the contaminated sediments beneath the cap must be developed and tested. Also, methods to monitor the rate and extent of contaminant degradation beneath the cap must also be developed and tested.



*Figure 2.5.* Effect of contaminant degradation in the underlying sediment on contaminant flux through the cap.

156

#### 2.9. CAPPING WITH HABITAT RESTORATION

The overall goal of an sediment remediation project is to improve the aquatic habitat, and often to restore the affected habitat back to precontamination conditions. Cap and treat scenarios offer the potential to simultaneously manage contaminated sediments, and to restore habit. For example, a sediment cap can be designed into a site-wide habitat redevelopment designed to restore wetlands and promote fish spawning. Substrates overlying the cap layer and contaminated sediment can be selected to promote the colonization of specific species of plants, fish, or macrofauna. A conceptual model of a habitat restoration cap is shown in Figure 2.6.



Figure 2.6. Conceptual design of sediment cap with habitat restoration.

### 2.10. SUMMARY AND KEY RESEARCH NEEDS

Sediment capping with "active" caps is a simple and effective in situ sediment remediation technique that can provide an alternative to dredging.

There is great potential for simultaneous containment and treatment of sediment contaminants, along with habitat restoration. Methods to accurately place thin layers of high value material within sand caps are available, which enables to potential use of high value "active" capping materials such as activated carbon and iron. Despite the great potential of "active" sediment caps and the decades of experience placing sand caps, basic research is still needed to elucidate the fundamental physical, chemical and biological processes that control contaminant fate and transport and the biogeochemical gradients within a cap. Tools to measure and monitor these processes (e.g. groundwater seepage, attenuation rates) are needed. The influence of seepage of non-aqueous phase liquids (NAPL) or gas ebullition on the transport of contaminants through the cap is needed.

Continued development of cap amendments that encourage sequestration or degradation of contaminants and can last for long times under in situ conditions are also needed. A better understanding of sediment slope stability, ice scour, and other erosion processes is needed to ensure that caps maintain their integrity throughout their designed life. Finally, the integration of habitat improvement into the design of sediment caps is in its infancy and must be explored further to fully reap the benefits of this approach.

# 3. Phytoremediation of sediments

# 3.1. INTRODUCTION

Many organic and inorganic compounds including polychlorinated biphenyls, phenols, polyaromatic hydrocarbons, heavy metals are priority soil contaminants because of their toxicity and tendency to persist in soils, sediments and to escape biological degradation. The fate of contaminants in the environment is controlled by a combination of interacting processes. They can be classified as physical, chemical or biological. Physical processes are responsible for the movement of contaminants through the soil and subsurface away from their source. Chemical and biological processes determine the extent to which compounds will be transformed. In some cases the most important transformations occur abiotically, while in other cases they are mediated by microorganisms with the help of other living organisms, for example with plants. A number of plant species are able to accumulate high amounts of heavy metals in their above-ground tissues or to degrade various organic soil pollutants. Therefore phytoremediation, i.e. bioremediation using plants has received during last ten years significant attention.

However, it has become clear that phytoremediation using conventional plants has some potential for use in remediation; but also that it is encumbered by limitations that prevent widespread benefit from its significant economics. The effects of plants that accumulate metals to a level high enough to make harvesting and metal recovery economical are hampered by their mostly very small biomasses. Some toxic heavy metals are immobilised in soils and taken up only poorly by known metalaccumulating plants (Meagher 200). Studies of the use of deep-rooted trees have shown that chlorinated hydrocarbons (CHCs), such as the premier groundwater pollutant, trichloroethylene (TCE), are taken up by trees and are degraded in tree plantations. The potential of genetic engineering for enhancing the biodegradation of xenobiotic toxic compounds has been recognised since the early '80s. Initially there was hope that transgenic
bacteria would provide powerful methods for remediation of environmental problems. However, this approach has failed due to two factors: inconsistent survival of genetically modified microorganisms in soil and water environments, and reluctance by regulatory authorities to approve the release of genetically modified microbes into the environment. The latter reluctance stems from the high growth rate and small size of bacteria and the ease of horizontal gene transfer among dissimilar prokaryotes. These problems can be overcome by using plants rather than microorganisms as the genetically engineered environmental cleanup biosystems. Primarily by virtue of their size, vascular plants are more easily controlled than microorganisms. By selecting sterile plants and controlling propagation by harvesting the plants prior to flowering, uncontrolled genetic releases from phytoremediation plantations can be prevented. Thus, by using plants, the promise of genetically engineered environmental cleanup systems can finally be realized (Macek et al., 2004a, b, Chaney et al., 2005, Burken et al., 2005).

During remediation of organic pollutants, it would be optimal, if the plants or plants together with a microbial consortium (Clemens et al., 2002) would be able to mineralise the compound fully to non-toxic products. It is generally accepted, that phytoremediation can be exploited on large areas, with shallow contamination accessible by the roots of species used. Decontamination of polluted water can be realized with the help of aquatic plants, different grasses are used for wetlands decontamination, contaminated sediment-derived soils usually polluted with many types of different xenobiotics can be cleaned ex-situ for example by *Salix* (Vandecasteele et al., 2005) or other plant and tree species.

Especially during the accumulation of heavy metals, but also during removal of organic contaminants the main limiting factor in large-scale exploitation of plants is the long time necessary for soil decontamination. For this reason, much effort is devoted to preparation of plants specifically phytoremediation purposes by breeding or tailored for genetic manipulations. Recently more information appeared in literature about possibilities to increase the expression of the genes already present, or introduction of bacterial or mammalian genes into plants, in order to increase the natural ability of plants to cope with xenobiotics (Krämer and Chardonnens 2001, Macek et al., 2002a, Krämer 2005). On real contaminated soil, only a few transgenic plants have been tested (Kärenlampi et al., 2000, Bizily et al., 1999, Macek et al., 2002b, Heaton et al., 1998, Pavlíková et al., 2004a).

Transgenic plant technology showed many obvious advantages over conventional plant breeding approaches for crop improvement. The discoveries allowed the engineering of new transgenic plants generating desirable products, such as enzymes, polymers and vaccines. Among new approaches, the use of transgenic plants specifically tailored for the bioremediation of organic pollutants and heavy metals have recently occurred. Beside modification of vascular plants, genetic transformation of trees (Vandecasteele et al., (2005) or monocotyledonous wetland grasses (Czako et al., 2005) was shown.

Not all plant species are equally well suited to metabolise or accumulate pollutants. Requirement for plants removing heavy metals is to grow fast in the contaminated environment, to be resistant, able to accumulate toxic metals and transfer cations or oxyanions into the harvestable (above ground) parts, or transform them into less-toxic forms (Krämer and Chardonnens 2001). At the end of the growth period, plant biomass is harvested, dried or incinerated, and the contaminant - enriched material is deposited in a special dump or added into a smelter (Kramer, 2005). The most important parameter for selection of suitable plants is not the tolerance of the plant to heavy metals, but the effectiveness in their accumulation. In addition to accumulation capacity, biomass production has to be considered in order to determine the total metal uptake. To achieve better economy of the whole process, use of energy plants or some technology crops is preferable, like hemp or flax as suggested by Griga et al. (2003). Recently also use of GM trees was described (Peňa and Séguin, 2001). Especially useful are plants able to remove more than one pollutant, because contamination, especially at industrial sites, is usually caused by a mixture of different toxic compounds. Many elemental pollutants enter to plants by basic transport systems designated for nutrient uptake. A number of xenobiotics are than stored in vacuoles as protection against their toxic effects.

Plants have developed their own systems for binding of heavy metals based largely on the synthesis of phytochelatins, described by Grill et al., (1989). Heavy metal binding in plants is normally achieved, as reviewed by Kotrba et al., (1999) by different peptides, phytochelatins and phytosiderophores. Phytoextraction of metals posses large economic opportunities because of the size and scope of environmental problems associated with metal-contaminated soils and the competitive advantages offered by a plant-based technology (Chaney et al., 1997). In case of radioactive compounds, showing high biological effect even at very low concentrations, the perfect decontamination by plants seems to be practically impossible (Soudek et al., 2003).

Complex interactions of transport and chelating activities control the rates of metal uptake and storage. In recent years, several key steps have been identified at the molecular level, enabling us to initiate transgenic approaches to engineer the transition metal content of plants (Kramer et al., 2005). Clemens et al. (2002) in excellent review summarise the determinants of metal accumulation, mobilisation, uptake, sequestration, xylem transport, discussing the role of different families of transporter proteins. Many problems are still unclear, the transport and storage forms of transition metals are largely unknown and many systems have yet to be analysed. Despite the unsolved questions, different attempts were already done to prepare transgenic plants with improved properties for phytoremediation purposes.

# 3.2. PREPARATION OF TRANSGENIC PLANTS

Transgenic plants used in everyday practice are nowadays represented especially by agricultural crops with increased resistance against insect pests, infections or herbicides (Koprowski and Yusibov 2001). As the next generation of genetically modified (GM) crops were introduced plants that can serve as a source of food products with increased nutritive value (e.g. with better composition and content of unsaturated fatty acids, vitamins etc.). Further, here belong plants with increased resistance against extreme weather conditions, or changes in their ability of ripening. In addition to these advantages, exploited in agricultural practice (wheat, corn, soy, rape, etc.) were prepared also GM plants able lemon. produce to pharmaceutically important products like antibodies or enzymes (McGarvey et al., 1995). Most recently as a new approach to plant protection GM plants have been prepared emiting insect sexual pheromone with the aim to disrupt mating behaviour of the target pest, thus decreasing the need for chemical pesticides (Nesnerová et al., 2004).

Genetic manipulations are thus finding increased use in plant biotechnology. Different transgenes influence the yield and qualities of harvested crops, genes regulate ripening, composition of fats and the quality of seed proteins. The studies of tissue specific promoters will allow genetic engineering to create new categories of GM plants producing the required products in a better-regulated way.

Until now most suitable vector for gene transfer into plants are the Ti plasmid of soil bacterium *Agrobacterium tumefaciens*, initiating tumours in plant tissue, and the plasmid Ri of *A. rhizogenes*, which induces hairy root formation. Hairy root cultures are exploited in phytoremediation studies for more than a decade; we used them for cadmium uptake studies (Macek et al., 1994, 1997a, b) and for PCB degradation studies (Mackova et al., 1997a, b). The exploitation of the agrobacterial gene transfer techniques is one of the main approaches for introduction of foreign genes into the plant genome (Ondrej et al., 1999), but even this already widespread technology has some inborn limitations. Agrobacterial plasmids as vectors are well

suited for transformation of dicotyledonous plants, but cases of successful use on monocotyledonous plants (corn, rice, wheat) were already also described (Gelvin 1998, Czako et al. 2005). At present, the natural wildtype strains of Agrobacterium are no more used for the genetic manipulations, replaced by strains with modified Ti plasmids. The genes responsible for plant hormone production, that cause the formation of tumours, and genes coding opine formation, were removed, being not necessary in constructed GM plants. In the same time, genes serving as selective markers were introduced into the original T-DNA. Among the most exploited marker genes belong bacterial β-glucuronidase (GUS) and luciferase genes, and recently also the gene coding the green fluorescent protein (GFP). Because the big size of the natural Ti-plasmid caused problems in handling, the T-DNA with border sequences was introduced into smaller plasmids derived from pBR 322, which are nowadays commercially available. Within the T-DNA region of these newly prepared plasmid unique restriction sites are present, allowing introduction of constructs prepared for transfer into plant genome. Genetic construct, intended to be transferred into plants, has to contain proper promoter region (most often constitutive promoter from cauliflower mosaic virus CaMV 35S, promoter controlling ubiquinone expression in corn, or other organ or tissue specific promoters active in plants), sequence coding the required gene and termination signal. At present much effort is devoted to isolation of novel promoters inducible e.g. by some environmental effector (like the presence of heavy metals). Another technique using "natural" gene transfer is the use of viral vectors, but this is leading only to transient, not stable transformation. In cases, where the above technique is inapplicable, other methods of transfer have to be exploited. Some were developed after finding of methods allowing regeneration of a whole plant from callus culture or from a single protoplast. These techniques include direct DNA transfer (Maliga 1990), electroporation, DNA vacuum infiltration, different chemical modifications, microinjection or ballistic methods (Old and Primrose, 1994).

# 3.3. INCREASED ACCUMULATION OF HEAVY METALS

Among other natural mechanisms, allowing plants to grow in heavy metal contaminated environment especially many studies dealt recently with the phytochelatins. In Plant Kingdom it is a general feature to complex heavy metals by phytochelatins. These peptides are formed by phytochelatin synthase from glutathione, when plants are challenged by Cd or other heavy metals. Much effort to explanation of their formation has been devoted by Zenk and co-workers (e.g. Chasaigne et al., 2001, Oven et al., 2001).

With the aim to improve metal uptake transgenic plants were prepared with increased formation of glutathione synthase or phytochelatin synthase, what in both cases led to increased Cd accumulation (Kärenlampi et al., 2000). Transgenic plants bearing foreign genes for proteins transporting metal across membranes have been also prepared, as summarised in a review by Krämer and Chardonnens (2001). In some laboratories genes coding different types of metallothionein (mammalian, yeast, insect and human) were introduced into plants. This effort led mostly to increase of the resistance towards some heavy metals (Kärenlampi et al., 2000, Macek et al., 1996, Macek et al., 1997b), but not yet to an increase of accumulation (Liu et al., 2000). Dorlhac de Borne et al. (1998) describe cadmium partitioning in transgenic tobacco plants expressing a mammalian metallothionein gene in detail.

Reserves are not only in better understanding, possibility of modifications and improvements of normal plant mechanisms themselves, but also in other ways to improve the overall yield of the remediation process. Improved bioavailability of metals caused by changes of exudate, increased excretion of organic acids, or cooperation with rhizospheric microorganisms might be itself target of genetic engineering, or increase the performance of transgenic plants.

In order to increase the metal-binding capacity the possibility of introducing an additional metal-binding domain with high affinity towards heavy metals into the implemented protein was studied (Macek et al., 1996). In this way by combining the gene CUP1 for yeast metallothionein from Saccharomyces cerevisiae and the gene for the histidine anchor with high affinity to heavy metals from the commercial plasmid pTrcHis (Invitrogene) a construct has been prepared and introduced by A. tumefaciens into tobacco (Macek et al., 2002b). The vector was plasmid pBI121, bearing CaMV 35S promoter from cauliflower mosaic virus, the reporter gene for B-glucuronidase (GUS) and all regions necessary for transformation using Agrobacterium, purchased successful from CLONTECH. The tested plants involve tobacco, Nicotiana tabacum L., var. Wisconsin 38, as the control (WSC-38), and genetically modified lines of the same variety, bearing the transgene HisCUP (gene coding a yeast metallothionein, CUP, combined with a polyhistidine tail), CUP (gene coding a yeast metallothionein), GUS (reporter gene for ß-glucuronidase), and HisGUS (reporter gene for ß-glucuronidase, combined with a polyhistidine tail) genes under a constitutive promoter, thus enabling us to follow the heavy metal tolerance and uptake changes as function of the transgene presence. Non-transformed control – WSC-38 and four constructs of transgenic tobacco (CUP, HisCUP, GUS and HisGUS) were tested for accumulation of Cd in above ground biomass and roots. The screening test was conducted on sand nutrient media with addition of Knop's nutrient solution modified by addition of cadmium (0.2 mg Cd.l<sup>-1</sup> as Cd(NO<sub>3</sub>)<sub>2</sub>.4H<sub>2</sub>O), The plants cultivated during the elongation period were harvested after 6 weeks of growth. Plant material was decomposed by dry ashing procedure and Cd concentrations in plants and roots were determined by atomic absorption spectrometry using Varian SpectrAA-300 (Pavlikova et al., 2004a, b).

The model pot experiment was set up using two soils with different Cd content – unpolluted Chernozem and Cambisol with contamination caused by the atmospheric emissions. In experiment, 5 kg of soil were thoroughly mixed with 2 g N applied as  $NH_4NO_3$ . P and K were added as  $K_2HPO_4$  at a rate of 0.44 g P and 1.1 g K per pot at the same time. The pots were irrigated by deionised water and humidity of the soil was kept on 60% of maximal water capacity for duration of the whole experiment. The plants were cultivated for three months. To evaluate statistical significance of the changes during experiment, each treatment was carried out in four replicates analysed separately.

# 3.3.1. Screening test in sand medium

Presence of introduced genes in plant tissue was proved by PCR, RT-PCR respectively (Macek et al., 2002b). The abilities of transgenic plants were tested first in sand medium containing defined amount of cadmium. The results showed that while control WSC-38 whole plants accumulated in average 15.8 mg Cd.kg<sup>-1</sup> in leaves and stems and 19.9 mg Cd.kg<sup>-1</sup> in roots (Pavlikova et al., 2004a) the presence of the CUP1 yeast metallothionein gene did not increase the accumulation much above the average of control. Cd concentration in HisGUS constructs did not differ from control treatment. The HisCUP construct proved to have positive effect on Cd accumulation. The Cd content in above ground biomass was increased by 90% comparing to the non-transformed control and the Cd content in roots was decreased by 49%. Distribution of this element among leaves, stems and roots showed majority of Cd in leaves of both treatments (87% from total uptake by HisCUP and 71% by control). Uptake by stems was the lowest - 5% control, 3% HisCUP. Uptake by roots also showed the difference between control and HisCUP construct (control 24%, HisCUP 10% from total uptake).

Cd dose	Cd content (mg . kg <sup>-1</sup> )			
$(mg Cd.l^{-1})$				
	WSC 38		HIS-CUP	
	Roots	Above-ground	Roots	Above-
		biomass		ground
				biomass
0.2	$13.6\pm2.9$	$16.6\pm5.9$	$11.6 \pm 6.2$	$30.1\pm3.5$
0.6	$34.4\pm4.17$	$46.8\pm4.6$	$36.8\pm2.9$	$74.7\pm3.3$
1.8	$142.5\pm0.7$	$125.5\pm2.1$	$150.0\pm6.4$	$144.6\pm1.3$
5.4	$520.0\pm15.5$	$259.0\pm4.5$	$514.0\pm23.6$	$336.0\pm5.9$
16.2	$1908.5\pm51.2$	$647.5 \pm 111.6$	$1886.8\pm100.5$	$736.5\pm269.2$

Table 3.1. Cd content in tobacco plants growing in sand medium (Pavlíková et al., 2004a).

The Cd content in above ground biomass of tobacco growing at Cd concentration of 0.2 mg Cd.1<sup>-1</sup> was increased by 80 % compared to the nontransformed control (Table 3.1). This result confirmed our previous experiments that accumulation of cadmium significantly increased in plants bearing the transgene coding the polyhistidine cluster, combined with yeast metallothionein (Macek et al., 2002b). Significantly increasing accumulation was also determined in HisCUP tobacco growing in Cd concentration of 0.6 mg Cd.l-1. Cadmium content in roots of control tobacco did not differ from transgenic plants and it was lower compared to the above ground biomass. Next three doses 1.8, 5.4 and 16.2 mg Cd.I-1 caused lower differences of Cd contents in plant above ground biomass and roots between control and HisCUP plants. Dose of 5.4 mg Cd.l-1 induced cadmium toxicity symptoms in leaves of control tobacco that resembled Fe chlorosis. The highest dose of Cd (16.2 mg Cd.l-1) in sand medium depressed root and plant growth and was lethal for this control treatment. HisCUP tobacco showed increasing plant resistance against stress response induced by cadmium. Both doses were not toxic for transgenic plants.

### 3.3.2. Model pot experiment in real contaminated soil

Experiments in contaminated soil gave better picture about properties of transgenic plants and showed their real capacity to accumulate heavy metal and survive in its presence, both affected primarily by bioavailability of Cd. This is different in sand medium and in real contaminated soil.

The bioavailability of Cd to plants can be more accurately predicted determining available portion of Cd than the soil total content (Adriano, 2001). Before the experiment plant-available Cd formed 8% from total Cd content in polluted Cambisol. Our results did not show changes of available Cd content after harvest of tobacco plants in this soil (Pavlikova et al.,

2004b). Unpolluted Chernozem contained 0.56% available Cd from total Cd content. Significant decrease of Cd was observed only in this soil after harvest of transgenic tobacco.

Both yield and content of dry matter of transgenic plants were not significantly differing compared to control. The most important factor affecting Cd content in tobacco plants cultivated in this experiment was the original Cd content in soils. Content of Cd in tobacco growing on polluted Cambisol with high portion of available Cd was 14-17 fold higher compared to Chernozem. The content of Cd in the above ground biomass of transgenic tobacco cultivated in non-polluted soil increased by 75% compared to control plants and by 45% in polluted soil (Table 3.2). Very important fact for phytoremediation is, that most of the accumulated cadmium was stored in the leaves, while the metal content in roots was lower comparing to the control plants. Previously it was shown by King and Hajjar (1990) that Cd concentrations in tobacco declined markedly with higher leaf position on the stalk. This general decline with increasing leaf height in the tobacco plant suggests that Cd has limited mobility within the plant and its concentration is related to the age of the leaf.

	Cd content (mg . kg <sup>-1</sup> )		
Soil	Treatment	Roots	Above ground
			biomass
Chernozem	WSC 38	$0.24\pm0.12$	$1.25\pm0.60$
	His CUP	$0.13\pm0.03$	$2.19\pm0.95$
Cambisol	WSC 38	$3.61\pm0.65$	$21.33 \pm 4.84$
	His CUP	$2.50\pm0.41$	$31.43 \pm 5.65$

Table 3.2. Cd content in tobacco plants growing in pot experiment (Pavlikova et al., 2004b).

Uptake of heavy metals by plants is characterized by transfer factors values (Kiekens and Camerlyck, 1982). Transfer factor is calculated as quotient of metal total content in plant biomass and in soil. According to Kloke et al. (1984) Cd transfer factor is 1-10. Higher values are desribed for leafy plants. Tlustos et al. (1998) calculated transfer factor values 1.50 - 3.59 for spinach leaves and 1.65 - 3.32 for spinach roots. Our results showed high transfer factor for tobacco, mainly for above ground biomass of transgenic plants 5.26 - 6.65 (Pavlikova et al., 2004a).

The comparison of genetically modified lines of tobacco with control plants showed an increase of resistance and good growth capability in cadmium contaminated soil, together with increased accumulation. Already before transformation of plants prepared constructs were tested in *E. coli* in

a series of growth experiments, and together with improved resistance a substantial increase of Cd accumulation was found (Macek et al., 1997b, 1998) already in bacterial biomass.

### 3.4. GM PLANTS FOR REMOVAL OF PCB

biphenyls represent another important xenobiotic Polychlorinated contaminating environment in many, especially in industrialised countries. Even now, 35 years after their production was banned in civilised countries, and 18 years after finished in former Czechoslovakia, these persistent compounds still represent a danger for biological diversity and for man. Being taken up by plants. PCBs can enter food chain and accumulate mostly in fat tissues. The ability of plants to metabolise them has been already proved, and is further studied (Kucerova et al., 2000). In our laboratories, we have a quite complex approach to PCB bioremediation (Demnerova et al., 2002, Mackova et al., 2001), which includes studies on enzymes involved (Chroma et al., 2003), products formed and attempts to introduce bacterial genes from the bacterial bph operon (Francova et al., 2003, Sura et al., 2005)). Starting with screening of different plant species in vitro (Mackova et al., 1997b), identification of plants growing on longterm contaminated soil (Ryslava et al., 2003), and characterisation of cooperation between degrading bacteria and plants, the best suited models were selected. Some of the products formed in the first, activation step of the detoxification pathway in plants were already identified as hydroxychlorobiphenyls (Kucerova et al., 2000). We proved that several derivatives of these hydroxylated monochlorobiphenyls can serve as substrates for bacterial enzymes of the bacterial biphenyl operon, and prepared plants bearing gene C from the bacterial biphenyl operon. responsible for opening of the ring of recalcitrant PCB molecules (Francova et al., 2004) (Fig. 3.1), could degrade them. Attempts to genetically improve the PCB-converting ability of plants are a further example of this approach in environmental remediation. While plants detoxify PCBs by hydroxylation and conjugation to hydrophilic molecules, followed by storage, some bacteria can open the biphenyl ring, thus allowing further steps leading to mineralization (see Fig 3.1, step 3). For this reason, as described in Borovka et al. (1998) and in Macek and Macková (1999), a bacterial gene (*bph C*) responsible for biphenyl ring opening (see Fig. 3.1) was chosen for these attempts.



Figure 3.1. Bacterial degradation of PCB – upper pathway (Hein et al., 1998, Pieper 2005).

I - biphenyl, II - 2,3-dihydroxy-4-phenyl-hexa-4,6-diene, formation catalyzed by gene *bphA*, III - 2,3-dihydroxybiphenyl , formation catalyzed by enzyme coded by gene bphB, IV - 2-hydroxy-6-oxo-6-phenyl-hexa-2,4-dienoic acid, formation catalyzed by enzyme of gene bphC, V - benzoic acid, VI - 2-hydroxy-penta-2,4-dienoic acid, formation catalyzed by enzyme coded by gene bphB.

This gene from *Comamonas testosteroni* B356 is coding the enzyme responsible for PCB ring opening, 2,3-dihydroxybiphenyl-1,2-dioxygenase, characterised by Sylvestre and co-workers (Hein et al., 1998). Cloning into *Agrobacterium tumefaciens* and plant transformation was performed by standard methods (Francova et al., 2003). Constructs harbored the constitutive promoter CaMV 35S.

For detection and other physiological studies constructs coding fusion proteins bphC and reporter genes for green fluorescent protein (GFP), luciferase (LUC) or glucuronidase (GUS) and His tail (6x histidines) were also designed and prepared (Sura et al., 2005).

The transgenic plants were selected after transferring the regenerants to selective medium. The best yield of regenerated transformed plantlets was obtained on medium according to Linsmeier and Skoog (LS medium) containing kanamycine or hygromycine (Sura et al., 2005) and supplemented by naphtaleneacetic acid and kinetin, as suggested by Macek et al. (1998).

### 3.4.1. Testing of the properties of GM plants containing bphC gene

Successful transformation of plant tissue was confirmed on different levels of gene expression:

A) First successfully transformed plantlets proved kanamycine or hygromycine resistance and regenerants grew on LS medium in presence of antibiotic (kanamycine or hygromycine). In particular also by root formation of selected regenerants on rooting-supporting LS medium in presence of kanamycine (hygromycine).

B) By isolation of plant DNA and RNA, identification of *bphC* fragment after PCR or RT-PCR amplification with specific primers (Sura et al., 2005) and electrophoresis in agarose gel and comparison of its size with bacterial gene (Fig. 3.2, 3.3),

C) Histochemical detection of reporter genes cloned in fusion with bphC (glucuronidase and luciferase) – see Fig 3.5.

D) Expressed fusion protein BphC/GFP was detected by Western blot analysis / see Fig. 3.4.

E) Growth and better resistance to high PCB concentration (see Fig. 3.6).



*Figure 3.2.* Agarose-gel electrophoresis after PCR of plant DNA transformed with bacterial *bphC* gene.



*Figure 3.3.* Agarose-gel electrophoresis after PCR of cDNA obtained after RT-PCR of plant RNA containing *bph C* gene. The plant RNA was isolated, treated by DNase I, and further RT-PCR proved presence of appropriate RNA (see Figure 3.3).



*Figure 3.4.* Western blot with commercial antibody against GFP. Protein extracts were prepared from different clones (R1A, R2D, R2G, R3A) grown after transformation with *Agrobacterium tumefaciens* bearing fused genes bphC/GFP and negative control from nontransformed tissue.



*Figure 3.5.* Histochemical detection of glucuronidase in transgenic plants (G1, G2, G3) and nontransformed plant used as a negative control.



*Figure 3.6.* Growth of transgenic (3KFC) and nontransgenic (WSC) tobacco on agar medium with 250 ppm PCB. The figure shows the growth of nontransgenic (WSC) and transgenic (3KFC) plants (vegetatively multiplied) on medium containing 250 ppm of PCB. It can be seen that transgenic culture exhibits much higher resistance to high PCB concentration than nontransgenic ones.

Transgenic plants originally grown on agar medium were transferred to normal soil and grew to adult plants during three months period. Grown seeds were harvested and then tested for the ability to germinate on medium with kanamycine and PCB to select those plants, which are highly resistant to PCB (Macek et al., 2005). Finally it can be concluded that engineered organisms used in this study were previously proven to be stable and to survive in contaminated media. At present, experiments are underway to evaluate the performance of GM plants on media with PCBs added, and to follow the formed products.

### 3.5. CONCLUSIONS

Not all waste types and site conditions are comparable. Each site must be individually investigated and tested. Engineering and scientific judgment must be used to determine if a technology is appropriate for a site, as EPA repeatedly states (Anonymous 1998). Engineered degradation by transgenic plants proved to be a possible remediation technology that leads to more favorable contaminant fate and reduced toxicological concerns.

The most important parameter for selection of suitable plants is not the tolerance of the plant to the contaminants, but the effectiveness in their accumulation or degradation. In addition to accumulation capacity, biomass production must be considered in order to determine the total metal uptake. We tried to address these problems by preparing GM plants with improved properties. Our results allow for moderate optimism, the GM tobaccos with *bphC* express the bacterial PCB cleavage gene, and the HisCUP plants exhibit better Cd accumulation, higher resistance and better transport into harvestable parts. Instead of tobacco as a useful laboratory model, other species can be considered to be tailored according to requirements of specific contaminated sites (Griga et al., 2003), especially technological crops are interesting candidates for this task.

Only the decreasing of the amount of environmental contamination can prevent the toxic compounds entering the food chain, help to conserve the biological diversity (which is surely more endangered by the contaminated environment than by the use of genetically modified plants) (Macek et al., 2002a), and this way contribute to the sustainable development. Other advantages of phytoremediation systems are still germane to transgenic modifications, including: the cost effectiveness for large areas and the natural aesthetic appeal (Burken et al., 2005). Our experiments illustrate two viable approaches to prepare plants with properties suitable for exploitation in biological remediation of soils contaminated by heavy metals or polychlorinated biphenyls.

### Acknowledgement

The work on phytoremediation was sponsored by the grants MSM 6046137305 and FRVS 1938/G4, research projects Z 40550506 and ICT 320080015

## References

- Adriano, D.C., 2001, *Trace elements in terrestrial environments*. Springer-Verlag New York Inc., pp. 867.
- Alowitz, M.J.; Scherer, M.M. (2002) Kinetics of nitrate, nitrite, and Cr(VI) reduction by iron metal. *Environ. Sci. Technol.*, 36, 299–306.
- Anderson, B.S., Hunt, J.W., Phillips, B.M., Stoelting, M., Becker, J., Fairey, R., Puckett, H.M., Stephenson, M., Tjeerdema, R.S. and Martin, M., 2000, Ecotoxicologic Change at a Remediated Superfund Site in San Francisco Bay, California, USA, *Environ. Toxicol. Chem.* **19**(4): 879–887.
- Anonymous (1998) A citizens guide to phytoremediation. In: EPA Technology Fact Sheet, US EPA 542-F-98–011. US Environmental Protection Agency, Technology Innovation Office, pp 1–6.
- Antrim, L.D. and Kohn, N.P., 2000, Post-Remediation Biomonitoring of Pesticides in Marine Waters Near the United Heckathorn Superfund Site, Richmond, California, Prepared for the U.S. Environmental Protection Agency by Battelle Marine Sciences Laboratory, Sequim, Washington; published by Pacific Northwest Laboratory, Richland, Washington.
- Battelle, Stanford and NAVFAC, 2004, Hunters Point Shipyard Parcel F Treatability Study Work Plan.
- Bizily, S., Rugh, C., Summers, A.O. and Meagher, R., 1999, Phytoremediation of methylmercury pollution: merB expression in *Arabidopsis thaliana* confers resistance to organomercurials, *Proc. Natl. Acad. Sci. USA*, 96: 6808–6813.
- Borovka, R., Szekeres, M., Macek, T., Kotrba, P., Sylvestre, M. and Macková, M., 1998, First steps in attempt to enhance the ability of plants to metabolise polychlorinated biphenyls by introduction of bacterial genes to plant DNA. *Int. Biodeter. Biodegrad.* 42: 243.
- Bruggers, J., February 3, 1997, Utah Takes Waste That Arizona Rejected, http://HighCountryOrg.com.
- Burken, J.G., Ma, X., Struckhoff, G.C. and Gilbertson, A.W., 2005, Volatile organic compound fate in phytoremediation applications: Natural and engineers systems. Z. *Naturforsch.* 60: 208–215.
- Chaney, R.L., Angle, M.S., McIntosh, M.S., Reeves, R.D., Li, Y.-M., Brewer, E.P., Chen, K.-Y., Roseberg, R.J., Prner, H., Synkowski, E.C., Broadhurst, C.L., Wang, S. and Baker, A.J.M., 2005, Using hyperaccumulator plants to phytoextract soil Ni and Cd. Z. *Naturforsch.* 60, 190–199.
- Chaney, R.L., Malik, M., Li, Y.M., Brown, S.L., Angle, J.S. and Baker, A.J.M., 1997, Phytoremediation of soil metals. *Curr. Opin. Biotechnol.* 8: 279–284.
- Chasaigne, H., Vacchina, V., Kutchan, T., Zenk, M.H., 2001, Identification of phytochelatin-related peptides in maize seedlings exposed to cadmium and obtained enzymatically in vitro. *Phytochemistry* 56: 657–668.
- Cho, Y., Smithenry, D.W. and Luthy, R.G., 2005, Preliminary Field Testing of Activated Carbon Mixing and In-Situ Stabilization of PCBs in Sediment, ACS National Meeting Extended Abstracts, American Chemical Society, Division of Environmental Chemistry.
- Chroma, L., Moeder, M., Kucerova, P., Macek, T. and Mackova, M., 2003, Plant enzymes in metabolism of polychlorinated biphenyls, *Fres. Environ. Bull.* 12: 291–295.
- Clemens, S., Palmgren, M. and Krämer, U. (2002), A long way ahead: understanding and engineering plant metal accumulation, *Trends Plant Sci.* 7: 309–315.

- Czako, M., Feng, Y., He, D., Lianh, D. and Marton, L., 2005, Genetic modification of wetland grasses for phytoremediation. Z. Naturforsch. 60: 285–291.
- Demnerova, K., Mackova, M., Pazlarova, J., Vosahlikova, M., Novakova, H., Jindrova, E., Ryslava, E., Macek, T., Vrchotova, N., Brenner, V., Pavlu, L., Totevova, S., Kristoffer, T., Focht, D.D., Fava, F., Di Gioia, D., Marchetti, L., Fletcher, J.S., Leigh, M.B., Kucerova, P., Stiborova, H., Mateju, V., Sobotka, M., Kastánek, F., Kastánek, P. and Kasak, L., 2002, PCB - approaches to the removal from the environment, In: *Inovative Approaches to the On-Site Assessment and Remediation of Contaminated Soils*, D. Reible and K. Demnerova, (eds.), Kluwer Academic Publishers, Dordrecht, pp. 217– 263.
- Dorlhac de Borne, F., Elmayan, T., de Roton, Ch., de Hys, L. and Tepfer, M., 1998, Cadmium partitioning in transgenic tobacco plants expressing a mammalian metallothionein gene. *Mol. Breeding* **4:** 83–90.
- Francova, K., Mackova, M., Macek, T. and Sylvestre, M., 2004, Ability of bacterial biphenyl dioxygenases from Burkholderia sp. LB400 and Comamonas testosteroni B-356 to catalyse oxygenation of ortho-hydroxychlorobiphenyls formed from PCBs by plants. *Environ. Pollut.* **127:** 41–48.
- Francova, K., Sura, M., Macek, T., Szekeres, M., Bancos, S., Demnerova, K., Sylvestre, M. and Mackova, M., 2003. Generation of plants carrying bacterial enzymes for degradation of polychlorinated biphenyls, *Fres. Environ. Bull.* **12**: 309–313.
- Gelvin, S.B., 1998, The introduction and expression of transgenes in plants. *Curr. Opinion Biotechnol.* **9:** 227–235.
- Ghosh, U., Zimmerman J.R. and Luthy, R.G., 2003, PCB and PAH Speciation among Particle Types in Contaminated Harbor Sediments and Effects on PAH Bioavailability, *Environ. Sci. Technol.* 37(10): 2209–2217.
- Ghosh, U., Talley J.W. and Luthy, R.G., 2001, Particle-Scale Investigation of PAH Desorption Kinetics and Thermodynamics from Sediment, *Environ. Sci. Technol.* 35(17): 3468–3475.
- Griga, M., Bjelkova, M. and Tejklova, E., 2003, Phytoextraction of heavy metals by fibre crops: *Linum usitatissimum* L. case study. In: *Proceedings of the 2nd European Bioremediation Conference*, N. Kalogerakis and E. Psillakis (eds.) Crete, TU Crete, pp 353–356.
- Grill, E., Loffler, S., Winnacker, E.-L. and Zenk, M.H., 1989, Phytochelatins, the heavymetal-binding peptides of plants, are synthesised from glutathione by a specific γglutamylcysteine dipeptidyl transpeptidase (phytochelatin synthase). *Proc. Natl. Acad. Sci* USA **84:** 6838–6846.
- Heaton, A.C.P., Rugh, C.L., Wang, N. and Meagher, R.B., 1998, Phytoremediation of mercury and methylmercury polluted soil using genetically engineered plants. J. Soil Contamin. 7: 497–509.
- Hein, P., Powlowski, J., Barriault, D., Hurtubise, Y., Ahmad, D., Sylvestre, M., 1998, Biphenyl-associated *meta*-cleavage dioxygenases from *Comamonas testosteroni* B-356. *Can. J. Microbiol.* 44: 42–49.
- Kärenlampi, S., Schat, H., Vangronsveld, J., Verkleij, J.A.C., van der Lelie, D., Mergeay, M. and Tervahauta, A.I., 2000, Genetic engineering in the improvement of plants for phytoremediation of metal polluted soils. *Environ. Pollut.* **107**: 225–231.
- Kiekens, L. and Camerlyck, R., 1982, Transfer characteristics for uptake of heavy metals by plants. *Landwirtschaftliche Forschung* **SH39**: 255–261.
- King, L.D. and Hajjar, L.M., 1990, The residual effect of sewage sludge on heavy-metal content of tobacco and peanut. *J. Environl Qual.* **19:** 738–748.

- Kloke, A., Sauerbeck, D.R. and Vetter, H., 1984, The contamination of soils and plants with heavy metals and the transport of heavy metals with terrestrial food chains. In: *Changing metal cycles and human health.* J.O. Nriagu ed., Springer-Verlag, Berlin, pp. 113–141.
- Kohn, N.P. and Evans, N.R., 2004, Year 6 Post-Remediation Biomonitoring and Phase II Source Investigation at the United Heckathorn Superfund Site, Richmond, California, Sequim, Washington, Prepared for the U.S. Environmental Protection Agency by Battelle Marine Sciences Laboratory, Sequim, Washington; published by Pacific Northwest Laboratory, Richland, Washington.
- Koprowski, H. and Yusibov, V., 2001, The green revolution: plants as heterologous expression vectors. *Vaccine* **19**: 2735–2740.
- Kotrba, P., Macek, T. and Ruml, T., 1999, Heavy-metal binding peptides and proteins in plants. A review. Coll. Czech Chem. Commun. 64: 1057–1086.
- Krämer, U. and Chardonnens, A., 2001, The use of transgenic plants in the bioremediation of soils contaminated with trace elements. *Appl. Microbiol. Biotechnol.* **55:** 661–672.
- Krämer, U., 2005, Phytoremediation: novel approaches to cleaning up polluted soils. *Curr. Opinion Biotechnol.* **16:** 133–141.
- Kucerova, P., Mackova, M., Chroma, L., Burkhard, J., Triska, J., Demnerová, K. and Macek, T., 2000, Metabolism of polychlorinated biphenyls by *Solanum nigrum* hairy root clone SNC-9O and analysis of transformation products. *Plant Soil* 225: 109–115.
- Liu Y., Majetich, S.A., Tilton, R.D., Sholl, D.S., Lowry, G.V. (2005) TCE Dechlorination Rates, Pathways, and Efficiency of Nanoscale Iron Particles with Different Properties, *Environ. Sci. Technol.* 39 (5) 1338–1345.
- Liu, J.R., Suh, M.Ch. and Choi, D., 2000, Phytoremediation of cadmium contamination: Overexpression of metallothionein in transgenic tobacco plants. *Bundesgesundheitsbl-Gesundheitsforsch-Gesundheitsschutz* 2,126–130.
- Lowry, G.V., Johnson, K.M. (2004) Congener Specific PCB Dechlorination by Micro scale and Nanoscale Zero-valent Iron in a Methanol/water solution. *Environ. Sci. Technol.*, 38(19) 5208–5216.
- Macek, T. and Macková, M., 1999, Phytoremediation the use of plants to remove xenobiotics and pollutants from the environment, including transgenic plants tailored for this purpose. *Biologia* 54: 70–73.
- Macek, T., Francova, K., Kochankova, L., Lovecka, P., Ryslava, E., Rezek, J., Sura, M., Triska, J., Demnerova, K. and Mackova, M., 2004b, Phytoremediation: Biological cleaning of a polluted environment. *Rev. Environ. Health* **19**: 63–82.
- Macek, T., Kotrba, P., Ruml, T., Skacel, F. and Mackova, M., 1997a, Accumulation of cadmium by hairy root cultures. In: Doran P.M. (ed.), *Hairy Roots: Culture and Application*, Harwood Academic Publishers, London, pp 133–138.
- Macek, T., Kotrba, P., Suchova, M., Skacel, F., Demnerová, K. and Ruml, T., 1994, Accumulation of cadmium by hairy root cultures. *Biotechnol. Lett.* **16**: 621–624.
- Macek, T., Macková, M., Burkhard, J. and Demnerová, K., 1998, Introduction of green plants for the control of metals and organics in remediation. In: Holm FW (ed.) *Effluents from Alternative Demilitarization Technologies*. NATO PS Ser. 1, Vol. 12, Kluwer Acad. Publishers, Dordrecht, pp 71–85.
- Macek, T., Mackova, M., Kotrba, P., Truksa, M., Singh-Cundy, A., Scouten, W.H. and Yancey, N., 1997b, Attempts to prepare transgenic tobacco with higher capacity to accumulate heavy metals containing yeast metallothionein combined with a polyhistidine, In: *Environmental Biotechnology*, Verachtert H. and Verstraete W. eds., Technological Institute Gent, pp 263–266.

- Macek, T., Mackova, M., Kucerova, P., Chroma, L., Burkhard, J. and Demnerova, K. (2002a) Phytoremediation, in M. Hofman and J. Anne (series eds.), *Focus on Biotechnology*, Vol. 3, Kluwer Academic Publishers, Dordrecht, pp 115–137.
- Macek, T., Macková, M., Pavlíková, D., Száková, J., Truksa, M., Cundy, A.S., Kotrba, P., Yancey, N., Scouten, W. H., 2002b, Accumulation of cadmium by transgenic tobacco. *Acta Biotechnol.* 22: 101–106.
- Macek, T., Macková, M., Truksa, M., Singh-Cundy, A., Kotrba, P., Yancey, N. and Scouten W.H., 1996, Preparation of transgenic tobacco with a yeast metallothionein combined with a polyhistidine tail. *Chem. Listy* **90**: 690.
- Macek, T., Pavliková, D. and Macková, M., 2004a, Phytoremediation of Metals and Inorganic Pollutants. In: *Soil Biology*, Applied Bioremediation and Phytoremediation, Vol. 1, A. Singh and O.P. Ward, eds., Springer Verlag Berlin, Heidelberg, pp. 135–157.
- Macek, T., Surá, M., Pavlíková, D., Francová, K., Scouten, W.H., Szekeres, M., Sylvestre, M. and Macková, M., 2005, Can Tobacco Have Potentially Beneficial Effect to our Health? Z. Naturforschung., Vol. 60c, 292–298.
- Mackova, M., Chroma, L., Kucerova, P., Burkhard, J., Demnerova, K. and Macek, T., 2001, Some aspects of PCBs metabolism by horseradish cells. *Int. J. Phytorem.* 3, 401–414.
- Macková, M., Macek, T., Burkhard, J., Ocenasková, J., Demnerová, K. and Pazlarová, J., 1997a, Biodegradation of polychlorinated biphenyls by plant cells. *Int. Biodeter. Biodegrad.* 39: 317–325.
- Mackova, M., Macek, T., Kucerova, P., Burkhard, J., Pazlarova, J., Demnerova, K., 1997b. Degradation of polychlorinated biphenyls by hairy root culture of *Solanum nigrum*, *Biotechnol. Lett.* **19**: 787–790.
- Maliga, P., 1990, Transformation of N. tabacum by Agrobacterium and particle gun, Molecular and Developmental Biology of Plants, Cold Spring Harbor Course.
- McGarvey, B., Hammond, J., Dienet, M., Hooper, D., Fu, Z., Dietzschold, B., Koprowski, H. and Michaels, F., 1995, Expression of the rabies virus glycoprotein in transgenic tomatoes, *Bio/Technology* 13: 1484–1489.
- McLeod, P.B., Van Den Heuvel-Greve, M.J., Allen-King, R.M., Luoma S.N. and Luthy, R.G., 2004, Effects of particulate carbonaceous matter on the bioavailability of benzo[a]pyrene and 2,2 ',5,5 '-tetrachlorobiphenyl to the clam, Macoma balthica, *Environ. Sci. Technol.* 38(17): 4549–4556.
- Meagher, R.B., 2000, Phytoremediation of toxic elemental and organic pollutants, *Curr. Opinion Plant Biol.* **3:** 153–162.
- Millward, R.N., Bridges, T.S., Ghosh, U., Zimmerman J.R. and Luthy, R.G., 2005, Addition of activated carbon to sediments to reduce PCB bioaccumulation by a polychaete (Neanthes arenaceodentata) and an amphipod (Leptocheirus plumulosus), *Environ. Sci. Technol.* 39(8): 2880–2887.
- Moore, A.M., De Leon, C.H., Young, T.M. (2003). Rate and Extent of Aqueous Perchlorate Removal by Iron Surfaces. *Environ. Sci. Technol.* 37(14); 3189–3198.
- Murphy, P.J., Marquette, A., Reible, D.D., Lowry, G.V. (2005). "Predicting the Performance of Activated Carbon-, Coke-, and Soil-amended Thin Layer Sediment Caps." *J. Environ. Eng.* (submitted).
- Murphy, P.J., Olsta, J., Zhu, Y., Reible, D., Lowry, G.V. (2005), Development and Placement of an "Active" Sorbent-amended Thin Layer Sediment Cap in the Anacostia River. *Journal Soil Sed. Contam., an International Journal* (submitted).
- Nesnerová, P., Sebek, P., Macek, T. and Svatos, A., 2004. First semi-synthetic preparation of sex pheromones. *Green Chemistry* **6**, 305–307.
- Old, R.W. and Primrose, S.B., 1994, Gene transfer to plants. *Principles of Gene Manipulation*, Blackwell Scientific Publications.

- Ondrej, M., Drobnik, J., Gartland, K.M.A. and Gartland, J.S., 1999, *Gene engineering*, (in Czech, textbook of TEMPUS PHARE programme), ICT Prague.
- Oven, M., Raith, K., Neubert, R.H.H., Kutchan, T.M. and Zenk, M.H., 2001, Homophytochelatins are synthesized in response to cadmium in Azuki beans. *Plant Physiol*. 126: 1275–1280.
- Palermo, M., Maynord, S., Miller, J. and Reible, D. (1998). "Guidance for In-Situ Subaqueous Capping of Contaminated Sediments," EPA 905-B96–004, Great Lakes National Program Office, Chicago, IL.
- Pavlíková, D., Macek, T., Macková, M., Surá, M., Száková, J., Tlustoš, P., 2004a, The evaluation of cadmium, zinc, and nickel accumulation ability of transgenic tobacco bearing different transgenes. *Plant Soil Environ.* 50, 513–517.
- Pavlíková, D., Macek, T., Macková, M., Száková, J., Balík, J., 2004b, Cadmium tolerance and accumulation in transgenic tobacco plants with yeast metallothionein combined with a polyhistidine tail. *Int. Biodeter. Biodegrad.*, 233–237.
- Peňa, L. and Séguin, A., 2001, Recent advances in the genetic transformation of trees. *Trends Biotechnol.* 19: 500–506.
- Pieper, D.H., 2005, Aerobic degradation of polychlorinated biphenyls. Appl. Microbiol. Biotechnol. 67: 170–191.
- Reible, D.D., Hayes, D., Lue-Hing, C., Patterson, J., Bhowmik, N., Johnson, M. and Teal J. (2003) "Comparison of the long-term risks of removal and In-Situ management of contaminated sediments in the Fox River,", *Journal Soil Sed. Contam.*, 12(3) 325–344.
- Ryslava, E., Krejčík, Z., Macek, T., Novakova, H. and Mackova, M., 2003, Study of PCB biodegradation in real contaminated soil. *Fres. Environ. Bull.* **12** (3): 296–301.
- Soudek, P., Tykva, R., Kalisova, I. and Vanek T., 2003, Phytoremediation of heavy metals by sunflower and corn plants. In: *Proceedings of the 2nd European Bioremediation Conference*, Kalogerakis N and Psillakis E eds., Crete, TU Crete, pp 353–356.
- Starr, R.C. and Cherry, J.A. (1994) In-Situ Remediation of Contaminated Ground Water: The Funnel-and-Gate System, *Ground Water*, 32, 3, 465–476.
- Surá, M., Macková, M., Szekeres, M., Sylvestre, M., Chrastilová, Z., Macek, A., Transgenic T. plants – tool for preferable phytoremediation of PCB and other organic pollutants. In: *Proceedings of the 2nd European Bioremediation Conference*. Kalogerakis N., ed., P71 Chania, Crete, 2005 CD-ROM, Tech. University of Crete
- Swartz, R.C., Cole, F.A., Lamberson, J.O., Ferraro, S.P., Schults, D.W., DeBen, W.A., Lee II, H. and Ozretich, R.J., 1994, Sediment Toxicity, Contamination and Amphipod Abundance at a DDT- and Dieldrin-Contaminated Site in San Francisco Bay, *Environ. Toxicol. Chem.* **13**(6): 949–962.
- Talbert, B., Thibodeaux, L.J., Valsaraj, K.T. (2001). "Effectiveness of Very Thin Soil Layers in Chemical Release from Bed Sediment." *Environ. Prog.* 20(2), 103–107.
- Talley, J.W., Ghosh, U., Tucker, S.G., Furey, J.S., Luthy, R.G. (2002). "Particle-scale understanding of the bioavailability of PAHs in sediment. *Environ. Sci. Technol.*, 36, 477–483.
- Thornann, R.V. and Connolly, J.P., 1984. Model of PCB in the Lake Michigan lake trout food chain. *Environ. Sci. Technol.* 18(2):65–71.
- Tlustos, P., Pavlikova, D., Balik, J., Szakova, J., Hanc, A. and Balikova, M., 1998, The accumulation of arsenic and cadmium in plants and their distribution. *Rostlinna vyroba* 44: 463–469.
- USEPA, 1996, EPA Superfund Record of Decision: United Heckathorn Co., Richmond, California. EPA ID# CAD981436363, U.S. Environmental Protection Agency, Region IX, San Francisco, California.

- Vandecasteele, B., Meers, E., Vervaeke, P., De Vos, B., Quataert, P., Tak, F.M.G., 2005, Growth and trace metal accumulation of two *Salix* clones on sediment-derived soils with increasing contamination levels. *Chemosphere* 58, 995–1002.
- Werner, D., Higgins C.P. and Luthy, R.G., 2005, The sequestration of PCBs in Lake Hartwell sediment with activated carbon, *Water Res* **39**(10): 2105–2113.
- Weston, D.P., Jarman, W.M., Cabana, G., Bacon, C.E. and Jacobson, L.A., 2002, An Evaluation of the Success of Dredging as Remediation at a DDT-Contaminated Site in San Francisco Bay, California, USA, *Environ. Toxicol. Chem.* 21(10): 2216–2224.
- Zimmerman, J.R., Werner, D., Ghosh, U., Millward, R.N., Bridges, T.S. and Luthy, R.G., 2005, Effects of dose and particle size on activated carbon treatment to sequester polychlorinated biphenyls and polycyclic aromatic hydrocarbons in marine sediments, *Environ. Toxicol. Chem.* 24(7): 1594–1601.
- Zimmerman, J.R., Ghosh, U., Millward, R.N., Bridges, T.S. and Luthy, R.G., 2004, Addition of carbon sorbents to reduce PCB and PAH bioavailability in marine sediments: Physicochemical tests, *Environ. Sci. Technol.* 38(20): 5458–5464.

# **BIOLOGICAL ASSESSMENT AND REMEDIATION**

# OF CONTAMINATED SEDIMENTS

SAÏD EL FANTROUSSI & SPIROS N. AGATHOS\*

Unit of Bioengineering, Faculty of Bioengineering, Agronomy & Environment, Catholic University of Louvain, Place Croix du Sud 2/19,B-1348 Louvain-la-Neuve, Belgium

DIETMAR H. PIEPER\*, ROBERT WITZIG, BEATRIZ CÁMARA, LOTTE GABRIEL-JÜRGENS & HOWARD JUNCA

*GBF German Research Centre for Biotechnology, Mascheroder Weg 1 38124 Braunschweig, Germany* 

GIULIO ZANAROLI & FABIO FAVA\* DICASM, Faculty of Engineering, Alma Mater Studiorum-University of Bologna, viale Risorgimento 2, 40126 Bologna, Italy

JOSÉ R. PÉREZ-JIMÉNEZ & LILY Y. YOUNG Department of Environmental Sciences and Biotechnology Center for Agriculture and the Environment, Cook College, Rutgers, The State University of New Jersey, New Brunswick, New Jersey 08901, USA

KELLY HAMONTS, RICHARD LOOKMAN, MIRANDA MAESEN, LUDO DIELS, AND WINNIE DEJONGHE<sup>\*</sup> Flemish Institute for Technological Research, Geel, Belgium

JOHN DIJK AND DIRK SPRINGAEL Catholic University of Leuven, Heverlee, Belgium

\* to whom correspondence should be addressed

**Abstract-** Various approaches to clean contaminated aquatic environments have been proposed. In recent years, natural attenuation has received increasing attention and it is generally accepted that microorganisms are the principal mediators of the natural attenuation of many pollutants. However, the complexity of environmental systems such as sediments requires a multifaceted approach to understand microbial processes and their potential. This is even more so under *in situ* conditions, where the activity of pollutant degrading microorganisms is generally slow, partial and

constrained spatially and/or temporally. Recent developments in molecular biology and genomics are offering tools to explore microbial processes at a level that encompasses the genetic characteristics of the local microbial players, culturable or not, as well as their organization into complex communities and their interactions both with each other and with the target chemicals. It is now possible to study microbes directly in their environments at the population level as well as at the single cell level and to link biology to geochemistry. Integrative knowledge from culture independent studies based on functional characters and assessment of the diversity and quantity of catabolic genes in response to pollution, will allow a deeper understanding of and a rational intervention in environmental processes. Moreover, the use of genomic libraries to retrieve genes from natural bacterial communities without cultivation will allow a breakthrough in accessing new microbial capabilities. In this chapter, the main features, advantages and limitations of these innovative approaches to the biomonitoring and analysis of intrinsic bioremediation potential of polluted environments and sediments are critically reviewed. Then, the potential of the same strategies in the integrated chemical, physical and biological monitoring and characterization of polluted sediments subjected to natural decontamination is shown through the description of the main results of case studies performed on a) polychlorinated biphenyl (PCB)-contaminated marine sediments of the Porto Marghera area of Venice Lagoon (Italy) in which the occurrence of PCB-reductive dechlorination processes has been demonstrated for the first time in the literature, b) sediments contaminated by chlorinated aliphatic hydrocarbons (CAHs) collected from different positions of the eutrophic river Zenne (Vilvoorde, Belgium), where they have been found to act as a natural biobarrier for the CAHs occurring in the groundwater that is passing through the sediment zone, hereby reducing the risk of surface water contamination, and c) other environmental contaminated systems subjected to ex-situ and in situ active bioremediation, where these processes are described on the basis of the experience accumulated in pilot and real-life systems.

**Keywords:** polychlorinated biphenyls (PCBs), chlorinated aliphatic hydrocarbons, bioaugmentation, biostimulation, reductive dechlorination, biosensors, PCR, DGGE, T-RFLP, metagenomics, sulfate-reducing bacteria, fingerprinting methods, *exsitu* treatment, *in situ* treatment, engineered bioremediation systems.

Microbes critically impact and mediate nearly every major biogeochemical cycle on Earth. Microbial interactions that occur on the scale of microns can regulate global processes on planet-wide scales. As an example, the decomposition and transformation of both organic and inorganic compounds in sediments is almost completely carried out by microbes. The crucial role of microbes in the environment had been underestimated since the first discovery of these microscopic living forms. The main reason for this was the difficulty to access the astonishing diversity of microbes since the classical approaches based on their cultivation in the laboratory are biased and inadequate. There is a wide consensus among microbiologists that more than 90% of microbes are refractory to the classical approaches of cultivation<sup>1</sup>. Recent advances in understanding the diversity and importance of microbes, using culture-independent approaches, has led without any ambiguity to the conclusions that microbes are by far the most dominant living forms in terms of number of species, number of individual cells, total biomass, diversity of habitats, and diversity of cellular chemistries. Thus the environmental impact of microbes is being taken seriously and stands in the forefront of studies aiming at understanding the global environmental changes. Microbes dominate animal life in sediments and aquifers, and they are responsible for major fluxes of organic and inorganic nutrients. They are involved in processes as different as anaerobic methane oxidation<sup>2</sup>, uranium accumulation<sup>3</sup>, pesticide degradation<sup>4</sup>, and polychlorinated biphenyl degradation<sup>5</sup>.

# 1.1. HOW TO MASTER INCREASING COMPLEXITY

The complexity of microbial life in both natural and man-impacted habitats requires a multidisciplinary approach. In addition to their large diversity, microbes live in interaction with the different components of their environments. Figure 1.1 shows schematically the inherent complexity of a polluted sediment undergoing bioremediation. Several factors affect the process of biodegradation (bioavailability, physico-chemical conditions, composition of microbial communities, etc.). The characterization of such a complex biotope is now feasible using tools offered by such disciplines as ecology, biogeochemistry, physiology, biochemistry, genetics, process engineering, molecular biology, and metagenomics. However the relevance



*Figure 1.1.* Schematic representation of the complexity of polluted sediments undergoing bioremediation. The anaerobic dechlorination of tetrachloroethene (PCE) is used as an example of productive microbial metabolism leading to site cleanup.

and efficiency of each discipline depends on the scale of study (pure culture, laboratory bioreactors, or field studies). Figure 1.2 shows the relevance of each discipline to the field studies (adapted from E.L.Madsen<sup>6</sup>). The most striking discipline that has recently emerged is metagenomics, defined as the culture-independent genomic analysis of entire microbial communities. Metagenomics provides access to the pool of genomes of a given environment using a comprehensive survey of nucleotide sequence, structure, regulation, and function. Furthermore, direct genomic cloning offers the possibility to retrieve unknown sequences or functions, whereas methods relying on PCR amplification require prior knowledge of the sequence of genes for the design of primers<sup>7,8</sup>.



*Figure 1.2.* Disciplines involved in understanding microbial processes in sediments and their relevance to different scales of studies.

### 1.2. EXPERIMENTAL APPROACHES

Studying microbes in sediments involves several approaches that can be summarised by the following flowchart.

# Phase 1: Environmental sampling: physiological studies (adapted from E. L. Madsen<sup>9</sup>)

- 1. Sediment, soil, water or sludge in field site.
- 2. Aseptically remove, contain, transport to laboratory.
- 3. Divide into replicate live and abiotic treatments.

4. If appropriate, add radiolabelled or unlabelled organic compound of interest.

5. Use analytical chemistry or physiology tools to periodically measure consumption or production.

6. Compare time courses of live and abiotic treatments.

7. Interpret and consider genetic studies in Phases 2 and 3.

# Phase 2: Specialized microbiological studies using culture-based methods

1. Isolate pure culture expressing activity found in Phase 1.

2. Characterize growth, cell yield, sequential induction, and other characteristics during pollutant metabolism.

3. Extract and identify metabolites, enzymes and cofactors linked to pollutant metabolism.

4. Study metabolites, enzymes and cofactors in cell-free system.

5. Determine portion of genomic or plasmid DNA coding for pollutant metabolism by screening a cloned DNA library, by transposon mutagenesis or by other procedure.

6. Do hybridization, restriction mapping and sequencing DNA analyses seeking ORFs, phylogenetic relationships to similar genes, etc.

7. Elucidate details of gene expression and regulation via various genetic & molecular techniques (transposon mutagenesis, expression clones, insertional inactivation, inducer/reporter experiments).

# Phase 3: Specialised microbiological studies using cultureindependent approaches

- 1. Using a few grams of sediments, extract the whole genomic material (DNA and/or RNA).
- 2. Clean up the nucleic acids.
- 3. Proceed through PCR or in situ hybridization to study species or consortia of interest.
- 4. Use fingerprinting methods (DGGE, TRFLP, etc.) and clone libraries to study microbial diversity.
- 5. For genomic and metagenomic studies shear the nucleic acid fragments and clone them in vectors such as fosmids and BACs.
- 6. Use the function-driven approach or the sequence-driven approach to explore the phylogenetic and physiological diversity of the sediment biotope.

# 1.3. MOLECULAR TOOLS TO DETECT AND MONITOR MICROBES

The development of molecular tools has allowed more precise and sensitive techniques to be applied for microbial detection in sediments compared to classical microbiological methods. The most popular and widely used molecular techniques are those based on polymerase chain reaction (PCR). PCR techniques targeting specific portions of genes have been used to detect microbes and to explore microbial diversity in sediments, aquifers and soil samples. Several studies have shown the advantage of using PCR to detect and quantify the presence of microbes. As an example El Fantroussi et al<sup>10</sup> have used 16S rRNA-based techniques to study the biodegradation of chloroaromatic compounds in a simulated aquifer. The utilisation of PCR in parallel with biodegradation in microcosms showed an unambiguous role of bioaugmentation as a strategy for the bioremediation

of polluted sediments under constraints of fluid flow patterns, depth of matrix and presence of competing indigenous microbial communities.

Hendrickson et al.<sup>11</sup> examined the environmental distribution of organisms belonging to the *Dehalococcoides* group, known to be major players in the anaerobic transformation of chlorinated compounds, and their association with chloroethene-contaminated sites. Samples from 24 chloroethene-dechlorinating sites scattered throughout North America and Europe were tested for the presence of members of the *Dehalococcoides* group by using a PCR assay developed to detect *Dehalococcoides* 16S rRNA gene (rDNA) sequences. Sequences identified by homology to the *Dehalococcoides* group were detected at 21 sites. Full dechlorination of chloroethenes to ethene occurred at these sites. *Dehalococcoides* sequences were not detected in samples from three sites at which partial dechlorination of chloroethene. Phylogenetic analysis of the 16S rDNA amplicons confirmed that *Dehalococcoides* sequences formed a unique 16S rDNA group<sup>11</sup>.

Other studies have demonstrated the power of combining molecular techniques with classical approaches to understand microbial processes in sediments. De Lipthay et al.<sup>12</sup> evaluated how the in situ exposure of a subsurface aquifer to phenoxy acid herbicides at low concentrations (<40 ug 1<sup>-1</sup>) changes the local microbial community composition. Sediment and groundwater samples collected from a herbicide-exposed area and unexposed controls were analyzed for the presence of general microbial populations, *Pseudomonas* bacteria, and specific phenoxy acid degraders. Both culture-dependent and culture-independent methods were applied. The abundance of microbial phenoxy acid degraders  $(10^{\circ} \text{ to } 10^{4} \text{ per gram of})$ sediment) was determined by most probable number (MPN) assays, and their presence was only detected in herbicide-exposed sediments. Similarly, PCR analysis showed that the 2,4-dichlorophenoxyacetic acid degradation pathway genes tfdA and tfdB were only detected in sediments from contaminated areas of the aquifer. Different populations of *tfd* genes have been found, suggesting that the in situ herbicide degradation was caused by the activity of a heterogeneous population of phenoxy acid degraders. The number of *Pseudomonas* bacteria was higher in sediments subjected to high levels of phenoxy acid. Furthermore, high numbers of CFU compared to direct counting of 4',6-diamidino-2-phenylindole-stained cells under the microscope suggested an increased culturability of the indigenous microbial communities from acclimated sediments. The findings of this study demonstrate that continuous exposure to low herbicide concentrations can markedly change the bacterial community composition of a subsurface aquifer.

Anaerobic methane oxidation is a process that effectively controls emission of methane from many anaerobic environments into the atmosphere and thus plays an important role in the global methane budget<sup>2</sup>. Microbially mediated oxidation of methane in anoxic marine systems is a globally significant process, with up to 90% of the oceanic methane production recycled in anaerobic marine sediments<sup>13</sup>. Other studies have shown that microorganisms living in anoxic marine sediments consume more than 80% of the methane produced in the world's oceans. In addition to single-species aggregates, consortia of metabolically interdependent bacteria and archaea were found in methane-rich sediments. A combination of fluorescence in situ hybridization and secondary ion mass spectrometry showed that cells belonging to one specific archaeal group associated with the Methanosarcinales were all highly depleted in C-13. This depletion indicates assimilation of isotopically light methane into specific archaeal cells<sup>2</sup>. Anaerobic methane oxidation was also investigated in marine sediment from Aarhus Bay, Denmark<sup>14</sup>. Measured concentration profiles for methane and sulfate, as well as in situ rates determined with isotope tracers. indicated that there was a narrow zone of anaerobic methane oxidation about 150 cm below the sediment surface. Methane could account for 52% of the electron donor requirement for the peak sulfate reduction rate detected in the sulfate-methane transition zone. Molecular signatures of organisms present in the transition zone were detected by using selective PCR primers for sulfate-reducing bacteria and for Archaea. One primer pair amplified the dissimilatory sulfite reductase (DSR) gene of sulfate-reducing bacteria, whereas another primer (ANME) was designed to amplify archaeal sequences found in another study of sediments from the Eel River Basin, as these bacteria have been suggested to be anaerobic methane oxidizers<sup>15</sup>. This study is a nice illustration of combining chemistry to molecular biology to bring into stark evidence globally significant biogeochemical processes that are mediated by diverse coexisting and interacting microbial populations<sup>14</sup>.

### 1.4. SYNTROPHY

Anaerobic oxidation of methane, also called reverse methanogenesis, in sediments illustrates once again the importance of syntrophy in the microbial world in general and in anaerobic processes in particular. As in the process of methane production, methane oxidation occurs in ecological niches where archaea, responsible for methane oxidation; live in association with sulfate reducers. Syntrophy was first discovered in 1967 as a synergistic relationship between an ethanol-oxidizing bacterium and a methanogen. In the late 1970s, the importance of syntrophic relationships became established through culture-based studies of methanogens interacting with bacteria that oxidized lactate, butyrate, or propionate. Biochemically, these bacteria dispose of reducing equivalents by reducing protons or bicarbonate to H<sub>2</sub> and formate, respectively. Both products serve as electron carriers in interspecies H<sub>2</sub> transfer and interspecies formate transfer between syntrophic acetate-producing bacteria and methanogens. Since their original discovery, many other syntrophic systems have been identified, and the central role of syntrophic interactions in environmental samples and engineered anaerobic bioreactors is now widely recognized. Syntrophy (or syntrophism) can be defined as the interaction of two or more populations that satisfy each other's nutritional needs<sup>16</sup>. This strategy of "feeding together" is involved in the anaerobic breakdown of various sugars, amino acids, and aromatic compounds. El Fantroussi et al.<sup>10</sup> used Desulfomonile tiediei for bioaugmentation of a simulated aquifer polluted with 3-chlorobenzoate. This dechlorinating bacterium was isolated from a syntrophic association with a benzoate oxidizer and a H2-consuming methanogen (Figure 1.3).



Figure 1.3. Example of a bacterial syntrophic association.

*D. tiedjei* is a scavenger which relies on electron donors and vitamins from other organisms and uses a broad range of electron acceptors including 3-chlorobenzote<sup>17</sup>. The establishment of this strain in the simulated aquifer was evaluated with physiological microcosms and PCR

detection of its 16S rRNA signatures. There was a close relationship between 3-chlorobenzoate dechlorination, methane production and PCR detection<sup>10</sup>.

### 1.5. ENVIRONMENTAL MONITORING – BIOSENSORS

To fully understand microbial processes in sediments there is a need for new, fast, specific, accurate, precise and cost effective methods for field assays to monitor both chemical species and 'effect related' processes (e.g. xenoestrogens). Several techniques have been developed for environmental monitoring of chemical species. These methods include ultra-sensitive techniques based on immunoassays as well as whole-cell (mainly by using genetically engineered microorganisms "GEM") biosensors to assess toxicity / bioavailability that can be applied to both metals and organics. However there are still some technical barriers related to sample treatment and cleanup prior to instrumental analysis. These procedures are also timeconsuming and often not suited for *in situ* field monitoring, e.g. sediments. To achieve this goal Acha et al.<sup>18</sup> have developed a novel attenuated total reflection-Fourier transform infrared (ATR-FTIR) sensor and applied it to the continuous on-line monitoring of a dechlorination process. This optical sensor was developed to measure noninvasively part-per-million (ppm) concentrations of trichloroethylene (TCE), tetrachloroethylene (PCE), and carbon tetrachloride (CT) in the aqueous effluent of a fixed-bed dechlorinating bioreactor, without any prior sample preparation (Figure 1.4).



*Figure 1.4.* ATR-FTIR sensor (shown inside the circle) coupled to an anaerobic dechlorinating bioreactor for continuous on-line monitoring of chlorinated organic compounds.

The sensor was based on an ATR internal reflection element (IRE) coated with an extracting hydrophobic polymer, which prevented water molecules from interacting with the infrared (IR) radiation. The selective diffusion of chlorinated compound molecules from aqueous solution into the polymer made possible their detection by the IR beam. With the exclusion of water the detection limits were lowered, and measurements down to 2, 3, and 2.5 mg/L (ppm) for TCE, PCE, and CT, respectively, became possible. Before coupling the ATR-FTIR sensor to the dechlorinating bioreactor, preliminary spectra of the chlorinated compounds were acquired on a laboratory scale configuration in stop-flow and flowthrough closed-loop modes, in order to study the direct response of the sensor to any arbitrary concentration change of the analytes. Subsequently, the bioreactor (a fixed bed which could also simulate a sediment matrix) was monitored with the infrared sensor coupled permanently to it. The sensor tracked the progression of the analytes' spectra over time without perturbing the dechlorinating process. The accuracy of this ATR-FTIR sensor was validated against gas chromatography (GC) measurements of the chlorocompounds<sup>17</sup>.

### 1.6. CONCLUSION

Microbial processes play crucial roles in the transformation of organic and inorganic compounds in sediments. The recent developments in molecular biology allowed a better understanding of the distribution, diversity, and functions of microbes in sediments. Novel key processes have been attributed to microbes thanks to molecular biology such us anaerobic methane oxidation. Coupling these molecular techniques with geochemistry involving sophisticated tools such as chemical and biochemical sensors opens up interesting horizons for comprehensive studies of microbial processes in sediments.

### 1.7. ACKNOWLEDGEMENTS

The work of the authors S. E. Fantroussi and Spiros N. Agathos was supported by grants CHLOREM (BIO 4-CT 1998-0303) and MADOX (QLK3-CT-2001-00345) of the European Commission.

# 2. Accumulating basic knowledge on biodegradation processes and use of this knowledge for analyzing catabolic potential and metabolic networks *in situ* by culture-independent approaches

Various approaches to clean contaminated environments have been proposed including chemical, physical or biological treatments. Among them, the biological treatment is considered an efficient and cost saving way to achieve remediation of contaminated sites<sup>19-21</sup>. During the previous years, natural attenuation, "naturally occurring processes in soil and groundwater that act without human intervention to reduce the mass, toxicity, mobility, volume or concentration of contaminants in those media"<sup>22</sup> has received increasing attention. It is generally accepted that microorganisms are the principal mediators of the natural attenuation of many pollutants. They transform or mineralize pollutants, thereby usually decreasing their masses and toxicities, in contrast to most other processes contributing to natural attenuation. However, in some cases pollutants may be transformed into more toxic products, as reported for the anaerobic transformation of trichloroethylene. The use of natural attenuation thus requires a detailed monitoring to determine how effective natural attenuation is for attaining site remediation goals. In order to properly evaluate natural attenuation at a site, it is necessary to know the location and concentration of the contaminants, how the contaminants move in the environment, and how their concentration changes over time. Reliance on biologically meditated natural attenuation thus requires information on whether it can occur, whether it is actually occurring at a significant rate, which mechanisms and pathways are involved and, specifically, how it will behave in the future. Active in situ remediation involves the stimulation of the indigenous microbial activity by the addition of nutrients and/or electron acceptors. As in the case for natural attenuation, an understanding of the response of the indigenous microbial community is necessary for successful stimulation. Even more unpredictable are bioaugmenation approaches, as the added microorganisms, which are supposed to increase the degradative performance on site, not only have to express these capabilities under in situ conditions, but also have to compete with the complex microflora inhabiting the site under study. It is thus evident that only a detailed understanding of the functioning and interactions in microbial communities will allow their rational manipulation, and the overcoming of factors limiting efficient bioremediation.

### 2.1. BACTERIAL DEGRADATION OF AROMATIC POLLUTANTS

Aromatic compounds have been discharged into the environment during the last years in increasing amounts. They have significant impact on natural microbial communities, and thus, on the global element mass fluxes. The functional diversity in nature shows that many microorganisms have the potential to degrade and recycle aromatic compounds<sup>23</sup>. This potential can be used for the bioremediation processes mentioned above.

Many microorganisms have evolved biodegradative pathways to use aromatic compounds as a sole carbon and energy source<sup>24-26</sup>. Even though the first observations were made using aerobic microorganisms<sup>27,28</sup>, it is well established that aromatics can be degraded under nitrate-, iron-, or sulfate-reducing and even under methanogenic conditions<sup>29,30</sup>.

#### 2.1.1. Rieske non-heme iron oxygenases

Aerobic microorganisms usually initiate degradation by activation of the aromatic nucleus through oxygenation reactions. The introduction of hydroxyl-groups, usually in *ortho*-position to one another, results in a few central intermediates such as catechols, protocatechuate, gentisate and hydroxyhydroquinones. These intermediates are subject to oxygenolytic ring cleavage followed by channeling of the ring-cleavage products into the central metabolism<sup>31</sup>.

The so called Rieske non-heme iron oxygenases are one of the key families of enzymes important for aerobic degradation of aromatics. These enzymes usually catalyze the incorporation of two oxygen atoms into the aromatic ring to form arene-*cis*-dihydrodiols<sup>32</sup>, a reaction which is followed by a dehydrogenation catalyzed by *cis*-dihydrodiol dioxygenases to give catechols or substituted catechols. The oxygenases, such as benzoate, naphthalene, biphenyl, or toluene dioxygenases are multicomponent enzyme complexes, composed of a terminal oxygenase component (iron-sulfur protein) and different electron transport proteins (a ferredoxin and a reductase or a combined ferredoxin-NADH-reductase). The catalytic iron-sulfur proteins are heteromultimers, comprising a large ( $\alpha$ ) and a small ( $\beta$ ) subunit with the former containing a Rieske-type [2Fe-2S] cluster, a mononuclear non-heme iron and a substrate binding site (Fig. 2.1).



*Figure 2.1.* Biochemical and genetic organization of toluene 2,3-dioxygenases as an example of Rieske non heme iron oxygenases. Two electrons are transferred from NADH via the electron transfer chain, consisting of reductase and ferredoxin (Fd) to the terminal oxygenase component, comprising the small  $\alpha$ -subunit and the large  $\beta$ -subunit. The incorporation of two oxygen atoms into the substrate results in the formation of a *cis*-dihydrodiol. Regions of the gene encoding residues of the  $\alpha$ -subunit responsible for substrate specificity are indicated in white.

Comparison of the amino acid sequences of the terminal oxygenase  $\alpha$ subunits revealed that they form a family of diverse but evolutionarily related sequences (Figure 2.2). Although none of the enzymes is completely specific, a broad correlation between the grouping in toluene/biphenyl, naphthalene, benzoate or phthalate families and the native substrates oxidized by the family members can be observed.



*Figure 2.2.* Dendrogram showing the relatedness of members of different families of Rieske non heme iron oxygenases ( $\alpha$ -subunits).

Enzyme engineering studies on biphenyl, benzene, chlorobenzene, and naphthalene dioxygenases showed that the  $\alpha$ -subunit of the terminal oxygenase determines substrate specificity and that only slight sequence differences in the amino acid sequence can be associated with dramatic changes in substrate specificity or regiospecificity<sup>33-36</sup>. As an example a acid difference in toluene dioxygenase results in single amino of tetrachlorobenzene<sup>33</sup> or transformation in a completely new regioselectivity of naphthalene dioxygenase<sup>36</sup>. Regioselectivity also determines whether or not a given substrate can subsequently be subject to mineralization. As an example, dibenzofuran can be mineralized after a so called angular attack at the oxygen bridge<sup>37</sup> and an adjacent carbon atom, whereas lateral attack results in the formation of dead-end products<sup>38</sup>. Consequently, dioxygenases crucially determine the range of substrates that are accessible to microbial degradation and the metabolic net in microbial communities. Crystal structures are available for some members of Rieske non-heme iron oxygenases<sup>39,40</sup>. Those allow the modeling of active site structures of newly discovered derivative enzymes and at least a partial explanation of enzyme performance, even though amino acid residues, which have no contacts with substrates also strongly changed enzyme nerformance<sup>41,42</sup>. Nevertheless. Rieske-type non-heme iron oxygenases are an enzyme family in which significant advances have been made to relate sequence information to function.

# 2.1.2. Toluene monooxygenases

Two groups of enzymes have been reported to attack the non-activated benzene nucleus by monooxygenation, the multicomponent aromatic monooxygenases and some members of the multicomponent phenol hydroxylases<sup>43</sup>, both groups belonging to the family of soluble diiron monooxygenases. These monooxygenases can perform regio- and stereospecific hydroxylations (Fig. 2.3), and in the case of transformation of toluene, 2-methylphenol<sup>44</sup>, 3-methylphenol<sup>45</sup> as well as 4-methylphenol<sup>46</sup>, were reported to be formed, however, recent results showed that 4-methylphenol but not 3-methylphenol is formed by the enzyme termed toluene 3-monooxygenase<sup>47</sup>. Further oxygenation of the intermediate cresols by phenol hydroxylase results in the formation of 3-methyl- or 4-methylcatechol, respectively.

Alternatively, 4-methylphenol can be subject to oxygenation of the methylsubstituent resulting finally in protocatechuate (Figure 2.3) as ringcleavage substrate<sup>48</sup>. Evidently, again, substrate specificity will significantly determine the metabolic net responsible for degradation.



*Figure 2.3.* Transformation of toluene by two successive monooxygenations results in the formation of methylcatechols which are usually mineralized via extradiol cleavage.

Moreover, dependent on the regioselectivity, xylenes will be transformed to dead-end products. The aromatic monooxygenases consist of an electron transport system comprising a reductase and a ferredoxin, a catalytic effector protein which is assumed to play a role in assembly of an active oxygenase and a terminal hydroxylase with a  $(\alpha\beta\gamma)_2$  quaternary structure and a diiron center contained in each  $\alpha$ -subunit<sup>43</sup>. Even though the effector protein is necessary not only for effective coupling, but also for a high regioselectivity<sup>49</sup>, the  $\alpha$ -subunit was majorly responsible for substrate specificity and regioselectivity<sup>43</sup>. Various mutagenesis and directed evolution studies have actually identified residues which control regioselectivity and activity<sup>50,51</sup>. The structure of toluene monooxygenase from *P. stutzeri* OX1 has recently been elucidated<sup>52</sup> which offers the opportunity to understand how the enzyme adjusts the active site pocket to control product regioselectivity.

# 2.1.3. Catechol dioxygenases

Whereas a relatively broad diversity of activation mechanisms is possible in aromatic degradation, these pathways usually converge in the formation of (substituted) catechols as central intermediates<sup>53</sup>. Whereas chlorinated aromatics are predominantly degraded via intradiol cleavage<sup>54</sup> methylsubstituted aromatics such as toluene or xylenes are usually degraded via the respective catechols and extradiol cleavage (Fig. 2.3). Extradiol dioxygenases are thus key enzymes in the degradation of aromatic compounds and many of such proteins and their coding sequences have been described, purified and characterized in the last decades. Examination
of their evolutionary relationships<sup>55</sup> showed that the majority of extradiol dioxygenases with preferences for monocyclic substrates were, at the protein level, phylogenetically closely related. However, members of this enzyme subfamily were reported to differ significantly in their substrate specificity<sup>42,56</sup> and, thus, the metabolic network in environmental communities will be significantly interconnected with the C23O diversity. Catechol transforming activities are specifically important, as catechols are highly toxic for microorganisms<sup>57</sup> and even low accumulation will have severe impact in community functioning<sup>58</sup>.

# 2.1.4. Anaerobic degradation of aromatics

In recent years our knowledge on anaerobic degradation of aromatics has been significantly enhanced and genetic determinants for various key steps in anaerobic aromatic degradation have been elucidated. Under anaerobic conditions, aromatic hydrocarbons are initially attacked by novel reactions and in most cases oxidized further to benzoyl-CoA, a common intermediate in anaerobic catabolic pathways<sup>29</sup>. Toluene degradation is initiated by an unusual addition reaction of the toluene methyl group to the double bond of fumarate to form benzylsuccinate<sup>59</sup> catalyzed by benzylsuccinate synthase, a unique type of glycyl-radical enzyme<sup>60</sup>. However, not only toluene degradation is initiated by fumarate addition. Similar reactions have been shown to be involved in the anaerobic activation of *m*-xylene, *m*- and *p*and 2-methylnaphthalene<sup>61,62</sup>. Even though cresol. *n*-alkanes the metabolism of toluene has mainly been elucidated in denitrifying Thauera and Azoarcus it became evident that benzylsuccinate forming activities are found across a wide range of phylogenetically and physiologically diverse bacteria, and have also been found in the iron-reducing Geobacter *metallireducens* strains<sup>63</sup>. In contrast to the information on degradation of toluene and xylenes, information on anaerobic degradation of naphthalene and benzene is scarce<sup>64,65</sup>.

# 2.1.5. Reductive dehalogenation

It is known since more than one decade ago that chloroaromatics can function as an alternative electron acceptor in a type of anaerobic respiration<sup>66</sup>. Several anaerobic bacteria have been identified as being able to reductively dehalogenate chlorinated aliphatics as well as chlorinated aromatics and to couple this reaction to the synthesis of ATP via a chemiosmotic mechanism. Bacterial dehalorespiration is currently considered to be the most important process for the detoxification of organohalogens under anaerobic conditions. The dehalorespiring bacteria known to belong to the low GC content Gram-positive bacteria (*Desulfitobacterium* and *Dehalobacter*), the Proteobacteria (*Desulfomonile*,

Desulfuromonas and Dehalospririllum, now transferred to the genus the genus *Dehalococcoides*<sup>67</sup>. Although Sulfurospirillum). and dehalorespiring bacteria are widespread, Dehalococcoides species in particular seem to be of major environmental importance. It has been shown that members of this group can grow on vinyl chloride as electron acceptor<sup>68</sup>, a process previously assumed to be co-metabolic (i.e. without deriving energy or carbon from the transformed substrate). In addition, Dehalococcoides strains are, thus far, the only microorganisms known to be capable of growing anaerobically on chlorobenzenes<sup>69</sup>, polychlorinated biphenyls and even chlorinated dioxins<sup>70</sup>. Reductive dehalogenases involved in tetrachloroethene or chlorophenol respiration have been intensively studied (Figure 2.4). This novel subclass of reductases shares some common features at the biochemical and genetic level. Except for the chlorobenzoate reductive dehalogenase of Desulfomonile tiedjei that contains a heme cofactor and that is produced as a heterodimer<sup>71</sup>, reductive dehalogenases contain two iron-sulfur clusters and a corrinoid<sup>67</sup>.



Figure 2.4. Reductive dehalogenation of TCE (trichloroethene) to cis-1,2-dichloroethene.

They are produced as preproteins with a twin-arginine signal peptide recognized by the export system specific for the export of periplasmic proteins containing cofactors<sup>72</sup>. The protein sequence also revealed consensus sequences characteristic for the binding of two iron–sulfur clusters. Also vinyl chloride reductive dehalogenase was recently identified as a novel member of the family of corrinoid/iron-sulfur cluster containing reductive dehalogenases<sup>73</sup>.

# 2.2. MOLECULAR TOOLS FOR ASSESSING BIODEGRADATION POTENTIAL

It has been shown that the *in situ* catabolic potential of microbial communities depends on the history of the site under study, the site characteristics (geochemistry and redox conditions) and, in cases where the contaminated soil is covered by vegetation, interactions between the

microbial community and plant species, which are up to now poorly defined. Moreover, it is well known that many environmental microorganisms may not be cultivated under laboratory conditions. Culturing-independent molecular techniques are, currently, rapidly increasing our understanding of microbial community structure and activity in the subsurface. Polymerase Chain Reaction (PCR) amplification of nucleic acids extracted from environmental samples is at present the most powerful cultivation-independent technique. PCR facilitates the sensitive and fast detection of low amounts of specific gene fragments. This is of importance for monitoring purposes, as subsurface environments are, in general, oligotrophic, hence characterized by low biomass from which low amounts of nucleic acids can be extracted.

Molecular microbial ecology has significantly increased our knowledge on the diversity and dynamics of microbial communities in nature. By different PCR approaches, predominantly using the 16S rRNA gene or the inter-spacer region of 16-23S rRNA genes, specific taxonomic groups responsible for degradation, and specific species, can be characterized, quantified and their spread followed over time<sup>74</sup>. However, as only in a subset of scenarios, a degradation capability is directly related to a specific taxonomic group, research in recent years has focused on the adaptation of molecular ecology methods for assessing community composition to characterize the makeup of catabolic genes, as the abundance of degradation genes can be assumed to indicate the biodegradation potential in contaminated soils.

# 2.2.1. Analysis of catabolic genes

Catalytic enzymes involved in aromatic degradation are encoded by genes, which have evolved to produce proteins performing specific actions. Members of the different gene families share a certain degree of similarity represented by conserved sequence regions. With only two short sequences conserved along two fairly divergent gene family members, it is possible to detect, by amplification, regions between the conserved parts, in DNA from isolates or from environmental samples. The initial step in this process is to identify the crucial gene and protein regions. Subsequently, conserved regions are identified within the different groups of given gene families and evaluated for primer design.

Thus far, analysis of catabolic genes has focused mainly on the characterization of the presence or abundance of a family of catabolic genes<sup>73,75,76</sup> without assessing the finer levels of sequence diversity. However, the diversity of catabolic gene sequences as described above, often reflects differences in substrate specificity or affinity. Single amino acid differences have been reported to drastically change substrate

specificity of catabolic enzymes (see above). A more detailed picture of the catabolic gene structure and sequence diversity in environmental samples will, thus, significantly increase our knowledge of the functional potential of microbial communities. Moreover, shifts in catabolic gene structure will allow the deduction of the evolutionary fitness of catabolic genes, operons and their respective hosts. A variety of molecular fingerprinting approaches previously developed to assess community structure, via the analysis of 16S rDNA or rRNA diversity, may be applied to define functional gene structure and we succeeded recently<sup>77</sup> to adapt fingerprinting methods to elucidate the diversity of various key families of catabolic genes. Together with promising methods to characterize functional gene expression in environmental samples, and an improved knowledge on structure/function relationships in catabolic genes high throughput fingerprinting methods would help future trends to identify catabolic gene diversity and structure. as well as active operons and pathways under changing environmental conditions.

# 2.2.2. Fingerprinting of catabolic genes

A promising fingerprinting technique that is able to resolve fragments of identical length but different sequence composition is SSCP (single strand chain polymorphism). This technique (Figure 2.5) takes advantage of the property of the sequence-dependent conformation acquired by single-strand DNA molecules separated in gels under non-denaturing conditions. The application was originally conceived to compare amplifications from single templates to rapidly identify single nucleotide polymorphisms in the spanned sequence fragment<sup>78</sup>. For each single amplification (of a dsDNA PCR product) two single strands are produced generating two different conformations. The SSCP method was optimized to discriminate sequence diversity of environmental 16S rRNA genes by amplification with one phosphorylated primer. allowing the degradation 5'end of the phosphorylated strands of each PCR amplicon by lambda exonuclease digestion. This simplifies and improves the resolution of the method for complex amplicon mixtures<sup>79</sup>. The SSCP-approach has been used in various studies identifying changes in predominant members and bacterial taxonomical composition<sup>80</sup>.

Because of their central role in aromatic catabolism, the catechol 2,3dioxygenases (C23O) subfamily I.2.A genes have been analysed in studies of diverse environments and their presence has been observed in various contaminated sites<sup>75,81</sup>.



*Figure 2.5.* Schematic representation of the PCR-SSCP fingerprinting technique, which can be used to analyze complex amplicons from environmental DNA.

Genes encoding catechol 2,3-dioxygenases (C23O) were used as functional targets to assess the catabolic gene diversity in differentially (benzene, toluene. ethvlbenzene. xvlene) BTEX contaminated environments by polymerase chain reaction-single-strand conformation polymorphism (PCR-SSCP)<sup>77</sup>. Site specific PCR-SSCP fingerprints were obtained, showing that gene diversity experienced shifts correlated to temporal changes and levels of contamination. Overall, the PCR-SSCP technique was shown to be a powerful tool for assessing the diversity of functional genes and the identification of predominant gene polymorphs in environmental samples as a prerequisite to understand the functioning of microbial communities. PCR-SSCP profiling of a defined catabolic gene family with limited diversity lower the possibility of different single strands having identical mobility, in contrast to the use of PCR-SSCP for detection of highly diverse 16S rRNA gene sequences where bands were found to contain a multitude of different sequence types<sup>80</sup>.

As described above, Rieske non-heme iron oxygenases are a key family defining aerobic degradation networks. The same contaminated site and various isolates were used to develop a SSCP fingerprinting method covering regions defining substrate specificity and gene variants with new active site structure could be observed to be predominant in contaminated sites and obviously selected for by benzene as contaminant (Witzig, unpublished).

#### 2.3. NEW METABOLIC ROUTES OBSERVED IN ISOLATES

The above described methods rely on nucleic acid probes and PCR primers designed on the base of information retrieved from isolates and can thus only cover a subset of the activities assumed to be present in environmental samples and will not cover new genes or gene products. However, even isolate-based approaches still recover a broad set of new metabolic diversity previously undescribed.

One example is a new pathway for aerobic aromatic metabolism initially observed in *Azoarcus evansii*<sup>82,83</sup>. It could be shown that in various bacteria benzoate is first converted to its coenzyme A thioester, benzoyl-CoA, which is subsequently attacked by an oxygenase, followed by a non-oxygenolytic ring-fission. Respective genes were also observed during sequencing of the genome of the biphenyl degrading organism *Burkholderia xenovorans* LB400 and recently it could be shown that benzoate, when produced during biphenyl metabolism by this strain, is actually degraded via the benzoyl-CoA pathway<sup>84</sup>.

Also new routes for the degradation of chlorocatechols as central intermediates in aerobic chloroaromatic degradation could be elucidated, which ressemble variations of or patchworks between archaetype catechol degradation pathways and chlorocatechol pathways. In *Rhodococcus opacus* 1CP, the dehalogenation reaction during 3-chlorocatechol degradation is not catalyzed during cycloisomerization, as described for chlorocatechol pathways, but by a specialized muconolactone isomerase similar to enzymes commonly involved in catechol degradation<sup>85</sup>. 4-Chlorocatechol degradation by *Pseudomonas* sp. MT1 is initiated by enzymes resembling those of the archaetype catechol degradation pathway, however, dehalogenation is catalyzed by a new type of hydrolase acting on an unstable pathway intermediate<sup>86</sup>. Recent sequencing of the gene revealed that the protein does not show significant sequence similarity to any gene of described function available in public databases (Cámara, unpublished).

#### 2.4. METAGENOMICS

One approach that relies neither on conserved nucleotide sequences nor on previous knowledge on isolates is the use of genomic libraries to retrieve genes from natural bacterial communities without cultivation. The field of metagenomics is currently rapidly developing. Genomic DNA is extracted and inserted into vectors, such as plasmids, cosmids or bacterial artificial chromosomes, which can maintain inserts of around 100 kb. These can then be propagated in appropriate bacterial strains, usually *E. coli* and be screened for expressed catabolic activities. Interestingly, the activity-based screening of metagenome libraries usually resulted in the discovery of gene

products, which were only distantly related to previously described ones<sup>87,88</sup>. However, there are two significant drawbacks of the current procedures: (a) the necessity to screen an immense set of clones, only a few of which encode the desired activity, especially if the screening system is time consuming, and (b) the poor expression in E. coli of many genes present in metagenome libraries. In an effort to quantify the accessibility of the metagenome, it was shown that only 40% of the genes present in the genome of 32 prokaryotes could be easily detected in E.  $coli^{89}$ . However, new Streptomyces and Pseudomonas strains that are optimized to express environmental gene libraries have recently been constructed<sup>90</sup>. One approach to overcome time-consuming screening assays is a procedure termed substrate induced gene expression screening<sup>91</sup> which is based on the knowledge that catabolic gene expression is generally induced by substrates and metabolites of catabolic enzymes and, in many cases, controlled by regulatory elements situated proximate to catabolic genes. A metagenomic library was constructed in a green fluorescent protein (GFP)-expression vector, and clones expressing GFP in the presence of a target substrate were selected by cell sorting. By this procedure, novel catabolic genes could successfully be isolated from a metagenome library.

Considering the well-known limitation to describe microbial diversity by culture-dependent approaches, the analysis of environmental metagenomes will probably reveal that actually only a fraction of the metabolic capacity of microorganisms is thus far documented. This also holds for the metabolic capacity to degrade aromatic pollutants. Specifically the mechanisms of anaerobic biodegradation need to be characterized and exploited to their finer details, as it has been done in the studies for aerobic degradation pathways, where more detailed biochemical and genetic information has been obtained thus far.

# 2.5. ACKNOWLEDGEMENTS

The work of the authors D.H. Pieper, R. Witzig, B. Camara, L. Gabriel-Jürgens & H. Junca was supported by grants ACCESS (EVK1-CT-1999-00023), AMICO (QLK3-2000-00731) and BIOTOOL (GOCE 003998) of the European Community and by the DFG-European Graduate College 653.

# 3. Detection and characterization of microbial processes associated to PCB degradation in marine contaminated sediments

PCBs are poorly biodegradable and highly toxic contaminants. Due to their high hydrophobicity, PCBs released into aquatic systems tend to strongly accumulate in anoxic freshwater, estuarine and marine subsurface sediments<sup>92</sup>, where half-lives of several months up to many years have been

estimated. Several studies have documented the occurrence of reductive dechlorination processes towards PCBs in anaerobic freshwater sediments. where highly chlorinated *meta* and *para*-substituted PCBs can be generally bioconverted, mainly under methanogenic conditions, into low-chlorinated ortho-substituted congeners<sup>93,94</sup>. On the contrary, little is known about the occurrence of reductive dehalogenation processes towards weathered PCBs in marine sediments<sup>95,96</sup>, where in general sulfidogenic conditions prevail over methanogenesis. To our knowledge, the reductive dechlorination of pre-existing PCBs was documented only once in sediments of New Bedford Harbor under methanogenic conditions<sup>95,97</sup>. Some other studies have documented the reductive dechlorination of PCBs in sediment slurries developed with salt rich media where, however, spiked PCBs and/or synthetic media were employed<sup>98,99,100</sup>. Thus, more information on the potential fate of aged PCBs in marine contaminated sediments is required. In particular, we need studies performed on real contaminated sediments suspended in their own real water under laboratory conditions that closely mimic those occurring in  $situ^{93,94}$ . This might allow us to collect information of some relevance for predicting, when combined to lines of biogeochemical evidence<sup>101</sup>, the potential of biological processes in the final *in situ* restoration of contaminated sediments. In this paper, the main results are reviewed of two studies in which the occurrence of microbially mediated reductive dechlorination processes towards aged and spiked PCBs has been detected and characterized in distinct sediments of the Porto Marghera area (Venice lagoon, Italy) suspended in water from the site. Preliminary data on the dechlorination activity and the microbial composition of a consortium enriched from the most active microcosms are also presented.

3.1. FIRST DETECTION AND CHARACTERIZATION OF REDUCTIVE DECHLORINATION PROCESSES IN THE SEDIMENTS OF THE VENICE LAGOON

# 3.1.1. Experimental approach

A series of slurry-phase anaerobic microcosms consisting of a PCB contaminated sediment of the industrialized area Porto Marghera (Brentella Canal, Venice lagoon, Italy) suspended at 25% (v/v) in the water collected from the same contaminated area were developed. Replicate microcosms were set up also under different conditions in order to select for various indigenous microbial populations and to indirectly determine their potential involvement in PCB biotransformation. Pasteurized microcosms, as well as molybdate-amended and 2-bromoethanesulfonate (BES)-amended microcosms, were established in order to select for spore-forming bacteria, to inhibit sulfate-reducing bacteria and to inhibit methanogenic bacteria,

respectively. In addition, sterile microcosms were set up under each condition. Finally, a parallel set of identical microcosms was also spiked with 2.3.4.5.6-pentachlorobiphenvl (20 mg/kg drv wt sediment) in order to study the possibility of "priming" the dechlorination of the sediment aged PCBs and the mechanism through which they were dechlorinated under each of the experimental conditions. Microcosms were set up in 30 ml serum bottles starting from a preliminary sediment slurry (25% v/v) that was prepared under strictly anaerobic conditions as described by Fava et al.<sup>102</sup>. BES and molybdate were added as stock solutions in sterile lagoon water to the final concentration of 30.0 mM and 20.0 mM, respectively. Six and a half  $\mu$ l of an acetone stock solution (70 mM) of 2,3,4,5,6pentachlorobiphenvl were added to each spiked microcosm under sterile conditions to vield a final concentration of about 20 mg (kg dry wt sediment)<sup>-1</sup>, whereas the same volume of fresh acetone was added to the non-spiked microcosms. Pasteurization and sterilization were performed before adding the exogenous PCB. Pasteurized microcosms were obtained with a 15 minutes treatment at 90°C in a water bath, whereas sterile microcosms underwent autoclaving treatment at 121°C for 1 h for 3 consecutive days with incubation at 28°C between each autoclaving treatment<sup>102</sup>. Under each culture condition, a set of triplicate biologically active and sterilized microcosms was prepared. All the developed microcosms were then incubated stationary at  $25 \pm 1^{\circ}$ C in the dark for 20 weeks. During this period they were periodically sampled and analyzed to determine the volume and the composition of the headspace gas, as well as the concentration of PCBs and inorganic anions (i.e.,  $SO_4^{-}$  and Br) according to Fava et al.<sup>102</sup>.

At the end of the experiment the Terminal-Restriction Fragment Length Polymorphism (T-RFLP) technique was used to investigate the complexity of the microbial population in the most active microcosms as compared to the original sediment. The DNA was extracted from 0.6 grams of original sediment (conserved at 4°C in the dark under strict anoxic conditions) and from the pellet of 1.5 ml of slurry according to Scala and Kerkhof<sup>103</sup>. DNA was purified by CsCl gradient and 16S rDNA was amplified using universal bacterial [6]-carboxyfluorescein-labelled 27F (5'-AGAGTTTGATCM-TGGCTCAG-3') and 1525R (5'-AAGGAGGTGWTCCAR-3') primers by Polymerase Chain Reaction (PCR). Five µl of the PCR products were then digested at 37°C for 2 hours with Mnll, Rsal and Hhal in three independent 20 µl reactions, precipitated and resuspended in 14.75 µl of deionized formamide and 0.25 µl of TAMRA 500 (Perkin-Elmer, Boston, MA, USA). Fluorescently labelled fragments were then separated and detected with an ABI Prism 310 capillary sequencer (PE Applied Biosystem, Foster City, CA, USA) using the GeneScan mode. T-RFLP profiles were compared by Sorensen's similarity analysis.

#### 3.1.2. Results and discussion

A total PCB concentration of  $0.784 \pm 0.351 \,\mu$ mol (kg dry wt sediment)<sup>-1</sup> was detected in the biologically-active and sterile microcosms at the 7<sup>th</sup> day of incubation. Significant changes of the initial PCB distribution profile were unequivocally observed in the untreated microcosms at the 20<sup>th</sup> week, where an extensive depletion of highly chlorinated biphenyls together with a stoichiometric accumulation of tri- and di-chlorinated, *ortho*-substituted biphenyls were observed (Fig. 3.1A).

A less extensive but significant transformation of endogenous PCBs was also observed in the pasteurized microcosms, where the accumulation of 2-chlorobiphenyl was also observed (Fig. 3.1B). On the contrary, the sediment PCBs were only poorly transformed in the microcosms supplemented with BES or molybdate (Figs. 3.1C and 3.1D, respectively). Comparable changes in the endogenous PCBs distribution profiles were observed in the 2,3,4,5,6-pentachlorobiphenyl-spiked microcosms<sup>102</sup>.



*Figure 3.1.* Bioconversion of sediment-carried PCBs after 20 weeks of incubation. White bar: sterile microcosms; Black bar: biologically active microcosms. A: untreated microcosms; B: pasteurized microcosms; C: BES amended microcosms; D: Molybdate amended microcosms. (1):2-CB; (2):4-CB; (3):2,6-/2,2'-CB; (4):2,4-/2,5-CB; (5):2,4'-/2,3-CB; (6):2,4,6-CB; (7):2,2',5-/2,2',4-/4,4'-CB; (8):2,3,6-/2,3',6-CB; (9):2,3,3'-/2',3,4-/2,2',5,6'-CB; (10):2,2',4,6'-/2,3,4'-CB; (11):2,2',5,5'-CB; (12):3,3',4-CB; (13):2,2',3,5-CB; (14):3,4,4'-/2,3,3',6-/2,2',3,4'-CB; (15):2,2',3,4',2,3,4',6-CB; (16):2,3',4',5-CB; (17):2,3',4,4'-/2,2',3,5',6-CB; (18):2,2',3,4',5-/2,2',4,5'-CB; (19):2',3,4,4',5-/2,2',3,4',5'-CB; (20):2,2',-3,4',5,5'-CB; (21):2,2',3,3',4,6'-/2,2',4,4',5,5'-/2,3,3',4,4'-CB; (22):2,3,3',4,5,6-/2,2',3,4,4',5/-2,3,3',4,4',6-CB; (23):2,2',3,3',4,5,6-/2,3,3',4,4',5-/2,2',3,3',4,5',6,6'-CB; (24):-2,2',3,4,4',5,5'-CB. CB; chlorobiphenvl.

A detectable production of gas was observed in all the biologically active microcosms during the 20 weeks of incubation (Table 3.1). In the molybdate-amended microcosms a significant methane production and no consumption of the initial  $2.1\pm0.1$  g l<sup>-1</sup> of SO<sub>4</sub><sup>-</sup> were observed (Table 3.1). On the contrary, a consumption of about 30-50 % of the initial SO<sub>4</sub><sup>-</sup> was observed in the untreated, pasteurized and BES-amended microcosms, where no methane production was detected. In addition, in the BES-amended microcosms a release of about 6 mg l<sup>-1</sup> of Br<sup>-</sup> was measured (Table 3.1). The same changes were observed in the PCB-spiked set of microcosms, except for sulfate consumption in the untreated microcosms, where it was slightly faster and more extensive than in the corresponding non-spiked microcosms (data not shown)<sup>102</sup>.

Microcosms	Sulfate	Produced Gas	Produced CH <sub>4</sub>	Released
	depletion (%)	(ml)	(ml)	Br (mg/l)
Sterile microcosms	$-19.3 \pm 9.3$	0	0	0
S + W**	$29.9 \pm 7.8*$	$1.25\pm0.21$	0	0
S + W, Pasteurized	$41.2 \pm 14.2*$	$0.65\pm0.21$	0	0
S + W + Molybdate	$-12.4 \pm 2.7*$	$2.40 \pm 1.00$	$0.17\pm0.01$	0
S + W + BES	$47.3 \pm 35.0*$	$1.55\pm0.64$	0	$6.01\pm0.88$

Table 3.1. Microbial activities detected in the microcosms after 20 weeks of incubation.

\*: values corrected for the percentage sulfate consumption in the sterile microcosms.

\*\*: S + W = sediment + water

Taken together, these results indicate that indigenous sulfate-reducing bacteria were responsible for the detected PCB-reductive dechlorination. A low dechlorination activity was detected in the microcosms supplemented with BES, where a marked consumption of  $SO_4^{=}$  was observed. A significant release of Br was also observed in these microcosms; this suggests that BES acted as the preferential electron acceptor for the dehalogenating bacteria, thus inhibiting PCB dechlorination. The detection of both reductive dechlorination activity and sulfidogenic activity in the pasteurized microcosms indicates that spore-forming, sulfate-reducing bacteria were involved in the process. A similar hypothesis has already been proposed in the literature<sup>93</sup> but it is the first time that it is formulated for the reductive dechlorination of PCBs pre-existing in a marine sediment resuspended in the site water. The exogenous 2,3,4,5,6-pentachlorobiphenyl was markedly bioconverted into its tetra-, tri- and di- ortho-chlorinated daughter products in the spiked untreated microcosms, showing that the process had a higher selectivity towards meta- and para- positions. A less extensive but significant 2,3,4,5,6-pentachlorobiphenyl transformation also occurred in the pasteurized microcosms, where 2-chlorobiphenyl was accumulated under these culture conditions, as also observed in the corresponding non-spiked microcosms<sup>102</sup>. Finally, the dechlorination of the spiked PCB did not significantly affect the onset of the pre-existing PCB dechlorination, probably because of the low concentration ( $\approx 1 \text{ mg/kg}$ ) of the weathered PCBs in the sediment and/or the inhibition by high concentrations of SO<sub>4</sub><sup>=</sup> and salt occurring in these microcosms.

At the end of the 20 weeks incubation, the spiked microcosms that exhibited the higher PCB dechlorination activity, i.e. the untreated and the pasteurized ones, were chosen for the analysis of the indigenous microbial population considering that the exogenous PCB may have favored the growth of PCB-dechlorinating microorganisms. For comparison, the analysis was also performed on the original sediment. Since different microbial taxons can be distinguished by the sequence of the 16S rRNA genes, we analyzed the polymorphisms of these genes using the T-RFLP technique. Three different restriction enzymes were chosen (Mnll, Rsal and Hhal) in order to better resolve the complexity of the microbial populations. Very different T-RFLP profiles were generated by the microcosms and the sediment (Fig. 3.2 for Mnll digestion), indicating that large modifications in the microbial community structure occurred in both the microcosms as compared to the original sediment. Sorensen's similarity analysis showed that the microbial population occurring in the untreated microcosms at the end of the experiment was identical to that detected in the original sediment only by the 26%, 29% and 32% when restriction digestion was performed with Mnll, Rsal and Hhal, respectively. A lower similarity was found between the microbial population detected in the pasteurized microcosm and in the sediment (8%, 10% and 25%, respectively). This low similarity is due to the higher complexity of the microbial population occurring in the microcosms, as revealed by the larger number of terminal fragments detected in these samples. The analysis of the relative abundance of each terminal fragment also showed that three major fragments were generated from the sediment DNA with each restriction enzyme. These were not detected or were detected at much lower level in the microcosms as compared to the sediment (Fig. 3.2 for Mnll digestion). In addition, many of the major peaks detected in the microcosms were not found or were poorly represented in the original sediment.



*Figure 3.2.* T-RFLP profile resulted from *Mnl*I digestion: percentage relative abundance of each terminal fragment calculated as peak area/total area ratio.

These data indicate that a selective enrichment of some microbial populations potentially involved in PCBs dechlorination occurred in both microcosms, probably as a consequence of the ongoing dechlorinating process towards PCBs. We cannot exclude the possibility that the observed effects could be also partly due to other factors, such as the incubation temperature, that could have favored indigenous mesophilic bacteria, or the sediment slurry mixing provided during the microcosms setup and sampling, that probably increased the bioavailability of organic substrates and sulfates. In addition, some components of the microbial population not detected in original sediment could have been derived from the site water used to prepare the slurry.

In conclusion, an extensive reductive dechlorination process towards pre-existing and spiked PCBs was observed in both untreated and pasteurized microcosms of sediment and site-water of the Venice lagoon. PCB dechlorination was selectively directed to *meta-* and *para-* positions and seemed to be mediated by sulfate-reducing and spore-forming indigenous bacteria. Finally, T-RFLP analysis showed that some microbial populations originally occurring in the sediment significantly enriched in the PCB dechlorinating microcosms. This could be the consequence of the specific environmental conditions established in the microcosms (e.g., an incubation temperature higher than that occurring at the site from which the sediment was collected, a good slurry homogeneity, that might have increased the bioavailability of sediment nutrients, etc.) but also to the PCB-reductive dechlorination processes that might have favored the specialized bacteria over the other indigenous ones.

#### 3.2. REDUCTIVE DECHLORINATION PROCESSES OF COPLANAR PCBs IN ANOTHER SEDIMENT OF THE VENICE LAGOON AND PRELIMINARY CHARACTERIZATION OF ITS MICROBIAL POPULATION

# 3.2.1. Experimental approach

The sediment employed in this second work was also collected from the Brentella Canal. It was black, silty mud and contained approximately 1.6 mg/kg (on dry wt basis) of a mixture of PCBs which could be partially ascribed to PCBs of Aroclor 1242 and Aroclor 1254. A set of 8 slurryphase anaerobic microcosms consisting of the sediment suspended at 25% (v/v) in water of the same contaminated area was developed under N<sub>2</sub>:CO<sub>2</sub> (70:30) atmosphere according to Fava et al.<sup>96</sup>. Four of them (2 biologically active and 2 autoclave-sterilized) were used to investigate the occurrence of reductive dechlorination processes towards sediment-carried PCBs whereas the other 4 microcosms (2 biologically active and 2 autoclave-sterilized) were spiked with 3,3',4,4'-tetrachlorobiphenyl, 3,3',4,4',5-pentachloro-2,3',4,4',5-pentachlorobiphenyl, 3,3',4,4',5,5'-hexachlorobibiphenyl, phenyl and 2,3,3',4,4',5-hexachlorobiphenyl (all from Ultra Scientific, Rhode Island, USA) in order to investigate: a) the possibility of "priming" (i.e. stimulating) the dechlorination of PCBs pre-existing in the sediment, and b) the potential biological fate of target coplanar dioxin-like PCBs under the geochemical conditions created in the microcosms. Each exogenous PCB was individually added (as solutions at 10,000 mg/l prepared in acetone) to a final concentration of 100 mg/kg of dry sediment; a volume of pure acetone identical of that employed to provide PCBs in the spiked microcosms was added to the parallel non spiked microcosms. prepared through autoclave-sterilization Sterile microcosms were performed at 121°C for 1 h in three consecutive days; in the case of spiked control microcosms, exogenous PCBs were added after autoclave sterilization.

The developed microcosms were then incubated stationary at  $25 \pm 1^{\circ}$ C in the dark for 16 months, during which they were periodically sampled and analyzed to determine the volume and the composition of the head-space gas, as well as the concentration of PCBs and of SO<sub>4</sub><sup>=</sup> according to Fava et al.<sup>96</sup>.

# 3.2.2. Results and discussion

A total amount of sediment carried PCBs corresponding to  $1.60 \pm 0.13$  mg/kg of dry sediment was found to occur in the non-spiked sterile and biologically active microcosms after 7 days of microcosm incubation. No transformation of such PCBs was detected in the sterile microcosms until

208

the end of the experiment. On the contrary, after a 5 months lag phase a significant dechlorination activity towards pre-existing PCBs was detected in the corresponding non spiked biologically active microcosms. The reductive dechlorination process was directed towards several hexa-, pentaand tetra-chlorinated congeners, that were significantly depleted and stoichiometrically bioconverted into less chlorinated congeners at the end of the 16 months experiment. 2,2',5/2,2',4/4,4'-chlorobiphenyl and 2,4/2,5-chlorobiphenyl, as well as 4-monochlorobiphenyl, were the dechlorination products more significantly accumulated in the biologically active microcosms at the end of the experiment (Fig. 3.3).



*Figure 3.3.* Changes in the PCB concentration in the biologically active non spiked microcosms (estimated vs. the sterile ones) at the end of the 16th month of incubation.

A detectable sulfate-reduction activity started to take place in the active microcosms since the first month of incubation. Sulfate was then quickly depleted by about 92% after 2 months of incubation and by 100% at the end of the 3<sup>rd</sup> month of experiment (Fig. 3.4). No significant biogas production was observed in the active microcosms until sulfate was completely depleted. A large amount of biogas ( $27.5 \pm 16.5$  ml) consisting of more than 47% of methane was detected in the same microcosms between the 3<sup>rd</sup> and the 5<sup>th</sup> month of incubation, i.e. immediately after complete sulfate depletion and before PCB dechlorination started. Methane production was detected at a lower rate over the whole experiment (up to the 16<sup>th</sup> month) (Fig. 3.4).



*Figure 3.4.* Sulfate consumption (solid line, empty symbols), biogas (solid line, solid symbols) and methane production (dashed line, solid symbols) in the biologically active non spiked (squares) and spiked (triangles) microcosms during the 16 months of incubation (standard deviation as error bars).

Very similar trends, both in terms of sulfate consumption and methane production, were observed in the parallel biologically active spiked microcosms, where the overall amount of produced methane was about 60% of that detected in the non spiked ones (Fig. 3.4). The biotransformation of several sediment-carried PCBs could not be quantified in the spiked microcosms, as some of them were produced from the spiked congeners dechlorination. However, the fate of about 30% of the GC-ECD peaks ascribed to pre-existing PCBs could be monitored and compared with that observed in the non spiked microcosms (Table 3.2). A significant transformation of such pre-existing PCBs and of the exogenous PCBs was detected starting from the 5<sup>th</sup> month of incubation. However, the bioconversion extent and pattern of pre-existing PCBs (Table 3.2), suggesting that the dechlorination of the latter was not significantly primed by the addition of the 5 exogenous coplanar congeners.

Changes in PCB concentrations – active vs. sterile microcosms (µmoles/kg of					
dry sediment)					
PCB congeners	Non spiked microcosms	Spiked microcosms			
2,6-2,2'-CB	$-0,129 \pm 0,072$	$-0,108 \pm 0,062$			
2,4'-2,3-CB	$-0,021 \pm 0,020$	$0,034 \pm 0,032$			
2,2',5-2,2',4-4,4'-CB	$0,021 \pm 0,015$	$-0,052 \pm 0,009$			
2,2',3-2,4',6-CB	$0,007 \pm 0,002$	$-0,016 \pm 0,002$			
2,2',3,6'-CB	$-0,010 \pm 0,006$	$-0,020 \pm 0,001$			
2,2',3,4-2,3,4',6-CB	$-0,018 \pm 0,008$	$-0,037 \pm 0,003$			
2,3,4',5-CB	$-0,004 \pm 0,007$	$-0,024 \pm 0,002$			
2,3',4,4'-2,2',3,5',6-CB	$-0,181 \pm 0,069$	$-0,239 \pm 0,041$			
2,2',3,4',6-CB	$-0,003 \pm 0,002$	$-0,006 \pm 0,002$			
2,2',3',4,5-CB	$-0,013 \pm 0,007$	$-0,024 \pm 0,001$			
2,2',3,4,5'-CB	$-0,001 \pm 0,005$	$-0,017 \pm 0,004$			
2,2',3,4,4',5'-CB	$-0,261 \pm 0,138$	$-0,289 \pm 0,144$			

*Table 3.2.* Changes in the PCB concentration ( $\mu$ moles/kg of dry sediment) in the biologically active non spiked and spiked microcosms (all estimated vs. the sterile ones) at the end of the 16th month of incubation.

The exogenous PCBs were markedly bioconverted into less chlorinated 3,3',5,5'-/2,3',4,4'-, 2,3',4',5congeners, such as and 2,4,4',5tetrachlorobiphenyl, 2,4,4'-, 2,3',4- and 2,3',5-trichlorobiphenyl and 3,4and 3.4'-dichlorobiphenvl between the 5<sup>th</sup> and the 8<sup>th</sup> month of incubation. and were found to be depleted by more than 80% at the end of the experiment (data not shown). This finding is of great relevance, as it indicates that indigenous microbial consortia selected in the spiked primary microcosms were able to rapidly and extensively dechlorinate some dioxinlike coplanar PCBs that are regarded as some of the most toxic PCBs reported in the literature<sup>104</sup>. The extensive dechlorination of the spiked PCBs only slightly intensified the rate and extent of the dechlorination of the PCBs pre-existing in the sediment. The lack of significant priming effects in aged PCB contaminated marine sediments has also been observed in a different contaminated sediment of the Venice Lagoon spiked with 2,3,4,5,6-pentachlorobiphenyl<sup>102</sup> and in a sediment of the LCP Chemicals Superfund site in coastal Georgia (USA)<sup>97</sup>.

Taken together, these data suggest that PCB pre-existing in the contaminated sediment of Porto Marghera employed in this study can undergo microbial reductive dechlorination. This finding supports previous observations on the occurrence of dechlorination processes in contaminated sediments of the Venice Lagoon<sup>96,102</sup>. As all these laboratory studies were performed under geochemical conditions that mimick those occurring *in* 

*situ,* it is possible to speculate that such processes might be also in progress *in situ* in the lagoon of Venice.

Both sulfate-reduction and methane production occurred sequentially in the biologically active microcosms; PCB reductive dechlorination became detectable when sulfate was completely depleted and methanogenesis started to be significant. Thus, sulfate-reducing bacteria capable of dechlorinating PCBs in the absence of sulfate or methanogenic bacteria were probably responsible for the detected PCB biodegradation processes. The first hypothesis is supported by the work of Zwiernik et al.<sup>105</sup> and by the results of our recent work carried out on another contaminated sediment of Venice Lagoon<sup>102</sup>, whereas the second one is consistent with the findings reported by Alder et al.<sup>95</sup> on the aged PCB-contaminated sediment of the New Bedford Harbor. However, it cannot be excluded that either sulfate reducing bacteria and methanogenic bacteria along with fermentative bacteria, that generally strictly interact with methanogenic and sulfate-reducing bacteria in anaerobic environments<sup>106</sup>, played a direct role in the PCB dechlorination, as already proposed by other authors<sup>94,107</sup>.

To gain deeper insights on the microorganisms potentially involved in the process, a culture enrichment program was started by developing new microcosm sets where the same sediment, spiked with the 5 coplanar PCBs, was progressively applied at lower percentages. Spiked PCB-dechlorination rate increased by about 70% by moving from the primary to the secondary microcosm (developed with sediment in site water at 25% v/v) and by over 100% to the tertiary microcosms (developed with 12% or 6% v/v of sediment). A remarkable increase in sulfate-reduction rates and a progressive decrease in the methanogenic activity were also detected during the serial transfers of the culture, suggesting that an enrichment of sulfatereducing bacteria likely occurred and supporting their possible involvement in the dechlorination process. Molecular analyses of the microbial communities occurring in these enriched microcosms are currently in progress.

# 3.3. CONCLUSIONS

The occurrence of microbially-mediated, reductive dechlorination processes towards weathered PCBs and spiked high chlorinated and co-planar PCBs has been demonstrated in 2 contaminated sediment of the Brentelle Canal of the Porto Marghera area (Venice lagoon, Italy). The detected processes exhibited *meta-* and *para-specificity* (apparently they proceed through dechlorination pattern H' and M) and were not significantly "primed" by the dechlorination of exogenous PCBs, that were rapidly and extensively dechlorinated. PCB dechlorination seemed to be mediated by sulfatereducing bacteria, that probably started to use PCBs as electron acceptors during sulfate reduction but in particular when their native electron acceptor was completely depleted.

Such activities were detected under geochemical conditions that closely mimic those occurring *in situ*, and this allows to speculate that similar processes are also in progress *in situ*. However, the detected PCB dechlorination processes were slow and partial. In general, microbial processes occurring *in situ* are very constrained temporally and spatially. Further, natural sediment systems are complex, heterogeneous, and subjected to (bio)turbation phenomena and low and variable temperatures. Thus, the findings described above do not necessarily prove that biodegradation will provide sufficient natural *in situ* decontamination of the site but for sure, if properly combined with other lines of microbial and biogeochemical evidence coming from the same site<sup>101</sup>, a strong indication of biodegradation *in situ*, and will justify further investigations.

The work of the authors G. Zanaroli, F. Fava, J.R. Pérez-Jiménez & L.Y. Young was supported by grants from the University of Bologna, the Italian MIUR (PRIN 1999) and the Rutgers University. G. Zanaroli was recipient of a fellowship provided by the EC-US Exchange Scientists program in Environmental Biotechnology (EC Accompanying Measure QLK3-CT-2002-30292)

#### 4. Sediments as a biobarrier for CAH-polluted groundwater

# 4.1. DEGRADATION OF CONTAMINANTS IN THE INTERFACE ZONE OF THE TRANSFER OF GROUNDWATER TO SURFACE WATER

Contaminated groundwater reaching surface waters such as rivers and lakes is considered to be an important source of continuous pollution of these surface water bodies. This is especially the case in industrial, urban and agricultural areas that are often located near surface water bodies. However, the fate of the infiltrating groundwater pollutants might be influenced by the sediment zone in eutrophic water bodies since such sediments possess characteristic biological and physico-chemical degradation properties<sup>108-111</sup>. Considering the generally high abundance of bacteria in sediments and the high bacterial activity therein, these microorganisms could play a major role in the degradation of groundwater pollutants infiltrating the sediment zone. Conant et al.<sup>112</sup> studied a tetrachloroethene (PCE) groundwater plume discharging into a river and showed that the near-river zone strongly modified the distribution, concentration and composition of the plume prior to discharging into the surface water. Of the various factors affecting the plume, the extensive anaerobic biodegradation that occurred in the shallow streambed deposits had the greatest effect because it altered the composition and potentially changed the overall toxicity of the plume<sup>112</sup>.

Knowledge on natural attenuation of passing pollutants and the potential to stimulate and sustain occurring degradation processes in the sediment zone are however scarce. This is especially due to the lack of appropriate monitoring devices and tools to measure *in situ* mass balances of pollutants and other reactants/products.

#### 4.2. STUDY OF THE SEDIMENTS OF THE RIVER ZENNE (BELGIUM) AS A BIOBARRIER AGAINST THE DISCHARGE OF CAH-CONTAMINATED GROUNDWATER INTO A EUTROPHIC RIVER

We evaluated the intrinsic capacity of eutrophic river sediment microbial communities to degrade Chlorinated Aliphatic Hydrocarbons (CAHs). The Belgian river Zenne at Vilvoorde was selected for this study. In this area, the Zenne drains groundwater polluted with vinyl chloride (VC) and cis-1,2-dichloroethene (cis-DCE). In a first instance, the concentrations of CAHs discharging into the Zenne were determined by the sampling of groundwater in monitoring wells and Geoprobe direct push sampling techniques (screenpoints) followed by a piston drill sampling of the interstitial water in the river bed sediment. In addition, anaerobic microcosms were constructed using sediment material obtained from different positions in the interface of the CAH-polluted part of the river Zenne. These microcosms, and molecular biological techniques were used to study the CAH-degradation potential of the microbial population of the eutrophic river sediments. The information obtained will indicate if the sediments can be used as a natural biobarrier against the infiltration of the CAH pollution into the surface water.

# 4.2.1. Determination of the CAH-influx zones in the zenne riverbed

In Vilvoorde (Belgium), a 1,2 km-wide CAH groundwater plume, originating from former industrial activities, extends downgradient of multiple source zones to the river Zenne. Previous research indicated that PCE, that was spilled at the sources, is successively converted to TCE, DCE and mainly VC in the aquifer while the groundwater is flowing towards the Zenne.

To determine the zones where the highest CAH concentrations are reaching the river, a screen point measurement campaign was organized. During this campaign, temporary monitoring wells (screen points) were installed along the side of the Zenne at a distance of approximately 5 m from the river and 125 m from each other. Water samples were taken at a depth of 7 and 10 m-bgs at these screen points and in selected monitoring wells. The obtained water samples were analyzed, determining the concentration of CAH, ethene and ethane in the laboratory by, respectively, GC-MS and GC-FID analyses.



*Figure 4.1.* Site map showing CAH groundwater plume that is discharging into the river Zenne.

Figure 4.1 represents the positions of the screen points and monitoring wells along the Zenne that were sampled. Fig. 4.2 indicates the measured concentration of CAH and ethene in the obtained water samples.

The results indicate that at a depth of 7 m-bgs, mainly VC and to a lesser extent DCE are flowing into the Zenne (Fig. 4.2). The highest concentrations of VC were measured in screen point n°4 (154  $\mu$ g/L), screen point n°5 (677  $\mu$ g/L) and 6 (127  $\mu$ g/L). Samples obtained at a depth of 10 m-bgs contained next to VC also a high concentration of DCE (Fig. 4.2). These compounds were again mainly found in screen points n°5 (VC: 2212  $\mu$ g/L; DCE: 150  $\mu$ g/L) and 6 (VC: 743  $\mu$ g/L; DCE: 0  $\mu$ g/L). From these results we concluded that the VC-plume is mainly flowing into the Zenne in the area between screen point n°5 and 6. To narrow the area of investigation, three new screen points were drilled (screen points 5.1, 5.2 and 5.3, approximately 40 m apart and located between screenpoints n°5 and 6). Based on the analysis results we selected a 50 m long test area in the river Zenne (containing screenpoints 5 and 5.1. See rectangle on Fig. 4.2) at which all further measurements and sampling were conducted.

Next to VC and DCE, also the degradation products ethene (Fig. 4.2) and ethane (data not shown) were detected. This indicates that complete degradation of the groundwater pollutants VC and *cis*-DCE to their non-toxic end products ethene and ethane is occurring. However, at some locations, this microbial reductive dechlorination rate clearly does not suffice to prevent the groundwater pollutants from infiltrating into the river.



*Figure 4.2.* Representation of the measured VC, *cis*-DCE and ethene concentrations in the water sampled in the screen points (S) and monitoring wells (MW) in the Zenne area. The rectangle indicates the selected 50 m long test field.

To determine the locations where groundwater discharges into the surface water and the respective CAH concentrations in the interstitial water at these locations, a piston drill sampling of the riverbed sediment was performed in the selected test area, stretching from 'mooring post 26' up to a location 50 m upstream of this post (Fig. 4.3). Every 5 m a sample was taken over a total length of 50 m and this at the right, middle and left side of the Zenne (Fig. 4.3). For the obtained sediment cores, the structure of the river core was described and sub-samples were taken at depths of appr. 10, 50 and 100 cm in the core to determine the concentration of CAHs (by GC-MS analysis), ethene, and ethane (by GC-FID analysis), pH and redox potential.

Apparently, the concentrations of the CAHs that are flowing into the Zenne depend on the sampling place in the river. The spatial distribution of the concentration of VC is shown in Fig. 4.4. At post 26, no VC is flowing into the river. Analysis of the texture of the samples that were taken in this



Figure 4.3. Piston drill sampling at multiple locations in the streambed of the river Zenne.

last area indicated that they contain layers with very fine sand that could prevent the infiltration of VC into the river. In the area from 15 m to 45 m upstream of post 26, VC is flowing into the Zenne and the highest concentrations are detected between 25 and 30 m upstream of this post (> 1000  $\mu$ g/L). While VC is detected throughout the studied Zenne area, *cis*-DCE is only present between 30 m and 40 m upstream of post 26 at a concentration of 27 to 300  $\mu$ g/L (data not shown).



*Figure 4.4.* Measured interpolated VC and *cis*-DCE concentrations in the pore water of the sampled sediment in the river Zenne at the three sampling lines (right, middle and left) and depths (10, 50 and 100 cm).

Depending on the position in the Zenne, the VC-concentration at the top of the piston drill samples (approximately 10 cm beneath the surface water) is lower, equal or higher than in the middle and the bottom of the cores (Fig. 4.4). However, at one location a VC concentration of 1139  $\mu$ g/kg d.m. is detected in the top of the sediment layer. This indicates that VC is discharging into the surface water of the river Zenne. At other locations, e.g. closer to post 26, much lower VC concentrations are present. The concentrations of ethene and ethane that are detected (data not shown) indicate that full dechlorination of VC to ethene and ethane is occurring, but apparently not efficiently enough to prevent VC from flowing into the river.

It can be concluded that the groundwater discharge spatial distribution is highly heterogeneous, mostly depending on local sediment permeability. At some locations (e.g. at post 26) full dechlorination was observed, while VC and *cis*-DCE reach the surface water and discharge into the river at spots with a high-velocity groundwater influx.

# 4.2.2. Determination of the degradation potential of the eutrophic river sediment microbial community by molecular analyses and anaerobic degradation tests

To determine the degradation potential of the eutrophic river sediment microbial community, anaerobic microcosms and molecular biological techniques were applied on sediment material obtained from different positions in the interface of the CAH-polluted test area of the river Zenne.

The presence of dehalogenating bacteria and chloroethene dehalogenase genes was investigated in surface water, groundwater and sediment slices obtained from macro-core sediment samples (Fig. 4.5) collected at the right, middle and left side of the Zenne located around 'mooring post 26'. The undisturbed sediment cores were frozen on dry ice immediately after sampling and divided into slices of approximately 0,5 cm in the laboratory, using an electrical saw. The presence of CAH degrading bacteria (*Dehalococcoides* and *Desulfuromonas* species) and genes involved in the first steps (*pceA* and *tceA*) and the last step (*tceA*, *vcrA* and *bvcA*) of the reductive dehalogenation of PCE to ethene was investigated by PCR<sup>113-116</sup>.



*Figure 4.5.* Macro-core sampling at three different places in the river bed and the obtained undisturbed sediment sample.

Only the PCR results of a sediment core obtained at the right side of the river are shown (Fig. 4.6). Dehalococcoides species, capable of degrading PCE completely to ethene, were present in the investigated groundwater and in all the sediment slices of the cores obtained from the right, middle and left side of the Zenne riverbed, but not in the surface water. Desulfuromonas species, capable of degrading PCE to cis-DCE, were also found in almost all sediment slices of the different cores and in the surface water of the river Zenne, but not in the groundwater. The  $pceA^{(1)}$  gene of Desulfitobacterium hafniense, Dehalobacter restrictus, and Desulfitobacterium sp. PCE1, and the  $pceA^{(2)}$  gene of Sulfurospirillum multivorans were only detected in some of the sediment slices throughout the river bed. The  $pceA^{(2)}$  gene was also detected in surface water of the river Zenne. The tceA gene of Dehalococcoides sp. strain FL-2 and D. ethenogenes strain 195 was present in the groundwater, surface water and some or all sediment slices from the cores collected at the three different sampling places. The encoded TCE reductive dehalogenase can transform TCE to non-toxic ethene, but the reduction of VC to ethene is co-metabolic. The VC reductive dehalogenase genes of Dehalococcoides sp. strain VS and BAV1, vcrA and bvcA, were only detected in the sediment core collected from the right side and not in the middle or the left side of the Zenne. Since these two Dehalococcoides species can derive energy for growth from the degradation of *cis*-DCE and VC to non-toxic ethene as opposed to D. ethenogenes strain 195 and FL-2, their presence indicates that a large microbial degradation potential is present in the river sediment where polluted groundwater is passing the sediment zone.

The actual anaerobic degradation of VC and *cis*-DCE by the eutrophic river sediment microbial community was studied in anaerobic microcosms constructed with fresh Zenne sediment obtained from the top (5-10 cm) or the bottom (10-20 cm) of undisturbed sediment cores that were taken in the test area. A mixture of sediment material (37,5 g) obtained from the left and right side of the Zenne was suspended in 70 ml of groundwater that was collected directly upstream from the river using a Geoprobe direct-push system. The degradation of VC and *cis*-DCE was followed in function of time by headspace GC-FID analysis. The results of these batch degradation tests (Fig. 4.7 representing results 10-20 cm depth sediment material) indicate that both the *in situ* concentrations of VC (around 300  $\mu$ g/L) and *cis*-DCE (around 150  $\mu$ g/L) were rapidly reduced to non-toxic ethene and ethane both in the top and the bottom river sediment. When 2000  $\mu$ g/L *cis*-DCE was re-added to the microcosms, it was again completely reduced to ethene within 1 month.



*Figure 4.6.* PCR results of the detection of dehalogenating bacteria and chloroethene dehalogenase genes in a sediment core obtained at the right side of the river Zenne. Boxes represent surface water, groundwater or the sediment slices that were obtained from the frozen sediment core. Empty boxes indicate that no PCR product was obtained, shaded boxes indicate the presence of the bacterium or catabolic gene.



*Figure 4.7.* Anaerobic degradation of VC and *cis*-DCE in batch degradation tests containing (10-20 cm depth) river sediment material and groundwater from the CAH-polluted test area of the river Zenne.

#### 4.3. CONCLUSIONS

A high microbial CAH degradation potential was detected in the river sediment of the Zenne in Belgium. The detected *Dehalococcoides* sp. strain VS and BAV1, which can actually grow on VC and *cis*-DCE as sole energy source, are probably responsible for the relatively fast and complete reductive dehalogenation of these groundwater pollutants that is observed in the microcosms. By degrading VC and *cis*-DCE, the sediments act as a natural biobarrier for the CAHs present in the groundwater that is passing through the sediment zone, hereby reducing the risk of surface water contamination.

However, some locations in the studied test area have high groundwater influx rates. At these locations VC and *cis*-DCE most likely reach the surface water and discharge into the river. At these locations, the observed microbial degradation potential of the river sediment apparently is inadequate to prevent the CAHs from reaching the surface water. Future research will therefore focus on how to stimulate the degradation of the CAHs, both in the river sediment and in the aquifer material upstream of the river Zenne.

#### 5. Ex-situ and in-situ biotreatment of contaminated sediments

#### 5.1. INTRODUCTION

Pollution of soils, sediments and groundwaters is a major problem worldwide. The total world hazardous waste remediation market is estimated at about US \$16 billion per year. There are at least 350 000 contaminated sites in Western Europe alone and it may cost as much as US \$ 400 billion to clean the most dangerous of these sites over the next 20-25 years<sup>117</sup>. Sediments provide essential habitat for many freshwater, estuarine, and marine organisms. In aquatic systems, most anthropogenic chemicals and waste materials, particularly persistent organic and inorganic chemicals, may accumulate in sediments. These sediments become repositories for many of the more toxic chemicals that are introduced into surface waters. These pollutants can attach to suspended particulates in the water, and subsequently settle out to the bottom. Through complex chemical, physical and biological interactions, these contaminants may be further transformed and transported to other parts of the aquatic system. At elevated concentrations, contaminated sediments lead to many problems in lakes, rivers and harbors, including fish advisories, habitat impairments and restrictions on dredging. Contaminated sediments may also pose an unacceptable risk to aquatic organisms, aquatic-dependent wildlife and humans. Contaminants that build up in the food chain are of particular concern, especially mercury, polychlorinated biphenyls (PCBs), dioxins and organochlorine pesticides. It has become clear that sediment cleanup is an important task and several remediation strategies are designed and implemented to eliminate toxic pollutants from these matrices. Physical and chemical treatments have been used to clean up sediments but they are usually very costly and they are not applicable to all types of environments. For instance, air or steam stripping is practiced primarily for the cleanup of unsaturated soils, whereas "pump and treat" technologies target mainly the groundwater (and sometimes saturated soil and sediment zones), involving pumping and capturing of contaminants (e.g. on a sorbent or by skimming, flotation, etc.) from the water, which is returned to the aquifer. This physical flushing of the pollutants from large volumes of groundwater or saturated sediment is extremely slow and costly. Washing of the solid matrix is also practiced, but is mainly of use on unsaturated soils and less so on saturated soils and sediments. Finally, incineration is a drastic measure, addressing soils and sediments that are contaminated with weathered, often intimately encrusted recalcitrant pollutants. Its cost is high and the environmental impacts can be severe.

Bioremediation strategies are less expensive and friendlier to the environment than physical and chemical treatments of contaminated soils and, to a lesser extent, sediments. This set of technologies uses biological, primarily microbial, activities and can be classified into three major techniques:

- 1. *In situ*: No excavation, the solids remain in place. This class of techniques can be subdivided into engineered in situ bioremediation and intrinsic bioremediation (or natural attenuation). Examples include bioventing, biosparging, etc.
- 2. *Ex situ*: Sediment is excavated and transferred to be treated in specialized installations far from the site. It is always an engineered solution. Examples: land farming, slurry reactors. A sub-category of ex situ bioremediation, occasionally considered as a separate class of techniques, is *on site* bioremediation, which implies treatment of a soil or sediment at the surface of the site (e.g., biopiling).

The applicability of different biological treatments depends on the susceptibility of the pollutants to biodegradation (Table 5.1) and on the type of environment targeted for cleanup. A number of prior steps are needed before deciding on the bioremediation strategy, including a thorough characterization of the site in terms of hydrology, extent and type of contamination, intrinsic microbial activity, and supply rates of key materials, such as electron acceptors or donors<sup>118</sup>.

Pollutant	Frequency of	Status of bio-	Limitations
	occurrence	remediation	
Polycyclic aromatic hydrocarbons	Common	Emerging	Sorb strongly to subsurface Solids
Chlorinated aliphatics	Very frequent	Emerging	Nonaqueous phase liquids (NAPL)
Chlorinated aromatics	Common	Emerging	NAPL and solids
Polychlorinated biphenyls	Infrequent	Emerging	Sorb strongly to subsurface Solids
Heavy metals & metalloids	Common	Possible to immobilize	Availability highly variable
Gasoline, fuel oil	Very frequent	Established	NAPL

Table 5.1.	Contaminant	susceptibility to	bioremediation.
10010 0.11.	Containinaint	susceptionity to	ororennearation.

#### 5.2. IN SITU BIOREMEDIATION

In situ bioremediation encourages growth and reproduction of indigenous microorganisms to enhance biodegradation of organic constituents in the saturated zone. In situ groundwater bioremediation can effectively degrade organic constituents which are dissolved and adsorbed onto the aquifer matrix. In an analogous manner, a sediment contaminated with organic pollutants can be cleaned up. In situ bioremediation can be effective for the range of xenobiotic compounds indicated in Table 5.2. In certain cases in situ bioremediation is more effective when combined with other saturated zone remedial technologies (e.g., air sparging) and, for the case of contaminated soils, with vadose zone remedial operations (e.g., soil vapor extraction, bioventing). Air sparging involves the injection of compressed air or oxygen directly into the contaminated subsurface. Injected air traverses horizontally and vertically in channels through the sediment column, creating an underground stripper that removes volatile and semivolatile organic contaminants by volatilization and entrainment in the gas stream. The injected air helps to flush the contaminants into the unsaturated zone. Soil Vapor Extraction (SVE) usually is implemented in conjunction with air sparging to remove the generated vapor-phase contamination from the vadose zone. Oxygen added to the contaminated groundwater and vadose-zone soils also can enhance biodegradation of contaminants below and above the water table by stimulating the local microbial activity. Bioventing is an in situ remediation technology that combines soil vapor extraction methods with bioremediation. It uses vapor extraction wells that induce air flow in the subsurface through the use of a vacuum. Bioventing can be effective in remediating emissions of petroleum products, such as gasoline, jet fuels, kerosene, and diesel fuel but is mostly applicable to unsaturated matrices. Since the contaminated sediment represents a partly or totally saturated zone, a water circulation strategy for the supply of metabolism stimulating materials can often be applied *in situ*, e.g. for the engineered bioremediation of a nonaqueous phase liquid such as petroleum. In this case, hydrogen peroxide can be used as a dissolved source of oxygen, via vertical wells and horizontal infiltration galleries<sup>118,119</sup> (see below).

# 5.2.1. Biostimulation and bioaugmentation

*In situ* bioremediation can be implemented in a number of treatment modes, including: Aerobic (oxygen respiration); anoxic (nitrate respiration); anaerobic (non-oxygen respiration); and co-metabolic. The aerobic mode has been proven most effective in reducing contaminant levels of aliphatic (*e.g.*, hexane) and aromatic petroleum hydrocarbons (*e.g.*, benzene,

224

naphthalene) typically present in gasoline and diesel fuel. In the aerobic treatment mode, the sediment and aquifer area is oxygenated by one of three methods: Direct sparging of air or oxygen through an injection well; saturation of water with air or oxygen prior to re-injection; or addition of hydrogen peroxide directly into an injection well or into reinjected water. Irrespective of the method of oxygenation, it is important to ensure that oxygen is being distributed throughout the area of contamination. Anoxic, anaerobic, and co-metabolic modes are sometimes used for remediation of other compounds, such as chlorinated solvents, but are generally slower than aerobic respiration in breaking down xenobiotics. The key parameters that determine the effectiveness of *in situ* bioremediation are: (a) hydraulic conductivity of the aquifer, which controls the distribution of electron acceptors and nutrients in the subsurface; (b) biodegradability of the xenobiotic constituents, which determines both the rate and degree to which constituents will be degraded by microorganisms; and (c) location of the xenobiotic contamination in the subsurface. In general, the sediment/aquifer system will determine hydraulic conductivity. Fine-grained media (e.g., clays, silts) have lower intrinsic permeability than coarse-grained media (e.g., sands, gravels). Bioremediation is generally effective in permeable media. However, depending on the extent of contamination, bioremediation also can be effective in less permeable silty or clayey media, which represent a considerable proportion of sediments. In general, a medium of lower permeability will require a longer period of time to clean up than a more permeable medium<sup>3</sup>. The biodegradability of a xenobiotic constituent is a measure of its ability to be metabolized (or co-metabolized) by microorganisms. The chemical characteristics of the contaminants will dictate their biodegradability. For example, heavy metals are not degraded by bioremediation but they can be sequestered as insoluble precipitates under specific redox conditions. The biodegradability of organic constituents depends on their chemical structures and physical/chemical properties (e.g., water solubility, water/octanol partition coefficient). Highly soluble organic compounds with low molecular weights will tend to be more rapidly degraded than slightly soluble compounds with high molecular weights. The low water solubilities of the more complex bioavailable compounds render them less to microorganisms. Consequently, the larger, more complex chemical compounds may be slow to degrade or may even be recalcitrant to biological degradation<sup>118,119</sup>.

When oxygen is introduced to the subsurface as a terminal electron acceptor, it can react with dissolved iron [Fe(II)] to form an insoluble iron precipitate, ferric oxide. This precipitate can be deposited in infiltration trenches or channels, reducing permeability. The effects of iron precipitation tend to be most noticeable around injection wells, where oxygen concentration in groundwater is highest and can render injection wells inoperable.

Extreme pH values (*i.e.*, less than 5 or greater than 10) are generally unfavorable for microbial activity. Typically, optimal microbial activity occurs under neutral pH conditions (*i.e.*, in the range of 6-8). The optimal pH is site specific. For example, aggressive microbial activity has been observed at lower pH conditions outside of this range (*e.g.*, 4.5 to 5) in natural systems. Because indigenous microorganisms have adapted to the natural conditions where they are found, pH adjustment, even toward neutral, can inhibit microbial activity. If man-made conditions (*e.g.*, releases of petroleum) have altered the pH outside the neutral range, pH adjustment may be needed. If the pH of the groundwater is too acidic, lime or sodium hydroxide can be added to increase the pH. If the pH is too alkaline, then a suitable acid (*e.g.*, hydrochloric) can be added to reduce the pH. Changes to pH should be closely monitored because rapid changes of more than 1 or 2 units can inhibit microbial activity and may require an extended acclimation period before the microbes resume their activity.

As a rule, microorganisms require carbon/energy sources (electron donors) and terminal electron acceptors (TEA) to enzymatically transform the target contaminants. Carbon source(s) and TEA are introduced in *in situ* engineered bioremediation systems via injection wells or infiltration galleries. Extraction wells ensure that the pollutant plume does not spread contaminants into clean areas or accelerate the movement toward receptors. Placement of extraction wells is critical, especially in systems that also use nutrient injection wells or infiltration galleries. These additional sources of water can alter the natural groundwater flow patterns which can cause the contaminant plume to move in an unintended direction or rate. Without adequate hydraulic control, this situation can lead to worsening of the original condition and complicate the cleanup or extend it.

Nutrient injection systems may be unnecessary if the sediment/ groundwater system contains adequate amounts of nutrients, such as N and P, to sustain microorganism growth and biodegradation. Nutrients may be available in sufficient quantities in the sediment/aquifer system (intrinsic *in situ* bioremediation, in which no external intervention is taking place other than a careful monitoring of the natural attenuation of the target contaminants) but, more frequently, nutrients need to be added to maintain adequate bacterial populations (biostimulation)<sup>118,119</sup>. Addition of N, P and other inorganic nutrients to optimize microbial growth is becoming established for in situ bioremediation of subsurface contaminated by hydrocarbons. Supply of electron donors (e.g. sugar or methane) or TEA (e.g. oxygen or hydrogen peroxide) in case of such limitations, or addition of detergents in case of low bioavailability of the pollutants, are instances of biostimulation<sup>119</sup>.

Sometimes the microbial community of a polluted site may not display the metabolic potential for degradation and complete mineralization of the target pollutant. This could be due to very low numbers of a single microorganism possessing the entire pathway for its eventual breakdown to harmless end products. More likely, the pollutant may be a complex molecule or a mixture of compounds which could be broken down only by a specific combination of microorganisms ('consortium') and pathways. In such cases, successful bioremediation may call for bioaugmentation, i.e. the inoculation of the polluted sediment with specific populations of microorganisms<sup>120</sup>. Bioaugmentation is not yet a mature technology for in situ bioremediation of soils and groundwater and, even less so, of sediments. With the exception of a few field applications most bioaugmentation reports concern laboratory-scale studies. It is then no surprise that bioaugmentation is mired in controversy, given the excessive claims of success, esp. by commercial companies, and the documented campaigns<sup>119,120</sup>. inoculation Typical failures of large-scale bioaugmentation options include: the addition of a pre-adapted pure bacterial strain (e.g. inoculating with chloroaromatic degraders); the addition of pre-adapted consortia (e.g. PCB-degrading enrichment cultures to contaminated sediments); addition of genetically modified bacteria for simultaneous mineralization of chloro- and methylaromatics; and transfer of biodegradation-relevant genes by conjugation into microorganisms present in the biotope to help remove PCBs or pesticides<sup>120</sup>. Ideally, the applicability and the limits of bioaugmentation should be tested in a situation comparable to the eventual intervention, i.e., at a mesocosm or pilot scale, after initial microcosm tests. In the first study of bioaugmentation of a realistic 500-liter pilot-scale reactor simulating an anaerobic aquifer/sediment system, we tested the capacity of an inoculum of the halorespiring anaerobe Desulfomonile tiedjei to dechlorinate 3chlorobenzoate (3-CB) continuously fed into the saturated zone together with acetate+formate as cosubstrate<sup>121</sup>. Heterogeneous distribution of 3-CB dechlorination activity correlated with the presence of the inoculated bacterium which was detected by its 16S rRNA gene signature using PCR. Denaturing gradient gel electrophoresis (DGGE) of PCR amplicons from the total DNA of different compartments in the reactor demonstrated distinct community fingerprints as a function of location and in response to process modifications<sup>121</sup>Reductive dehalogenation of organic pollutants in groundwater following biostimulation of native microbial communities is often seen in the field, where chlorinated ethenes (PCE, TCE) are completely transformed to ethene<sup>122</sup>. According to industrial practitioners of in situ bioremediation in the field, bioaugmentation with specific halorespiring anaerobes of the genus Dehalococcoides, which degrades chloroethenes to ethene without stopping at *cis*-dichloroethene (cDCE) or at vinyl chloride (VC), may shorten the lag phase close to the injection wells, but is not required to assure complete in situ dechlorination. Moreover, because it is not necessary to constrain hydrogen levels for the benefit of inoculated Dehalococcoides sp., it is also not necessary to limit rates of electron donor (cosubstrate) consumption by using commercial "designer" slow-release cosubstrates<sup>122</sup>. In a pilot investigation consistent with this standpoint, we examined on a demonstration (pilot) scale the dynamics of in situ bioremediation of a sediment co-contaminated with trichloroethene (TCE) and nickel<sup>123,124</sup> in preparation for the field scale *in situ* intervention at the former metallurgical industrial site of Bunnik near Utrecht (The Netherlands), where the contaminated anaerobic sediment was located 20-30 m below ground. The 680-liter "sandbox-type" reactor filled with the contaminated sediment was fed under strict anaerobic conditions a 10 mg/l TCE-containing stream simulating the plume together with a methanol+lactate cosubstrate mixture. The cosubstrate chemical oxygen demand (COD) was progressively diminished from 1000 mg/l down to 100 mg/l as the redox potential reached levels consistent with sulfate reduction and methanogenesis while the feed COD/SO4 ratio was fixed at 100/0.6  $(w/w)^{123}$ . In previous small-scale fed-batch sediment column tests, this cosubstrate mixture had been shown to be a good electron donor to sustain complete TCE dechlorination, and a level of sulfate up to 10 mM not only did not interfere with the complete transformation of TCE to ethane, but also contributed to Ni sequestration in the form of NiS<sup>124</sup>. Results of the pilot reactor showed that TCE was dechlorinated to harmless ethene via cDCE and VC by indigenous microorganisms in the sediment, whereas concurrent stimulation of sulfate-reducing anaerobic bacteria with the fed sulfate led to the generation of sulphide which was efficient in precipitating nickel within the sediment<sup>123</sup>. Remarkably, probing with 16s rDNA -based PCR gave no indication of the presence of dechlorinating bacteria of the Dehalococcoides group, which indicates that the complete dechlorination was brought about either by as yet unknown halorespirers or cometabolically by local anaerobic trophic groups including sulphate reducers<sup>124</sup>. The dynamics of the process is shown in Fig. 5.1.



*Figure 5.1.* Time course of the concentrations of pollutant and products in the demonstration-scale sediment bioreactor. Chloroethene concentrations at the reactor outlet in the gas phase (upper panel) and in the liquid phase (lower panel).

#### 5.3. EX SITU BIOREMEDIATION

Above ground treatment of contaminated sediments is an option in the case of limited of quantities of heavily contaminated sediments, and whenever regulatory compliance or other constraints impose a rapid cleanup<sup>118,119</sup>. This set of technologies, which includes slurry bioreactor treatment, land farming and composting, should only be practiced when the usually high costs of excavation and exposure to workers and residents can be more than compensated by a major risk reduction stemming from the removal of the pollutant source.

Slurry bioreactor treatment involves vigorous agitation of the sediment which may constitute 5-20% of the solids in a two- or three-phase configuration (depending on whether there is aeration or not) in the presence of native or, occasionally, externally introduced competent microorganisms. The degree of control and optimization that can be exercised on this system is considerable: in addition to vigorous inocula, nutrients, TEA and other additives can be introduced as needed. Therefore, the endpoints of this treatment are also far more predictable than in *in situ* bioremediation. Because of the high degree of convective mass transfer in this setup, the rates of biotransformation and mineralization of the target contaminants are several-fold higher than what is observed *in situ*. In addition, mass transfer into the aqueous phase and, hence, overall rates of biodegradation in the case of less bioavailable, poorly water soluble compounds (such as PCB, PAH etc.), can be enhanced with the addition of solubilizing agents such as surfactants<sup>117,118</sup>.

Land farming involves the mixing of contaminated sediment into the surface layer of topsoil, to which there is an occasional addition of nutrients and moisture, the aim being an aerobic cometabolic degradation of the target pollutants by the soil microorganisms<sup>118</sup>. This empirical, generally uncontrolled, "low" technology can be upgraded by mixing the contaminated solids with fresh organic residues, resulting in composting<sup>117</sup>. The addition of air by appropriate channeling or turning as well as the regulation of moisture (and nutrient or cosubstrate input) can enhance the rates of pollutant degradation and also increase the overall degree of control and predictability of the treatment of sediments. Nonetheless, both these techniques can be cumbersome, time-consuming and expensive: in addition to the costs of excavation, transportation and mechanical intervention (e.g. turning, tilling, etc.), they do require ample land areas. Moreover, certain very recalcitrant contaminants, even if immobilized or rendered nonbioavailable by sorption and/or covalent binding to the humic acids of the compost or to other components of the soil matrix may present a risk of eventually leaching out under extreme local conditions.

#### Acknowledgements

The work of the authors S. E. Fantroussi and Spiros N. Agathos was supported by grants CHLOREM (BIO 4-CT 1998-0303) and MADOX (QLK3-CT-2001-00345) of the European Commission.
#### References

- N.R. Pace, A molecular view of microbial diversity and the biosphere, *Science* 276, 734–740 (1997).
- V.J. Orphan, C.H. House, K.U. Hinrichs, K.D. McKeegan, and E.F. DeLong, Methane-consuming archaea revealed by directly coupled isotopic and phylogenetic analysis, *Science* 293, 484–487 (2001).
- K.P. Nevin, K.T. Finneran, and D.R. Lovley, Microorganisms associated with uranium bioremediation in a high-salinity subsurface sediment, *Appl. Environ. Microbiol.* 69, 3672–3675 (2003).
- 4. S. El Fantroussi, W. Verstraete, and E.M. Top, Enrichment and molecular characterization of a bacterial culture that degrades methoxy-methyl urea herbicides and their aniline derivatives, *Appl. Environ. Microbiol.* 66, 5110–5115 (2000).
- M.R. Harkness, J.B. McDermott, D.A. Abramowicz, J.J. Salvo, W.P. Flanagan, M.L. Stephens, F.J. Mondello, R.J. May, J.H. Lobos, K.M. Carroll, et al., In situ stimulation of aerobic PCB biodegradation in Hudson river sediments, *Science* 259, 503–507 (1993).
- 6. E.L. Madsen, Epistemology of environmental microbiology, *Environ. Sci. Technol.* 32, 429–439 (1998).
- L. Eyers, I. George, L. Schuler, S.N. Agathos, and S. El Fantroussi, Environmental metagenomics: exploring the unmined richness of microbes to degrade xenobiotics. *Appl. Microbiol.Biotechnol.* 66, 123–130 (2004).
- P.D. Schloss, and J. Handelsman, Biotechnological prospects from metagenomics, *Curr. Opin. Microbiol.* 14, 303–310 (2003).
- E.L. Madsen, Methods for determining biodegradability. In: C.J. Hurst, R.L. Crawford, G.R. Knudsen, M.J. McInerney and L.D. Stetzenbach (Eds) Manual of Environmental Microbiology, 2<sup>nd</sup> Edition, ASM Press, Washington DC. (2002).
- S. El Fantroussi, M. Belkacemi, E.M. Top, J. Mahillon, H. Naveau, and S.N. Agathos, Bioaugmentation of a soil bioreactor designed for pilot scale anaerobic bioremediation studies, *Environ. Sci. & Technol.* 33, 2992–3001 (1999).
- E.R. Hendrickson, J.A. Payne, R.M. Young, M.G. Starr, M.P. Perry, S. Fahnestock, D.E. Ellis, and R.C. Ebersole, Molecular analysis of *Dehalococcoides* 16S ribosomal DNA from chloroethene-contaminated sites throughout North America and Europe, *Appl. Environ. Microbiol.* 68, 485–495 (2002).
- J.R. de Lipthay, N. Tuxen, K. Johnsen, L.H. Hansen, H.J. Albrechtsen, P.L. Bjerg and J. Aamand, In situ exposure to low herbicide concentrations affects microbial population composition and catabolic gene frequency in an aerobic shallow aquifer, *Appl. Environ. Microbiol.* 69, 461–7 (2003).
- W.S. Reeburgh, S.C. Whalen, and M.J. Alperin, The role of methylotrophy in the global methane budget, p. 1–14. *In* J.C. Murrell, and D.P. Kelly (ed.), Microbial growth on C<sub>1</sub> compounds. Proceedings of the 7th International Symposium. American Society for Microbiology, Washington, D.C. (1993).
- T.R. Thomsen, K. Finster, and N.B. Ramsing, Biogeochemical and molecular signatures of anaerobic methane oxidation in a marine sediment, Appl. Environ. Microbiol. 67, 1646–1656 (2001).
- K.U. Hinrichs, J.M. Hayes, S.P. Sylva, P.G. Brewer, and E.F. DeLong, Methaneconsuming archaebacteria in marine sediments, *Nature* 398, 802–805 (1999).
- R.M. Atlas and R. Bartha, Microbial Ecology: Fundamentals and Applications. Menlo Park, CA: Benjamin/Cummings (1998).

- W.W. Mohn and J.M. Tiedje, Microbial reductive dehalogenation, *Microbiol. Rev.* 56, 482–507. (1992).
- V. Achá, M. Meurens, H. Naveau, and S.N. Agathos, ATR-FTIR sensor development for continuous on-line monitoring of chlorinated aliphatic hydrocarbons in a fixed-bed bioreactor, *Biotechnol. Bioeng.* 68, 473–487 (2000).
- 19. R.M. Atlas, Microbial hydrocarbon degradation bioremediation of oil spills, *J. Chem. Technol. Biotechnol.* 52, 149–156 (1991).
- F.H. Chapelle, Bioremediation of petroleum hydrocarbon-contaminated ground water: The perspectives of history and hydrology, *Ground Water* 37, 122–132 (1999).
- D.C. Dobbins, C.M. Aelion, and F. Pfaender, Subsurface, terrestrial microbial ecology and biodegradation of organic chemicals: A review, *Crit. Rev. Environ. Ctrl.* 22, 67– 136 (1992).
- 22. D.F. Pope, and J.N. Jones, 1999, Monitored natural attenuation of chlorinated solvents, edited. (EPA, Washington, 1999), pp. 1–3.
- 23. K. Watanabe, H. Futamata, and S. Harayama, Understanding the diversity in catabolic potential of microorganisms for the development of bioremediation strategies, *Antonie Van Leeuwenhoek* 81, 655–663 (2002).
- D.H. Pieper, V. dos Santos, and P.N. Golyshin, Genomic and mechanistic insights into the biodegradation of organic pollutants, *Curr. Opin. Biotechnol.* 15, 215–224 (2004).
- 25. K. Watanabe, Microorganisms relevant to bioremediation, *Curr. Opin. Biotechnol.* 12, 237–241 (2001).
- 26. C.S. Harwood, and R.E. Parales, The beta-ketoadipate pathway and the biology of self-identity, *Annu. Rev. Microbiol.* 50, 553–590 (1996).
- 27. N. Söhngen, Benzin, Petroleum, Paraffinöl und Paraffin als Kohlenstoff-und Energiequelle für Mikroben, *Zentr. Bacteriol. Parasitenk. Abt II* 37, 595–609 (1913).
- R.Y. Stanier, The oxidation of aromatic compounds by fluorescent *Pseudomonas*, J. Bacteriol. 55, 477–494 (1947).
- 29. M. Boll, G. Fuchs, and J. Heider, Anaerobic oxidation of aromatic compounds and hydrocarbons, *Curr. Opin. Chem. Biol.* 6, 604–611 (2002).
- J. Heider, A.M. Spormann, H.R. Beller, and F. Widdel, Anaerobic bacterial metabolism of hydrocarbons, *FEMS Microbiol. Rev.* 22, 459–473 (1998).
- 31. M.R. Smith, The biodegradation of aromatic hydrocarbons by bacteria, *Biodegradation* 1, 191–206 (1990).
- D.T. Gibson, and R.E. Parales, Aromatic hydrocarbon dioxygenases in environmental biotechnology, *Curr. Opin. Biotechnol.* 11, 236–243 (2000).
- 33. S. Beil, J.R. Mason, K.N. Timmis, and D.H. Pieper, Identification of chlorobenzene dioxygenase sequence elements involved in dechlorination of 1,2,4,5-tetrachlorobenzene, *J. Bacteriol.* 180, 5520–5528 (1998).
- K. Furukawa, H. Suenaga, and M. Goto, Biphenyl dioxygenases: functional versatilities and directed evolution, *J. Bacteriol.* 186, 5189–5196 (2004).
- 35. M. Zielinski, S. Backhaus, and B. Hofer, The principal determinants for the structure of the substrate- binding pocket are located within a central core of a biphenyl dioxygenase alpha subunit, *Microbiology* 148, 2439–2448 (2002).
- R.E. Parales, S.M. Resnick, C.L. Yu, D.R. Boyd, N.D. Sharma, and D.T. Gibson, Regioselectivity and enantioselectivity of naphthalene dioxygenase during arene cisdihydroxylation: Control by phenylalanine 352 in the alpha subunit, *J. Bacteriol.* 182, 5495–5504 (2000).
- J. Armengaud, B. Happe, and K.N. Timmis, Genetic analysis of dioxin dioxygenase of Sphingomonas sp. strain RW1: Catabolic genes dispersed on the genome, J. Bacteriol. 180, 3954–3966 (1998).

- D.T. Gibson, R.L. Roberts, M.C. Wells, and V.M. Kobal, Oxidation of biphenyl by a Beijerinckia species, *Biochem. Biophys. Res. Comm.* 50, 211–215 (1973).
- Y. Furusawa, V. Nagarajan, M. Tanokura, E. Masai, M. Fukuda, and T. Senda, Crystal structure of the terminal oxygenase component of biphenyl dioxygenase derived from *Rhodococcus* sp strain RHA1, *J. Mol. Biol.* 342, 1041–1052 (2004).
- X.S. Dong, S. Fushinobu, E. Fukuda, T. Terada, S. Nakamura, K. Shimizu, H. Nojiri, T. Omori, H. Shoun, and T. Wakagi, Crystal structure of the terminal oxygenase component of cumene dioxygenase from *Pseudomonas fluorescens* IP01, *J. Bacteriol.* 187, 2483–2490 (2005).
- 41. M. Zielinski, S. Kahl, H.J. Hecht, and B. Hofer, Pinpointing biphenyl dioxygenase residues that are crucial for substrate interaction, *J. Bacteriol.* 185, 6976–6980 (2003).
- 42. H. Junca, I. Plumeier, H. Hecht, and D. Pieper, Difference in kinetic behaviour of catechol 2,3-dioxygenase variants from a polluted environment, *Microbiology* 150, 4181–4187 (2004).
- 43. J.G. Leahy, P.J. Batchelor, and S.M. Morcomb, Evolution of the soluble diiron monooxygenases, *FEMS Microbiol. Rev.* 27, 449–479 (2003).
- M.S. Shields, S.O. Montgomery, P.J. Chapman, S.M. Cuskey, and P.H. Pritchard, Novel pathway of toluene catabolism in the trichloroethylene-degrading bacterium G4, *Appl. Environ. Microbiol.* 55, 1624–1629 (1989).
- 45. R.H. Olsen, J.J. Kukor, and B. Kaphammer, A novel toluene-3-monooxygenase pathway cloned from *Pseudomonas pickettii* PKO1, *J. Bacteriol.* 176, 3749–3756 (1994).
- 46. G.M. Whited, and D.T. Gibson, Toluene-4-monooxygenase, a three-component enzyme system that catalyzes the oxidation of toluene to *p*-cresol in *Pseudomonas mendocina* KR1, *J. Bacteriol.* 173, 3010–3016 (1991).
- 47. A. Fishman, Y. Tao, and T.K. Wood, Toluene 3-monooxygenase of *Ralstonia pickettii* PKO1 is a para-hydroxylating enzyme, *J. Bacteriol.* 186, 3117–3123 (2004).
- 48. G.M. Whited, and D.T. Gibson, Separation and partial characterization of the enzymes of the toluene-4-monooxygenase catabolic pathway in *Pseudomonas mendocina* KR1, *J. Bacteriol.* 173, 3017–3020 (1991).
- 49. K.H. Mitchell, J.M. Studts, and B.G. Fox, Combined participation of hydroxylase active site residues and effector protein binding in a *para* to *ortho* modulation of toluene 4-monooxygenase regiospecificity, *Biochemistry* 41, 3176–3188 (2002).
- L. Rui, K.F. Reardon, and T.K. Wood, Protein engineering of toluene *ortho*monooxygenase of *Burkholderia cepacia* G4 for regiospecific hydroxylation of indole to form various indigoid compounds, *Appl. Microbiol. Biotechnol.* 66, 422–429 (2005).
- 51. K.A. Canada, S. Iwashita, H. Shim, and T.K. Wood, Directed evolution of toluene *ortho*-monoxygenase for enhanced 1-naphthol synthesis and chlorinated ethene degradation, *J. Bacteriol.* 184, 344–349 (2002).
- M.H. Sazinsky, J. Bard, A. Di Donato, and S.J. Lippard, Crystal structure of the toluene/o-xylene monooxygenase hydroxylase from *Pseudomonas stutzeri* OX1 -Insight into the substrate specificity, substrate channeling, and active site tuning of multicomponent monooxygenases, *J. Biol. Chem.* 279, 30600–30610 (2004).
- 53. W. Reineke, Development of hybrid strains for the mineralization of chloroaromatics by patchwork assembly, *Annu. Rev. Microbiol.* 52, 287–331 (1998).
- D. Pieper, and W. Reineke, 2004, Degradation of chloroaromatics by Pseudomona(d)s, in: *The Pseudomonads* Vol III. Biosynthesis of Macromolecules and Molecular metabolism, edited by J.L. Ramos, (Kluwer Academic/Plenum Publishers, New York, 2004), pp. 509–574.

- L.D. Eltis, and J.T. Bolin, Evolutionary relationships among extradiol dioxygenases, J. Bacteriol. 178, 5930–5937 (1996).
- A. Kitayama, T. Achioku, T. Yanagawa, K. Kanou, M. Kikuchi, H. Ueda, E. Suzuki, H. Nishimura, T. Nagamune, and Y. Kawakami, Cloning and characterization of extradiol aromatic ring-cleavage dioxygenases of *Pseudomonas aeruginosa* JI104, *J. Ferm. Bioeng.* 82, 217–223 (1996).
- 57. N. Schweigert, A.J.B. Zehnder, and R.I.L. Eggen, Chemical properties of catechols and their molecular modes of toxic action in cells, from microorganisms to mammals, *Environ. Microbiol.* 3, 81–91 (2001).
- D. Perez-Pantoja, T. Ledger, D.H. Pieper, and B. Gonzalez, Efficient turnover of chlorocatechols is essential for growth of *Ralstonia eutropha* JMP134(pJP4) in 3chlorobenzoic acid, *J. Bacteriol.* 185, 1534–1542 (2003).
- 59. H.R. Beller, and A.M. Spormann, Anaerobic activation of toluene and o-xylene by addition to fumarate in denitrifying strain T, *J. Bacteriol.* 179, 670–676 (1997).
- B. Leuthner, C. Leutwein, H. Schulz, P. Horth, W. Haehnel, E. Schiltz, H. Schagger, and J. Heider, Biochemical and genetic characterization of benzylsuccinate synthase from *Thauera aromatica*: a new glycyl radical enzyme catalysing the first step in anaerobic toluene metabolism, *Mol. Microbiol.* 28, 615–628 (1998).
- 61. F. Widdel, and R. Rabus, Anaerobic biodegradation of saturated and aromatic hydrocarbons, *Curr. Opin. Biotechnol.* 12, 259–276 (2001).
- 62. E. Annweiler, A. Materna, M. Safinowski, A. Kappler, H.H. Richnow, W. Michaelis, and R.U. Meckenstock, Anaerobic degradation of 2-methylnaphthalene by a sulfate-reducing enrichment culture. *Appl. Environ. Microbiol.* 66, 5329–5333 (2000).
- 63. S.R. Kane, H.R. Beller, T.C. Legler, and R.T. Anderson, Biochemical and genetic evidence of benzylsuccinate synthase in toluene-degrading, ferric iron-reducing *Geobacter metallireducens*, *Biodegradation* 13, 149–154 (2002).
- C.D. Phelps, J. Battistelli, and L.Y. Young, Metabolic biomarkers for monitoring anaerobic naphthalene biodegradation in situ, *Environ. Microbiol.* 4, 532–537 (2002).
- 65. J.D. Coates, R. Chakraborty, and M.J. McInerney, Anaerobic benzene biodegradation a new era, *Res. Microbiol.* 153, 621–628 (2002).
- 66. W.W. Mohn, and J.M. Tiedje, Evidence for chemiosmotic coupling of reductive dechlorination and ATP synthesis in *Desulfomonile tiedjei*, *Arch. Microbiol.* 157, 1–6 (1991).
- C. Holliger, C. Regeard, and G. Diekert, 2003, Dehalogenation by anaerobic bacteria, in: *Dehalogenation: Microbial Processes and Environmental Applications*, edited by M.M. Häggblom, and I.D. Bossert. (Kluwer Academic Publishing, Boston, USA, 2003), pp. 115–157.
- J.Z. He, K.M. Ritalahti, K.L. Yang, S.S. Koenigsberg, and F.E. Loffler, Detoxification of vinyl chloride to ethene coupled to growth of an anaerobic bacterium, *Nature* 424, 62–65 (2003).
- L. Adrian, U. Szewzyk, J. Wecke, and H. Gorisch, Bacterial dehalorespiration with chlorinated benzenes, *Nature* 408, 580–583 (2000).
- M. Bunge, L. Adrian, A. Kraus, M. Opel, W.G. Lorenz, J.R. Andreesen, H. Gorisch, and U. Lechner, Reductive dehalogenation of chlorinated dioxins by an anaerobic bacterium, *Nature* 421, 357–360 (2003).
- S. Ni, J.K. Fredrickson, and L. Xun, Purification and characterization of a novel 3chlorobenzoate-reductive dehalogenase from the cytoplasmic membrane of *Desulfomonile tiedjei* DCB-1, *J. Bacteriol.* 177, 5135–5139 (1995).

- B.C. Berks, F. Sargent, E. De Leeuw, A.P. Hinsley, N.R. Stanley, R.L. Jack, G. Buchanan, and T. Palmer, A novel protein transport system involved in the biogenesis of bacterial electron transfer chains, *Biochim. Biophys. Acta* 1459, 325–330 (2000).
- J.A. Muller, B.M. Rosner, G. von Abendroth, G. Meshulam-Simon, P.L. McCarty, and A.M. Spormann, Molecular identification of the catabolic vinyl chloride reductase from *Dehalococcoides* sp strain VS and its environmental distribution, *Appl. Environ. Microbiol.* 70, 4880–4888 (2004).
- 74. V. Torsvik, and L. Ovreas, Microbial diversity and function in soil: from genes to ecosystems, *Curr. Opin. Microbiol.* 5, 240–245 (2002).
- P. Wikstrom, A. Wiklund, A.C. Andersson, and M. Forsman, DNA recovery and PCR quantification of catechol 2,3-dioxygenase genes from different soil types, *J. Biotechnol.* 52, 107–120 (1996).
- H.R. Beller, S.R. Kane, T.C. Legler, and P.J. Alvarez, A real-time polymerase chain reaction method for monitoring anaerobic, hydrocarbon-degrading bacteria based on a catabolic gene, *Environ. Sci. Technol.* 36, 3977–3984 (2002).
- H. Junca, and D.H. Pieper, Functional gene diversity analysis in BTEX contaminated soils by means of PCR-SSCP DNA fingerprinting: comparative diversity assessment against bacterial isolates and PCR-DNA clone libraries, *Environ. Microbiol.* 6, 95–110 (2004).
- M. Orita, H. Iwahana, H. Kanazawa, K. Hayashi, and T. Sekiya, Detection of polymorphisms of human DNA by gel electrophoresis as single-strand conformation polymorphisms, *Proc. Natl. Acad. Sci. U.S.A.* 86, 2766–2770 (1989).
- F. Schwieger, and C.C. Tebbe, A new approach to utilize PCR-single-strandconformation polymorphism for 16s rRNA gene-based microbial community analysis, *Appl. Environ. Microbiol.* 64, 4870–4876 (1998).
- A. Schmalenberger, and C.C. Tebbe, Bacterial diversity in maize rhizospheres: conclusions on the use of genetic profiles based on PCR-amplified partial small subunit rRNA genes in ecological studies, *Mol. Ecol.* 12, 251–261 (2003).
- A. Okuta, K. Ohnishi, and S. Harayama, PCR isolation of catechol 2,3-dioxygenase gene fragments from environmental samples and their assembly into functional genes, *Gene* 212, 221–228 (1998).
- J. Gescher, A. Zaar, M. Mohamed, H. Schagger, and G. Fuchs, Genes coding for a new pathway of aerobic benzoate metabolism in *Azoarcus evansii*, *J. Bacteriol.* 184, 6301–6315 (2002).
- A. Zaar, J. Gescher, W. Eisenreich, A. Bacher, and G. Fuchs, New enzymes involved in aerobic benzoate metabolism in *Azoarcus evansii*, *Mol. Microbiol.* 54, 223–238 (2004).
- V.J. Denef, J. Park, T.V. Tsoi, J.M. Rouillard, H. Zhang, J.A. Wibbenmeyer, W. Verstraete, E. Gulari, S.A. Hashsham, and J.M. Tiedje, Biphenyl and benzoate metabolism in a genomic context: outlining genome-wide metabolic networks in Burkholderia xenovorans LB400, *Appl. Environ. Microbiol.* 70, 4961–4970 (2004).
- O.V. Moiseeva, I.P. Solyanikova, S.R. Kaschabek, J. Groning, M. Thiel, L.A. Golovleva, and M. Schlomann, A new modified *ortho* cleavage pathway of 3-chlorocatechol degradation by *Rhodococcus opacus* 1CP: Genetic and biochemical evidence, *J. Bacteriol.* 184, 5282–5292 (2002).
- P. Nikodem, V. Hecht, M. Schlomann, and D.H. Pieper, New bacterial pathway for 4and 5-chlorosalicylate degradation via 4-chlorocatechol and maleylacetate in *Pseudomonas* sp strain MT1, *J. Bacteriol.* 185, 6790–6800 (2003).

- J. Yun, S. Kang, S. Park, H. Yoon, M. Kim, S. Heu, and S. Ryu, Characterization of a novel amylolytic enzyme encoded by a gene from a soil-derived metagenomic library, *Appl. Environ. Microbiol.* 70, 7229–7235 (2004).
- S. Lee, K. Won, H. Lim, J. Kim, G. Choi, and K. Cho, Screening for novel lipolytic enzymes from uncultured soil microorganisms, *Appl. Microbiol. Biotechnol.* 65, 720– 726 (2004).
- E.M. Gabor, W.B.L. Alkema, and D.B. Janssen, Quantifying the accessibility of the metagenome by random expression cloning techniques, *Environ. Microbiol.* 6, 879– 886 (2004).
- A. Martinez, S.J. Kolvek, C.L.T. Yip, J. Hopke, K.A. Brown, I.A. MacNeil, and M.S. Osburne, Genetically modified bacterial strains and novel bacterial artificial chromosome shuttle vectors for constructing environmental libraries and detecting heterologous natural products in multiple expression hosts, *Appl. Environ. Microbiol.* 70, 2452–2463 (2004).
- T. Uchiyama, T. Abe, T. Ikemura, and K. Watanabe, Substrate-induced geneexpression screening of environmental metagenome libraries for isolation of catabolic genes, *Nat. Biotechnol.* 23, 88–93 (2005).
- 92. J.F. Jr. Brown and R.E. Wagner, PCB movement, dechlorination and detoxification in the Acushnet estuarine, *Environ. Toxicol. Chem.* 9, 1215–1233 (1990).
- D.L. Bedard and J.F. Quensen III, Microbial reductive dechlorination of polychlorinated biphenyls, in: *Microbial Transformation and Degradation of Toxic Organic Chemicals*, edited by L.Y. Young and C.E. Cerniglia (Wiley-Liss Division, John Wiley and Sons, New York, 1995), pp.127–216.
- 94. J. Wiegel and Q. Wu, Microbial reductive dehalogenation of polychlorinated biphenyls, *FEMS Microbiol. Ecol.*, 32, 1–15 (2000).
- A.C. Alder, M.M. Häggblom, S. Oppenheimer and L.Y. Young, Reductive dechlorination of polychlorinated biphenyls in anaerobic sediments, *Environ. Sci. Technol.* 27, 530–538 (1993).
- F. Fava, S. Gentilucci, and G. Zanaroli, Anaerobic biodegradation of weathered polychlorinated biphenyls (PCBs) in contaminated sediments of Porto Marghera (Venice Lagoon, Italy), *Chemosphere* 53, 101–109 (2003).
- 97. L.D. Palekar, K.A. Maruya, J.E. Kostka and J. Wiegel, Dehalogenation of 2,6dibromobiphenyl and 2,3,4,5,6-pentachlorobiphenyl in contaminated estuarine sediment, *Chemosphere* 53, 593–600 (2003).
- J.L. Lake, R.J. Pruell and F.A. Osterman, An examination of dechlorination process and pathways in new Bedford Harbor sediments, *Mar. Environ. Res.* 33, 31–47 (1992).
- 99. G.D. Øfjord, J.A. Puhakka, and J.F. Ferguson, Reductive dechlorination of Aroclor 1254 by marine sediment cultures, *Environ. Sci. Technol.* 28, 2286–2294 (1994).
- M. Berkaw, K.R. Sowers and H.D. May, Anaerobic *ortho* dechlorination of PCBs by estuarine sediments from Baltimore Harbor, *Appl. Environ. Microbiol.* 62, 2534–2539 (1996).
- 101. S.E. Apitz, B.P. Ayers and V.J. Kirtay, Use of data on contaminant/sediment interactions to streamline sediment assessment and management. http://www.spawar.navy.mil/sti/publications/pubs/tr/1918/tr1918cond.pdf (2004).
- 102. F. Fava, G. Zanaroli and L.Y. Young, Microbial reductive dechlorination of preexisting PCBs and spiked 2,3,4,5,6-pentachlorobiphenyl in anaerobic slurries of a contaminated sediment of Venice Lagoon (Italy), *FEMS Microbiol. Ecol.* 44, 309–318 (2003).

- 103. D.J. Scala and L.J. Kerkhof, Nitrous oxide reductase (nosZ) gene-specific PCR primers for detection of denitrifiers and three nosZ genes from marine sediments", *FEMS Microbiol. Lett.* 162, 61–68 (1999).
- 104. R.D. Kimbrough, Polychlorinated biphenyls and human health: an update, *Crit. Rev. Toxicol.* 25, 133–163 (1995).
- M.J. Zwiernik, J.F. III Quensen and S.A. Boyd, FeSO<sub>4</sub> amendments stimulate extensive anaerobic PCB dechlorination, *Environ. Sci. Technol.* 32, 3360–3365 (1998).
- D.M. Ward and M.R. Wrinfey, Interactions between methanogenic and sulfatereducing bacteria in sediments, *Adv. Microbiol. Ecol.* 3, 141–175 (1985).
- 107. J. Kim and G.-Y. Rhee, Reductive dechlorination of polychlorinated biphenyls: interactions of dechlorinating microorganisms with methanogens and sulfate reducers, *Environ. Toxicol. Chem.* 18, 2696–2702 (1999).
- G. Bretschko, and H. Moser, Transport and Retention of Matter in Riparian Ecotones, *Hydrobiologica* 251(1–3): 95–102 (1993).
- 109. B.G. Fraser, and D.D. Williams, Seasonal Boundary Dynamics of a Groundwater/Surface Water Ecotone, *Ecology* 79(6): 2019–2031 (1998).
- 110. M.W. Naegeli., and U. Uehlinger, Contribution of the Hyporheic Zone to Ecosystem Metabolism in a Prealpine Gravel-Bed River, *Journal of the North American Benthological Society 16*(4): 794–804 (1997).
- 111. M. Push, D. Fiebig, I. Brettar, H. Eisenmann, B.K. Ellis, L.A. Kaplan, M.A. Lock, M.W. Naegeli, and W. Traunspurger, The Role of Micro-Organisms in the Ecological Connectivity of Running Waters, *Freshwater Biology* 40(3): 453–495 (1998).
- 112. B. Jr. Conant, J.A. Cherry, and R.W. Gillham, A PCE Groundwater Plume Discharging into a River: Influence of the Streambed and Near-River Zone on Contaminant Distributions, *Journal of Contaminant Hydrology* 73(1–4): 249–279 (2004).
- 113. R. Krajmalnik-Brown, T. Hölscher, I.N. Thomson, F.M. Saunders, K.M. Ritalahti, and F.E. Löffler, Genetic Identification of a Putative Vinyl Chloride Reductase in *Dehalococcoides* sp. Strain BAV1, *Applied and Environmental Microbiology* 70(10): 6347–6351 (2004).
- 114. F.E. Löffler, Q. Sun, J. Li, and J.M. Tiedje, 16S rRNA Gene-Based Detection of Tetrachloroethene-Dechlorinating *Desulfuromonas* and *Dehalococcoides* Species, *Applied and Environmental Microbiology* 66(4): 1369–1374 (2000).
- 115. J.A. Müller., B.M. Rosner, G. von Abendroth, G. Meshulam-Simon, P.L. McCarty, and A.M. Spormann, Molecular Identification of the Catabolic Vinyl Chloride Reductase from *Dehalococcoides* sp. Strain VS and Its Environmental Distribution, *Applied and Environmental Microbiology* 70(8):4880–4888 (2004).
- 116. C. Regeard, J. Maillard, and C. Holliger, Development of Degenerate and Specific PCR Primers for the Detection and Isolation of Known and Putative Chloroethene Reductive Dehalogenase Genes, *Journal of Microbiological Methods* 56(1): 107–118 (2004).
- 117. P. Vandevivere, and W. Verstraete, Environmental applications, p. 531–557. In C. Ratledge, and B. Kristiansen (ed.), Basic Biotechnology, 2<sup>nd</sup> Edition. Cambridge University Press, Cambridge, U.K. (2001).
- 118. B.E. Rittmann, and P.L. McCarty, Environmental Biotechnology: Principles and Applications, Chapter 15, p. 695–728. McGraw-Hill, New York (2001).
- J.C. Philp, and R.M. Atlas, Bioremediation of Contaminated Soils and Aquifers, p. 139–236. *In* R.M. Atlas, and J.C. Philp (ed.) Bioremediation. American Society for Microbiology Press, Washington, DC (2005).

- 120. S. El Fantroussi, and S.N. Agathos, Is bioaugmentation a feasible strategy for pollutant removal and site remediation? *Curr. Opin. Microbiol.* 8, 268–275 (2005).
- 121. S. El Fantroussi, M. Belkacemi, E.M. Top, J. Mahillon, H. Naveau, and S.N. Agathos, Bioaugmentation of a soil bioreactor designed for pilot-scale anaerobic bioaremediation studies, *Environ. Sci. Technol.* 33, 2992–3001 (1999).
- 122. E.K. Nyer, F. Payne, and S. Sutherson, Environment vs. bacteria or let's play 'name that bacteria', *Ground Wat. Monitor. Remed.* 23, 36–45 (2003).
- 123. R. El Mamouni, R. Jacquet, P. Gerin, and S.N. Agathos, Influence of electron donors and acceptors on the bioremediation of soil contaminated with trichloroethene and nickel: Laboratory and pilot-scale study. *Wat. Sci. Technol.* 45, 49–54 (2002).
- 124. O. Drzyzga, R. El Mamouni, S.N. Agathos and J.C. Gottschal, Dehalogenation of chlorinated ethenes and immobilization of nickel in anaerobic sediment columns under sulfidogenic conditions. *Environ. Sci. Technol.* 36, 2630–2635 (2002).

### UNCERTAINTY AND RESEARCH NEEDS IN THE AREA OF THE BIOLOGICAL RESTORATION OF CONTAMINATED SEDIMENTS

FABIO FAVA & SPYROS N. AGATHOS

DICASM, Faculty of Engineering, University of Bologna, Bologna, viale Risorgimento 2, 40126 Italy Unit of Bioengineering, Faculty of Bioengineering, Agronomy & Environment, Catholic University of Louvain, Croix du Sud 2, 1348 Louvain-la-Neuve, Belgium

# **1.** Background: Microbial processes in sediments and their potential role in the contaminated sediments remediation

A large array of microbial processes are taking place in aerobic or subsurface anaerobic sediments, where they are responsible for the turnover of naturally occurring organic matter and the N, P, S geochemical cycles (Kafkewitz and Togna, 1998). Oxidizing conditions generally exist only at the top few centimeters of surface sediments, providing a limited zone available electron acceptor for where oxygen is as aerobic biotransformation of organic and inorganic chemical compounds. Beneath the aerobic zone, sediments are typically dominated by a single electron acceptor process such as sulfate-reduction in marine sediments or methanogenesis in freshwater sediments. Bacteria and eukaryotic organisms are coexisting in such matrices, by strictly interacting and cooperating through complex and often not fully elucidated mechanisms. The microbial population might be responsible for the biodegradation/biotransformation of several chlorinated priority pollutants, such polychlorinated-biphenyls, dibenzodioxins, -dibenzofurans, -phenols and -benzenes, as well as some hydrocarbons, in particular under anaerobic conditions. Under sulfidogenic conditions, it might be also responsible for the precipitation/immobilization of some toxic heavy metals (Lloyd and Lovley, 2001). These microbial processes might be in turn responsible for a significant and cost effective decontamination/detoxification of polluted sediments (Bedard and Ouensen, 1995; Lloyd and Lovley, 2001; Haggblom et al., 2003). This potential becomes of special relevance when they are actively taking place *in situ*, where they might contribute to a significant mitigation of contamination (Natural Attenuation, NA) (Apitz et al., 2004). NA often results in a significant reduction of the area (volume of contaminated sediment) to be dredged or managed through suitable *in situ* physical-chemical treatments.

However, many among such observations are based on preliminary and/or incomplete/inadequate experimental evidence (Bedard and Ouensen. 1995; Lloyd and Loyley, 2001). In fact, relatively little is known vet about the actual relevance of microbially mediated degradation/detoxification processes, such as those mentioned above, in situ, and in particular in the large number of marine contaminated habitats (Bedard and Ouensen, 1995; Haggblom et al., 2003), often impacted by marked advective processes (Apitz et al., 2004). The few data coming from in situ monitoring (Monitored Natural Attenuation. MNA) generally indicate that microbiologically mediated biodegradation processes are slow, partial and very often constrained spatially and/or temporally (Apitz et al., 2004). Almost nothing is currently known about possible strategies/approaches suitable to efficiently and safely stimulate such processes in situ. These and other gaps of information on several other basic issues related to biological removal of pollutants from sediments (issues that are listed and discussed below) and on their actual relevance in situ, have dramatically reduced and are still adversely affecting the opportunities and perspectives of biological approaches in the management of the huge amounts of contaminated sediments.

# 2. Uncertainties in the area of biological treatment of contaminated sediments

### 2.1. UNCERTAINTIES ON BASIC ASPECTS OF MICROBIAL TRANSFORMATION OF POLLUTANTS IN SEDIMENTS

The biotransformation potential of microorganisms occurring in contaminated sediments is largely controlled by the pollutant bioavailability (Bedard and Quensen, 1995; Alexander, 1999; Haggblom et al., 2003). The value of this parameter is generally very low in sediments, in particular for highly hydrophobic pollutants, because of the generally very high content of sorbing or otherwise sequestering materials within the matrices. Microbes are also important determinants for this parameter, as they significantly contribute to the sediment redox potential, pH, precipitation and dissolution of minerals that compose the surface of sediments. Microbial degradation processes are probably able to remove the pollutants *per se* (assuming a relative abundance of competent microorganisms) and

the fraction of each pollutant loosely bound to the matrix (Alexander, 1999; Apitz et al., 2004).

Thus, a first point of uncertainty is whether we have to spend time and financial resources aiming at enhancing/speeding up the release of tightly bound contaminants out of sediments and if we can ever have an effective and biocompatible strategy to do so. Another uncertainty is related to the actual toxicity of the non-bioavailable pollutant fraction that, due to this feature, is remaining in the sediments. A further discussion of bioavailability and of its possible implications on sediment toxicity and biological restoration is available at the following web-site: http://www.sediments.org./sedmgt.pdf.

In the majority of contaminated sediments there are occurring very complex mixtures of organic pollutants, often also associated to heavy metals, which can frequently act as inhibitors of microbial activities and/or sources of highly toxic and bioaccumulable biotransformation byproducts (e.g., methylmercury generated from the microbial methylation of mercury in surface sediments) (Lloyd and Lovley, 2001; Apitz et al., 2004). Further, the metabolism of several halogenated pollutants and high-molecularweight hydrocarbons results in the accumulation of a number of potentially toxic, recalcitrant and often more water soluble (and therefore mobile) intermediates (Bedard and Quensen, 1995; Adriaens and Vogel. 1995: Haggblom et al., 2003). The uncertainties here are: a) how and how much does the coexistence of similar or different pollutants affect the biological fate of each single pollutant in terms of biotransformation rate and extent. breakdown mechanism and accumulation of transformation intermediates: and b) how and by how much do daughter products resulting from pollutant biotransformation contribute to the final sediment toxicity. There is a need to look not just at the disappearance of priority pollutants but also at the net changes in toxic effects in sediments before and after biological treatment. In addition, a very few studies have been aimed at understanding the interactions among microorganisms directly and indirectly involved in pollutant biodegradation in real sediments (Abraham et al., 2002).

Finally, new mixtures of chemicals have entered recently the large market of personal care products. Pharmaceuticals, hormones and a number of new endocrine disruptors are detected at growing concentrations (from ng to mg/l) in several aquatic systems (Lee et al., 2003; Pedersen et al., 2005). The uncertainties here are on a) our ability to detect and characterize them and their biotransformation products in sediments in a reliable manner through the currently available analytical procedures and b) on their potential biodegradability and degradation pathways in aerobic and anaerobic sediments.

#### 2.2. UNCERTAINTIES ON THE APPLICATION/EXPLOITING OF MICROBIAL CAPACITIES IN THE RESTORATION OF CONTAMINATED SEDIMENTS, IN PARTICULAR UNDER IN SITU CONDITIONS

The majority of information available on the biotransformation of pollutants in sediments and on the microorganisms potentially responsible for it has come from studies performed in laboratory-scale microcosms very often developed with sediments spiked with well defined target pollutants and suspended in synthetic culture media (Bedard and Quensen, 1995; Adriaens and Vogel, 1995; Haggblom et al., 2003). Thus, little is known about the biological fate of several priority pollutants in real sediments under actual site geochemical conditions. In addition, a very few studies have been aimed at a) investigating the occurrence and the main features in marine sediments of several biodegradation processes already elucidated in freshwater sediments, which markedly differ in many geochemical and biological properties from the marine ones (Haggblom et al., 2003; Apitz et al., 2004), b) developing strategies/approaches to stimulate pollutant biodegradation processes in sediments and c) understanding the interactions between microorganisms directly and indirectly involved in the pollutant biodegradation and the eukaryotic organisms, such as benthic organisms, co-occurring in real sediments.

any case, laboratory-scale evidence does not prove In that biodegradation will provide sufficient natural attenuation at the site (Apitz et al., 2004). Indeed, the coexistence of several different known and unknown pollutants along with the occurrence of lower and variable temperatures, (bio)turbation phenomena, etc, in the actual site sediment might markedly affect the occurrence and the route of biotransformation processes under in situ conditions (Bedard and Quensen, 1995; Adriaens and Vogel, 1995; Apitz et al., 2004). A lack of integrated metabolic knowledge within the in situ biogeochemical context might further contribute to making bioremediation processes unpredictable. Finally, tidal events, creeks, streams and the other features that move the sediments through transport may further adversely affect the onset and predictability of pollutant biodegradation in situ. Therefore, the main uncertainties are focused on a) how and to what extent the information currently available from laboratory experiments can be extrapolated and used for an effective prediction of microbial process potential and features in situ; b) how the spatial and temporal interactions between the large variety of native microorganisms with distinct metabolisms (reductive dechlorinators, methanogens, sulfate reducers, methanotrophic organisms) directly or indirectly involved in pollutant biodegradation can be managed and optimized in an ex-situ or an in situ treatment.

242

### **3.** Needs in the area of biological treatment of contaminated sediments

Too much laboratory-scale research has been performed on spiked sediments suspended in artificial mineral media (Bedard and Quensen, 1995; Haggblom et al., 2003). We need studies performed on real contaminated sediments suspended in their own real water under laboratory conditions that closely mimic those occurring *in situ* or those under which the sediments are subjected to *ex-situ* treatment (Apitz et al., 2004). This might allow us to collect information of some relevance for predicting, when combined to lines of biogeochemical evidences (Apitz et al., 2004), the actual potential of biological processes in the final *in situ* restoration of contaminated sediments.

A lot of work needs to be done under these laboratory-scale conditions in order to collect reliable information on a) the rate, extent and mechanism of biodegradation of aged priority pollutants occurring in a representative number of both freshwater and marine sediments, and b) how these parameters might change by enhancing the bioavailability of pollutants through the increase (via spiking) of the concentration of these same pollutants in the same sediments or through the addition of specific nutrients, electron donors/acceptors, specialized microorganisms, etc. The microcosms thus developed need to be monitored through an integrated chemical, molecular and ecotoxicological analytical methodology, able to provide holistic information on a) the fate of the parent pollutants and of the metabolites generated from their biotransformation and the potential role of bioavailability on such parameters, b) the basic microbial processes (nitrate-, sulfate-, Fe(III) or Mn (IV)-consumption and volatile fatty acids, CH<sub>4</sub> or H<sub>2</sub> production) in progress in the sediment and the structure and key catabolic potential of indigenous microbial community and c) the toxicity of the sediment throughout the whole treatment.

These data, in turn, can be valuable in predicting: i) the potential fate of common mixtures of pollutants in different types of sediments, ii) the main background products that might be expected from their biodegradation and their impacts on the final sediment toxicity, iii) the nutrients or inocula useful for the process and iv) the possible dynamics through which the main members of the indigenous microbial community interact temporally and are potentially involved in the pollutant removal.

The last group of findings might be of great relevance in order to develop sediment-specific biostimulation or bioaugmentation strategies. However, because microbial metabolism in sediments is still poorly understood and only little information on key catabolic genes involved in such processes is available yet, for this aim there is a need to develop better, more robust monitoring tools, including molecular and other culture-independent approaches.

More knowledge on the interactions between indigenous microorganisms and benthic organisms might be also useful. The latter organisms can be responsible for an improved pollutant bioavailability, partial pollutant biodegradation and the establishment of geochemical conditions favorable for pollutant-degrading microorganisms. On the other hand, microorganisms might support benthic organisms by removing toxic pollutants, complementing their metabolism towards useful sediment substrates.

Laboratory scale investigations performed as described above can provide a body of reliable knowledge essential for a preliminary evaluation of the actual potential/relevance of an *in situ* natural recovery (NA/MNA) for a contaminated area. As stated above, such experimental evidence does not necessarily prove that biodegradation will provide sufficient natural decontamination of the site. However, it is a key line of evidence that, properly combined with other lines of microbial and biogeochemical evidence coming from the same site (Apitz et al., 2004), can permit the assessment of the role of biodegradation in the eventual natural recovery of the site.

Information obtained from microcosm studies performed with nutrients and inocula might be also useful in the preliminary design of a site-specific biostimulation/biaugmentation strategy. Here a lot of new knowledge is needed, in terms of commercially available nutrients, electron donors/acceptors, suitable inocula and the strategies/technologies through which to efficiently incorporate them into the contaminated sediment/water system.

The site conceptual modelling is also of great relevance for a reliable understanding/prediction of the susceptibility of a site to *in situ* natural recovery (Apitz et al., 2004). Here we need monitoring which not only provides qualitative descriptions of the *in situ* processes, but that also has the potential to supply parameter values for mathematical modelling of these processes. This is crucial in order to link conceptual understanding to performance models of biotreatment.

The hyporheic zone or interface between groundwater and surface water seems to play an important role in the natural attenuation of groundwater pollutants flowing from the sediment into surface water. This zone is a unique niche created by the sediment in which (bio)reductive processes can take place. For instance several chlorinated aliphatic hydrocarbons (PCE, TCE, etc) can be dehalogenated in this zone. Heavy metals such as Cd and Zn, can be precipitated as metal sulfides (formed by sulfate reducing bacteria, SRB). However some of these zones do have higher groundwater influx rates than others leading to shorter residence times. If the residence time is shorter than the time needed to perform the dehalogenation process, incompletely dehalogenated compounds are produced and enter the surface water. The need here is to develop site specific strategies, which might also consist of injecting redox manipulating compounds, to enhance the reactivity in the hyporheic zone.

Ex-situ treatment is successfully applied to bioremediation of PAH- and mineral oil-contaminated sediments under aerobic conditions (landfarming or bioreactors). The residual pollutant concentration represents a key limiting factor in this treatment and it is generally responsible for prolonged treatment times (Harmsen, 2004). Concentration limit values and standard guidelines for sediment reuse should be developed and applied at the international level.

Here more information on the possibility of detoxifying sediments through a biologically or chemically mediated pollutant binding to the sediment inorganic/organic components or combining biodegradation and catalyst-enhanced chemical degradation would contribute considerably to the development of improved or innovative highly effective ex-situ remediation strategies. Furthermore, to accomplish the same aim, it would be very useful to select biogenic, biodegradable and non-toxic pollutant "solubilizing" agents able to intensify in a totally biocompatible way the bioavailability and therefore the biodegradation of aged hydrophobic pollutants during sediment conventional ex-situ bioremediation (via bioreactor systems) or the pollutant mobilization during sediment washing operations (Abraham et al., 2002). This choice should be modulated by a prior assessment of the local microbial community, since, at least in some cases, addition of exogenous "solubilizing" or surfactant agents may interfere with the natural tendency of specific bacteria (e.g. *Mycobacteria*) to directly adhere on target pollutants (e.g. PAHs). The laboratory-scale microcosms approach described above should be the preferential strategy of choice for performing such investigations.

#### References

- Abraham, W.R., Nogales, B., Golyshin, P.N., Pieper, D.H., and Timmis, K.N., 2002, Polychlorinated biphenyl-degrading microbial communities in soils and sediments, *Curr. Opin. Microbiol.* 5: 246–253.
- Adriaens, P., and Vogel, T.M., 1995, Biological treatment of chlorinated organics, in: *Microbial Transformation and Degradation of Toxic Organic Chemicals*, L.Y. Young and C.E. Cerniglia, eds., Wiley-Liss, USA, Chapter 12, pp 435.

- Alexander, M., 1999, *Biodegradation and Bioremediation*, 2nd ed., Academic Press, CA, USA.
- Apitz, S.E., Ayers, B.P., and Kirtay, V.J., 2004, Use of data on contaminant/sediment interactions to streamline sediment assessment and management. http://www.spawar.navy.mil/sti/publications/pubs/tr/1918/tr1918cond.pdf.
- Bedard, D.L., and Quensen, J.F. III., 1995, Microbial reductive dechlorination of polychlorinated biphenyls, in: *Microbial Transformation and Degradation of Toxic Organic Chemicals*, L.Y. Young and C.E. Cerniglia, eds., Wiley-Liss, USA, Chapter 4, pp 127.
- Haggblom, M.M., Ahn, Y.B., Fennel, D.E., Kerkhof, L.J., and Rhee, S.K., 2003, Anaerobic dehalogenation of organohalide contaminants in the marine environment, *Adv. Appl. Microbiol.* 53: 61–84.
- Harmsen, J., 2004, Landfarming of Polycyclic Aromatic Hydrocarbons and Mineral Oil Contaminated Sediments, Alterra Scientific Contribution 14, Wageningen, The Netherlands.
- Kafkewitz, D., and Togna, M.T., 1998, Microbes in the muck: a look into the anaerobic world, in: *Biological Treatment of Hazardous Wastes*, G.A. Lewandowski and L.J. DeFilippi, eds., John Wiley & Sons, Inc, USA, Chapter 12 pp. 327.
- Lee, L.S., Strock, T.J., Sarmah, A.K., and Rao, P.S., 2003, Sorption and dissipation of testosterone, estrogens, and their primary transformation products in soils and sediment, *Environ. Sci. Technol.* 37: 4098–4105.
- Lloyd, J.R., and Lovley, D.R., 2001, Microbial detoxification of metals and radionuclides, *Curr. Opin. Biotechnol.* 12: 248–253.
- Pedersen, J.A., Soliman, M., and Suffet, I.H., 2005, Human pharmaceuticals, hormones and personal care product ingredients in runoff from agricultural fields irrigated with treated wastewater, J. Agric. Food Chem. 53: 1625–1632.